Taking Back the Wheel: Transition of Control From Automated Cars and Trucks to Manual Driving

Bo Zhang

### TAKING BACK THE WHEEL: TRANSITION OF CONTROL FROM AUTOMATED CARS AND TRUCKS TO MANUAL DRIVING

**Bo Zhang** 

University of Twente

This research was funded by the Marie Curie Initial Training Network (ITN) project HFAuto – Human Factors of Automated Driving (PITN-GA- 2013-605817), TNO, and the Department Centre for Transport Studies, Faculty of Engineering Technology, University of Twente.



Cover illustration by Bo Zhang, inspired by the H-Metaphor that driving automated vehicles is like riding horses (Flemisch et al., 2003).

### TAKING BACK THE WHEEL: TRANSITION OF CONTROL FROM AUTOMATED CARS AND TRUCKS TO MANUAL DRIVING

DISSERTATION

to obtain

the degree of doctor at the Universiteit Twente, on the authority of the rector magnificus, prof. dr. ir. A. Veldkamp, on account of the decision of the Doctorate Board to be publicly defended on Friday 12 February 2021 at 16.45 hours

by

**Bo Zhang** 

born on the 28th of August, 1988 in Heilongjiang, China This dissertation has been approved by: supervisor: Prof. dr. M.H. Martens co-supervisor: Prof. dr. ir. E.C. van Berkum

Composition of the doctoral committee: Prof. dr. ir. H.F.J.M. Koopman chairperson Prof. dr. M.H. Martens supervisor Prof. dr. ir. E.C. van Berkum co-supervisor

Independent members:Prof. dr. L. BoyleUniversity of WashingtonProf. dr. K. BenglerTechnical University of MunichProf. dr. M. HagenziekerDelft University of TechnologyProf. dr. M.C. van der VoortUniversity of TwenteProf. dr. W.B. VerweyUniversity of Twente

#### TRAIL Thesis Series no. T2021/11, the Netherlands Research School TRAIL

TRAIL P.O. Box 5017 2600 GA Delft The Netherlands E-mail: info@rsTRAIL.nl

ISBN: 978-90-5584-284-1

Copyright © 2021 by Bo Zhang

All rights reserved. No part of the material protected by this copyright notice may be reproduced or utilized in any form or by any means, electronic or mechanical, including photocopying, recording or by any information storage and retrieval system, without written permission from the author.

Printed in the Netherlands

Dedicated to

My Mother Yuehua Zhang, who named her daughter 'Bo (博)' hoping that she would become a PhD (博士) one day. Your dream now has come true.

### 献给我的妈妈张月华:

亲爱的妈妈,您的女儿张博终于成为张博士了,您的梦想实现了!

### Preface

This thesis is the result of a PhD research carried out at University of Twente, Department Centre for Transport Studies from 2016 to 2020. Now this long journey has finally come to an end, filled with treasurable experiences that have shaped me into a qualified scientific researcher, a persevering person, and a critical thinker. Through all the ups and downs I was never alone, and I would like to express my sincere gratitude to all people who made this possible.

First of all, I would like to thank my promotors Marieke Martens and Eric van Berkum. They believed in my potentials and provided me the great opportunity to join their Department and develop my personal and professional skills. I am particularly grateful to Marieke for her all-round guidance and support. Thank you, Marieke, for sharing your invaluable experience and perspective as a successful female scientist, for helping me through the obstacles when the road got bumpy, and for all the inspiring and encouraging talks that lit up my day and kept me going steadily towards completing my PhD. My special thanks also go to Riender Happee and Joost de Winter at TU Delft, who welcomed me into the HFauto family, hosted my secondment in the early stage of my PhD, and provided continuous supervision and mentoring throughout my studies. Thank you Riender for reaching out to me and introducing me to Marieke when I was a fresh master graduate; my journey started because of you. Joost, your high standard of academic research pushed me further than I could imagine, and I will forever benefit from what I have learned from you. I am very grateful to Klaus Bengler for hosting my secondment at my old school TU Munich. Thank you for introducing me to the fantastic realm of Human Factors, and for supporting every important decision in my study and career.

An essential part of my research concerns driver behaviour in automated platooning scenarios, and this could not be completed without generous support from the amazing colleagues at TNO. I am grateful to have participated in the TNO Early Research Program (ERP) Human Enhancement: Adaptive Automation, and to have the access to large amounts of experimental data that formed the base for four chapters in this thesis. I want to express my deep gratitude to Ellen Wilschut, Dehlia Willemsen, and Jeroen Hogema for their invaluable scientific advice and contributions as co-authors, to Ingmar Stel for technical support, and to Bart Joosten for inspiring brainstorm sessions. My thanks also go to Tom Alkim at Rijkswaterstraat for giving me the opportunity to work on the see-through truck study and for revising the manuscript for publication.

Many thanks to Linda Boyle, Klaus Bengler, Marjan Hagenzieker, Mascha van der Voort, and Willem Verwey for being my doctoral committee members. Thank you for your time and patience, and for your intellectual contributions to my development as a scientist.

I would like to thank all my HFauto research fellows for sharing their knowledge and expertise in diverse research areas, and for the fantastic time at the consortium meetings. Many thanks to Silvia Varotto and Zhenji Lu at TU Delft for their invaluable contributions to two important chapters of my thesis, and for their encouragement and companion as friends. Thanks to Marjan Hagenzieker, Christopher Cabrall, Pavlo Bazilinskyy, and Miltos Kyriakidis at TU Delft, Daniel Heikoop, Alexander Eriksson and Neville Stanton at the University of Southampton, Joel Gonçalves, Bastiaan Petermeijer and Klaus Bengler at the TU Munich, Matt Sassman, Thierry Bellet and Marie-Pierre Bruyas at IFSTTAR Lyon, Alberto Morando and Marco Dozza at Chalmers University of Technology, Ignacio Solis, Veronika Petrovych, Katja Kircher, Jan Andersson, and Magnus Hjälmdahl at VTI Linköping. I feel lucky to be part of the big family and to learn from you at the early stage of my training, which helped kick-start my research. I'm also grateful to fellow researchers outside the HFauto consortium, Anna Feldhütter and Andre Dietrich at TU Munich, for supporting the experiment during my joint secondment with Zhenji.

Many thanks also go to my colleagues at the Department Centre for Transport Studies, University of Twente: Amelia Huang, IG Ayu Andani, Anika Boelhouwer, Francesco Walker, Tom Thomas, Kasper van Zuilekom, Mariska van Essen, Oskar Eikenbroek, Karst Geurs, Lissy La Paix Puello, John Pritchard, Tiago Fioreze, and others. Thank you for all the good time we spent together and for the inspiring discussions during lunch and coffee breaks. Many thanks to our secretary Dorette Olthof for making me feel at home from day one.

My life would be pale and I would never come this far without wonderful friendship developed along the way. Special thanks go to my paranymphs Ellen Wilschut and Silvia Varotto. Ellen, thank you for your strong support in every possible aspect in my career and personal life, and for your dedicated Dutch translation of the *Samenvattig* of my thesis. Silvia, thank you for your company and support throughout my journey. We share happiness and sorrow, and grow stronger together. My sincere gratitude also goes to my teachers and classmates at the Gerrit Rietveld Academy. It takes great courage to develop my Alter Ego as a visual artist besides being a scientific researcher; thank you all for keeping me going this far and staying true to myself. Thanks to all those who dropped a random "how's going" that distracted me from a blue, unproductive day.

I cannot express how grateful I am to my beloved family. Thank you, Fanyu, my husband and best friend in life, for your love, encouragement, patience, tolerance, and understandings, for your constructive and practical suggestions as a mature PhD researcher, and for respecting and supporting my bold decisions and wildest dreams. Thank you, my mom Yuehua Zhang, for your unconditional love and for being my role model as a decent, modest, strong, and independent woman. I'm always proud of being your daughter and I hope at this moment you feel proud too. To Amélie, Heidi, Penny, Pavarotti, Yellow, Grey, and Pipi, the cutest little living creatures keeping my company through the lonely days.

I'm so lucky and proud to have you all along my journey.

Bo Zhang Amersfoort, The Netherlands, January 2021

### Content

1. I	Introduction	1
1.1 (	General introduction and problem statement	1
1.2 I	Development of automated driving	2
1.3 I	Levels of driving automation and human factors issues	4
1.4 0	Control transitions in driving automation	9
1.5 H	Research objectives and research questions	14
1.6 1	Thesis structure	16
Refe	erences	18
<b>2. I</b> 2.1 I	Determinants of take-over time from automated driving	<b>27</b> 28
2.2 N	Methods	30
2.3 H	Results	36
2.4 I	Discussion	46
2.5 \$	Supplementary material	50
Ack	nowledgement	50
Refe	erences	50
3. ]	Fransitions to manual control from highly automated driving in non-critical truck platooning scenarios (with hand-held NDT)	65
3.1 I	Introduction	66
3.2 N	Methods	69
3.3 H	Results	73
3.4 I	Discussion	78
3.5 (	Conclusion	81

Acknowledgement	
Reference	
4. Transitions to manual control from highly automated driving in non-cr platooning scenarios (with mounted NDT)	itical truck 87
4.1 Introduction	
4.2 Methods	
4.3 Results	91
4.4 Discussion	
4.5 Conclusion	
Acknowledgement	
References	
5. Taking back manual control after automated platooning: A comparison and truck driver's behaviour	ı between car 101
5.1 Introduction	
5.2 Methods	
5.3 Results	
5.4 Discussion	
5.5 Conclusion	
Acknowledgement	
References	
6. The effect of see-through truck on driver monitoring patterns and respo	onses to
critical events in truck platooning	<b>129</b>
6.2 Mathad	
6.3 Results and Discussion	
6.4 Conclusion and Outlook	
Acknowledgement	137
References	137
7. The effects of monitoring requests on driver attention, take-over perfor	mance. and
acceptance	
7.1 Introduction	
7.2 Methods	144
7.3 Results	
7.4 Discussion	
7.5 Conclusion	
Acknowledgements	
References	
<ol> <li>Conclusions, discussion and implication for practice</li> <li>8.1 Implications for practice</li> </ol>	<b>171</b> 177

8.2 Recommendations for future research	178
References	179
Summary	181
Samenvatting in het Nederlands	185
About the author	189
TRAIL Thesis Series	192

### List of Abbreviations and Acronyms

ACC	Adaptive Cruise Control
ADAS	Advanced Driver Assistance System
AD	Automated Driving
ADS	Automated Driving System
AOI	Area of Interest
BE	Brake Event
BRT	Brake Response Time
CACC	Cooperative Adaptive Cruise Control
CAV	Connected and Automated Vehicle
DDT	Dynamic Driving task
DOF	Degree of Freedom
HMI	Human-Machine Interface
ICC	Interclass Correlation Coefficient
ITS	Intelligent Transportation Systems
LKAS	Lane Keeping Assist System
MD	Manual Driving
MR	Monitoring Request

MRT	Movement Response Time
NDT	Non-Driving Task
ODD	Operational Design Domain
OEDR	Object and Event Detection and Response
PRT	Perception Response Time
RT	Response Time
SAE	Society of Automotive Engineers
SD	Standard Deviation
SDLP	Standard Deviation Lane Position
SPD	Speed
STS	See-Through System
SWA	Steering Wheel Angle
THW	Time Headway
ТО	Take-over
ТОТ	Take-over Time
TRT	Total Response Time
TTC	Time to Collision

### 1. Introduction

#### 1.1 General introduction and problem statement

Motor vehicles radically changed the way we live and work, ever since its introduction 130 years ago. Although they have often been a symbol of independence and freedom (Steg, 2005), times are changing and the negative impacts are getting more and more attention, such as road accidents, traffic congestion, and pollutant emissions. According to the World Health Organization (2018), every year more than 1.3 million people worldwide are killed, and up to 50 million are injured in road traffic crashes. The National Highway Traffic Safety Administration (NHTSA) estimates that "dangerous choices or errors people made behind the wheel" contribute to more than 90% of serious crashes, which can be caused by various humanrelated factors such as intentional violations, attention lapses, distraction, fatigue, and alcohol use. Automated driving technology has great potential to fundamentally solve road traffic issues and improve our quality of life (Fragnant & Kockelman, 2015; Kyriakidis et al., 2019). By assisting or replacing human operation in normal and critical driving tasks, serious crashes that are dominantly caused by human errors are assumed to be largely reduced. If connected and coordinated through advanced communication systems in the form of a platoon, automated vehicles are able to travel safely even with much smaller headways, which can largely increase road capacity, improve traffic flow efficiency, and reduce energy consumption (Coppola & Silvestri, 2019; Rios-Torres & Malikopoulos; 2017, Shladover, 2018; Talebpour & Mahmassani, 2016).

Decades of effort and technology advancement seem to bring automated vehicles from science fiction fantasy closer to reality, but a long and windy road is still ahead to the realization of full automation where no human intervention would be needed in any road situation. Safe human-system interactions, particularly at transitions of control to the driver when the system cannot cope with the current driving situation, pose key challenges for a successful deployment of automated intelligent systems at different stages of development. When the system is less capable and reliable, the driver has to closely monitor the system and take over imminent manual control when necessary. This challenges humans' inherent weak point of staying vigilant over a prolonged period of time (Mackworth, 1948; Davies & Parasuraman, 1982;

Parasuraman 1987), and drivers' capability to respond adequately within a short time budget (Banks, Eriksson, O'Donoghue, & Stanton, 2018; Casner, Hutchins, & Norman, 2016). When the technology becomes more mature, the system is supposed to largely replace human operation in most situations. However, even in these cases, driver interventions due to system limitations or failures, or the exceedance of the system's operational limits are still needed. A main challenge at this stage is how drivers respond in these conditions and how they can be supported in taking back control in a safe and smooth manner.

Since the past decade, an increasing number of studies have addressed human factors issues related to control transitions (for an overview, see Lu, Happee, Cabrall, Kyriakidis, & De Winter, 2016 and Kyriakidis et al., 2019), and suggest that multiple factors related to the driver, the automation system, and the situation potentially influence driver readiness to take back control. This means that no single criterion for an optimal take-over time budget exists today that fits all drivers in all situations, and design solutions are called for to support individual drivers in taking over control. To achieve this goal, a good understanding of driver take-over process and the variability between and within drivers is needed, which requires further efforts because the large majority of studies merely focus on mean take-over response times measured in stand-alone automated car scenarios. This thesis tackles the issues stated above and investigates driver behaviour and performance at control transitions and the variability, particularly in automated platooning scenarios. The aim is to contribute to designing safe and comfortable control transitions to manual.

The following part of this chapter first provides a state of the art on development of automated driving (Section 1.2), then introduces levels of driving automation and related human factors challenges (Section 1.3), followed by fundamental knowledge on transitions of control, driver take-over process, and driver take-over performance (Section 1.4). In Section 1.5, the research objectives and research questions are formulated and explained. The overall structure of the thesis is outlined in Section 1.6.

### 1.2 Development of automated driving

Even long before the wish to solve traffic safety issues by means of automated vehicles, people started to dream of cars driving by themselves. One very early prototype of "driverless" cars dates back to mid-1920s when Houdina Radio Control demonstrated "American Wonder" in New York – a 1926 Chandler controlled by the following car via radio impulses (Time Magazine, 1925). At the 1939 World's Fair in New York, General Motors sponsored the "Futurama" exhibit to envision American lifestyle 20 years in the future. The highlight was an infrastructure system that could guide radio-controlled cars through electromagnetic fields embedded within the roadways, which was generally seen as the first proposal of an automated highway system in the world. In the late 1950s, General Motor together with the Radio Corporation of America (RCA) brought this idea to life and demonstrated a full-size electric guide-wire system on a test track that enabled automated lateral and longitudinal control of vehicles. Research and development that revolved around this concept continued for another 20-30 years in the United States, UK, Germany, and Japan.

From the 1980s, the rapid advances in electronics, computers, communications, controls, and sensor technologies started to shape modern automated driving systems that operate mainly based on onboard sensors and control units without dedicated infrastructure support (also known as stand-alone automation systems). In the early 1980s, Ernst Dickmanns and his team at Bundeswehr University Munich developed the first automated vehicle of this type: a Mercedes van incorporating a vision-based system that was capable of detecting road markings

and controlling steering wheel, throttle, and brakes of the van based on real-time evaluation of image sequence. In successive years, various projects tackling technological challenges in automated driving and practical road traffic problems were launched worldwide, including the EUREKA Prometheus project of the European Union (Williams and Preston, 1987), the DARPA Autonomous Land Vehicle (ALV) project in the United States (Schefter, 1985), and the Super-Smart Vehicle Systems program in Japan (Tsugawa, 1991). These intensive research and development efforts largely increased the capability and efficiency of vehicle automation, as evidenced by a series of successful demonstrations and challenges conducted during the 1990s and 2000s, such as DARPA Grand Challenges (Buehler, Iagnemma, & Singh, 2007) and the Urban Challenge (Buehler, Iagnemma, & Singh, 2009).

In the late 2000s and the early 2010s, Google and many major automotive manufacturers initiated commercial research on automated driving systems and began various testing on public roads. Meanwhile, several core advanced driver assistance systems (ADAS) such as adaptive cruise control (ACC) and lane keeping assist systems (LKAS) were gradually introduced to the market. Driving automation began to receive considerable attention in mass media, raising increasing interest in publicity and industry. In 2015, Tesla became one of the first car manufacturers to release partial automated driving features (Autopilot) to its customers, followed by other major automakers including BMW, Mercedes Benz, Audi, and VOLVO. Incorporating multiple advanced sensors (e.g., stereo camera, radar, and ultrasonic sensor) and enhanced processing capabilities, these commercialized automation systems are able to conduct longitudinal and lateral control of the vehicle in simple traffic situations under non-adverse weather conditions. Despite the image that is being presented by some users, industry or the media, the driver of these commercially available vehicles has to constantly monitor the driving environment and be prepared to take immediate control when necessary since its functioning is not reliable yet. Around 2015, at the time of the introduction of Autopilot by Tesla, the concept of automated driving reached its peak in expectation in the Hype Cycle (Figure 1.1 Left), implying an expected mainstream adoption within 5 to 10 years

As more time passed, the high expectations have diminished to a more realistic level in 2019 (Figure 1.1 Right), due to increasing real-world experience with automated driving technology and a better understanding of its capabilities and limitations (for an overview of expert opinions, see Bazilinskyy, Kyriakidis, Dodou, & De Winter, 2019). For example, it is now more widely recognized that currently, commercially available automation systems requiring constant driver supervision involve safety risks (for an overview of expert opinions, see Kyriakidis et al., 2019), as evidenced by a number of fatalities involving Autopilot that occurred in recent years (Mider, 2019). Also in the Netherlands, the Dutch Safety Board has identified a number of new road safety risks associated with current commercially available vehicles with automated functions (Dutch Safety Board, 2019).



Figure 1.1: Gartner Hype Cycle for Emerging Technologies illustrated for the year 2015 (Left) and 2019 (Right). The red circles highlight the position of automated driving technology.

In parallel with the development of stand-alone automated vehicles, extensive research on connected automated vehicles began in the late 1980s with the California PATH (Partners for Advanced Transportation Technology) program. The core concept is to operate vehicles in platoons, in which virtually connected vehicles travel closely together as one cooperative system, with its primary goal to maximize highway capacity, energy efficiency, and safety (Shladover, 2006). The early research of PATH focused on passenger cars platoons, revolving around the idea that all vehicles (including the platoon leader) would be fully automated on dedicated lanes to eliminate negative impact caused by human error. Since the late 1990s, research and development interest has shifted towards platooning of heavy-duty trucks, largely driven by the fuel economy in freight transportation (Tsugawa, 2013). Advanced longitudinal control functionalities that combine onboard sensors and vehicle-to-vehicle (V2V) communication, such as cooperative adaptive cruise control (CACC), have also become subjects of intensive research to achieve more flexible and reliable platooning systems. Although platooning has not yet been deployed in commercial use, current efforts are made towards operating truck platooning in real life cases and implementation of multi-brand platooning (e.g., ENSEMBLE, see Willemsen et al., 2018). A milestone is the European Truck Platooning Challenge initiated by the Dutch EU Presidency, in which six European truck manufactures brought truck platoons onto public roads for the first time, travelling from various European cities to the final destination of the port of Rotterdam in the Netherlands in April 2016. Due to safety concern and legal issues, the trucks participating in the challenge only performed automated longitudinal control, despite the capability of automated steering as demonstrated on test tracks, leaving an important role for the human in the platooning scenarios.

#### 1.3 Levels of driving automation and human factors issues

It can be seen that automated vehicles of various forms are merging into our roadways more or less along an evolutionary path. As the capability of a driving automation system increases, a wider range of driving tasks could be carried out by the system, leading to a reduction in human engagement and a change in the role of the human driver. It is highly important to provide common classifications that facilitate the exchange of knowledge across domains, and to avoid confusion and imprecisions when describing system functionalities and limitations (Shladover, 2018).

Michon's (1985) hierarchical structure is typically referenced to categorize manual driving tasks, which comprises three levels. The lowest level (operational level) concerns longitudinal and lateral motion control to maintain the vehicle's lane position in traffic, which are normally carried out with little cognitive effort. At the intermediate level (tactical level), driving manoeuvres are planned and executed based on the pre-defined goals and in response to the objects and events in the driving environment, which normally requires greater mental efforts, and more elaborate physical movement. Examples of such manoeuvres are a lane change, obstacle avoidance, and overtaking. At the highest level (strategical level), the general planning of a trip is conducted, including scheduling of the trip and selection of destinations and routes. At the current stage, automated vehicles are mainly expected to carry over driving tasks (SAE, 2018; Merat et al., 2019).

The German Federal Highway Research Institute (BASt; Gasser & Westhoff, 2012), the United States National Highway Traffic Safety Administration (NHTSA, 2013), and the Society of Automotive Engineers (SAE, 2018) have each developed a taxonomy of levels of driving automation. Despite differences in definitions and terminologies, the three taxonomies share similar criteria to categorize the automated driving systems, mainly based on how primary dynamic driving tasks (i.e., longitudinal and lateral vehicle motion control, and monitoring of the driving environment) are distributed between the human driver and the automation system (Lu et al., 2016). Because the SAE taxonomy provides relatively more precise definitions and is the most-cited reference for automated-vehicle capabilities in both industry and academic research (Shuttleworth, 2019), it is adopted in this thesis and is explained below (according to the latest version released in 2018).

As depicted in Table 1.1, six levels of driving automation are defined by SAE (2018), ranging from SAE L0 (no driving automation) to SAE L5 (full driving automation). Differences between levels are primarily determined by means of who is performing the dynamic driving task, who is the fallback-ready agent (i.e., who performs the dynamic driving tasks in case of system failures), the limit of the operational design domain (ODD, the specific conditions under which the system is supposed to function), and the role that is required of the driver in that specific level.

SAE L0 is equivalent to manual driving, which means that the driver executes all dynamic driving tasks, possibly assisted by lower levels of ADAS that provide warnings or momentary assistance (e.g., emergency braking assistance).

In SAE L1 (Driver assistance), the driving automation system executes either longitudinal or lateral control of the vehicle, while the driver performs the remaining dynamic driving tasks. Examples of such systems are the Adaptive Cruise Control (ACC) or Lane-Keeping System (LKS).

In SAE L2 (Partial driving automation), vehicle motion control in both dimensions are executed by the system (e.g., combining ACC and LKS), but the driver is required to constantly monitor the driving environment and supervise the system, and intervene as necessary to maintain safe operation of the vehicle even without notification. From this level on (L3 and higher), the human driver's role as an active operator is fundamentally changed.

In SAE L3 (Conditional driving automation) and L4 (High driving automation), the system performs all dynamic driving tasks within the limit of the ODD, so the driver does not need to permanently monitoring the driving environment and is allowed to engage in non-driving tasks. This is the level where automated driving becomes interesting for users, since they can make

better use of travel time for work and relaxation. A SAE L3 system expects the driver to intervene in case the system approaches the limits of its ODD (for instance when approaching a workzone or extreme weather conditions) or when system failures occur (i.e., the driver serves as the fallback-ready user within a reasonable time budget because the camera is obstructed). The system is capable to determine the necessity for driver intervention and issues a timely request to intervene (also known as take over request (TOR) in a rich body of literature). In SAE L4, the system does not primarily rely on the driver to be the fallback-ready user, even though a request to take back control may be provided. In case a system failure occurs or the driver does not respond to the issued request, the system performs the fallback itself and transitions automatically to a minimal risk condition (e.g., conducting an emergency stop at a 'safe' spot). Currently, SAE L4 automation is predominantly developed for public transport concepts such as last mile transits and automated shuttle buses on limited trajectories. Public transport solutions will not be part of this thesis since there is no transition of control back to manual driving.

In SAE L5 (Full driving automation), the system is capable of all dynamic driving tasks in all situations (i.e., the ODD is unlimited) without involvement from human drivers. However even there, some situations may occur where the car cannot continue, such as flooded roads or extreme snow storms. Since L5 vehicles are not designed for a transition of control, this level will also not be discussed in this thesis.

There have been discussions that the taxonomies with numbered levels may induce misinterpretation and false expectation among the public, because the ascending levels do not necessarily correspond to the actual evolution of the technology (Templeton, 2014). For instance, a L3 system may be only available during traffic congestions with speeds under 50km/h in the operational design domain, whereas on a trip without any congestion, it will not have this level available at all. And even with a L4 system, it still may be the case that it may only work on motorways for 10% of the time, and not be available on other roads or under adverse weather conditions. Some also argue that the SAE levels are not sufficient to describe the variety of automation systems, for instance public transport on pre-defined routes only, or systems that are related to connected and cooperative technology (e.g., automated platooning systems). Due to the very short inter-vehicular gaps, it is very risky for drivers in a platoon to respond to longitudinal critical events in case of system failures, so the platooning system should perform the fallback even in lower SAE levels. In addition, the lead vehicle is normally operated by a professional driver in a lower level of automation (e.g., in SAE L0 or L1) than the following vehicles (e.g., in SAE L3 or L4), which makes it difficult to apply the SAE taxonomy to the whole platooning system.

Despite the limitations of the SAE levels to describe all possible types of automation, the key element remains the role of the human driver in driving tasks. Irrespective of whether a vehicle does or does not use cooperative technology or has an extended or a limited operational design domain, it is important to know if the driver needs to monitor the road or not, and what will happen when a driver needs to intervene under various conditions. While it may be true that only fully automated vehicles that completely remove driver responsibilities throughout the ride could maximize safety benefits of driving automation, experts generally believe that there is still a long way to go until this ultimate goal can be achieved (Kyriakidis et al., 2019; Shladover, 2016; Gomes, 2014; Underwood, 2014; Yoshida, 2014). As long as human intervention in any form is still expected, a safe human-automation interaction would play a central role in a successful deployment of driving automation on public roads (Carsten & Martens, 2019).

# Table 1.1: Summary of levels of driving automation defined by SAE (2016). DDT = Dynamic driving task, OEDR = Object and event detection and response (incl. monitoring the driving environment and the automation system), ODD = Operational design domain, ADS = Automated driving system.

	Name	Narrative definition	DDT			
Level			Sustained lateral and longitudinal vehicle motion control	OEDR	DDT fallback	ODD
Driver per	forms part or a	ll of the DDT				
0	No Driving Automation	The performance by the driver of the entire DDT, even when enhanced by active safety systems.	Driver	Driver	Driver	n/a
1	Driver Assistance	The sustained and ODD-specific execution by a driving automation system of either the lateral or the longitudinal vehicle motion control subtask of the DDT (but not both simultaneously) with the expectation that the driver performs the remainder of the DDT.	Driver and system	Driver	Driver	Limited
2	Partial Driving Automation	The sustained and ODD-specific execution by a driving automation system of both the lateral and longitudinal vehicle motion control subtasks of the DDT with the expectation that the driver completes the OEDR subtask and supervises the driving automation system.	System	Driver	Driver	Limited
ADS ("Sys	tem") performs	s the entire DDT (while engaged)				
3	Conditional Driving Automation	The sustained and ODD-specific performance by an ADS of the entire DDT with the expectation that the DDT fallback-ready user is receptive to ADS-issued requests to intervene, as well as to DDT performance-relevant system failures in other vehicle systems, and will respond appropriately.	System	System	Fallback- ready user (becomes the driver during fallback)	Limited
4	High Driving Automation	The sustained and ODD-specific performance by an ADS of the entire DDT and DDT fallback without any expectation that a user will respond to a request to intervene.	System	System	System	Limited
5	Full Driving Automation	The sustained and unconditional (i.e., not ODD-specific) performance by an ADS of the entire DDT and DDT fallback without any expectation that a user will respond to a request to intervene.	System	System	System	Unlimited

An increasing number of studies have addressed human factors issues that hinder drivers' capabilities to maintain safe control at different levels of automation (Saffarian, De Winter, &

Happee, 2012; Van den Beukel; & Martens, 2013; Cunningham, & Regan, 2015; Kyriakidis et al., 2019; Navarro, 2019). In partial automation (SAE L2), the current state of technology requires the driver to be prepared for imminent intervention and does not allow drivers to be able to take their eyes off the road. Therefore, it is crucial to constantly keep the driver in the monitoring loop, so that a high level of situation awareness (Endsley, 1995) can be maintained. The problem with this is that humans are by nature poor at vigilance tasks (i.e., to sustain concentrated attention and respond to irregular and infrequent target stimuli for extended period), as is the case with monitoring driving automation systems (Onnasch, Wickens, Li, & Manzey, 2014; Norman, 2015). Accumulating evidence suggests that vigilance performance (e.g., detection accuracy and response speed) inevitably declines over time on task due to the depletion of attentional resources (Mackworth, 1948; Davies & Parasuraman, 1982; Parasuraman 1987; Scerbo, 2001). The monotonous nature of monitoring tasks would also induce intentional or unintentional attention switching towards task-unrelated thoughts and stimuli (Scerbo, 1998; Helton et al., 2005; Casner & Schooler, 2015), and consequently cause a loss of situation awareness.

In addition, overreliance (or complacency), resulting from inappropriately high trust (overtrust) in automation, is another major cause for monitoring failures (Ensley, 2017; Singh, Molloy& Parasuraman, 1993; Parasuraman & Riley, 1997; Carsten & Martens, 2019). Poor trust calibration is directly associated with an inaccurate mental model of system capabilities and limitations, which can be caused by insufficient information provided about system functionalities, little or no feedback on the system status, and a lack of prior experience with such systems (Lee & See, 2014; Endsley, 2017; Walker, Wang, Martens, & Verwey, 2018; Carsten & Martens, 2019). Also, for people it may seem counter-intuitive that they are driving cars with automated functions, without any of the benefits of vehicle automation such as being able to (temporarily) do something else.

In higher levels of automation (SAE L3 and L4), the driver would be allowed to be temporarily out of the monitoring loop and to engage in a wide range of non-driving tasks (Naujoks, Befelein, Wiedemann, & Neukum, 2017), which would result in a large variability in drivers' activities, mental states, and body postures. How the systems can adapt to broad variations within and between drivers in taking over control, becomes the main challenge at this stage. Great caution is also needed to minimize automation surprises caused by unexpected system performance (Sarter & Woods, 1997), which are more prone to occur with increasing system reliabilities (Carsten & Martens, 2019; Endsley, 2017). Although ideally the system should only be allowed to be activated under conditions that it can cope with, accidents cannot always be avoided. Another potential issue induced by extensive use of automation is the loss of manual control skills, which has been frequently found among pilots that have become accustomed to autopilot systems (Veillette, 1995; Young, Fanjoy, & Suckow, 2006; Haslbeck & Zhang, 2017). Safety critical situations would occur when the driver has to take over control due to system failures, while he/she is no longer proficient in manual driving.

The human factors challenges described above largely revolve around transitions of control from automation to the driver and vice versa. A good understanding of drivers' performance at control transitions in various states and conditions remains substantial for the development of safe driving automation platforms. In the following section, we introduce the mechanism of control transitions in driving automation, driver take-over process, and the measures for driver take-over performance.

#### 1.4 Control transitions in driving automation

Transitions of driving tasks between the human driver and the automation system may occur at different automation levels, due to various reasons such as drivers' personal preferences, entering or exiting the ODD, sensor limits or in extreme cases malfunctioning This can be further differentiated between transitions of control and transitions of monitoring activity (Lu et al., 2016). A transition of control mainly involves a reallocation of vehicle motion control tasks, while a monitoring transition concerns a change in status between driver (temporarily) monitoring and system (temporarily) monitoring. An overview of all possible transitions is illustrated in Figure 1.2.



Figure 1.2: All possible transitions between the driver and the automation system at different levels of automation, adapted from Flemisch, Kelsch, Löper, Schieben, & Schindler, (2008) and Lu et al., (2016). The solid green lines indicate transitions of both longitudinal and lateral vehicle motion control; the dashed green lines indicate transitions of vehicle motion control in only one dimension; the orange lines indicate monitoring transitions.

Martens et al., (2008) outlined three fundamental questions to classify control transitions: 1) who has it (who conducts the control task at the start of the transition); 2) who should get it (who should conduct the control task after the transition); and 3) who initiates the transition. Based on this classification scheme, Lu et al., (2016) further integrated the underlying reason of the transition into their framework, (i.e., whether the transition is *mandatory* based on certain decision rules or requirements, or *optional* based on the driver's voluntary intention while both agents are capable of the control task), yielding six types of control transitions between the driver and the automation as depicted in Figure 1.3.



Figure 1.3: Classification tree of transitions of control (Lu et al., 2016).

*Control transitions to the automation* are normally performed within the ODD of the system. In optional Driver-Initiated transitions, the driver decides to activate the automation system because he/she feels like doing so, such as switch on the ACC on a normal highway. In turn, the driver may choose to deactivate the automation and start driving manually even though the system is functioning well. Mandatory Driver-Initiated transitions concern situations where the driver has to switch on the automation to avoid or minimize undesirable consequences or safely-critical situations. For example, the driver has to activate the automation in order to enter an area specially equipped for automated vehicles, or to join in an automated platoon. Another possible use case is that the driver hands over control to the automation because he/she is no longer able to drive safely (e.g. due to emergency health issues or with driver fatigue). Mandatory transitions to automation could be automatically initiated by the automation system as well, depending on the implemented transition strategies. Currently, these transitions do not exist yet in current levels of automation of commercially available vehicles.

*Mandatory control transitions to the driver* (also referred to as *driver take-over*) are generally seen as a major challenge facing automotive engineers and human factors researchers, which would occur upon exceeding the system's ODD or due to system failures. In such situations, the driver is forced to take over control because the automation can no longer drive safely or cannot drive safely under conditions that are coming up soon. Safety critical situations would occur if the driver cannot intervene adequately in time.

In Automation-Initiated transitions, the system is able to determine the necessity to transfer control to the driver and issue a TOR, which allows the driver to respond within a certain time budget before causing adverse consequences (e.g., a collision) or triggering the system fallback performance. The available time budget may vary largely between different types of take-over scenarios (Gold, Naujoks, Radlmayr, Bellem, & Jarosch, 2018). If, when or where the system reaches its functional limit can be estimated in advance from system backend, map or V2X communication (i.e., scheduled take-over). Therefore, the system is able to provide a timely TOR with a longer time budget. The take-over scenario is normally more critical when it is related to the behaviours of other road users or system failures (i.e., unscheduled take-over). The available time budget depends on the predictability of the unfolding situation and the capabilities of the onboard sensors.

In Driver-Initiated transitions, the driver diagnoses that the system can no longer handle the current situation based on kinematic feedback from the vehicle, cues in the driving environment and his/her expectations, and takes over control without being requested by the system. This type of transition would occur when the system is not acting according to what a driver expects, or when a system is not able to detect (in time) a critical event or fails to diagnose its malfunction (i.e., silent automation failures, Louw, Kuo, Romano, Radhakrishnan, Lenné, & Merat, 2019), which is more likely in lower levels of driving automation. The hazardous situation could have already become highly critical when it is detected by the driver.

#### 1.4.1 Driver take-over process and take-over performance

As with most of the human executed activities in dynamic environments, taking over control in response to external input involves multiple psychological and motor processes. Referencing Wickens' information processing model (Wickens, Hollands, Banbury, & Parasuraman, 2015), the take-over process covers 1) the detection and perception of the take-over stimulus (i.e., a TOR or an environmental event that can initiate a driver take-over), 2) cognitive processing of the stimulus (to comprehend the necessity to take over control) and the current driving situation (to determine how to take over control), 3) establishment of motor readiness by repositioning

the hands on the wheel and foot on the pedal, and 4) execution of an action that influences vehicle motion control (by steering, pressing the braking/gas pedal, or pressing a button to disengage the automation).

Wickens et al. (2015) described speed, accuracy, and attentional demand as "the big three" measures of human performance. Generally speaking, the faster, the more accurately, and the more effortlessly a task is being conducted, the better the performance. Correspondingly, driver take-over performance can be evaluated using measures related to the response time (RT) to complete the take-over process, the quality of the manoeuvre that is required for a specific take-over scenario, as well as the workload involved in the process. They are introduced in turn below.

#### Take-over time

Driver take-over response time, or in short *take-over time* (TOT), is an essential parameter to evaluate driver take-over performance. A successful take-over first requires the driver to respond before the situation exceeds his/her controllability (Nilsson, Falcone, & Vinter, 2015). TOT is generally measured from the onset of the take-over stimulus (a TOR or an environmental event) until the driver makes a conscious intervention (Gold & Bengler, 2014).

Additional response time metrics can be measured for a sequence of actions to break down the take-over process, in order to analyse driver behaviour at a fine-grained level. An overview of measurable RTs is given in Figure 1.4. In a few studies, gaze reaction time, eyes-on-road time, and hands-on wheel time were registered to reflect the moments when the driver senses the take-over stimulus, starts perceptual and cognitive processing of the driving environment, and establishes motor readiness, respectively (e.g., Gold, Damböck, Lorenz, & Bengler, 2013; Körber, Gold, Lechner, & Bengler, 2016; Feldhütter, Gold, Schneider, & Bengler, 2017). Less commonly registered is the initial start of the driver's hand movement (Kerschbaum, Omozik, Wagner, Levin, Hermsdörfer, & Bengler, 2017; Kerschbaum, Lorenz, & Bengler, 2015), possibly because time consuming video annotation is needed. RTs measured until the first hand movement provide some insight in the time elapsed until the driver has comprehended the necessity to take over control and starts to regain motor readiness. This is similar to perception time in the concept perception-response time to analyse drivers' braking response in the manual driving context (Olson, 1986; Green, 2000). Movement time, or motor response time (corresponding to the second component of *perception-response time*), can be measured from the start of the movement until the moment when the driver grasps the steering wheel, which reflects the time it takes to execute the actual response to establish motor readiness.

It has to be noted that the RTs above may only apply to the analysis of more basic responses towards the take-over stimulus. How long it takes for the driver to fully comprehend the different aspects of the driving situation and how the process of selecting a specific driving manoeuvre is difficult to observe and determine. These processes may overlap with other information processing stages and may even continue after the start of the manoeuvre. The driver may respond inadequately or even incorrectly if he/she takes over control before acquiring a sufficient level of situation awareness, particularly in complex and critical situations.



### Figure 1.4. Measurable response times (RT) within the take-over process in temporal order, adapted from Gold et al., (2013) with modifications.

#### Take-over quality

Besides take-over time, the quality of the take-over performance is also important to be measured. The evaluation of take-over quality depends on required actions to handle a certain take-over scenario (Gold et al., 2018; International Organization for Standardization [ISO], 2020). In some simple scenarios, such as taking over at the end of an ODD zone or due to the absence of lane markings, operational actions to stabilise the vehicle in its lane is sufficient. Desired performance is to maintain a steady lane position, a smooth, appropriate speed, and a safe following distance to the front vehicle. Correspondingly, metrics related to the variability of lane position (e.g., Standard Deviation Lane Position, SDLP), steering activity (e.g., the number of steering wheel reversals per minute), driving speed, and time headway can be employed to assess the quality of lateral and longitudinal driving performance. The performance data are often assessed in small windows after the control transition to explore the course of manual driving performance "recovery" as a function of time or travelling distance (Merat, Jamson, Lai, Daly & Carsten, 2014; Skottke, Debus, Wang, & Huestegge, 2014; Pfromm, Khan, Oppelt, Abendroth, & Brudera, 2015; Eriksson & Stanton, 2017). In order to estimate when the carryover effects of automated driving have sufficiently diminished and the driver could continue driving in a safe manner, driving performance after a take-over is often compared to driving performance in a baseline manual driving condition or reference threshold (ISO, 2020). More complex take-over scenarios require the driver to perform tactical manoeuvres according to pre-defined goals or rules, such as lane changing and turning, stopping at a traffic light, and adjusting speed to a new speed limit. These types of scenarios are less addressed in empirical studies. Whether the driver was able to respond correctly and timely to achieve the specific goal is the main criterion used in studies for the assessment of take-over quality.

In most challenging scenarios, the driver has to perform imminent tactical actions to achieve the goal of avoiding a collision with obstacles or other road users. Quality assessment for collision avoidance scenarios mainly concerns if the driver can operate the vehicle in a relatively safe manner and as a minimum can prevent a collision without endangering him or herself and other road users. First it has to be evaluated if crashes or other non-controllable events occur (skidding, rotating, swerving across several lanes or off paved road), which selfevidently would indicate take-over failures (Naujoks, Wiedemanna, Schömig, Jarosch, & Gold, 2018). For successful take-overs, low risk should be involved in the manoeuvre, which can be estimated using surrogate safety metrics such as minimum time to collision (TTC) and minimum clearance towards the obstacle or other road users (Gettman & Head, 2003; Happee, Gold, Radlmayr, Hergeth, & Bengler, 2017; Tarko, 2018). A higher value indicates a larger time gap or spatial safety distance and a lower degree of endangerment (Naujoks et al., 2018). Measures related to acceleration profiles such as maximum longitudinal and lateral acceleration are commonly used to reflect the intensity of the executed manoeuvres. Lower accelerations generally suggest smoother and safer manoeuvring. Both TOT and take-over quality should be taken into consideration to determine take-over performance. While a faster response is generally preferable, it can be assumed that drivers are more prone to making errors. Such a speed-accuracy trade-off (Fitts, 1966; Wickelgren, 1977) in control transitions has been suggested in a number of empirical studies that reported a higher take-over quality when drivers responded later (e.g., Damböck, Weissgerber, Kienle, & Bengler, 2013; Gold et al, 2013; Clark & Feng, 2017; Ito, Takata, & Oosawa, 2016). To find a "sweet spot", balancing take-over quality and TOT is a key aspect in determining a desirable take-over time budget and providing implications for the development of driving automation systems.

#### Workload

Workload can be understood as the amount of work or effort necessary to perform a task, which concerns both physical and mental aspects (Meijman & Mulder, 1998). When taking over control, the driver experiences physical workload to reposition his/her hands back on the wheel and feet back on the pedals, and mental workload to process the take-over stimulus and the take-over situation, and to make decisions on a take-over action. The physical load would be relatively small, while the mental workload may vary largely depending on task difficulty, situation complexity, and the driver's capability and states (Lee, Regan, & Horrey, 2020).

There are three main categories of workload measurement techniques: subjective measures, physiological measures, and performance measures (De Waard, 1996; O'Donnell & Eggemeier, 1986). Subjective measures assess and quantify personal judgements of experienced workload, which are usually well-established rating scales, such as the NASA-TLX (Task Load Index) (Hart & Staveland, 1988), the SWAT (Subjective Assessment Technique) (Reid & Nygren, 1988), and the RSME (Rating Scale Mental Effort) (Zijlstra & Van Doorn, 1985). Physiological measures are used to infer variations in workload through changes in an individual's physiological states. Commonly used measures concern brain activity (EEG), cardiac activity (heart rate, heart rate variability, blood pressure), respiratory activity (respiration rate), eye activity (pupil diameter, eye fixation), and galvanic skin response (De Waard, 1996; Charles & Nixon, 2019). Performance-based measurement techniques are developed based on the assumption that the human operator has limited attentional resources (Kahneman, 1973; Yeh & Wickens, 1988). Decrements in primary task (i.e., the task of interest) performance can be an indicator that the workload is too high or too low, as both overload and underload can diminish performance. The performance on an additional, low-priority task, also called secondary task, can reflect the remaining capacity of the operator while performing the primary task (see e.g., Jahn, Oehme, Krems, & Gelau, 2005; Martens & Winsum, 2001; and Verwey, 2000 for applications of secondary task measures in the manual driving context).

In the context of automated driving, driver workload during the use of automation systems is a frequently recurring research topic (e.g., Stanton, Young, & McCaulder, 1997; Heikoop, De Winter, Van Arem, & Stanton, 2019; Stapel, Mullakkal-Babu, & Happee, 2019; see De Winter, Happee, Martens, & Stanton, 2014 for a review), while workload directly related to control transitions is hardly touched upon. One possible reason is a lack of suitable workload measures. Physiological and performance measures are usually aggregated over time and consequently

not applicable for assessing workload in a fragment of seconds (De Waard, 1996; Verwey & Veltman, 1996). They are also likely to interfere with the take-over process involving rapid physical movements. Subjective assessment causes least intrusion, but may induce biases if the driver cannot precisely recall workload experienced during one specific control transition, especially when multiples transitions are performed within one drive. Focusing on safety aspects of control transitions, this thesis will mainly assess time and quality aspects of take-over performance, which are direct indicators of transition safety and can be objectively assessed in simple, and non-intrusive manners.

### 1.5 Research objectives and research questions

This thesis focusses on Human Factors issues related to the transition of control from SAE L2-L4 vehicle automation to manual control, in both stand-alone automated driving scenarios and automated platooning scenarios. The research in this thesis addresses the challenges of take-over times and take over quality in various conditions. In order to contribute to a better understanding of behaviour during transitions of control to manual driving, the following research objectives are proposed:

## • Obj. 1: Explore determinants of TOT and TOT variability in normal and critical take-over scenarios and gain a deeper insight in the actual driver take-over process.

The first objective is to study the factors that affect TOT under various circumstances, in order to get more insight in inter- and intra-individual differences. Knowing the factors that affect TOT may help support drivers in taking back control in a safe and smooth manner. Although existing research provides useful insight into some factors that affect TOTs, findings of the individual studies are hardly generalizable across different driving contexts, because only a small number of variables are manipulated per study. Up to now, little effort has been made to quantitatively synthesize all the available TOT studies for a more holistic picture. In addition, most studies only focus on mean or median TOT values without addressing intra- and interindividual differences (Dinparast, Djadid, Lee, Domeyer, Schwarz, Brown, & Gunaratne, 2019; Eriksson & Stanton, 2017; Mole et al., 2020). Nevertheless, outliers in the response time distribution are most prone to safety critical accidents and other adverse situations (Horrey & Wickens, 2007). This would indicate that taking mean TOT of one study as the basis for understanding how long it takes before a driver takes back control may lead to an unsafe design. More research efforts are needed to focus on variability in TOTs in various conditions, and to investigate the cause for large outliers. Wickens & Corlett (2015) even claim that the ultimate goal is to design safe automation systems that can accommodate as close to 100% of the target driver population.

### • Obj. 2: Study driver take-over time and performance with professional drivers in automated truck platooning scenarios.

The second objective is to fill in the research gap of TOT in platooning scenarios, particularly that of professional truck drivers. Truck platooning is considered as a first step towards automated freight transportation in an open and uncontrolled environment (Bhoopalam, Agatz, & Zuidwijk, 2018; Janssen, Zwijnenberg, Blankers, & De Kruijff, 2015; World Maritime University, 2019). Nevertheless, the large majority of human factors studies merely focus on passenger car scenarios in a non-platooning situation. Research on professional truck drivers'

take-over behaviour is very limited, and at the start of this thesis, we were not aware of any attempts that were made to systematically study driver take-over performance when leaving an automated platoon. It still remains to be understood if the specific features of platooning, such as the very short inter-vehicular distance and blocked front view, and the driver categories (professional or non-professional drivers), influence the way drivers take over control. If we are able to identify the specific requirements of platoon drivers, and the differences between professional truck drivers and normal passenger car drivers in take-over behaviour, we can deliver valuable input for designing safe control transitions in platooning systems.

### • Obj. 3: Explore potential approaches that prime drivers for a safe and smooth take-over.

The third objective is to design and evaluate solutions that prime drivers for a safe and smooth take-over. As mentioned in the previous sections, a TOR with a large time budget cannot always be provided, particularly in on-road settings where critical yet unpredictable situations may occur. However, it is ineffective and unrealistic to require the driver to sustain attention and stay prepared to take back control throughout the ride. It is important to explore possible countermeasures that adapt to the uncertainty and complexity of the road situations and allow the driver to allocate attention accordingly. As mentioned above (Obj. 1), several previous studies pointed to a large variation between individual drivers in taking over control, and no single take-over time budget exists that fits all situations. The feasibility of an adaptive and personalized control transition approach will also be discussed in this thesis.

Based on the research objectives, the following research questions are formulated:

### • RQ1: What factors influence driver response times in taking back control from automated to manual driving?

This research question focuses on TOT and mainly links to research **Obj. 1.** To answer this question, a comprehensive literature search and meta-analysis has been conducted to provide the state of the art on driver take-over research, and to explore determinants of TOT on an aggregated level (i.e., what determined the mean/median TOTs). Data collected from empirical driving simulator studies performed in this thesis provide additional insight into influencing factors that were not included in the meta-analysis.

## • RQ2: How do car drivers and professional truck drivers perform when decoupling from highly automated platoons in normal, non-critical situations under the influence of various task conditions?

This research question focuses on driver behaviour at control transitions in platooning scenarios, and mainly links to research objective 2. This question is answered by analysing car and truck drivers' performance data when taking over in car and truck platooning driving simulator studies.

#### • RQ3: Could a monitoring request help driver respond more adequately with takeover performance in critical take-over situations?

This research question focuses on designing and evaluating an innovative HMI concept that may prepare the driver for potential critical takeovers, which mainly links to research **Obj. 3**. To answer this question, an empirical car driving simulator study is conducted that compares drivers' take-over performance and subjective ratings when using the innovative system to the conventional system that only issues TORs.

## • RQ4: What explains variability in driver take-over times and is an adaptive approach tuned to a specific driver or conditions a feasible solution for a safe and smooth transition to manual driving?

This RQ focuses on variability in TOT between and within drivers (**Obj. 1**), based on which the feasibility of adaptive driving automation tuned to a specific driver's states is discussed (**Obj. 3**). This question is answered combining the meta-review study, from which the correlation between the mean and standard deviation of TOT can be yielded, and all empirical studies performed in this thesis, which allow inspection into individual drivers' takeover response.

This research contributes to the literature on Human Factors in transport, primarily in the domain of highly automated driving in the context of driver behaviour at transitions of control. Especially, this research is one of the initial studies that focuses on automated truck platooning and adaptive automation. The results contribute to the development of an improved human-system interaction for comfortable and safe transitions of control.

### 1.6 Thesis structure

This thesis consists of eight chapters. The structure of the thesis and the links between the chapters and the research questions are depicted in Figure 1.5, and explained below.

In Chapter 1, the general research background, research objectives and research questions have been introduced.

Chapter 2 presents an exhaustive meta-review of 129 driver take-over time studies. The aim of this meta-review is to provide the state of the art on the relevant research on TOT, and to explore the effects of a wide range of factors on driver TOT and its variability on an aggregated level. This study mainly aims to answer **RQ 1**. Three complementary meta-analytical approaches were employed: (1) a within-study analysis, in which differences in mean TOTs were assessed for pairs of experimental conditions, (2) a between-study analysis, in which correlations between experimental conditions and mean TOTs were assessed, and (3) a linear mixed-effects model combining between study and within-study effects.

Chapter 3, 4, 5 and 6 present empirical studies that systematically investigate driver take-over performance in platooning scenarios. Data for driver performance analyses were generated from three simulator-based studies conducted in the project Adaptive Virtue Tow-Bar (A-VTB) within TNO's Early Research Program Human Enhancement. Chapter 3 and 4 describe two truck platooning studies with professional truck drivers in non-critical scenarios. In Chapter 5, we compared a passenger car platooning study with one truck platooning study to explore the difference between car and truck drivers. Chapter 6 concerns truck drivers' take-over performance in a critical system failure scenario.

The four chapters together provide a holistic picture of drivers' take-over performance in platooning scenarios and address **RQ 2**. In addition, we conducted manual video annotations to break down the total TOT. TOT was divided into the perception response time and the hand movement response, in order to analyse driver take-over processes at a fine-grained level. Video recordings of participants' behaviours during the take-over process were also analysed to explore individual differences. This contributes to a better understanding of the driver take-over process and contributes to **RQ 1**. A more detailed overview of each chapter is presented below.

- Chapter 3 and Chapter 4 describe two truck platooning experiments with professional truck drivers in a truck driving simulator. The aim is to investigate how truck drivers take over control to leave the truck platoon in normal situations under three task conditions: monitoring the driving situation, performing non-driving tasks on a tablet PC, and resting with the eyes closed. The difference between the two experiments lies in whether the tablet PC was hand-held or mounted on the centre console. In both experiments, driver TOTs and take-over quality in terms of vehicle stabilization and responses to an emergency brake event were analysed and compared between task conditions
- Chapter 5 presents a comparison study. A car platooning experiment was conducted in which identical experimental designs were used as in the first truck experiment (Chapter 4), but with the driving simulator in car configuration and with 18 passenger car drivers as participants. Car drivers' take-over performance was compared to that of the truck drivers to explore potential differences between two driver types and vehicle types.
- Chapter 6 describes a critical truck platooning scenario in which a system failure occurred and the participants had to take over control within a short time budget. Besides investigating truck drivers' take-over performance in abnormal and emergency situations, this study also explored the effects of a "see-through" screen that was developed in the driving simulation that allowed the driver to obtain images of the road situation in front of the lead vehicle. The see-through screen was simulated as a large LED screen attached at the back of the lead simulated vehicle. For drivers in the truck platoon, monitoring surrounding traffic environment and foreseeing upcoming hazardous situations is very difficult due to very short inter-vehicular distances and consequently a heavily blocked front view. It is therefore meaningful to explore whether providing drivers in a truck platoon with additional visual information of the front view can influence their monitoring pattern and increase awareness for an upcoming event.

Chapter 7 presents an empirical study to evaluate an approach that stimulates a dynamic allocation of monitoring tasks to human and automation, addressing **RQ 3.** First, an HMI concept was designed that provides a monitoring request (MR) when approaching a location where driver take-over may or may not be requested. The MR asked the driver to pause the non-driving task, monitor the traffic environment, and be prepared for a potential take-over. If a critical event was detected, the system provided a take-over request (TOR) as well. The effects of the MR+TOR system were assessed in a driving simulator study with 41 participants by comparing driver take-over performance and gaze behaviour to a conventional system that only issued a TOR. Because only a small portion out of all MRs required an actual driver take-over, an additional analysis was conducted to investigate how drivers' compliance with MRs was associated with previously experienced scenarios. The compliance level was measured based on drivers' eye, hand, and foot preparatory behaviours retrieved from manual video observation.

The empirical studies described in Chapter 3 - 7 also allow a deeper insight into individual drivers' take-over behaviour and performance under various conditions, which complement the knowledge obtained from the meta-review (Chapter 2). All together they answer **RQ 1 and RQ 4**.

Chapter 8 discusses the main research findings, conclusions, implications for practice, and recommendations for further research.



Figure 1.5: Overview of the thesis structure.

### References

Banks, V. A., Eriksson, A., O'Donoghue, J., & Stanton, N. A. (2018). Is partially automated driving a bad idea? Observations from an on-road study. *Applied ergonomics*, *68*, 138-145. doi:10.1016/j.apergo.2017.11.010

Bazilinskyy, P., Kyriakidis, M., Dodou, D., & De Winter, J. (2019). When will most cars be able to drive fully automatically? Projections of 18,970 survey respondents. *Transportation Research Part F: Traffic Psychology and Behaviour*, 64, 184-195. doi:10.1016/j.trf.2019.05.008

Bhoopalam, A. K., Agatz, N., & Zuidwijk, R. (2018). Planning of truck platoons: A literature review and directions for future research. *Transportation Research Part B: Methodological*, *107*, 212-228. doi:10.1016/j.trb.2017.10.016

Buehler, M., Iagnemma, K., & Singh, S. (Eds.) (2007). *The 2005 darpa grand challenge: The great robot race*. Berlin: Springer.

Buehler, M., Iagnemma, K., & Singh, S. (Eds.) (2009). *The darpa urban challenge:* Autonomous vehicles in city traffic. Berlin: Springer.

Carsten, O., & Martens, M. H. (2019). How can humans understand their automated cars? HMI principles, problems and solutions. *Cognition, Technology & Work, 21, 3-20.* doi:10.1007/s10111-018-0484-0

Casner, S. M., Hutchins, E. L., & Norman, D. (2016). The challenges of partially automated driving. *Communications of the ACM*, 59, 70-77. doi:10.1145/2830565

Casner, S. M., & Schooler, J. W. (2015). Vigilance impossible: Diligence, distraction, and daydreaming all lead to failures in a practical monitoring task. *Consciousness and cognition*, *35*, 33-41.

Clark, H., & Feng, J. (2017). Age differences in the takeover of vehicle control and engagement in non-driving-related activities in simulated driving with conditional automation. *Accident Analysis & Prevention*, *106*, 468–479. doi:10.1016/j.aap.2016.08.027.

Cunningham, M., & Regan, M. A. (2015). Autonomous vehicles: human factors issues and future research. *Proceedings of the Australasian Road Safety Conference, Australia.* 

Damböck, D., Weissgerber, T., Kienle, M., & Bengler, K. (2013). *Requirements for cooperative vehicle guidance*. *Proceedings of the 16th International IEEE Annual Conference on Intelligent Transportation Systems, The Hague, The Netherlands* (pp. 1656–1661). doi:10.1109/itsc.2013.6728467

Davies, D. R., & Parasuraman, R. (1982). *The psychology of vigilance*. London: Academic Press.

De Waard, D., (1996). *The measurement of drivers' mental workload* (Doctoral thesis, University of Groningen, Groningen, The Netherlands). Retrieved from: <u>https://www.researchgate.net/publication/237132452 The Measurement of Drivers' Mental</u> <u>Workload</u>

De Winter, J. C. F., Happee, R., Martens, M. H., & Stanton, N. A. (2014). Effects of adaptive cruise control and highly automated driving on workload and situation awareness: A review of the empirical evidence. *Transportation Research Part F: Traffic Psychology and Behaviour*, 27, 196–217. doi:10.1016/j.trf.2014.06.016

DinparastDjadid, A., Lee, J. D., Domeyer, J., Schwarz, C., Brown, T. L., & Gunaratne, P. (2019). Designing for the Extremes: Modeling Drivers' Response Time to Take Back Control From Automation Using Bayesian Quantile Regression. *Human factors*. doi:10.1177/0018720819893429.

Dutch Safety Board. (2019). *Who is in control? Road safety and automation in road traffic.* Retrieved from: <u>https://www.onderzoeksraad.nl/en/media/attachment/2019/11/28/who\_is\_in\_control\_road\_safety and automation in road traffic.pdf</u>

Endsley, M. R. (2017). Autonomous driving systems: A preliminary naturalistic study of the Tesla Model S. *Journal of Cognitive Engineering and Decision Making*, *11*, 225-238. doi:10.1177/1555343417695197

Endsley, M. R. (1995). Measurement of situation awareness in dynamic systems. *Human factors*, 37, 65-84. doi:10.1518/001872095779049499

Eriksson, A., & Stanton, N. A. (2017). Driving performance after self-regulated control transitions in highly automated vehicles. *Human factors*, *59*, 1233-1248. doi:10.1177/0018720817728774

Fagnant, D. J., & Kockelman, K. (2015). Preparing a nation for autonomous vehicles: opportunities, barriers and policy recommendations. *Transportation Research Part A: Policy and Practice*, 77, 167-181. doi:10.1016/j.tra.2015.04.003

Feldhütter, A., Gold, C., Schneider, S., & Bengler, K. (2017). How the duration of automated driving influences take-over performance and gaze behavior. In C. M. Schlick, S. Duckwitz, F. Flemisch, M. Frenz, & S. Kuz (Eds.), *Advances in ergonomic design of systems, products and processes* (pp. 309–318). Berlin, Heidelberg: Springer. doi:10.1007/978-3-662-53305-5\_22

Fitts, P. M. (1966). Cognitive aspects of information processing: III. Set for speed versus accuracy. *Journal of experimental psychology*, 71, 849-857. doi:10.1037/h0023232

Flemisch, F., Kelsch, J., Löper, C., Schieben, A., & Schindler, J. (2008). Automation spectrum, inner/outer compatibility and other potentially useful human factors concepts for assistance and automation. In D. de Waard, F. O. Flemisch, B. Lorenz, H. Oberheid, & K. A. Brookhuis (Eds.), *Human factors for assistance and automation* (pp. 1–16). Maastricht, the Netherlands: Shaker Publishing. Retrieved from: https://elib.dlr.de/57625/1/08HumanFactorsForAssistanceAndAutomation\_FlemischEtAl\_AutomationSpectrum.pdf

Gasser, T., & Westhoff, D. (2012). *BASt-study: Definitions of automation and legal issues in Germany*. Irvine, CA, USA: TRB Road Vehicle Automation Workshop. Retrieved from: http://onlinepubs.trb.org/onlinepubs/conferences/2012/Automation/presentations/Gasser.pdf

Gettman, D., & Head, L. (2003). Surrogate safety measures from traffic simulation models. *Transportation Research Record*, 1840, 104-115. doi:10.3141/1840-12

Gold, C., & Bengler, K. (2014). *Taking over control from highly automated vehicles*. *Paper* presented at the 5th international conference on applied human factors and ergonomics (pp. 3662–3667).

Gold, C., Damböck, D., Lorenz, L., & Bengler, K. (2013). "Take over!" How long does it take to get the driver back into the loop? *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *57*, 1938–1942. doi:10.1177/1541931213571433

Gold, C., Naujoks, F., Radlmayr, J., Bellem, H., & Jarosch, O. (2018). Testing scenarios for human factors research in level 3 automated vehicles. In N. A. Stanton (Ed.), *Advances in human aspects of transportation* (pp. 551–559). Cham: Springer. doi:10.1007/978-3-319-60441-1\_54

Gomes, L. (2014). *Google self-driving car: It may never actually happen*. Retrieved from: <u>http://www.slate.com/articles/technology/</u>

Green, M. (2000). "How long does it take to stop?" Methodological analysis of driver perception-brake times. *Transportation Human Factors*, 2, 195–216. doi:10.1207/STHF0203\_1.

Happee, R., Gold, C., Radlmayr, J., Hergeth, S., & Bengler, K. (2017). Take-over performance in evasive manoeuvres. *Accident Analysis & Prevention*, *106*, 211-222. doi:10.1016/j.aap.2017.04.017

Haslbeck, A., & Zhang, B. (2017). I spy with my little eye: Analysis of airline pilots' gaze patterns in a manual instrument flight scenario. *Applied ergonomics*, 63, 62-71. doi:10.1016/j.apergo.2017.03.015

Hart, S. G., and Staveland, L. E. (1988). Development of NASA-TLX (Task Load Index): Results of empirical and theoretical research. In P. A. Hancock and N. Meshkati (Eds.), *Human* 

mental workload (pp. 139-183). Amsterdam: North-Holland. doi:10.1016/S0166-4115(08)62386-9

Heikoop, D. D., De Winter, J. C., van Arem, B., & Stanton, N. A. (2019). Acclimatizing to automation: Driver workload and stress during partially automated car following in real traffic. *Transportation research part F: traffic psychology and behaviour*, 65, 503-517. doi:10.1016/j.trf.2019.07.024

Helton, W. S., Hollander, T. D., Warm, J. S., Matthews, G., Dember, W. N., Wallaart, M., ... & Hancock, P. A. (2005). Signal regularity and the mindlessness model of vigilance. *British Journal of Psychology*, *96*, 249-261. doi:10.1348/000712605X38369

Horrey, W., & Wickens, C. (2007). In-vehicle glance duration: Distributions, tails, and model of crash risk. *Transportation Research Record: Journal of the Transportation Research Board*, 2018, 22–28. doi:10.3141/2018-04

International Organization for Standardization. (2020). *Road Vehicles: Human Performance and State in the Context of Automated Driving: Part 2 – Considerations in designing experiments to investigate transition processes* (ISO/TR 21959-2). Retrieved from: https://www.iso.org/standard/73928.html

Ito, T., Takata, A., & Oosawa, K. (2016). *Time required for take-over from automated to manual driving* (No. 2016-01-0158). SAE Technical Paper. doi:10.4271/2016-01-0158.

Jahn, G., Oehme, A., Krems, J. F., & Gelau, C. (2005). Peripheral detection as a workload measure in driving: Effects of traffic complexity and route guidance system use in a driving study. *Transportation Research Part F: Traffic Psychology and Behaviour*, 8, 255-275. doi:10.1016/j.trf.2005.04.009

Janssen, R., Zwijnenberg, H., Blankers, I., & De Kruijff, J. (2015). *Truck platooning: Driving the future of transportation*. Technical report, TNO. Retrieved from: https://www.logistiekprofs.nl/uploads/content/file/janssen-2015-truck.pdf

Kahneman, D. (1973). Attention and effort. New Jersey, U.S.A.: Prentice Hall.

Kerschbaum, P., Lorenz, L., & Bengler, K. (2015). *A transforming steering wheel for highly automated cars*. Paper presented at the proceedings of the 2015 IEEE intelligent vehicles symposium (IV), Seoul, South Korea (pp. 1287–1292). doi:10.1109/IVS.2015.7225893.

Kerschbaum, P., Omozik, K., Wagner, P., Levin, S., Hermsdörfer, J., & Bengler, K. (2017). How does a symmetrical steering wheel transformation influence the take-over process? In D. de Waard, A. Toffetti, R. Wiczorek, A. Sonderegger, S. Röttger, P. Bouchner, T. Franke, S. Fairclough, M. Noordzij, & K. Brookhuis (Eds.) *Proceedings of the human factors and ergonomics society Europe chapter 2016 annual conference*, Prague, Czech Republic (pp. 96– 106). Retrieved from: <u>https://www.hfes-europe.org/wp-</u> content/uploads/2016/11/Kerschbaum2017.pdf

Körber, M., Gold, C., Lechner, D., & Bengler, K. (2016). The influence of age on the take-over of vehicle control in highly automated driving. *Transportation Research Part F: Traffic Psychology and Behaviour*, *39*, 19–32. https://doi.org/10.1016/j.trf.2016.03.002.
Kyriakidis, M., De Winter, J. C., Stanton, N., Bellet, T., van Arem, B., Brookhuis, K., ... & Reed, N. (2019). A human factors perspective on automated driving. *Theoretical Issues in Ergonomics Science*, *20*, 223-249. doi:10.1080/1463922X.2017.1293187

Lee, J. D., Regan, M. A., & Horrey, W. J. (2020). Workload, Distraction, and Automation. In D. L. Fisher, W. J. Horrey, J. D. Lee, M. A. Regan (Eds.) *Handbook of Human Factors for Automated, Connected, and Intelligent Vehicles.* CRC Press.

Lee, J. D., & See, K. A. (2004). Trust in automation: Designing for appropriate reliance. *Human factors*, *46*, 50-80. doi:10.1518/hfes.46.1.50\_30392

Louw, T., Kuo, J., Romano, R., Radhakrishnan, V., Lenné, M. G., & Merat, N. (2019). Engaging in NDRTs affects drivers' responses and glance patterns after silent automation failures. *Transportation research part F: traffic psychology and behaviour*, 62, 870-882. doi:10.1016/j.trf.2019.03.020

Lu, Z., Happee, R., Cabrall, C. D. D., Kyriakidis, M., & De Winter, J. C. F. (2016). Human factors of transitions in automated driving: A general framework and literature survey. *Transportation Research Part F: Traffic Psychology and Behaviour, 43*, 183–198. doi:10.1016/j.trf.2016.10.007

Luximon, A., & Goonetilleke, R. S. (2001). Simplified subjective workload assessment technique. *Ergonomics*, 44, 229-243. doi:10.1080/00140130010000901

Mackworth, N. H. (1948). The breakdown of vigilance during prolonged visual search. *Quarterly Journal of Experimental Psychology*, *1*, 6-21. doi:0.1080/17470214808416738

Martens, M., Pauwelussen, J., Schieben, A., Flemisch, F., Merat, N., Jamson, S., & Caci, R. (2008). *Human Factors' aspects in automated and semi- automatic transport systems: State of the art.* Retrieved from: <u>http://www.citymobil-project.eu/downloadables/Deliverables/D3.2.1-</u> PU-Human%20Factors%20aspects-CityMobil.pdf

Martens, M., & Van Winsum, W. (2001). *Effects of speech versus tactile driver support messages on driving behaviour and workload* (No. 2001-06-0080). SAE Technical Paper.

Martens, M. H., & Van den Beukel, A. P. (2013). *The Road to Automated Driving: Dual Mode and Human Factors Considerations*. Paper presented at the 16th International IEEE Conference on Intelligent Transportation Systems (ITSC 2013), The Hague, the Netherlands. doi:10.1109/ITSC.2013.6728564

Meijman, T. F., & Mulder, G. (1998). Psychological aspects of workload. In P. J. D. Drenth & C. J. de Wolff (Eds.), *Handbook of work and organizational psychology, Vol. 2: Work psychology* (pp. 5–33). Hove, UK: Psychology Press.

Merat, N., Jamson, A. H., Lai, F. C., Daly, M., & Carsten, O. M. (2014). Transition to manual: Driver behaviour when resuming control from a highly automated vehicle. *Transportation research part F: traffic psychology and behaviour*, 27, 274-282. doi:10.1016/j.trf.2014.09.005

Merat, N., Seppelt, B., Louw, T., Engström, J., Lee, J. D., Johansson, E., ... & McGehee, D. (2019). The "out-of-the-loop" concept in automated driving: Proposed definition, measures and implications. *Cognition, Technology & Work*, *21*, 87-98. doi:10.1007/s10111-018-0525-8

Mider, Z. (2019). Tesla's Autopilot Could Save the Lives of Millions, But It Will Kill Some People First. Retrieved from <u>https://www.bloomberg.com/news/features/2019-10-09/tesla-s-autopilot-could-save-the-lives-of-millions-but-it-will-kill-some-people-first</u>

Michon, J. A. (1985). A critical view of driver behavior models: What do we know, what should we do?. In L. Evans and R. Schwing (Eds.), *Human Behavior and Traffic Safety*. Springer, Boston, MA. doi:10.1007/978-1-4613-2173-6\_19

Naujoks, F., Befelein, D., Wiedemann, K., & Neukum, A. (2017). A review of non-drivingrelated tasks used in studies on automated driving. In N. A. Stanton (Ed.), *Advances in human aspects of transportation*. doi:10.1007/978-3-319-60441-1\_52.

Naujoks, F., Wiedemann, K., Schömig, N., Jarosch, O., & Gold, C. (2018). Expert-based controllability assessment of control transitions from automated to manual driving. *MethodsX*, *5*, 579-592. doi:10.1016/j.mex.2018.05.007

Navarro, J. (2019). A state of science on highly automated driving. *Theoretical issues in ergonomics science*, 20, 366-396. doi:10.1080/1463922X.2018.1439544

NHTSA (2013). Preliminary statement of policy concerning automated vehicles. United States National Highway Traffic Safety Administration (NHTSA). Retrieved from: http://www.nhtsa.gov/staticfiles/rulemaking/pdf/Automated Vehicles Policy.pdf

Nilsson, J., Falcone, P., & Vinter, J. (2014). Safe transitions from automated to manual driving using driver controllability estimation. *IEEE Transactions on Intelligent Transportation Systems*, *16*, 1806-1816. doi:10.1109/TITS.2014.2376877

Norman, D. A. (2015). The human side of automation. In: G. Meyer, S. Beiker (Eds.) *Road Vehicle Automation 2. Lecture Notes in Mobility* (pp. 73-79). Springer, Cham. doi:10.1007/978-3-319-19078-5\_7

O'Donnell, R.D. & Eggemeier, F.T. (1986). Workload assessment methodology. In K.R. Boff, L. Kaufman & J.P. Thomas (Eds.), *Handbook of perception and human performance. Volume II, cognitive processes and performance.* (pp 42/1-42/49). New York: Wiley.

Olson, P. L., & Sivak, M. (1986). Perception-response time to unexpected roadway hazards. *Human factors*, 28, 91-96. doi:10.1177/001872088602800110

Onnasch, L., Wickens, C. D., Li, H., & Manzey, D. (2014). Human performance consequences of stages and levels of automation: An integrated meta-analysis. *Human factors*, *56*, 476-488. doi:10.1177/0018720813501549

Parasuraman, R. (1987). Human computer monitoring. *Human factors*, 29, 695–706. doi:10.1177/001872088702900609

Parasuraman, R., & Riley, V. (1997). Humans and automation: Use, misuse, disuse, abuse. *Human factors*, *39*, 230-253. doi:10.1518/001872097778543886

Paxion, J., Galy, E., & Berthelon, C. (2014). Mental workload and driving. *Frontiers in psychology*, *5*, 1344. doi:10.3389/fpsyg.2014.01344

Pfromm, M., Khan, R., Oppelt, S., Abendroth, B., & Bruder, R. (2015). Investigation of Take-Over Performance of Driving Tasks by the Driver due to System Failure of Semi-automated and Assisted Driving. *Proceedings of the 19th Triennial Congress of the IEA* (pp. 1–8). Reid, G. B., & Nygren, T. E. (1988). The subjective workload assessment technique: A scaling procedure for measuring mental workload. *Advances in psychology*, *52*, 185-218. doi:10.1016/S0166-4115(08)62387-0

Rios-Torres, J., & Malikopoulos, A. A. (2017). *Impact of connected and automated vehicles on traffic flow*. Paper presented at 2017 IEEE 20th International Conference on Intelligent Transportation Systems (ITSC) (pp. 1-6), Yokohama, Japan. doi:10.1109/ITSC.2017.8317654

SAE International. (2018). *Taxonomy and definitions for terms related to driving automation systems for on-road motor vehicle* (J3016). Warrendale, PA: SAE International. Retrieved from: https://saemobilus.sae.org/content/J3016\_201806

Saffarian, M., De Winter, J. C. F., & Happee, R. (2012). Automated driving: Human-factors issues and design solutions. In *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, Vol. 56 (pp. 2296–2300). doi:10.1177/1071181312561483.

Sarter, N. B., Woods, D. D., & Billings, C. E. (1997). Automation surprises. *Handbook of human factors and ergonomics*, 2, 1926-1943. Retrieved from: https://www.ida.liu.se/~769A09/Literature/Automation/Sarter,%20Woods,%20Billings\_1997. pdf

Scerbo, M. W. (2001). Stress, workload, and boredom in vigilance: A problem and an answer. In P. A. Hancock & P. A. Desmond (Eds.), *Stress, workload, and fatigue* (pp. 267–278). Mahwah, NJ: Erlbaum.

Scerbo, M. W. (1998). What's so boring about vigilance? In R. R. Hoffman, M. F. Sherrick, & J. S. Warm (Eds.), *Viewing psychology as a whole: The integrative science of William N. Dember* (pp. 145–166). Washington, DC: American Psychological Association. doi:10.1037/10290-006

Science: Radio Auto (1925). *Time Magazine*. Retrieved from: http://content.time.com/time/magazine/article/0,9171,720720,00.html

Shladover, S. E. (2018). Connected and automated vehicle systems: Introduction and overview. *Journal of Intelligent Transportation Systems*, 22, 190-200. doi:10.1080/15472450.2017.1336053

Shladaver, S. E. (2016). The truth about "self-driving" cars. Scientific American, 314, 52-57.

Shladover, S. E. (2007). PATH at 20—History and major milestones. *IEEE Transactions on intelligent transportation systems*, 8, 584-592. doi:10.1109/TITS.2007.903052

Shuttleworth, J. (2019). SAE Standards News: J3016 automated-driving graphic update. Retrieved from: <u>https://www.sae.org/news/2019/01/sae-updates-j3016-automated-driving-graphic</u>

Singh, I. L., Molloy, R., & Parasuraman, R. (1993). Automation-induced" complacency": Development of the complacency-potential rating scale. *The International Journal of Aviation Psychology*, *3*, 111-122. doi:10.1207/s15327108ijap0302\_2

Skottke, E. M., Debus, G., Wang, L., & Huestegge, L. (2014). Carryover effects of highly automated convoy driving on subsequent manual driving performance. *Human factors*, *56*, 1272-1283. doi:10.1177/0018720814524594

Stanton, N. A., Young, M., & McCaulder, B. (1997). Drive-by-wire: The case of driver workload and reclaiming control with adaptive cruise control. *Safety science*, 27, 149-159. doi:10.1016/S0925-7535(97)00054-4

Stapel, J., Mullakkal-Babu, F. A., & Happee, R. (2019). Automated driving reduces perceived workload, but monitoring causes higher cognitive load than manual driving. *Transportation research part F: traffic psychology and behaviour*, *60*, 590-605.doi:10.1016/j.trf.2018.11.006

Steg, L. (2005). Car use: lust and must. Instrumental, symbolic and affective motives for car use. *Transportation Research Part A: Policy and Practice*, *39*, 147-162. doi:10.1016/j.tra.2004.07.001

Talebpour, A., & Mahmassani, H. S. (2016). Influence of connected and autonomous vehicles on traffic flow stability and throughput. *Transportation Research Part C: Emerging Technologies*, 71, 143-163. doi:10.1016/j.trc.2016.07.007

Tarko, A. (2018). Surrogate Measures of Safety. In Lord, D. and Washington, S. (Eds.) *Safe Mobility: Challenges, Methodology and Solutions Vol. 11* (pp. 383-405). Emerald Publishing Limited. doi:10.1108/S2044-994120180000011019

Templeton, B. (2014). A Critique of NHTSA and SAE "Levels" of self-driving [Web log post]. Retrieved from: templetons.com/brad/robocars/levels.html

Tsugawa, S. (2013). An overview on an automated truck platoon within the energy ITS project. *IFAC Proceedings Volumes*, *46*, 41-46. doi:10.3182/20130904-4-JP-2042.00110

Tsugawa, S., Watanabe, N., & Fujii, H. (1991). *Super smart vehicle system-its concept and preliminary works*. Paper presented at the Vehicle Navigation and Information Systems Conference (pp. 269-277). doi:10.1109/VNIS.1991.205772

Underwood, S. E. (2014). Automated vehicles forecast vehicle symposium opinion survey. Presentation at the 2014 Automated Vehicles Symposium, San Francisco, CA.

Veillette, P. R. (1995). Differences in aircrew manual skills in automated and conventional flight decks. *Transportation Research Record*, *1480*, 43–50. Retrieved from: http://onlinepubs.trb.org/Onlinepubs/trr/1995/1480/1480-006.pdf

Verwey, W. B. (2000). On-line driver workload estimation. Effects of road situation and age on secondary task measures. *Ergonomics*, 43, 187-209. doi:10.1080/001401300184558

Verwey, W. B., & Veltman, H. A. (1996). Detecting short periods of elevated workload: A comparison of nine workload assessment techniques. *Journal of experimental psychology: Applied*, *2*, 270-285. doi:10.1037/1076-898X.2.3.270

Walker, F., Wang, J., Martens, M. H., & Verwey, W. B. (2019). Gaze behaviour and electrodermal activity: Objective measures of drivers' trust in automated vehicles. *Transportation Research Part F: Traffic Psychology and Behaviour*, 64, 401-412. doi:10.1016/j.trf.2019.05.021

Wickelgren, W. A. (1977). Speed-accuracy tradeoff and information processing dynamics. *Acta psychologica*, *41*, 67-85. doi:10.1016/0001-6918(77)90012-9

Wickens, C. D., Hollands, J. G., Banbury, S., & Parasuraman, R. (2015). *Engineering psychology and human performance*. Psychology Press.

Wilson, J. R., & Corlett, N. (Eds.). (2005). Evaluation of human work. CRC press.

Willemsen, D. M. C., Schmeitz, A., Fusco, M., Van Ark, E. J., Van Kempen, E., Söderman, M., Atanassow, B., Sjöberg, K., Nordin., H, Dhurjati, P., Schmidt, C., Daems, F., & Allard, J. (2018). *Requirements Review from EU projects. D2.1 of H2020 project ENSEMBLE* (GA No. 769115). www.platooningensemble.eu

World Health Organization. (2018). *Global status report on road safety 2018*. World Health Organization. Retrieved from: https://www.who.int/violence\_injury\_prevention/road\_safety\_status/2018/en/

World Maritime University. (2019). *Transport 2040: Truck platooning: The Australian case* - *Technology and transformation*. World Maritime University, Malmö. Retrieved from: https://commons.wmu.se/cgi/viewcontent.cgi?article=1074&context=lib\_reports

Yeh, Y.Y. & Wickens, C.D. (1988). Dissociation of performance and subjective measures of workload. *Human Factors*, *30*, 111-120. doi:10.1177/001872088803000110

Yoshida, J. (2014). *Fully autonomous car? Don't buy shotgun yet*. Retrieved from: http://www.eetimes.com/author.asp?section\_id=36&doc\_id=1322246

Young, J. P., Fanjoy, R. O., & Suckow, M. W. (2006). Impact of Glass Cockpit Experience On Manual Flight Skills. *Journal of Aviation Aerospace Education and Research*, *15*, 27–32. doi:10.15394/jaaer.2006.1501

Zijlstra, F.R.H. & Van Doorn, L. (1985). *The construction of a scale to measure perceived effort*. Delft, The Netherlands: Department of Philosophy and Social Sciences, Delft University of Technology. Retrieved from: https://www.researchgate.net/publication/266392097 The Construction of a Scale to Mea sure Perceived Effort

# 2. Determinants of take-over time from automated driving

This chapter is based on the following publication: Zhang, B., De Winter, J. C. F., Varotto, S., Happee, R., & Martens, M. H (2019). Determinants of take-over time from automated driving: a meta-analysis of 129 studies. *Transportation Research Part F: Traffic Psychology and Behaviour*, *64*, 285–307.

# 2.1 Introduction

Until automated driving systems are capable of performing all driving tasks under all road conditions (i.e., full automation as defined by SAE International, 2016), drivers will have to take over control when the automation fails or reaches its operational limits. Partial automation (SAE L2), which is already made available by several car manufacturers, requires drivers to monitor the road and to be prepared for immediate intervention at all times. At higher levels of automation (SAE L3 'conditional automation' and L4 'high automation'), drivers are allowed to engage in non-driving activities, while the automation executes the monitoring task and issues a take-over request (TOR) when the driver has to intervene. How long it takes drivers to reclaim manual control and what factors determine take-over time are important questions for both scientific researchers and automobile manufacturers.

#### 2.1.1 Driver take-over process and response times

The driver take-over process comprises several information-processing stages: perception of visual, auditory, and/or vibrotactile stimuli, cognitive processing of the information, response selection (decision making), resuming motor readiness (by repositioning the hands and feet on the steering wheel and pedals), and the actual action (e.g., steering and braking input to the vehicle) (Gold & Bengler, 2014; Gold, Damböck, Lorenz, & Bengler, 2013; Petermeijer, De Winter, & Bengler, 2016; Zeeb, Buchner, & Schrauf, 2015; Zhang, Wilschut, Willemsen, & Martens, 2019). Gold, Damböck et al. (2013) described four response (RT) measures: (1) gaze response time, (2) eyes-on-road time, (3) hands-on-wheel response time, and (4) take-over time (i.e., intervention time). In addition, researchers have used task-specific measures, such as hand movement response time (e.g., Kerschbaum, Lorenz, & Bengler, 2015; Kerschbaum, Omozik, Wagner, Levin, Hermsdörfer, & Bengler, 2017; Zhang et al., 2019), mirror check response time (e.g., Gold, Damböck et al., 2013; Vogelpohl, Kühn, Hummel, Gehlert, & Vollrath, 2018) and lane change response time (e.g., Petermeijer, Cieler, & De Winter, 2017; Eriksson et al., 2019). Although different response time measures can be distinguished, take-over time (TOT), defined as the time that drivers take to resume control from automated driving after a critical event in the environment or after having received a TOR, appears to be the most frequently used measure in the literature.

The temporal sequence of the take-over process is illustrated in Figure 2.1. Typically, the driver has to take over within the 'time budget' available until the system limit of the automation is reached. Such system limits may comprise an upcoming collision (e.g., with a stationary vehicle in the ego lane) or operational limits of the automated driving system (e.g., due to missing lane markings). If drivers do not take over within the available time budget, serious safety issues may occur.



Figure 2.1: Illustration of the take-over procedure. The present meta-analysis focuses on the take-over time (TOT), defined as the time between the take-over stimulus (take-over request or critical event in the environment) and the intervention by the driver.

#### 2.1.2 Previous Review Studies

The empirical literature reports a wide range of TOT values. For example, De Winter, Stanton, Price, and Mistry (2016) reported a mean TOT of 0.87 s (SD = 0.24 s) when the participants were required to brake in response to a salient red stop sign, whereas, in Politis et al. (2018), participants took over control on average 19.8 s (SD = 9.3 s) after the onset of a 60 s countdown TOR.

Several researchers have provided narrative reviews of TOT studies. Radlmayr and Bengler (2015) summarised eleven studies that investigated the effect of take-over time budget and concluded that longer time budgets are associated with longer TOTs and better take-over quality. A time budget smaller than 7 s was regarded as insufficient for a fully distracted driver to successfully take over control. Vogelpohl, Vollrath, Kühn, Hummel, and Gehlert (2016) provided an overview of 22 TOT studies involving a transition from highly automated driving (SAE L3) to manual driving and identified potentially influential factors related to the environment, the driver, the human-machine interface, and the vehicle. The authors suggested that the complexity of the take-over situation, the modality of the TOR, and the non-driving task (NDT) performed at the moment of the TOR are important factors. Furthermore, a traffic situation of high complexity and engagement in NDTs were argued to lead to slow responses, whereas multi-modal TORs shortened the TOT and improved the take-over quality. In another literature survey, Walch et al. (2017) discussed 17 take-over studies, focusing on the effect of the time budget, traffic complexity, NDT, and driver age. The authors concluded that 10 s seems an adequate time budget, while pointing out that the driver state and situational circumstances affect the driver's ability to take over control. Vogelpohl et al. (2016) and Walch et al. (2017) both noted that outcomes were sometimes inconsistent between the surveyed studies. For example, it was observed that Gold, Berisha, and Bengler (2015) and Petermann-Stock, Hackenberg, Muhr, and Mergl (2013) reported significantly longer TOTs when the participants were engaged in visual-motor NDTs compared to cognitive-auditory NDTs, whereas this effect was not statistically significant in the experiment by Radlmayr, Gold, Lorenz, Farid, and Bengler (2014). Another example of an inconsistency is that a negative effect of higher traffic complexity on TOT was found in Radlmayr et al. (2014) and Gold, Lorenz, and Bengler (2014), but not in Shen and Neyens (2014). This heterogeneity suggests that a larger number of studies need to be reviewed to draw reliable conclusions.

Although a number of narrative reviews exist, little effort has been devoted to quantitatively synthesising the available TOT studies. Eriksson and Stanton (2017) reviewed 25 take-over studies; they extracted 43 take-over time budgets (lead times) which varied between 0 and 30 s (mean = 6.37, SD = 5.36 s), and 87 TOTs from 1.14 s to 15 s (mean = 2.96, SD = 1.96 s). The authors noted that 3 s, 4 s, 6 s, and 7 s were the most frequently used time budgets and that the corresponding mean TOTs were 1.14, 2.05, 2.69, and 3.04 s. Apart from the time budget, Eriksson and Stanton (2017) did not review the effect of study variables that may affect the TOT. Gold, Happee, and Bengler (2018) provided a predictive model of TOTs based on the datasets obtained from six driving simulator experiments. Out of the seven variables considered in the model, the time budget, traffic density, and repetition (i.e., prior experience) turned out to be significant predictors, whereas driver age, physical and cognitive load of NDT, and the lane in which the ego car was driving (i.e., left, right, or middle) showed only minor effects. A limitation of Gold, Bengler et al. (2018) is that only six experiments were analysed and that the experimental settings were similar (i.e., in all the experiments, the take-over scenario was represented by two crashed vehicles blocking the ego lane).

The above reviews suggest that the time budget potentially affects TOTs. However, the existing reviews have several limitations. First, the available reviews analysed only a small number of study variables. Second, most reviews did not numerically synthesise the effects of study variables. Third, the number of reviewed studies is small: The maximum number of studies reviewed was 25 (Eriksson & Stanton, 2017), while this study included 129 TOT studies.

### 2.1.3 Research Objectives

As pointed out above, there is a need for a new quantitative synthesis of the various TOT studies, having a higher statistical power (i.e., a larger number of included studies) and broader scope (i.e., multiple variables examined simultaneously) as compared to previous reviews. We conducted a comprehensive search of empirical studies and employed meta-analytic methods to examine the predictors of TOT.

Cronbach (1975) discussed two disciplines of scientific psychology: experimental psychology, which is concerned with studying the effects of experimental manipulations, and correlational psychology, which is concerned with understanding differences between individuals and groups. In his work, Cronbach called for combining these two disciplines. A similar approach was followed in the present paper.

First, a within-study meta-analysis was performed to summarise studies that compared pairs of experimental conditions. The within-study analysis describes how mean TOTs are affected by a particular study variable when holding all other study variables constant, thus allowing for statements about causal effects.

Second, because individual experiments typically manipulate only a small number of variables, while experimental conditions differ across studies, we also examined the associations between experimental conditions and TOTs. The second approach concerned a correlation analysis to examine the relationships between the mean TOTs and a comprehensive list of study variables (related to the driver, the automation system, the human-machine interface, the take-over situation, and the experimental set-up) across all studies. The between-study analysis allows for predicting under which experimental conditions the mean TOTs will be low or high.

Third, the within-study experimental approach and the between-study correlational approach were united in a linear mixed-effects model. The mixed model allows for a powerful analysis of the effects of study variables while controlling for the confounding effect of the other study variables.

At the end of the paper, we discuss the similarity in the outcomes of the three methods. Consistent results across all three methods suggest high robustness and generalizability.

# 2.2 Methods

This study was performed in accordance with the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) guidelines (Moher, Liberati, Tetzlaff, & Altman, 2009). No protocol was generated or registered.

#### 2.2.1 Information sources and search strategy

We conducted a literature search with the aim to retrieve as many take-over studies as possible, including grey literature records to minimize publication bias (Rothstein, Sutton, & Borenstein, 2005). Multiple search strategies were used, such as database searching (Google Scholar,

ResearchGate), scanning the reference lists of papers, using the 'cited by' feature of Google Scholar, snowballing strategies (Jalali & Wohlin, 2012), and asking fellow researchers for relevant studies. Most searches were conducted using Google Scholar, as it is the most comprehensive search engine, especially for works of the 21st century (De Winter et al., 2014; Martín-Martín et al., 2018). Typically used keywords were 'takeover', 'take over', and 'transition of control' in combination with domain-specific keywords, to minimize false positives (e.g., 'automated driving', 'driverless', or the names of often-cited authors such as 'Bengler' or 'Merat'). The searches were performed between October 2, 2016 and December 17, 2018.

#### 2.2.2 Eligibility criteria

To be included in this review, studies had to fulfil the following six criteria:

- 1. The study had to involve a transition from partially, conditionally, or highly automated driving (i.e., SAE L2 and above; hands off the steering wheel and feet off the pedals) to manual driving.
- 2. The study had to involve an automation-to-manual take-over performed by a human (e.g., braking, steering, button pressing).
- 3. The study had to involve a transition in response to a TOR or a critical event in the environment. That is, this meta-analysis includes only 'mandatory driver-in-control (DC) transitions' as defined by Lu, Happee, Cabrall, Kyriakidis, and De Winter (2016). Studies in which more than one take-over stimulus (i.e., both a TOR and a critical event in the environment) was presented at different moments were not included, because in such cases it cannot be determined to which stimulus the participants responded. For example, in Körber, Baseler, and Bengler (2018), the obstacle was visible three seconds before the TOR, and the drivers were, therefore, able to take over control before the TOR.
- 4. The study had to report a mean or median take-over time (TOT), or the mean/median TOT should be calculable from the information reported.
- 5. In the text of the paper, the TOT had to be defined as the time interval between the initiation of the take-over stimulus (i.e., the onset of the TOR or the start of an environmental event that can initiate driver take-over) and the moment of driver intervention (by means of braking, steering, or button pressing).
- 6. The study had to be written in English or German.

All types of studies were eligible, including journal publications, papers from conference proceedings, theses, reports, posters, and presentation slides. If a publication contained more than one experiment, we considered each experiment as a separate study. We applied no restriction on the year of publication.

#### 2.2.3 Study selection and data extraction

After initial scanning and filtering, we retrieved 299 potentially relevant full-text records, which were further reviewed for eligibility. After removing 30 duplicate records and 1 record written in a language other than English or German (Japanese), 149 records were excluded for the following reasons: no TOT was measured (103 records), no mean/median TOT was reported (8 records), TOT was not defined according to the fifth eligibility criterion or the TOT measurement was not clearly described (18 records), participants were required to have at least one hand on the wheel during automated driving (4 records), or multiple take-over stimuli were presented at different moments (16 records). In the end, 129 studies from 119 records met the inclusion criteria. These studies comprised 520 mean or median TOT observations. In studies

where mean TOT values were only available in figures, the numerical values were extracted using the online tool WebPlotDigitizer (Rohatgi, 2017). Besides the mean and median TOT, which are measures of central tendency and our primary meta-analysis outcome of interest (see Section 2.1.1), we also extracted the standard deviation of the TOT as an index of variability.

The data extraction and variable coding (described in Section 2.2.4.2) were conducted by the first author. The second author supervised the process using a file hosting service (Dropbox), conducted multiple manual inspections of the annotated values, and corrected errors.

Ten of the included studies reported no mean TOT, but only the median TOT. Since the distribution of human response times is right skewed, using unadjusted medians together with means would induce bias. To reduce this bias, we applied a multiplication factor of 1.123 to the median TOTs to obtain an estimate of the corresponding mean TOTs. This correction factor was established from the means and medians from 14 included studies in which both values were reported.

An examination of the included studies showed similar experimental methods. That is, almost all studies used a virtual driving simulator and measured the TOT from the simulator sensor data (i.e., brake pedal depression or steering wheel angle). Although the studies involved simulators of different fidelity levels (e.g., motion base vs. no motion base), different experimental designs (e.g., between-subjects vs. within-subjects), and different experimental protocols (in terms of e.g., participant training, instruction, duration, and breaks), these differences provided no meaningful basis for assessing the study quality. Hence, we did not code the quality of individual studies and considered them of equal importance.

Furthermore, we did not apply weights that depend on sample size (e.g., Hedges & Vevea, 1998; Schmidt & Hunter, 2015). A preliminary analysis showed a substantial skewness of the sample size distribution, with a few large-sample studies including over 100 participants and many moderate-sample studies of about 20 participants. The use of unit weights has been recommended when the sample sizes are unequal, to avoid that the meta-analysis outcome is dominated by a small number of large-sample studies (Osburn & Callender, 1992). Our choice for unit weights is in line with simulation studies showing that unit weights offer similar or sometimes even superior predictive validity as compared to procedures that involve weighting (Bobko, Roth, & Buster, 2007; Einhorn & Hogarth, 1975). Unit weights are not estimated from the data and therefore do not have standard errors, as a result of which they can contribute to reduced estimation error as compared to a weighted average, especially when sample sizes and effect sizes are unequal (Bonett, 2008; Einhorn & Hogarth, 1975).

#### 2.2.4 Analysis methods

#### 2.2.4.1 Within-study analysis

In the within-study analysis, pairs of experimental conditions were categorised (e.g., no NDT vs. NDT, young participants vs. old participants, etc.). A meta-analysis was performed for a category when at least four studies were available in that category, following the recommendation by Fu et al. (2011). The 21 identified categories are shown in Figure 2.3.

Because all studies used the same unit to measure TOTs (seconds), the meta-analyses were performed on the raw (unstandardized) difference between mean TOTs (D). The use of Ds allows for intuitive interpretations as compared to standardised effect size measures (Bond, Wiitala, & Richard, 2003; Higgins & Green, 2005). In other words, we described the effect of an independent variable in seconds instead of a dimensionless index such as Cohen's d. The

use of seconds as a unit allows for easy interpretation in regard to practical applications (e.g., time budgets, look-ahead time of sensors) and the scientific literature in general (e.g., literature about brake response times, psychometric literature about reaction times).

The outcome of the meta-analysis was the unweighted average D per category. An absolute average D of 1 s was interpreted as a strong effect, and an absolute average D of 0.5 s was interpreted as a moderate effect.

In addition, we examined whether the D values differ from 0 (i.e., no effect), using a two-sided Wilcoxon signed-rank test with an alpha value of 0.05. This statistical test is conservative, because a significant effect (p < 0.05) can only be obtained when six or more studies (D values) are available (i.e., 2 \* 0.56 = 0.031 < 0.05).

#### 2.2.4.2 Between-study analysis

In the between-study analysis, we examined the correlations between 18 study variables (Table 2.1) and the mean TOTs. The 18 selected study variables were related to the driver, the automation system, the human-machine interface, the take-over situation, and the experimental set-up. More specifically, the variables concerned the mean age of participants, simulator fidelity, the level of automation, the modality of the TOR, the non-driving task (the modality of the task and if a device needed to be held in the hands), and the take-over situation (urgency of the scenario, complexity of the required driver response, and interaction with other road users). The selection of the study variables was based on the narrative reviews introduced above, studies providing guidelines for human factors research in the automated driving domain (e.g., De Winter, Happee, Martens, & Stanton, 2014; Gold, Naujoks, Radlmayr, Bellem, & Jarosch, 2018; Naujoks, Befelein, Wiedemann, & Neukum, 2017), and previous studies concerning driver response times in manual driving (e.g., Green, 2000; Summala, 2000). These variables were also selected based on whether they were available from the papers. For example, the physical intensity of the TOR, the level of drowsiness of the driver, and the duration of automated driving were not included as study variables, because these variables were often not documented, even though they are likely to affect the mean TOT.

We used Pearson product-moment correlations (equivalent to point-biserial correlations if the study variable is binary) and Spearman rank-order correlations to describe the relationships between the mean TOT and the study variables. The Spearman rank-order correlation is robust to tailed distributions and outliers.

A standard technique for assessing publication bias is to create a scatter plot showing the study outcome measures on the x-axis and a measure of sample size or precision on the y-axis, also called a funnel plot. An asymmetric relationship, where there exists a correlation between sample size and outcome measure, can be indicative of publication bias (Begg & Mazumdar, 1994; Egger, Smith, Schneider, & Minder, 1997; Deeks, Macaskill, & Irwig, 2005). We expected no effects of publication bias regarding the mean TOT, as high and low TOTs could be regarded as equally interesting to authors, publishers, and editors. Nonetheless, we assessed whether the mean TOT was correlated with the corresponding sample size.

Additionally, we computed correlations between sample size and all study variables (Table 2.1). These correlations allowed us to determine whether larger studies were associated with specific types of study design.

Study variable	Coding	Description
1. Age (AGE)	Years	Mean age of the participant group.
2. Level of automated drivi (LAD)	0 = L2; 1 = L3 and above ng	The level of automated driving (SAE International, 2016) as reported by the authors of the paper. In L2 automated driving, participants are in charge of the monitoring task. In L3 automated driving and above, the drivers are not supposed to monitor the driving environment.
3. Simulator (SIM)	$0^1 = 1$ low fidelity	A desktop-based simulator or a simulator providing the environment through computer monitors.
	1 = medium fidelity	An instrumented-cabin simulator or a simulator providing more than 120 deg horizontal field of view.
	2 = high fidelity	A simulator with motion platform or a real car.
4. Visual TOR (TOR_V)	0 = no; 1 = yes	Whether the TOR contains a visual stimulus.
5. Auditory TOR (TOR_A)	0 = no; 1 = yes	Whether the TOR contains an auditory stimulus. We applied no differentiation between vocal and acoustic TORs.
6. Vibrotactile TO (TOR_VT)	R $0 = \text{no}; 1 = \text{yes}$	Whether the TOR contains a vibrotactile stimulus (i.e., vibrations are provided on one or more locations on the human body).
7. Presence of TOI (TOR_P)	R $0 = \text{no}; 1 = \text{yes}$	Whether a TOR is implemented.
8. Visual NDT (NDT_V)	0 = no; 1 yes	Whether the NDT is visual (e.g., reading, watching a movie).
9. Auditory NDT (NDT_A)	0 = no; 1 = yes	Whether the NDT is auditory (e.g., listening to the radio, watching movies with sound, communicating with the instructors and answering questions verbally).
10. Motoric NDT (NDT_M)	0 = no; 1 = yes	Whether dynamic operation by hand is needed to perform the NDT (e.g., texting and tapping).
11. Cognitive load (NDT_C)	0 = normal cognitive load; 1 = high cognitive load	NDTs that require working memory (e.g., N- back task) were assigned to the high cognitive load category. Otherwise, the task was assumed to involve normal cognitive load.
12. Hand holding a device (HAND)	0 = hands-free (No non- driving task, the non- driving task does not require a device, or the non-driving task is performed using a fixed device);	Whether a device is handheld when undertaking the non-driving task.

# Table 2.1: Study variables and coding

	1 = handheld	
13. Presence of a non- driving task (NDT_P)	0 = no; 1 = yes	Whether a non-driving task is performed.
14. Time budget to collision (TBTC) <sup>1</sup>	Ratio variable	The available time budget for a response from the initiation of the take-over stimulus until the collision with an obstacle.
15. Time budget to other boundaries (TBTB) <sup>1</sup>	Ratio variable	The time from the initiation of the take-over stimulus until reaching the boundaries of the automated driving system other than collisions (e.g., due to the end of the automated zone, missing lane markers, or system failure).
16. Urgency (URG) <sup>1</sup>	0 = low urgency	No foreseeable collision risk or a high time budget to collision ( $\geq 15$ s).
	1 = medium urgency	Potential collision risk or disturbance to other road users if no response was made (e.g., the ego car drifting to the adjacent lane containing traffic), or a medium time budget (between 8 s and 15 s).
	2 = high urgency	Immediate risk of collision (time budget $\leq 8$ s) if no response was made, or the participants were instructed to react to a stimulus as quickly as possible.
17. Driver response (DRE) <sup>1</sup>	1 = low complexity	The participant had to take over control on a straight road by stabilising the vehicle in its lane.
	2 = medium complexity	The take-over scenarios required a specific driver response (braking or steering), such as when encountering a road narrowing, road constructions, or decelerating vehicles ahead, or when having to take over control on a curvy road.
	3 = high complexity	The take-over scenario requires complex driver decision making. The participant had to decide whether to brake or steer in response to the event.
18. Interaction with other road users (IRU)	0 = no	There were no other road users around, or other road users could not affect the driver's decision-making.
	1 = yes	The participants had to take into account one or more other road users when choosing their optimal take-over action. For example, participants had to take over control while driving in the middle lane while the left lane contained traffic.

<sup>1</sup>*Note.* The fidelity of the simulator was identified according to De Winter et al. (2014). The classification of URG and DRE was adapted from Gold, Naujoks et al. (2018). URG combines the TBTC and TBTB variables. TOR = Take-over request, NDT = Non-driving task.

#### 2.2.4.3 Linear mixed-effects model

We estimated a linear mixed-effects model describing the impact of the study variables on the mean TOTs, using the same dataset as the between-study analysis. A study-specific error term  $\vartheta$  was introduced to capture unobserved effects that affect mean TOTs within a study (i.e., the intercept differs between studies).

The model was estimated using the 'Mixed Model' command in SPSS 24 (IBM Corporation, 2016) with the estimation method restricted maximum-likelihood (REML) (Molenberghs, Kenward, & Verbeke, 2009; Zuur, Leno, Walker, Saveliev, & Smith, 2009). Goodness-of-fit measures (log likelihood) and information criteria (AIC, BIC) were used to compare alternative model specifications. The SPSS script is provided in the supplementary materials.

A log-normal probability density function was found to fit the mean TOT distribution better than the normal probability density function. All variables listed in Table 2.1 were tested as potential explanatory factors. The variables included in the final specification were selected based on their meaning (i.e., we selected non-redundant variables) and statistical significance (p < 0.05). One parameter was associated with each level of the explanatory variables and differences between levels were tested comparing alternative model specifications. Levels that did not differ significantly were merged. Variables that had a non-significant impact on the mean TOTs were excluded one by one. When a variable could not be extracted from one or more studies, a dummy variable was created to indicate the missing values. This variable was included in the model equation in addition to the original variable (dummy variable adjustment method). The advantage of this approach is that all observations could be analysed.

### 2.3 Results

#### 2.3.1 Study characteristics

The 129 included studies yielded 520 mean TOT observations from 4556 participants. 45 studies were conducted in a high-fidelity driving simulator with motion platform (40 studies) or in a real car (five studies). The 129 studies comprised 68 papers from conference proceedings, 40 journals articles, 3 technical reports, 16 chapters from a PhD or master thesis, and 2 posters. The year of publication ranged between 2000 and 2018, with the majority of the studies being published in and after 2015 (116 out of 129). A list of the included studies, the scores per study variable, and a MATLAB script that processes these data are provided in the supplementary materials.

The mean TOT across studies and conditions ranged from 0.69 s to 19.79 s, and the average mean TOT was 2.72 s (SD = 1.45, n = 520). Figure 2.2 shows the distribution of the mean TOTs, which is right-skewed.



Figure 2.2: Distribution of 520 mean take-over times (TOTs) reported in the included studies. The bin width is 0.25 s.

#### 2.3.2 Within-study analysis

Twenty-one categories with four or more studies were identified in the experimental analysis. Figure 2.3 shows the average difference in mean TOTs (D) for each category (squares), as well as the Ds from each study (smaller circles). The presence of non-driving tasks and the modality of the take-over request (TOR) were frequently used independent variables.

The following statistically significant findings stand out from Figure 2.3:

- A strong effect of time budget was found, with a higher mean TOT (average D = 1.35 s) for a large time budget compared to a small time budget.
- The mean TOT was substantially lower when taking over control for the second time (when asked to take over twice in the same driving session or perform the same scenario in a second driving session) compared to the first time (average D = -1.00 s).
- The use of a handheld device strongly increased the mean TOT (average D = 1.33 s).
- For hands-free non-driving tasks, performing a visual non-driving task slightly increased the mean TOT compared to not performing a non-driving task (average D = 0.29 s).
- The presentation of a TOR moderately decreased response times compared to when no TOR was provided (average D = -0.58 s).

In addition, the following findings are noteworthy, although based on five or less studies.

- Having eyes closed before taking over control strongly increased TOTs compared to not performing a non-driving task and staying alert (average D = 1.19 s).
- Strongly reduced TOTs were found when an auditory or vibrotactile TOR was provided compared to a visual-only TOR (average D = -1.41 and -1.41 s, respectively).
- Being able to anticipate the TOR (e.g., when the TOR was periodically scheduled or could be anticipated from environmental cues such as the traffic and weather) moderately shortened the mean TOT (average D = -0.54 s).
- The effect of traffic compared to no traffic was moderate (average D = 0.49 s).

No NDT vs. Visual-auditory NDT (hands-free)		C D D	Av. D = 0.40 s, k = 8, p = 0.039
No NDT vs. Visual-motor NDT (hands-free)		ø	Av. D = 0.29 s, k = 10, p = 0.004
No NDT vs. Auditory-cognitive NDT (hands-free)		000	Av. D = 0.23 s, k = 7, p = 0.297
No NDT vs. NDT (handheld)		0000 (1000)	Av. D = 1.33 s, k = 17, p < 0.001
No NDT vs. Eyes closed		<b>o</b>	Av. D = 1.19 s, k = 4, p = 0.125
Young vs. Old		ത്തം	Av. D = 0.10 s, k = 6, p = 0.688
No traffic vs. Traffic		o <mark>@</mark> p	Av. D = 0.49 s, k = 5, p = 0.125
Unanticipatable vs. Anticipatable take-over		0 🚳	Av. D = -0.54 s, k = 4, p = 0.125
1st take-over vs. 2nd take-over			Av. D = -1.00 s, k = 6, p = 0.031
Smaller vs. Larger time budget		000	<b>O</b> Av. D = 1.35 s, k = 10, p = 0.004
Shorter vs. Longer automated driving duration		<b>D</b>	Av. D = -0.26 s, k = 4, p = 0.625
Without vs. With TOR		0 🚳	-Av. D = -0.58 s, k = 6, p = 0.031
Visual vs. Auditory TOR	- o	്വത്ത	Av. D = -1.41 s, k = 5, p = 0.125
Visual vs. Vibrotactile TOR	- o	Do	Av. D = -1.41 s, k = 5, p = 0.063
Auditory vs. Visual-auditory TOR			Av. D = 0.11 s, k = 4, p = 0.875
Auditory vs. Vibrotactile TOR		<b>D</b>	Av. D = 0.19 s, k = 5, p = 0.188
Auditory vs. Auditory-vibrotactile TOR		<b>o</b> 👼 o	Av. D = -0.14 s, k = 7, p = 0.297
Vibrotactile vs. Auditory-vibrotactile TOR			Av. D = -0.35 s, k = 4, p = 0.125
Visual-auditory vs. Visual-auditory-vibrotactile TOR		oœo	Av. D = -0.11 s, k = 7, p = 0.656
Non-directional vs. Directional TOR		<b>o</b>	Av. D = -0.04 s, k = 5, p = 0.625
Without vs. With peripheral visual stimuli		<b>d</b>	Av. D = -0.15 s, k = 4, p = 0.250
-	6 -4	-2 0 2 4	6
Γ	Difference	in mean take-over time	(D) (s)

Figure 2.3: Within-study effects. The circles represent the difference in mean take-over time between two conditions (D) of a particular study. A positive D indicates that a larger mean TOT was obtained from the latter condition compared to the former condition. The large square markers represent the average D in the category. k represents the number of studies (D values) in the category. p is the p-value from a two-sided signed rank test for the hypothesis that the D values come from a distribution having a median of 0. TOR = Take-over request, NDT = Non-driving task. A directional TOR is a TOR that is informative about the location of the hazard. Peripheral visual stimuli are stimuli that indicate the status of the automation or the environment (e.g., using ambient LEDs).

#### 2.3.3 Between-study analysis

The correlations between the 18 study variables and the mean TOT are shown in Table 2.2. As in the within-study analysis, urgency of the take-over scenario and holding a handheld device showed strong correlations with the mean TOT: higher-urgency levels and shorter time budgets were associated with lower mean TOT (URG: r = -.44;  $\rho = -.42$ ; TBTC: r = .53,  $\rho = .43$ ; TBTB: r = .73,  $\rho = .31$ ), and holding a handheld device (HAND) yielded higher mean TOT (r = .30,  $\rho = .35$ ). The correlations between mean TOT and other categorical variables were weak to moderate, with absolute values below 0.3. That is, the mean TOT did not substantially correlate with the modality of the TOR or the type of non-driving task.

	Corre	Correlation				
	with	mean				
Study	TOT					
variable					Average	SD
, unuore	r	0	Study variable conditions	n	mean	mean
		F			TOT (s)	TOT
					101(0)	(s)
1. AGE	.22	.24	-	485	-	-
	_		0 (low fidelity)	81	2.67	2.39
2. SIM	04	.02	1 (medium fidelity)	268	2.84	1.33
	$\frac{104}{2}$ (hi		2 (high fidelity)	171	2.57	0.95
3 LAD	15	19	0 (L2) 67 1 (L3 and above) 45		2.14	1.17
5. EAD	.15	.17	1 (L3 and above)	458	2.80	1.47
4.	04	08	0 (no visual TOR)		2.63	1.75
TOR_V	.04	.00	1 (with visual TOR)	360	2.77	1.29
5.	12	12	0 (no auditory TOR)	84	2.33	1.00
TOR_A	.12	.12	1 (with auditory TOR)	436	2.80	1.51
6.	-	10	0 (no vibrotactile TOR)	447	2.79	1.50
TOR_VT	.11	10	1 (with vibrotactile TOR)	73	2.35	0.99
7.	06	0.0	0 (no TOR)	34	2.40	1.08
TOR_P	.06	.06	1 (TOR present)	486	2.75	1.47
8.	12	12	0 (no visual NDT)	204	2.49	1.12
NDT_V	.13	.13	1 (the NDT is visual)	309	2.89	1.62
9.	-	07	0 (no auditory NDT)	384	2.75	1.29
NDT_A	.03	07	1 (the NDT is auditory)	129	2.67	1.88
10.	-	0.4	0 (no motoric NDT)	289	2.74	1.27
NDT_M	.01	04	1 (the NDT requires a motoric manoeuvre)	224	2.72	1.67
11.	-		0 (without highly cognitively demanding NDT)	385	2.78	1.28
NDT_C	.05	11	1 (with highly cognitively demanding NDT)	128	2.60	1.90
12.	20		0 (no handheld device)	371	2.54	1.42
HAND	.30	.35	1 (NDT device held in the hands)	108	3.61	1.46
13.			0 (no NDT present at the moment of TOR)	143	2.46	1.17
NDT P	.11	.11	1 (NDT present at the moment of TOR)	377	2.82	1.53
			0 (low urgency)	83	3.95	2.35
14. URG		42	1 (medium urgency)	114	3.03	1.32
	.44		2 (high urgency)	295	2.25	0.81
			0 (low response complexity)	108	3.43	2.21
15 DRE	-	07	1 (medium response complexity)	134	2.34	1 16
10. DILL	.16	.07	2 (high response complexity)	253	2.66	1.10
			0 (no interaction with other road user)	344	2.60	1.51
16. IRU	.08	.14	1 (interaction with other road user)	141	2.07	1.33
17			1 (interaction with other road user)	240	2.75	1.27
TBTC	.53	.43	—	240		
18				160		
TBTR	.73	.31	—	100		
	<b>m</b> 1					

#### Table 2.2: Correlations between the study variables and mean TOTs in the betweenstudy analysis. Both Pearson product-moment correlations (r) and Spearman rankorder correlations ( $\rho$ ) are reported.

*Note.* TOR = Take-over request, NDT = Non-driving task.

Additionally, we calculated the correlation between the mean and the standard deviation of the TOTs and found a strong association (Figure 2.4; r = .82;  $\rho = .73$ , n = 397). The correlations between the mean TOTs and the three continuous study variables: AGE, TBTC, and TBTB are depicted in Figure 2.5.

A weak to moderate correlation was observed between sample size and mean TOT (r = .21,  $\rho = .14$ , n = 520), see Figure 2.6. The correlations between sample size and all study variables were weak with an absolute r and  $\rho$  smaller than 0.20, except for the correlation with the time budget to collision (TBTC, r = .26,  $\rho = 27$ , n = 240) and simulator fidelity where  $\rho$  was smaller than -0.20 (SIM; r = .13,  $\rho = .25$ , n = 520). In other words, studies in high-fidelity simulators involved smaller sample sizes than studies in low-fidelity simulators.

Table 2.3 shows the correlations between the study variables, providing insight into the patterns of the experimental design. For a higher level of automation, the studies tended to implement a TOR ( $\rho = .65$ ), instruct the participants to perform a non-driving task ( $\rho = 0.34$ ), and provide longer time budgets to collision ( $\rho = .35$ ). These observations are in accordance with the definition of SAE International (2016).

Concerning the modalities of the non-driving task and TOR, strong positive correlations were found between the presence of a motoric and visual non-driving task ( $\rho = 0.68$ ). Visual and auditory TORs tended to be combined ( $\rho = 0.35$ ), which was not the case for the auditory and vibrotactile modalities ( $\rho = -0.35$ ).

A complex driver response was more likely to be required when the take-over situation was more urgent ( $\rho = 0.52$ ), and when other road users were involved in the take-over process ( $\rho = 0.46$ ). Also, it is worth noting that studies that used higher fidelity simulators tended to employ older participants ( $\rho = 0.38$ ).

Ial	16 2.J:	opear		oruer rai	uk correis	(d) suom	neween	nue preu	lictof va	riables su		1 anie 2.2	•					
	1. AGE	2. SIM	3. LAD	4. TOR_V	5. TOR_A	6. TOR_VT	7. TOR_P	8. NDT_V	9. NDT_A	10. NDT_M	11. NDT_C	12. HAND	13. NDT_P	14. URG	15. DRE	16. IRU	17. TBTC	18. TBTB
1.AGE																		
2.SIM	.38																	
3.LAD	07	.03	ī															
4.TOR_V	.07	.26	.24	ı														
5.TOR_A	.18	.18	.34	.35														
6.TOR_VT	21	30	.13	13	35													
7.TOR_P	.05	.08	.65	.40	.60	.11												
8.NDT_V	12	08	.30	.12	.08	.10	.19											
9.NDT_A	.04	17	.08	-00	04	.07	.04	.03										
10.NDT_M	18	01	.28	.13	.07	.13	.18	.68	21									
11.NDT_C	-00	21	.16	06	.01	60.	.10	.13	.27	.29								
12.HAND	05	.03	.21	60.	.10	.04	.11	.31	14	.24	12							
13.NDT_P	08	08	.34	.08	.08	II.	.22	LL.	.36	.55	.33	.35	ı					
14.URG	27	.01	16	04	14	.10	13	07	02	.12	.13	18	06					
15.DRE	10	.05	.06	00.	01	60.	01	.05	04	.17	.10	.00	60.	.52	,			
16.IRU	.12	.03	.08	.03	.01	06	00 <sup>.</sup>	.01	07	.05	.16	.04	.02	.10	.46	,		
17.TBTC	17	30	.35	.13	.02	.20	.22	60.	.01	.02	03	.21	.17	61	12	01		
18.TBTB	24	.08	.22	19	12	.13	.01	.12	03	.21	06	.28	.16	33	.03	01	N/A	

Tahla 3 3: Snaarman-order rank correlations (a) hetween the nredictor variables shown in Tahla 3 3

41



Figure 2.4 Scatter plot of the standard deviation of the take-over time (SD TOT) as a function of the mean take-over time (mean TOT), with a fitted least-squares regression line (n = 397).



Figure 2.5 Scatter plot of the mean take-over time (mean TOT) as a function of a) mean age of the participant group (n = 485), b) take-over time budget to collision (TBTC, n = 240), c) take-over time budget to other boundaries (TBTB, n = 160), with a fitted least-squares regression line. A TBTB of 0 s means that the automation was deactivated at the moment of the take-over stimulus.



Figure 2.6 Scatter plot of the sample size of the participant group as a function of the corresponding mean take-over time (mean TOT), with a fitted least-squares regression line (n = 520). The vertical line represents the grand mean TOT.

#### 2.3.4 Linear mixed-effects model

The goodness of fit indicators of the linear mixed-effects model are shown in Table 2.4. Table 2.5 shows the estimation results where effects for most study variables were strongly statistically significant (i.e., low p-values). The model predicting "ln(TOT)" (i.e., the logarithm of the mean TOT ) is specified according to Eq. (1):

$$ln(\text{TOT}) = \alpha + \beta_{\text{LAD}} \cdot \text{LAD} + \beta_{\text{TOR}_{A}} \cdot \text{TOR}_{A} + \beta_{\text{TOR}_{VT}} \cdot \text{TOR}_{VT} + \beta_{\text{NDT}_{V}} \cdot \text{NDT}_{V} + \beta_{\text{MissNDT}_{V}} \cdot \text{MissNDT}_{V} + \beta_{\text{HAND}} \cdot \text{HAND} + \beta_{\text{MissHAND}} \cdot \text{MissHAND} + \beta_{\text{URG}_{\text{High}}} \cdot \text{URG}_{\text{High}} + \beta_{\text{URG}_{\text{Med}}} \cdot \text{URG}_{\text{Med}} + \beta_{\text{MissURG}} \cdot \text{MissURG} + \beta_{\text{IRU}} \cdot \text{IRU} + \beta_{\text{MissIRU}} \cdot \text{MissIRU} + \gamma \cdot \vartheta + \sigma \cdot \varepsilon$$
(1)

where  $\alpha$  is the intercept,  $\beta_{LAD}$ ,  $\beta_{TOR_A}$ ,  $\beta_{TOR_{VT}}$ ,  $\beta_{NDT_V}$ ,  $\beta_{HAND}$ ,  $\beta_{URG_{High}}$ ,  $\beta_{URG_{Med}}$ ,  $\beta_{IRU}$  are with the parameters associated the study variables listed Table in 2.5. $\beta_{\text{MissNDT}_V}$ ,  $\beta_{\text{MissHAND}}$ ,  $\beta_{\text{MissURG}}$ ,  $\beta_{\text{MissIRU}}$  are parameters associated with the dummy variables indicating the missing values,  $\gamma$  is the parameter associated with the study-specific error term  $\vartheta \sim N(0,1)$ , and  $\sigma$  is the parameter associated with the observation-specific error term  $\varepsilon \sim N(0,1)$ . The study-specific error term captures between-study variance and the observationspecific error term captures residual variance between observations. We selected the study variables based on statistical significance. The other study variables listed in Table 2.1 did not significantly influence the mean TOT.

Studies carried out with high levels of automation ( $\geq$  SAE level 3) showed longer mean TOTs than studies with partial automation (SAE Level 2). The fidelity of the driving simulator did not significantly influence mean TOTs (i.e., was not a sufficiently strong predictor to be included in the model as a predictor variable). Auditory and vibrotactile TOTs were associated with shorter mean TOTs, whereas visual warnings did not have a statistically significant impact on mean TOTs. Participants were slower in taking over control when they were engaged in

visual non-driving tasks. Other types of non-driving tasks (auditory, motoric, and cognitive) did not significantly influence mean TOTs. Participants showed longer mean TOTs when holding a handheld device. Take-over situations with high and medium levels of urgency were related to shorter TOTs than situations with a low level of urgency. Finally, taking into account one or more other road users was associated with a longer mean TOT as compared to when no road users were driving in the vicinity.

Table 2.4: Characteristics of the li	inear mixed-effects model
--------------------------------------	---------------------------

Number of parameters	14
Number of studies	129
Number of observations	520
-2 Restricted log likelihood	132.2
Akaike's Information Criterion (AIC)	136.2
Schwarz's Bayesian Criterion (BIC)	144.6

# Table 2.5: Results of the linear mixed-effects model. The parameters represent the unconditional marginal effects of the study variables on the logarithm of the mean TOT.

Variable	Description	Parameter	Estimate	df	t	р
	Intercept	α	1.06	203.39	11.12	9.39*10 <sup>-23</sup>
LAD	Equal to 1 when the level of automated driving (SAE International, 2016) is L3 or above as reported by the authors of the paper	$\beta_{LAD}$	0.285	140.51	3.11	2.24*10 <sup>-3</sup>
TOR <sub>A</sub>	Equal to 1 when the TOR contains an auditory stimulus	$\beta_{TOR_A}$	-0.213	451.50	-5.29	1.89*10 <sup>-7</sup>
TOR <sub>VT</sub>	Equal to 1 when the TOR contains a vibrotactile stimulus	$\beta_{TOR_{VT}}$	-0.180	450.54	-4.00	7.53*10 <sup>-5</sup>
NDT <sub>V</sub>	Equal to 1 when the NDT is visual	$\beta_{NDT_V}$	0.0975	450.68	3.32	9.56*10 <sup>-4</sup>
MissNDT <sub>V</sub>	Equal to 1 when it is not mentioned whether the NDT is visual	$\beta_{MissNDT_V}$	-0.155	151.99	-0.62	5.36*10-1
HAND	Equal to 1 when a device is handheld when undertaking the NDT	$\beta_{HAND}$	0.231	461.09	6.32	6.22*10 <sup>-10</sup>
MissHAND	Equal to 1 when it is not mentioned whether a device is handheld when undertaking the NDT	$\beta_{MissHAND}$	-0.0559	507.00	-0.63	5.29*10 <sup>-1</sup>

URG <sub>High</sub>	Equal to 1 when there is an immediate risk of collision (time budget $\leq 8$ s), or the participants were instructed to react to a stimulus as quickly as possible.	$\beta_{URG_{High}}$	-0.466	491.88	-8.14	3.17*10 <sup>-15</sup>
URG <sub>Med</sub>	Equal to 1 when there is potential collision risk or disturbance to other road users if no response is made, or a medium time budget (between 8 s and 15 s).	$\beta_{URG_{Med}}$	-0.219	506.95	-3.62	<b>3.21</b> *10 <sup>-4</sup>
MissURG	Equal to 1 when it is not mentioned whether there is an immediate risk	$\beta_{Missurg}$	-0.142	145.68	-0.77	4.44*10-1
IRU	Equal to 1 when the participants had to take into account one or more other road users when choosing their optimal take-over action.	$\beta_{IRU}$	0.182	502.89	4.57	6.13*10 <sup>-6</sup>
MissIRU	Equal to 1 when it is not mentioned whether the driver had to take into account one or more other road users when choosing their optimal take- over action.	$\beta_{MissIRU}$	-0.0220	336.49	-0.18	8.59*10 <sup>-1</sup>
Error term	Description	Parameter	Estimate		Wald- Z	р
θ <sub>s</sub>	Study-specific error term (between-study variance)	γ	0.136		6.89	5.40*10 <sup>-12</sup>
ε <sub>s</sub>	Observation-specific error term (between-observation variance)	σ	0.0375		13.73	6.73*10 <sup>-43</sup>

p < 0.05 is indicated in boldface

The model coefficients in Table 2.5 are defined on a logarithmic scale, which enhances the model fit, but complicates the interpretation. To illustrate the impact of the study variables on the mean TOT in seconds, we used the linear mixed-effects model to calculate the mean TOT in a baseline observation and the mean TOT where one variable was changed while keeping all the other variables fixed. In the baseline observation, the level of automation was high ( $\geq$  SAE level 3), the TOR was auditory, the NDT was visual, and the level of urgency was high. In addition, drivers did not use a handheld device and did not have to take into account other road users. These baseline values were selected because they represent the majority of the conditions available. The impact of the study variables on the mean TOTs is shown in Table 2.6. The level

of urgency, holding a device in the hands, and the use of an auditory TOR had the largest impact on the mean TOTs.

Table 2.6: Effect of the study variables on the baseline TOTs (average baseline TOT = 2.15 s).

Variable	Estimated mean TOT (s)
LAD = 0	1.62
$TOR_A = 0$	2.66
$TOR_VT = 1$	1.80
$NDT_V = 0$	1.95
Hand = 1	2.71
$URG\_Low = 1$	3.43
URG_Med =1	2.75
IRU = 1	2.58

#### 2.4 Discussion

#### 2.4.1 Findings from this Meta-Analysis

This meta-analysis quantified the determinants of mean take-over time (TOT) as observed in 129 experiments using three complementary approaches: a within-study analysis, a betweenstudy analysis, and a linear mixed-effects model. The within-study analysis provides a synthesis of causal experimental effects (Figure 2.3). The between-study analysis is based on correlations that involve hundreds of mean TOT values (Table 2.2), and may, therefore, feature higher generalizability than the within-study analysis. However, the between-study analysis is a synthesis of correlations rather than causal effects, and may therefore be susceptible to various confounding factors. The linear mixed-effects model is statistically powerful because it uses the dataset of the between-study analysis while taking into account whether the mean TOT values were obtained from the same study. Although all models are wrong if taken literally (Box, 1976), we would argue that our three-fold complementary approach provides a good picture of the current take-over literature.

Several main findings stand out from the three meta-analysis methods. First, the urgency of the situation, defined in terms of (1) the hand-coded urgency level (URG), (2) time budget to collision with an obstacle (TBTC), and (3) time budget to boundaries (TBTB), has substantial associations with the mean TOT. In other words, if more time is available, drivers use more time to take over. This observation is consistent with previous reviews (see Section 2.1.2) and can be interpreted using a literature review by Summala (2000) on brake reaction times in manual driving. Summala argued that a distinction exists between drivers' ability to intervene quickly and their motivation to intervene, and explained that "it is not always necessary to react as soon as possible". If there is sufficient time, drivers do not take-over as quickly as they can, but first assess the situation (e.g., by checking the mirrors) (Gold, Damböck et al., 2013) and resume an optimal driving posture (e.g., by adjusting the seating position) (Zhang et al., 2019) before taking over.

The second finding is that performing a non-driving task with a handheld device strongly increases the mean TOT, as confirmed by each of the three analyses. Among the studies without a handheld device, performing a visual NDT yielded a moderate increase in mean TOT as compared to not performing such a task. The mixed-effects model confirmed that engagement

in a visual NDT increased mean TOTs. The other modalities of the NDT, that is, whether the task demand is auditory or cognitive, did not show significant associations with the mean TOTs. Zhang et al. (2019) examined driver perception and movement response times during take-over process, and found that physically switching arm posture from the current NDT to the driving control task requires more time than perceiving and cognitively processing the take-over stimuli, especially when the arm movement amplitude is high (cf. Fitts, 1954).

Third, a high level of automation (SAE L3 and above) showed higher mean TOTs compared to partial automation (SAE L2), possibly due to a combined effect of a longer time budget, lower urgency, and more involvement in (handheld) NDTs, consistent with the definition provided by the SAE International (2016).

Fourth, as shown in the within-study analysis, prior experience with taking over has a strong effect: drivers responded about 1 second faster if the take-over scenario occurred the second time compared to the first time. The majority of the included studies (92 out of 129) used a within-subject design. In a review about brake response times in manual driving, Green (2000) pointed out that in most studies, participants performed multiple trials to generate more data. The repeated trials would contribute to shorter response times, which calls for caution when interpreting the results. Our meta-analysis also showed that drivers responded about 0.5 seconds faster when the TOR could be anticipated from task-related or environmental cues. This finding is in line with publications indicating that expectancy is an important factor influencing brake response times (Green 2000; Warshawsky-Livne & Shinar, 2002; Young & Stanton, 2007).

Fifth, visual-only TORs showed longer mean TOTs than auditory or vibrotactile TORs. The mixed-effects model further showed that auditory and vibrotactile TORs reduce the mean TOTs as compared to when such TORs are not present or the TOR is visual only. Petermeijer, Doubek, and De Winter (2017) argued that a visual-only warning is not suitable as a TOR, as drivers may overlook a visual signal (especially if they are performing a visually distracting NDT) or may not interpret a visual signal as urgent. Auditory warnings, on the other hand, are well established due to their omnidirectional characteristics (Bazilinskyy & De Winter, 2015). Vibrotactile TORs are effective as well, as they can attract the driver's attention when the driver is performing a visual or auditory NDT (Petermeijer et al., 2016).

Sixth, we found no clear effect of age in the within-study analysis or the multi-level model, which is interesting because age is known to be associated with a slower speed of processing (e.g., Salthouse, 2009). One possible explanation is that TOTs largely reflect motivational processes, not biological limitations, as pointed out above. For example, although older drivers have a slower simple reaction time, they could have a more cautious driving style and are likely to take over quickly even when not strictly necessary (Körber, Gold, Lechner, & Bengler, 2016). Compensatory behaviours, such as a less intensive involvement in NDTs, may also alleviate ageing effects (Clark & Feng, 2017). Furthermore, the positive correlation between age and mean TOT in the between-study analysis may point to a confounding effect, where older drivers have participated in different types of experiments, as discussed below. We argue that the lack of observed age effects in the within-study analysis is not due to range restrictions, as the differences in mean age for the six included studies were substantial (23 vs. 67 years, 20 vs. 70 years, 34 vs. 60 years, 18 vs 70 years, 18 vs. 37 years, 26 vs. 71 years). It has been recommended that future take-over studies include even older drivers, above 80 years of age (e.g., Körber et al., 2016, Li, Blythe, Guo, & Namdeo, 2018). Additionally, we would recommend that future research on the effect of biological age on TOT should try to obtain a more in-depth understanding by examining the effects of covariates, such as years of driving experience, psychometric performance (e.g., simple, reaction time, perceptual speed), and sensation seeking scores.

Finally, we found a moderate effect of surrounding traffic (IRU) on the mean TOT. This can be explained by the fact that drivers, in case of surrounding traffic, need time for visual scanning and situation assessment before taking over. However, we found that a more complex driver response (i.e., higher DRE) was associated with a shorter mean TOT in the between-study analysis. This counterintuitive finding could be due to the strong positive correlation between DRE and urgency. In other words, although complex responses require cognitive processing time (see e.g., Gold, Naujoks et al., 2018, and Green, 2000 for discussion in manual driving context), such responses are more likely to be performed in urgent situations, which are associated with lower mean TOTs.

#### 2.4.2 Limitations

The current study was performed using mean TOTs. In the end, collision risk is not determined by the mean TOT, but by outliers in the TOT distribution (Horrey & Wickens, 2007). We found that the mean and standard deviation of TOT are highly correlated (r = .82;  $\rho = .73$ ), indicating that mean TOTs are informative about the tail of the TOT distribution. However, we note that accidents may be due to extreme values (e.g., TOTs that exceed the 99.999th percentile, such as due to a driver being asleep behind the wheel). Our meta-analysis is not suitable for making inferences about crash likelihood or for proposing generic guidelines about what time budget constitutes safety.

A second limitation is that this meta-analysis investigated take-over *time*, not take-over *quality*. A number of studies found fast but also hazardous responses (severe braking or steering) under higher mental workload and short time budgets (e.g., Gold, Damböck et al., 2013; Clark & Feng, 2017; Ito, Takata, & Oosawa, 2016). Put differently, a short TOT does not necessarily indicate a safe situation, but could actually be a sign of hazard, because short TOTs typically occur in urgent situations for which an evasive manoeuvre may be needed. We found that directional TORs (i.e., TORs that are informative about the location of the hazard) have no beneficial effects on mean TOT, but this does not imply that directional TORs are ineffective, as they could be useful to enhance take-over quality (e.g., to enhance situation awareness). We recommend that researchers publish not only the mean and standard deviation of the TOT, but also provide data files with TOT values per event. This would allow meta-analysts to make inferences about the TOT distribution. Additional response variables, such as minimum time to collision and maximum longitudinal/lateral acceleration would enable the assessment of take-over quality.

Third, the starting moment of the take-over response is a source of ambiguity (Liu & Green, 2017). While some researchers used criteria such as "the moment the driver gave an input either on the pedals or the steering wheel" (Payre, Cestac, Dang, Vienne, & Delhomme, 2017), other researchers provided exact criteria. For example, Gold, Damböck et al. (2013) adopted a 2 degree steering angle or 10% brake pedal position, which was employed in a number of subsequent studies (Feldhütter, Gold, Schneider, & Bengler, 2017; Gold et al., 2015; Gold, Körber, Lechner, & Bengler, 2016; Gold et al., 2014; Gold, Lorenz, Damböck, & Bengler, 2013; Kerschbaum et al., 2015; Körber et al., 2016; Radlmayr et al., 2014). Somewhat different criteria can be found in other studies, such as absolute steering acceleration larger than 5 deg/s<sup>2</sup> (Zeeb, Buchner, & Schrauf, 2015, 2016). Another issue is that the TBTC posed an upper limit to the TOTs that could be observed; if a participant would not react at all (which sometimes

happened, see Gold, Lorenz et al., 2013; Young and Stanton, 2007), their results were not taken into account in the reported mean TOT, which would underestimate the mean TOT.

Fourth, although a large number of study variables were investigated, there may still be unobserved study variables that affect TOT. Examples of hidden moderators are driving speed, the intensity of the TOR, and the state of the operator (e.g., whether he or she is fatigued or impaired by alcohol, see Wiedemann et al., 2018). Hidden moderators may also explain why the study-specific error term is strong in the mixed-effects model.

Fifth, nearly all included studies were conducted in a driving simulator. Despite advantages such as controllability and safety, driving simulators have limited fidelity, which raises the issue about behavioural validity (De Winter, Van Leeuwen, & Happee, 2012; Green, 2000; Kaptein, Theeuwes, & Van der Horst, 1996; Riener, 2010; Risto & Martens, 2014). Also, the driver's level of perceived risk perception may be low in simulators as compared to on-road conditions (Carsten & Jamson, 2011), which could discourage a fast take-over response. While the TOTs measured in simulator studies may not accurately reflect the numeric values of the TOTs in the real world, the results may still be valid concerning the direction of the effects (Kaptein et al., 1996).

Sixth, although our meta-analysis is much more comprehensive compared to previous reviews on the same topic (see Introduction), the within-study analysis would still benefit from a larger sample of studies. The Ds in Figure 2.3 are based on 4 to 17 studies. Although each of these individual studies may present credible findings, more studies should be conducted to examine whether the experimental effects are generalizable.

Finally, as in any meta-analysis, there may be sources of bias or confounding effects. In a previous review on automated driving, De Winter et al. (2014) observed a confounder, namely that young participants are overrepresented in lower-fidelity driving simulators. The authors explained that lower-fidelity simulators are available at universities where participants are usually students, whereas companies with high-fidelity simulators tend to recruit middle-age drivers. A similar association between simulator fidelity level and participant age was observed in the between-study meta-analysis ( $\rho = 0.38$ ). We also found studies in lower-fidelity simulators involved larger sample sizes ( $\rho = -0.25$ ), which could be explained because students participate in relatively large amounts, e.g., for course credit. Such confounds affect some of the correlations in the between-study analysis (Table 2.2), but are controlled for in the mixed-effects model. In the between-study analysis, a weak-to-moderate correlation was observed between sample size and mean TOT, which may be related to the confounding effect of simulator fidelity discussed above.

We expect that publication bias regarding mean TOT in the between-study analysis is small as compared to other types of research such as drug trials where researchers and sponsors may favour a positive drug efficacy. That is, we are not aware of a mechanism by which the mean TOT would affect the likelihood of publication. The scatter plot of sample size vs. mean TOT (Figure 2.6) showed no characteristic funnel shape, likely because the observed spread in mean TOT reflects study heterogeneity rather than imprecision of the mean TOT values.

Regarding the within-study analysis, where we assessed differences in mean TOT (D), publication bias is possible but not evident from our findings. For example, innovative types of TORs where publication bias may be expected (e.g., directional and peripheral TORs) showed near-zero effects (Figure 2.3), thus indicating that small (null) effects were published. The number of studies per category of the within-subject analysis was too small to create funnel plots or perform a formal test of publication bias.

#### 2.4.3 Conclusion and Recommendations for Future Research

The meta-analysis included 129 studies that measured driver TOTs when resuming manual control after automated driving and investigated the effect of multiple factors related to the driver, the automation system, the human-machine interface, the driving situation, and the experimental setup. Notable findings are that the available time, a lack of experience with TORs, and using a handheld device were associated with substantially increased mean TOT. Although providing a take-over request yields a lower mean TOT than no take-over request at all (or a visual-only take-over request), the modality of the TORs had relatively minor effects on the mean TOT.

These findings have important implications for future research and design. In particular, instead of designing new types of take-over requests that may have only incremental effects on mean TOT, efforts could be made towards ensuring that drivers are prepared and trained to take over. Also, drivers should not be permitted to engage in handheld non-driving tasks if take-over situations can be urgent. Conducting non-driving tasks on a mounted (head-up) display could be a safer option in such cases.

Finally, our meta-analysis suggests that achieving a low mean TOT should not necessarily be a design target. We showed that drivers take more time (i.e., the mean TOT is higher) when they have more time (i.e., when the urgency is lower). Future engineering efforts should be directed towards ensuring that drivers actually have sufficient time, which could be done by building better sensors with larger look-ahead time or by using vehicle-to-vehicle communication.

# 2.5 Supplementary material

Supplementary materials are accessible via this link: <u>https://doi.org/10.4121/uuid:75c28abe-6559-4273-85f4-927e969c1c59</u>

#### Acknowledgement

The authors were involved in the Marie Curie Initial Training Network (ITN) project HFAuto – Human Factors of Automated Driving (PITN-GA-2013-605817).

#### References

References marked with an asterisk indicate studies included in the meta-analysis.

Bazilinskyy, P., & De Winter, J. C. F. (2015). Auditory interfaces in automated driving: an international survey. *PeerJ Computer Science*, *1*, e13. doi:10.7717/peerj-cs.13

\*Befelein, D., Boschet, J., & Neukum, A. (2018). *Influence of non-driving-related tasks' motivational aspects and interruption effort on driver take-over performance in conditionally automated driving*. Paper presented at the 6th International Conference on Driver Distraction and Inattention, Gothenburg, Sweden. Retrieved from: http://ddi2018.org/wp-content/uploads/2018/10/S9.3-Befelein.pdf

\*Befelein, D., Naujoks, F., & Neukum, A. (2016). *How do naturalistic non-driving-related tasks influence Take-over time and quality during highly automated driving*. Poster session presented at Human Factors and Ergonomics Society Europe Chapter 2016 Annual Conference. Prague, Czech Republic.

Begg, C. B., & Mazumdar, M. (1994). Operating characteristics of a rank correlation test for publication bias. *Biometrics*, *50*, 1088–1101. doi:10.2307/2533446

\*Belderbos, C. A. G. (2015). Authority transition interface: A human machine interface for taking over control from a highly automated truck (MSc thesis). Delft University of Technology, Delft, The Netherlands.

\*Belwadi, A., Loeb, H., Shen, M., Shaikh, S., & McIntosh, C. W. (2018). *Emergency autonomous to manual takeover in a driving simulator: teen vs. adult drivers–a pilot study* (No. 2018-01-0499). SAE Technical Paper. doi:10.4271/2018-01-0499

\*Berghöfer, F., Purucker, C., Naujoks, F., Wiedemann, K., & Marberger, K. (2018). *Prediction of Take-Over Time Demand in Highly Automated Driving. Results of a Naturalistic Driving Study.* Poster session presented at the Human Factors and Ergonomics Society Europe Chapter 2018 Annual Conference.

\*Biedermann, A., & Cieler, S. (2017). Bedeutung und Herausforderungen der Fahrerzustandserkennung im Kontext des vollautomatisierten Fahrens [Importance and challenges of driver condition detection in the context of fully automated driving]. In *VDI-Tagung Der Fahrer im 21. Jahrhundert 2017, Der Mensch im Fokus technischer Innovationen* (pp. 91–106). Braunschweig, DE: VDI-Gesellschaft Fahrzeug- und Verkehrstechnik.

\*Blommer, M., Curry, R., Kochhar, D., Swaminathan, R., Talamonti, W., & Tijerina, L. (2015). The effects of a scheduled driver engagement strategy in automated driving. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *59*, 1681–1685. doi:10.1177/1541931215591363

\*Blommer, M., Curry, R., Swaminathan, R., Tijerina, L., Talamonti, W., & Kochhar, D. (2017). Driver brake vs. steer response to sudden forward collision scenario in manual and automated driving modes. *Transportation research part F: traffic psychology and behaviour*, *45*, 93-101. doi:10.1016/j.trf.2016.11.006

Bobko, P., Roth, P. L., & Buster, M. A. (2007). The usefulness of unit weights in creating composite scores: A literature review, application to content validity, and metaanalysis. *Organizational Research Methods*, *10*, 689–709. doi:10.1177/1094428106294734

Bond, C. F., Jr., Wiitala, W. L., & Richard, F. D. (2003). Meta-analysis of raw mean differences. *Psychological Methods*, *8*, 406–418. doi:10.1037/1082-989X.8.4.406

Bonett, D. G. (2008). Meta-analytic interval estimation for bivariate correlations. *Psychological Methods*, *13*, 173–181. doi:10.1037/a0012868

Box, G. E. (1976). Science and statistics. *Journal of the American Statistical Association*, 71, 791–799. doi:10.2307/2286841

\*Braunagel, C., Rosenstiel, W., & Kasneci, E. (2017). Ready for take-over? A new driver assistance system for an automated classification of driver take-over readiness. *IEEE Intelligent Transportation Systems Magazine*, *9*, 10–22. doi:10.1109/MITS.2017.2743165

\*Bueno, M., Dogan, E., Selem, F. H., Monacelli, E., Boverie, S., & Guillaume, A. (2016). *How different mental workload levels affect the take-over control after automated driving.* Paper presented at 2016 IEEE 19th International Conference on Intelligent Transportation Systems (ITSC) (pp. 2040–2045), Rio de Janeiro, Brazil. doi:10.1109/ITSC.2016.7795886

\*Çapalar, J., & Olaverri-Monreal, C. (2017). *Hypovigilance in limited self-driving automation: Peripheral visual stimulus for a balanced level of automation and cognitive workload.* Paper presented at 2017 IEEE 20th International Conference on Intelligent Transportation Systems (ITSC) (pp. 27–31). IEEE.

Carsten, O., & Jamson, A. H. (2011). Driving simulators as research tools in traffic psychology. In B. E. Porter (Ed.), *Handbook of traffic psychology* (pp. 87–96). Academic Press. doi:10.1016/B978-0-12-381984-0.10007-4

\*Clark, H., & Feng, J. (2017). Age differences in the takeover of vehicle control and engagement in non-driving-related activities in simulated driving with conditional automation. *Accident Analysis & Prevention*, *106*, 468–479. doi:10.1016/j.aap.2016.08.027

\*Cohen-Lazry, G., Borowsky, A., & Oron-Gilad, T. (2017). The effects of continuous drivingrelated feedback on drivers' response to automation failures. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *61*, 1980–1984. doi:10.1177/1541931213601974

\*Cohen-Lazry, G., & Katzman, N. (2018). Ipsilateral Versus Contralateral Tactile Alerts for Take-Over Requests in Highly-Automated Driving. Paper presented at the 6th Humanist Conference, The Hague, Netherlands. Retrieved from http://www.humanist-vce.eu/fileadmin/contributeurs/humanist/TheHague2018/1-cohen-lazry.pdf

Cronbach, L. J. (1975). Beyond the two disciplines of scientific psychology. *American Psychologist*, *30*, 116–127. doi:10.1037/h0076829

\*Cui, W., Zhou, R., Yan, Y., Ran, L., & Zhang, X. (2017). Effect of warning levels on drivers' decision- making with the self-driving vehicle system. In N. A. Stanton (Ed.), *Advances in human aspects of transportation* (pp. 720–729). Cham: Springer. doi:10.1007/978-3-319-60441-1\_69

\*Damböck, D., Weissgerber, T., Kienle, M., & Bengler, K. (2013). *Requirements for cooperative vehicle guidance*. Paper presented at the 16th International IEEE Conference on Intelligent Transportation Systems-(ITSC) (pp. 1656–1661). The Hague, Netherlands. doi:10.1109/ITSC.2013.6728467

Deeks, J. J., Macaskill, P., & Irwig, L. (2005). The performance of tests of publication bias and other sample size effects in systematic reviews of diagnostic test accuracy was assessed. *Journal of clinical epidemiology*, *58*, 882–893. doi:10.1016/j.jclinepi.2005.01.016

De Winter, J. F. C., Van Leeuwen, P. M., & Happee, R. (2012). Advantages and disadvantages of driving simulators: A discussion. In A.J. Spink, F. Grieco, O.E. Krips, L.W.S. Loijens, L.P.J.J. Noldus, & P.H. Zimmerman (Eds.), *Proceedings of Measuring Behavior* (pp. 47–50). Retrieved from: http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.388.1603&rep=rep1&type=pdf#pa ge=47

De Winter, J. C. F., Happee, R., Martens, M. H., & Stanton, N. A. (2014). Effects of adaptive cruise control and highly automated driving on workload and situation awareness: A review of the empirical evidence. *Transportation Research Part F: Traffic Psychology and Behaviour*, 27, 196–217. doi:10.1016/j.trf.2014.06.016

\*De Winter, J. C. F., Stanton, N. A., Price, J. S., & Mistry, H. (2016). The effects of driving with different levels of unreliable automation on self-reported workload and secondary task

performance. International Journal of Vehicle Design, 70, 297–324. doi:10.1504/IJVD.2016.076736

De Winter, J. C., Zadpoor, A. A., & Dodou, D. (2014). The expansion of Google Scholar versus Web of Science: a longitudinal study. *Scientometrics*, *98*, 1547–1565. doi:10.1007/s11192-013-1089-2

\*Dogan, E., Deborne, R., Delhomme, P., Kemeny, A., & Jonville, P. (2014). *Evaluating the shift of control between driver and vehicle at high automation at low speed: The role of anticipation.* Paper presented at the Transport Research Arena (TRA) 5th Conference: Transport Solutions from Research to Deployment, Paris, France.

Egger, M., Smith, G. D., Schneider, M., & Minder, C. (1997). Bias in meta-analysis detected by a simple, graphical test. *BMJ*, *315*, 629–634. doi:10.1136/bmj.315.7109.629

Einhorn, H. J., & Hogarth, R. M. (1975). Unit weighting schemes for decision making. *Organizational Behavior and Human Performance*, 13, 171–192. doi:10.1016/0030-5073(75)90044-6

\*Eriksson, A., Petermeijer, S. M., Zimmerman, M., De Winter, J. C. F., Bengler, K. J., & Stanton, N. (in press). Rolling out the red (and green) carpet: supporting driver decision making in automation to manual transitions. *IEEE Transactions on Human-Machine Systems*.

\*Eriksson, A., & Stanton, N. A. (2017). Takeover time in highly automated vehicles: noncritical transitions to and from manual control. *Human Factors*, *59*, 689–705. doi:10.1177/0018720816685832

\*Feldhütter, A., Gold, C., Schneider, S., & Bengler, K. (2017). How the duration of automated driving influences take-over performance and gaze behavior. In C. M. Schlick, S. Duckwitz, F. Flemisch, M. Frenz, & S. Kuz (Eds.), *Advances in ergonomic design of systems, products and processes* (pp. 309–318). Berlin, Heidelberg: Springer. doi:10.1007/978-3-662-53305-5\_22

\*Feldhütter, A., Hecht, T., & Bengler, K. (2018). Fahrerspezifische Aspekte beim hochautomatisierten Fahren [Driver-specific aspects in highly automated driving]. *FAT Schriftenreihe*.

\*Feldhütter, A., Kroll, D., & Bengler, K. (2018). *Wake up and take over! The effect of fatigue on the take-over performance in conditionally automated driving*. Paper presented at the 21th International Conference on Intelligent Transportation Systems (ITSC) (pp. 2080–2085). doi:10.1109/ITSC.2018.8569545

\*Feldhütter, A., Segler, C., & Bengler, K. (2017). Does shifting between conditionally and partially automated driving lead to a loss of mode awareness? In N. A. Stanton (Ed.), *Advances in human aspects of transportation* (pp. 730–741). Cham: Springer. doi:10.1007/978-3-319-60441-1\_70

Fitts, P. M. (1954). The information capacity of the human motor system in controlling the amplitude of movement. *Journal of Experimental Psychology*. 47, 381–391. doi:10.1037/h0055392

Fu, R., Gartlehner, G., Grant, M., Shamliyan, T., Sedrakyan, A., Wilt, T. J., ... & Santaguida, P. (2011). Conducting quantitative synthesis when comparing medical interventions: AHRQ

and the Effective Health Care Program. *Journal of clinical epidemiology*, 64, 1187–1197. doi:10.1016/j.jclinepi.2010.08.010

\*Funkhouser, K., & Drews, F. (2016). Putting the brakes on autonomous vehicle control: responding to system breakdowns. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 60, 1859–1863. doi:10.1177/1541931213601424

Gold, C., & Bengler, K. (2014). *Taking over control from highly automated vehicles*. Paper presented at the 5th International Conference on Applied Human Factors and Ergonomics (pp. 3662–3667).

\*Gold, C., Berisha, I., & Bengler, K. (2015). Utilization of Drivetime – Performing non-driving related tasks while driving highly automated. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *59*, 1666–1670. doi:10.1177/1541931215591360

\*Gold, C., Damböck, D., Lorenz, L., & Bengler, K. (2013). "Take over!" How long does it take to get the driver back into the loop? *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *57*, 1938–1942. doi:10.1177/1541931213571433

\*Gold, C., Körber, M., Lechner, D., & Bengler, K. (2016). Taking over control from highly automated vehicles in complex traffic situations the role of traffic density. *Human Factors*, *58*, 642–652. doi:10.1177/0018720816634226

\*Gold, C., Lorenz, L., & Bengler, K. (2014). *Influence of automated brake application on takeover situations in highly automated driving scenarios*. Paper presented at the FISITA 2014 World Automotive Congress, Maastricht, The Netherlands.

\*Gold, C., Lorenz, L., Damböck, D., & Bengler, K. (2013). Partially automated driving as a fallback level of high automation. *Tagung Fahrerassistenzsysteme. Der Weg zum automatischen Fahren*, 6.

Gold, C., Happee, R., & Bengler, K. (2018). Modeling take-over performance in level 3 conditionally automated vehicles. *Accident Analysis & Prevention*, *116*, 3–13. doi:10.1016/j.aap.2017.11.009

Gold, C., Naujoks, F., Radlmayr, J., Bellem, H., & Jarosch, O. (2018). Testing scenarios for human factors research in level 3 automated vehicles. In N. A. Stanton (Ed.), *Advances in human aspects of transportation*, 597, 551–559. Cham: Springer. doi:10.1007/978-3-319-60441-1\_54

Green, M. (2000). "How long does it take to stop?" Methodological analysis of driver perception-brake times. *Transportation Human Factors*, 2, 195–216. doi:10.1207/STHF0203\_1

\*Helldin, T., Falkman, G., Riveiro, M., & Davidsson, S. (2013). *Presenting system uncertainty in automotive UIs for supporting trust calibration in autonomous driving*. Paper presented at the 5th International Conference on Automotive User Interfaces and Interactive Vehicular Applications (pp. 210–217), Eindhoven, The Netherlands. doi:10.1145/2516540.2516554

Hedges, L. V., & Vevea, J. L. (1998). Fixed-and random-effects models in metaanalysis. *Psychological Methods*, *3*, 486–504. \*Hergeth, S., Lorenz, L., & Krems, J. F. (2017). Prior familiarization with takeover requests affects drivers' takeover performance and automation trust. *Human Factors*, *59*, 457–470. doi:10.1177/0018720816678714

Higgins, J. P. T., & Green, S. (Eds.). (2005). Cochrane handbook for systematic reviews of interventions. The Cochrane Collaboration.

\*Homma, R., Wakasugi, T., & Kodaka, K. (2016). *Human-factors issues of transition from an automated driving system to manual driving*. Paper presented at the 23rd ITS World Congress, Melbourne, Australia.

Horrey, W., & Wickens, C. (2007). In-vehicle glance duration: distributions, tails, and model of crash risk. *Transportation Research Record: Journal of the Transportation Research Board*, 2018, 22–28. doi:10.3141/2018-04

\*Ito, T., Takata, A., & Oosawa, K. (2016). *Time required for take-over from automated to manual driving* (No. 2016-01-0158). SAE Technical Paper. doi:10.4271/2016-01-0158

IBM Corporation (2016). IBM SPSS Statistics for Windows, Version 24.0. Armonk, NY: IBM Corp.

Jalali, S., & Wohlin, C. (2012). Systematic literature studies: database searches vs. backward snowballing. *Proceedings of the ACM-IEEE International Symposium on Empirical Software Engineering and Measurement* (pp. 29–38). ACM. doi:10.1145/2372251.2372257

\*Jarosch, O., Kuhnt, M., Paradies, S., & Bengler, K. (2017). It's out of our hands now! Effects of non-driving related tasks during highly automated driving on drivers' fatigue. Paper presented at the Proceedings of the Ninth International Driving Symposium on Human Factors in Driver Assessment Training and Vehicle Design (pp. 319–325). Public Policy Center, University of Iowa. doi:10.17077/drivingassessment.1653

\*Josten, J., Zlocki, A., & Eckstein, L. (2016). Untersuchung der Bewältigungsleistung des Fahrers von kurzfristig auftretenden Wiederübernahmesituationen nach teilautomatischem, freihändigen Fahren [Investigation of drivers' handling performance in sudden take-over situations after semi-automated, hands-off driving]. *FAT Schriftenreihe 289*, Juli 2016.

Kaptein, N., Theeuwes, J., & Van der Horst, R. (1996). Driving simulator validity: Some considerations. *Transportation Research Record: Journal of the Transportation Research Board*, 1550, 30–36. doi:10.3141/1550-05

\*Kerschbaum, P., Lorenz, L., & Bengler, K. (2015). *A transforming steering wheel for highly automated cars*. Paper presented at the Proceedings of the 2015 IEEE Intelligent Vehicles Symposium (IV), Seoul, South Korea (pp. 1287–1292). doi:10.1109/IVS.2015.7225893

\*Kerschbaum, P., Omozik, K., Wagner, P., Levin, S., Hermsdörfer, J., & Bengler, K. (2017). How does a symmetrical steering wheel transformation influence the take-over process? In D. de Waard, A. Toffetti, R. Wiczorek, A. Sonderegger, S. Röttger, P. Bouchner, T. Franke, S. Fairclough, M. Noordzij, & K. Brookhuis (Eds.) *Proceedings of the Human Factors and Ergonomics Society Europe Chapter 2016 Annual Conference, Prague, Czech Republic* (pp. 96–106).

\*Kim, J., Kim, H. S., Kim, W., & Yoon, D. (2018). Take-over performance analysis depending on the drivers' non-driving secondary tasks in automated vehicles. Paper presented at 2018

International Conference on Information and Communication Technology Convergence (ICTC) (pp. 1364–1366). doi:10.1109/ICTC.2018.8539431

\*Kim, H. J., & Yang, J. H. (2017). Takeover requests in simulated partially autonomous vehicles considering human factors. *IEEE Transactions on Human-Machine Systems*, 47, 735–740. doi:10.1109/THMS.2017.2674998

\*Ko, S. M., & Ji, Y. G. (2018). How we can measure the non-driving-task engagement in automated driving: comparing flow experience and workload. *Applied Ergonomics*, 67, 237–245. doi:10.1016/j.apergo.2017.10.009

Körber, M., Baseler, E., & Bengler, K. (2018). Introduction matters: manipulating trust in automation and reliance in automated driving. *Applied Ergonomics*, 66, 18–31. doi:10.1016/j.apergo.2017.07.006

\*Körber, M., Gold, C., Lechner, D., & Bengler, K. (2016). The influence of age on the takeover of vehicle control in highly automated driving. *Transportation Research Part F: Traffic Psychology and Behaviour, 39*, 19–32. doi:10.1016/j.trf.2016.03.002

\*Körber, M., Schneider, W., & Zimmermann, M. (2015). *Vigilance, boredom proneness and detection time of a malfunction in partially automated driving*. Paper presented at the 2015 International Conference on Collaboration Technologies and Systems (CTS), Atlanta, GA (pp. 70–76). doi:10.1109/CTS.2015.7210402

\*Kreuzmair, C., Gold, C., & Meyer, M.-L. (2017). *The influence of driver fatigue on take-over performance in highly automated vehicles*. Paper presented at the 25th International Technical Conference on the Enhanced Safety of Vehicles (ESV), Detroit, MI.

\*Kuehn, M., Vogelpohl, T., & Vollrath, M. (2017). *Takeover times in highly automated driving* (*level 3*). Paper presented at the 25th International Technical Conference on the Enhanced Safety of Vehicles (ESV), Detroit, MI.

\*Langlois, S., & Soualmi, B. (2016). Augmented reality versus classical HUD to take over from automated driving: An aid to smooth reactions and to anticipate maneuvers. Paper presented at the 2016 IEEE 19th International Conference on In Intelligent Transportation Systems (ITSC), Rio de Janeiro, Brazil (pp. 1571–1578). doi:10.1109/ITSC.2016.7795767

\* Lapoehn, S., Dziennus, M., Utesch, F., Kelsch, J., Schieben, A., Dotzauer, M., ... & Köster, F. (2016). Interaction design for nomadic devices in highly automated vehicles. In Weyers, B., & Dittmar, A. (Eds.), *Mensch und Computer 2016 – Workshopband*. Aachen: Gesellschaft für Informatik e.V.. doi:10.18420/muc2016-ws08-0006

\*Lau, C.P., Harbluk, J.L., Burns, P.C., & El-Hage, Y. (2018). The Influence of Interface Design on Driver Behavior in Automated Driving. CARSP: The Canadian Association of Road Safety Professionals, Victoria, BC.

\*Li, S., Blythe, P., Guo, W., & Namdeo, A. (2018). Investigation of older driver's takeover performance in highly automated vehicles in adverse weather conditions. *IET Intelligent Transport Systems*, *12*, 1157-1165. doi:10.1049/iet-its.2018.0104

Liu, K., & Green, P. (2017). The conclusion of a driving study about warnings depends upon how response time is defined. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *61*, 1876–1880. doi:10.1177/1541931213601949

\*Lorenz, L., Kerschbaum, P., & Schumann, J. (2014). *Designing take over scenarios for automated driving How does augmented reality support the driver to get back into the loop?* Paper presented at the Proceedings of the Human Factors and Ergonomics Society Annual Meeting, *58*, 1681–1685. doi:10.1177/1541931214581351

\*Louw, T., Merat, N., & Jamson, H. (2015). Engaging with highly automated driving: to be or not to be in the loop? *Proceedings of the Eighth International Driving Symposium on Human Factors in Driver Assessment, Training and Vehicle Design, Salt Lake City, UT* (pp. 190–196). doi:10.17077/drivingassessment.1570

\*Lu, Z, Zhang, B., Feldhütter, A., Happee, R., Martens, H. M., & De Winter, J. C. F. (2019). Beyond mere take-over requests: The effects of monitoring requests on driver attention, take-over performance, and acceptance. *Transportation Research Part F: Traffic Psychology and Behaviour*, *63*, 22 - 37. doi:10.1016/j.trf.2019.03.018

Lu, Z., Happee, R., Cabrall, C. D. D., Kyriakidis, M., & De Winter, J. C. F. (2016). Human factors of transitions in automated driving: A general framework and literature survey. *Transportation Research Part F: Traffic Psychology and Behaviour, 43*, 183–198. doi:10.1016/j.trf.2016.10.007

\*Manawadu, U. E., Hayashi, H., Ema, T., Kawano, T., Kamezaki, M., & Sugano, S. (2018). *Tactical-Level Input with Multimodal Feedback for Unscheduled Takeover Situations in Human-Centered Automated Vehicles*. Paper presented at 2018 IEEE/ASME International Conference on Advanced Intelligent Mechatronics (AIM) (pp. 634–639). doi:10.1109/AIM.2018.8452227

Martín-Martín, A., Orduna-Malea, E., Thelwall, M., & López-Cózar, E. D. (2018). Google Scholar, Web of Science, and Scopus: A systematic comparison of citations in 252 subject categories. *Journal of Informetrics*, *12*, 1160–1177. doi:10.1016/j.joi.2018.09.002

\*McNabb, J. C. (2017). Warning a distracted driver: smart phone applications, informative warnings and automated driving take-over requests (Doctoral dissertation, Arizona State University).

\*Melcher, V., Rauh, S., Diederichs, F., Widlroither, H., & Bauer, W. (2015). Take-over requests for automated driving. *Procedia Manufacturing*, *3*, 2867–2873. doi:10.1016/j.promfg.2015.07.788

\*Miller, D., Sun, A., Johns, M., Ive, H., Sirkin, D., Aich, S., & Ju, W. (2015). Distraction becomes engagement in automated driving. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 59, 1676–1680. doi:10.1177/1541931215591362

Moher, D., Liberati, A., Tetzlaff, J., & Altman, D. G. (2009). Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. *Annals of Internal Medicine*, *151*, 264–269. doi:10.7326/0003-4819-151-4-200908180-00135

Molenberghs, G., Kenward, M. G., & Verbeke, G. (2009). Discussion of likelihood inference for models with unobservables: another view. *Statistical Science*, *24*, 273–279. doi:10.1214/09-STS277B

\*Molnar, L. J., Pradhan, A. K., Eby, D. W., Ryan, L. H., St. Louis, R. M., Zakrajsek, J., . . . Zhang, L. (2017). Age-related differences in driver behavior associated with automated vehicles and the transfer of control between automated and manual control: a simulator
*evaluation* (Report No. UMTRI 2017-4). Ann Arbor, MI: University of Michigan Transportation Research Institute.

\*Morgan, P. L., Alford, C., Williams, C., Parkhurst, G., & Pipe, T. (2017). Manual takeover and handover of a simulated fully autonomous vehicle within urban and extra-urban settings. In N. A. Stanton (Ed.), *Advances in human aspects of transportation* (pp. 760–771). Cham, Springer. doi:10.1007/978-3-319-60441-1\_73

\*Mueller, A., Ogrizek, M., Bier, L., & Abendroth, B. (2018). *Design concept for a visual, vibrotactile and acoustic take-over request in a conditional automated vehicle during non-driving-related tasks*. Paper presented at the 6th International Conference on Driver Distraction and Inattention, Gothenburg, Sweden. Retrieved from: http://ddi2018.org/wp-content/uploads/2018/10/S2.3-Mueller.pdf

Naujoks, F., Befelein, D., Wiedemann, K., & Neukum, A. (2017). A review of non-drivingrelated tasks used in studies on automated driving. In N. A. Stanton (Ed.), *Advances in human aspects of transportation*. Cham: Springer. doi:10.1007/978-3-319-60441-1\_52

\*Naujoks, F., Höfling, S., Purucker, C., & Zeeb, K. (2018). From partial and high automation to manual driving: Relationship between non-driving related tasks, drowsiness and take-over performance. *Accident Analysis & Prevention*, *121*, 28–42. doi:10.1016/j.aap.2018.08.018

\*Naujoks, F., Mai, C., & Neukum, A. (2014). The effect of urgency of take-over requests during highly automated driving under distraction conditions. In N. Stanton, S. Landry, G. Di Bucchianico, & A. Vallicelli (Eds.), *Advances in human aspects of transportation, 7 (Part I)*, 431–438. Retrieved from: https://www.researchgate.net/publication/264734018\_The\_Effect\_of\_Urgency\_of\_Take-Over\_Requests\_During\_Highly\_Automated\_Driving\_Under\_Distraction\_Conditions

\*Olaverri-Monreal, C., Kumar, S., & Dìaz-Àlvarez, A. (2018). Automated Driving: Interactive Automation Control System to Enhance Situational Awareness in Conditional Automation. Paper presented at 2018 IEEE Intelligent Vehicles Symposium (IV) (pp. 1698–1703). doi:10.1109/IVS.2018.8500367

Osburn, H. G., & Callender, J. (1992). A note on the sampling variance of the mean uncorrected correlation in meta-analysis and validity generalization. *Journal of Applied Psychology*, 77, 115–122. http://dx.doi.org/10.1037/0021-9010.77.2.115

\*Othersen, I. (2016). Vom Fahrer zum Denker und Teilzeitlenker: Einflussfaktoren und Gestaltungsmerkmale nutzerorientierter Interaktionskonzepte für die Überwachungsaufgabe des Fahrers im teilautomatisierten Modus [From driver to thinker and part-time driver: Influencing factors and design features of user-oriented interaction concepts for the driver's monitoring task in semi-automated mode] (Doctoral dissertation, Technische Universität Braunschweig, Braunschweig, Germany).

\*Othersen, I., Petermann-Stock, I., Schoemig, N., & Fuest, T. (2017). Methoden und Wechselwirkung Kognitive Übernahmefähigkeit nach einer pilotierten Fahrt [Methods and interaction cognitive adoption after a pilot drive]. *ATZelektronik*, *12*, 26–31. doi:10.1007/s35658-017-0068-4

\*Payre, W., Cestac, J., Dang, N.-T., Vienne, F., & Delhomme, P. (2017). Impact of training and in-vehicle task performance on manual control recovery in an automated car.

Transportation Research Part F: Traffic Psychology and Behaviour, 46, 216–227. doi:10.1016/j.trf.2017.02.001

\*Payre, W., Cestac, J., & Delhomme, P. (2016). Fully automated driving: Impact of trust and practice on manual control recovery. *Human Factors: The Journal of the Human Factors and Ergonomics Society*, *58*, 229–241. doi:10.1177/0018720815612319

\*Petermann-Stock, I., Hackenberg, L., Muhr, T., & Mergl, C. (2013). Wie lange braucht der Fahrer--eine Analyse zu Übernahmezeiten aus verschiedenen Nebentätigkeiten während einer hochautomatisierten Staufahrt [How long does the driver need - an analysis of take-over times from various non-driving tasks during highly automated traffic jam driving]. *6. Tagung Fahrerassistenzsysteme. Der Weg zum automatischen Fahren.* TÜV SÜD Akademie GmbH.

\*Petermeijer, S. M., Cieler, S., & De Winter, J. C. F. (2017). Comparing spatially static and dynamic vibrotactile take-over requests in the driver seat. *Accident Analysis & Prevention*, *99*, 218–227. doi:10.1016/j.aap.2016.12.001

\*Petermeijer, S. M., Bazilinskyy, P., Bengler, K., & De Winter, J. C. F. (2017). Take-over again: Investigating multimodal and directional TORs to get the driver back into the loop. *Applied Ergonomics*, *62*, 204–215. doi:10.1016/j.apergo.2017.02.023

\*Petermeijer, S. M., Doubek, F., & De Winter, J. C. F. (2017). Driver response times to auditory, visual, and tactile take-over requests: A simulator study with 101 participants. *Proceedings of the IEEE International Conference Systems, Man, and Cybernetics, Banff, AB, Canada.* doi:10.1109/SMC.2017.8122827

Petermeijer, S. M., De Winter, J. C. F., & Bengler, K. (2016). Vibrotactile displays: A survey with a view on highly automated driving. *IEEE Transactions on Intelligent Transportation Systems*, *17*, 897–907. doi:10.1109/TITS.2015.2494873

\*Pfromm, M. (2016). *Menschbezogener Umgang mit Systemstörungen bei teilautomatisierter Fahrzeugführung* [Human-related handling of system errors in semi-automated vehicle guidance] (Doctoral dissertation, Technische Universität Darmstadt, Darmstadt, Germany).

\*Politis, I., Brewster, S., & Pollick, F. (2015). *Language-based multimodal displays for the handover of control in autonomous cars.* Paper presented at the 7th International Conference on Automotive User Interfaces and Interactive Vehicular Applications, Nottingham, UK (pp. 3–10). doi:10.1145/2799250.2799262

\*Politis, I., Brewster, S., & Pollick, F. (2017). Using multimodal displays to signify critical handovers of control to distracted autonomous car drivers. *International Journal of Mobile Human Computer Interaction*, *9*, 1–16. doi:10.4018/ijmhci.2017070101

\*Politis, I., Langdon, P., Adebayo, D., Bradley, M., Clarkson, P. J., Skrypchuk, L.,... Stanton, N. (2018). *An evaluation of inclusive dialogue-based interfaces for the takeover of control in autonomous cars.* Paper presented at the 23rd International Conference on Intelligent User Interfaces, Tokyo, Japan (pp. 601–606). doi:10.1145/3172944.3172990

Radlmayr, J., & Bengler, K. (2015). Literaturanalyse und Methodenauswahl zur Gestaltung von Systemen zum hochautomatisierten Fahren [Literature analysis and method selection for designing systems for highly automated driving]. *FAT Schriftenreihe*. doi:10.13140/RG.2.1.2222.4401

\*Radlmayr, J., Fischer, F.M., & Bengler, K. (2019). The Influence of Non-driving Related Tasks on Driver Availability in the Context of Conditionally Automated Driving. In S. Bagnara, R. Tartaglia, S. Albolino, T. Alexander, & Y. Fujita (Eds), *Proceedings of the 20th Congress of the International Ergonomics Association (IEA 2018). Advances in Intelligent Systems and Computing: Vol. 823.* Springer, Cham. doi:10.1007/978-3-319-96074-6\_32

\*Radlmayr, J., Gold, C., Lorenz, L., Farid, M., & Bengler, K. (2014). How traffic situations and non-driving related tasks affect the take-over quality in highly automated driving. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *58*, 2063–2067. doi:10.1177/1541931214581434

\*Radlmayr, J., Weinbeer, V., Löber, C., Farid, M., & Bengler, K. (2017). How automation level and system reliability influence driver performance in a cut-in situation. In N. A. Stanton (Ed.): *Advances in human aspects of transportation* (pp. 684–694). Cham: Springer. doi:10.1007/978-3-319-60441-1\_66

Riener, A. (2010). *Simulating on-the-road behavior using a driving simulator*. Paper presented at the 2010 Third International Conference on Advances in Computer-Human Interactions, Saint Maarten, Netherlands Antilles (pp. 25–31). doi:10.1109/ACHI.2010.10

Risto, M., & Martens, M. H. (2014). Driver headway choice: A comparison between driving simulator and real-road driving. *Transportation Research Part F: Traffic Psychology and Behaviour*, 25, 1–9. doi:10.1016/j.trf.2014.05.001

\*Roche, F., & Brandenburg, S. (2018). *Should the urgency of auditory-tactile takeover requests match the criticality of takeover situations?* Paper presented at the 21st International Conference on Intelligent Transportation Systems (ITSC) (pp. 1035–1040). doi:10.1109/ITSC.2018.8569650

Rohatgi, A. (2017). WebPlotDigitizer [Web based tool]. Retrieved from: https://automeris.io/WebPlotDigitizer/index.html

Rothstein, H. R., Sutton, A. J., & Borenstein, M. (Eds.). (2005). *Publication bias in metaanalysis: Prevention, assessment and adjustments*. Chichester, England: John Wiley & Sons.

\*Sadeghian Borojeni, S., Boll, S. C. J., Heuten, W., Bülthoff, H. H., & Chuang, L. (2018). *Feel the movement: real motion influences responses to take-over requests in highly automated vehicles.* Paper presented at the 2018 CHI Conference on Human Factors in Computing Systems, Montreal QC, Canada, 246. doi:10.1145/3173574.3173820

\*Sadeghian Borojeni, S., Chuang, L., Heuten, W., & Boll, S. (2016). *Assisting drivers with ambient take-over requests in highly automated driving*. Paper presented at the 8th International Conference on Automotive User Interfaces and Interactive Vehicular Applications, Ann Arbor, MI (pp. 237–244). doi:10.1145/3003715.3005409

\*Sadeghian Borojeni, S., Weber, L., Heuten, W., & Boll, S. (2018). *From reading to driving: priming mobile users for take-over situations in highly automated driving*. Paper presented at the 20th International Conference on Human-Computer Interaction with Mobile Devices and Services, *14*. ACM. doi:10.1145/3229434.3229464

SAE International. (2016). *Taxonomy and definitions for terms related to driving automation systems for on-road motor vehicle* (J3016) [Standard]. Warrendale, PA: SAE International.

Salthouse, T. A. (2009). When does age-related cognitive decline begin? *Neurobiology of Aging*, *30*, 507–514. doi:10.1016/j.neurobiolaging.2008.09.023

\*Schartmüller, C., Riener, A., & Wintersberger, P. (2018). *Steer-By-WiFi: Lateral Vehicle Control for Take-Overs with Nomadic Devices*. Paper presented at the 10th International Conference on Automotive User Interfaces and Interactive Vehicular Applications (pp. 121–126). ACM. doi:10.1145/3239092.3265954

\*Schartmüller, C., Riener, A., Wintersberger, P., & Frison, A. K. (2018). *Workaholistic: on balancing typing-and handover-performance in automated driving*. Paper presented at the 20th International Conference on Human-Computer Interaction with Mobile Devices and Services, *16*. ACM. doi:1145/3229434.3229459

\*Schittenhelm, H., & Nöcker, G. (2017). Übernahmequalität bei Level 3 Systemen [Drivers take-over reaction times and quality with Level 3 systems]. Poster session presented at the 8th Tagung Fahrerassistenz, Einführung hochautomatisiertes Fahren, München, Germany. Retrieved from: https://www.researchgate.net/publication/321905188\_Ubernahmequalitat\_bei\_Level\_3\_Syste

men\_Drivers\_take-over\_reaction\_times\_and\_quality\_with\_Level\_3\_Systems

Schmidt, F., & Hunter, J. (2015). *Methods of meta-analysis*. 55 City Road, London: SAGE Publications, Ltd. doi:10.4135/9781483398105

\*Schmidt, J. (2018). *Detektion der Reaktionsbereitschaft beim hochautomatisierten Fahren* [Detection of reaction readiness in highly automated driving] (Doctoral dissertation, Technical University of Munich, Munich, Germany).

\*Schmidt, J., Dreißig, M., Stolzmann, W., & Rötting, M. (2017). The influence of prolonged conditionally automated driving on the take-over ability of the driver. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *61*, 1974–1978. doi:10.1177/1541931213601972

\*Shen, S. (2016). *Quantifying drivers' responses to failures of semi-autonomous vehicle systems* (Doctoral dissertation, Clemson University).

Shen, S., & Neyens, D. M. (2014). Assessing drivers' performance when automated driver support systems fail with different levels of automation. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 58, 2068–2072. doi:10.1177/1541931214581435

\*Sportillo, D., Paljic, A., Boukhris, M., Fuchs, P., Ojeda, L., & Roussarie, V. (2017). An immersive Virtual Reality system for semi-autonomous driving simulation: a comparison between realistic and 6-DoF controller-based interaction. Paper presented at the 9th International Conference on Computer and Automation Engineering (pp. 6–10). ACM. doi:10.1145/3057039.3057079

\*Sportillo, D., Paljic, A., & Ojeda, L. (2018). Get ready for automated driving using virtual reality. *Accident Analysis & Prevention*, *118*, 102–113. doi:10.1016/j.aap.2018.06.003

Summala, H. (2000). Brake reaction times and driver behavior analysis. *Transportation Human Factors*, 2, 217–226. doi:10.1207/STHF0203\_2

\*Talamonti, W. J. (2017). Human-Machine Interface Development For Modifying Driver Lane Change Behavior In Manual, Automated, And Shared Control Automated Driving (Doctoral dissertation, Wayne State University). Retrieved from: https://digitalcommons.wayne.edu/cgi/viewcontent.cgi?referer=http://scholar.google.nl/&http sredir=1&article=2746&context=oa\_dissertations

\*Telpaz, A., Rhindress, B., Zelman, I., & Tsimhoni, O. (2015). *Haptic seat for automated driving: preparing the driver to take control effectively*. Paper presented at the 7th International Conference on Automotive User Interfaces and Interactive Vehicular Applications, Nottingham, UK (pp. 23–30). doi:10.1145/2799250.2799267

\*Telpaz, A., Rhindress, B., Zelman, I., & Tsimhoni, O. (2017). Using a vibrotactile seat for facilitating the handover of control during automated driving. *International Journal of Mobile Human Computer Interaction*, *9*, 17–33. doi:10.4018/ijmhci.2017070102

\*Toffetti, A., Wilschut, E., Martens, M., Schieben, A., Rambaldini, A., Merat, N., & Flemisch, F. (2009). CityMobil: Human factor issues regarding highly automated vehicles on eLane. *Transportation Research Record: Journal of the Transportation Research Board*, 2110, 1–8. doi:10.3141/2110-01

\*Van den Beukel, A. P., Van der Voort, M. C., & Eger, A. O. (2016). Supporting the changing driver's task: Exploration of interface designs for supervision and intervention in automated driving. *Transportation Research Part F: Traffic Psychology and Behaviour, 43*, 279–301. doi:10.1016/j.trf.2016.09.009

\*Van der Meulen, H., Kun, A. L., & Janssen, C. P. (2016). *Switching back to manual driving: how does it compare to simply driving away after parking?* Paper presented at the 8th International Conference on Automotive User Interfaces and Interactive Vehicular Applications, Ann Arbor, MI (pp. 229–236). doi:10.1145/3003715.3005452

\*Vlakveld, W., van Nes, N., de Bruin, J., Vissers, L., & van der Kroft, M. (2018). Situation awareness increases when drivers have more time to take over the wheel in a Level 3 automated car: a simulator study. *Transportation research part F: traffic psychology and behaviour*, 58, 917–929. doi:10.1016/j.trf.2018.07.025

\*Vogelpohl, T., Kühn, M., Hummel, T., Gehlert, T., & Vollrath, M. (2018). Transitioning to manual driving requires additional time after automation deactivation. *Transportation Research Part F: Traffic Psychology and Behaviour*, *55*, 464–482. doi:10.1016/j.trf.2018.03.019

\*Vogelpohl, T., Kühn, M., Hummel, T., & Vollrath, M. (in press). Asleep at the automated wheel—Sleepiness and fatigue during highly automated driving. *Accident Analysis & Prevention.* doi:10.1016/j.aap.2018.03.013

Vogelpohl, T., Vollrath, M., Kühn, M., Hummel, T., & Gehlert, T. (2016). *Übergabe von hochautomatisiertem Fahren zu manueller Steuerung* [Transition from highly automated driving to manual control]. Berlin: GDV.

\*Walch, M., Lange, K., Baumann, M., & Weber, M. (2017). Autonomous driving: investigating the feasibility of car-driver handover assistance. *International Journal of Mobile Human Computer Interaction*, *9*, 58–74. doi:10.4018/IJMHCI.2017040104

Walch, M., Mühl, K., Kraus, J., Stoll, T., Baumann, M., & Weber, M. (2017). From car-driverhandovers to cooperative interfaces: visions for driver–vehicle interaction in automated driving. In G. Meixner & C. Müller (Eds.), *Automotive user interfaces: creating interactive experiences*  in the car (pp. 273–294). Cham: Springer International Publishing. doi:10.1007/978-3-319-49448-7\_10

\*Wan, J., & Wu, C. (2018a). The Effects of Lead Time of Take-Over Request and Nondriving Tasks on Taking-Over Control of Automated Vehicles. *IEEE Transactions on Human-Machine Systems*, 48, 582–591. doi:10.1109/THMS.2018.2844251

\*Wan, J., & Wu, C. (2018b). The Effects of Vibration Patterns of Take-Over Request and Non-Driving Tasks on Taking-Over Control of Automated Vehicles. *International Journal of Human–Computer Interaction*, *34*, 987–998. doi:10.1080/10447318.2017.1404778

\*Wandtner, B. (2018). Non-driving related tasks in highly automated driving-Effects of task characteristics and drivers' self-regulation on take-over performance (Doctoral dissertation, University of Würzburg, Würzburg, Germany). Retrieved from: https://opus.bibliothek.uni-wuerzburg.de/opus4-

 $wu erz burg/front door/deliver/index/docId/17395/file/Wandtner\_Bernhard\_AutomatedDriving.pdf$ 

\*Wandtner, B., Schmidt, G., Schömig, N., & Kunde, W. (2018). *Non-driving related tasks in highly automated driving – Effects of task modalities and cognitive workload on take-over performance.* Paper presented at the AmE 2018 – Automotive meets Electronics; 9th GMM-Symposium, Dortmund, Germany (pp. 8-13).

\*Wandtner, B., Schömig, N., & Schmidt, G. (2018a). Effects of non-driving related task modalities on takeover performance in highly automated driving. *Human Factors, 60*, 870–881. doi:10.1177/0018720818768199

\*Wandtner, B., Schömig, N., & Schmidt, G. (2018b). Secondary task engagement and disengagement in the context of highly automated driving. *Transportation research part F: traffic psychology and behaviour*, *58*, 253-263. doi:10.1016/j.trf.2018.06.001

Warshawsky-Livne, L., & Shinar, D. (2002). Effects of uncertainty, transmission type, driver age and gender on brake reaction and movement time. *Journal of safety research*, *33*, 117–128.

\*Weinbeer, V., Baur, C., Radlmayr, J., & Bill, J., Muhr, T., & Bengler, K., (2017). *Highly automated driving: How to get the driver drowsy and how does drowsiness influence various take-over-aspects?* Paper presented at the 8th Tagung Fahrerassistenz, Einführung hochautomatisiertes Fahren, München, Germany.

\*Wiedemann, K., Naujoks, F., Wörle, J., Kenntner-Mabiala, R., Kaussner, Y., & Neukum, A. (2018). Effect of different alcohol levels on take-over performance in conditionally automated driving. *Accident Analysis & Prevention*, *115*, 89–97. doi:10.1016/j.aap.2018.03.001

\*Wintersberger, P., Riener, A., Schartmüller, C., Frison, A. K., & Weigl, K. (2018). *Let Me Finish before I Take Over: Towards Attention Aware Device Integration in Highly Automated Vehicles.* Paper presented at the 10th International Conference on Automotive User Interfaces and Interactive Vehicular Applications (pp. 53–65). ACM. doi:10.1145/3239060.3239085

\*Yang, Y., Gerlicher, M., & Bengler, K. (2018). How does relaxing posture influence take-over performance in an automated vehicle? *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *62*, 696–700. doi:10.1177/1541931218621157

\*Yang, Y., Götze, M., Laqua, A., Dominioni, G. C., Kawabe, K., & Bengler, K. (2017). *A method to improve driver's situation awareness in automated driving*. Paper presented at the Human Factors and Ergonomics Society Europe Chapter 2017 Annual Conference (pp. 29–47). Retrieved from: https://www.hfes-europe.org/wp-content/uploads/2017/10/Yang2017.pdf

\*Yang, Y., Karakaya, B., Dominioni, G. C., Kawabe, K., & Bengler, K. (2018). An HMI Concept to Improve Driver's Visual Behavior and Situation Awareness in Automated Vehicle. Paper presented at the 21st International Conference on Intelligent Transportation Systems (ITSC) (pp. 650–655). doi:10.1109/ITSC.2018.8569986

\*Yoon, S. H., Kim, Y. W., & Ji, Y. G. (2019). The effects of takeover request modalities on highly automated car control transitions. *Accident Analysis & Prevention*, *123*, 150–158. doi:10.1016/j.aap.2018.11.018

\*Young, M. S., & Stanton, N. A. (2000). Journey's end: Will vehicle automation make skilled drivers redundant? *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 44, 89–92. doi:10.1177/154193120004400124

\*Young, M. S., & Stanton, N. A. (2007). Back to the future: Brake reaction times for manual and automated vehicles. *Ergonomics*, *50*, 46–58. doi:10.1080/00140130600980789

\*Zeeb, K., Buchner, A., & Schrauf, M. (2015). What determines the take-over time? An integrated model approach of driver take-over after automated driving. *Accident Analysis & Prevention*, 78, 212–221. doi:10.1016/j.aap.2015.02.023

\*Zeeb, K., Buchner, A., & Schrauf, M. (2016). Is take-over time all that matters? The impact of visual-cognitive load on driver take-over quality after conditionally automated driving. *Accident Analysis & Prevention*, *92*, 230–239. doi:10.1016/j.aap.2016.04.002

\*Zeeb, K., Härtel, M., Buchner, A., & Schrauf, M. (2017). Why is steering not the same as braking? The impact of non-driving related tasks on lateral and longitudinal driver interventions during conditionally automated driving. *Transportation Research Part F: Traffic Psychology and Behaviour, 50*, 65–79. doi:10.1016/j.trf.2017.07.008

\*Zhang, B., Wilschut, E. S., Willemsen, D. M. C., Alkim, T., & Martens, M. H. (2017). The effect of see-through truck on driver monitoring patterns and responses to critical events in truck platooning. In N. A. Stanton (Ed.), *Advances in human aspects of transportation* (pp. 842–852). Cham: Springer. doi:10.1007/978-3-319-60441-1\_81

\*Zhang, B., Wilschut, E. S., Willemsen, D. M. C., & Martens, M. H. (2019). Transitions to Manual Control from Highly Automated Driving in Non-critical Truck Platooning Scenarios. *Transportation Research Part F: Traffic Psychology and Behaviour*, *64*, 84-97. doi:10.1016/j.trf.2019.04.006

Zuur, A. F., Leno, E. N., Walker, N. J., Saveliev, A. A., & Smith, G. M. (2009). *Mixed effects models and extensions in ecology with R*. New York: NY: Spring Science and Business Media.

# **3.** Transitions to manual control from highly automated driving in non-critical truck platooning scenarios (with hand-held NDT)

This chapter is based on: Zhang, B., Wilschut, E. S., Willemsen, D. M. C., & Martens, M. H. (2019). Transitions to manual control from highly automated driving in non-critical truck platooning scenarios. Transportation Research Part F: Traffic Psychology and Behaviour, 64, 84-97. (adapted with minor textual changes)

### 3.1 Introduction

Over the past decade, the rapid advancement of intelligent transport systems (ITS) and sensor technology has been dramatically changing the automotive industry. The first (semi-)automated systems already available are designed for specific driving situations. For example, some cars already have systems such as traffic jam assist that work at low speeds, or automated driving on highways. These systems are primarily based on Adaptive Cruise Control (ACC), which allows a vehicle to maintain a predefined headway to the vehicle in front, combined with Lane Keeping Assist Systems (LKAS) that take control of the lateral position of a car within its lane primarily on the basis of road markings. In a recently published naturalistic driving study, Endsley (2017) reported her experience with the Tesla Model S over a period of six months, during which considerable number of trips were performed in autopilot mode. Besides automated passenger cars, automated truck platooning, defined as a team of virtually connected trucks traveling together at short following distances, is also getting increasing attention for its potential positive effects on energy saving, driver workload, traffic flow, and safety (Alam, Besselink, Turri, Martensson, & Johansson, 2015; Bergenhem, Shladover, Coelingh, Englund, & Tsugawa, 2012; Hjälmdahl, Krupenia, & Thorslund, 2017; Willemsen, Stuiver, & Hogema, 2015). The lead truck is normally driven by a human driver, and the trucks behind follow the lead truck with automated longitudinal control (e.g., via Cooperative Adaptive Cruise Control (CACC), as described in Ploeg, Van de Wouw, & Nijmeijer, 2014) or with both automated longitudinal and lateral control (e.g., combining CACC and LKAS). Although truck platooning has not yet been implemented as a commercial product, various demonstrations and pilots already showed its potential. One example is the European Truck Platooning Challenge (2016), in which six European truck manufacturers brought platoons of trucks equipped with various automation technologies on public roads, travelling from various European cities to the final destination of the Port of Rotterdam.

In these real-life applications and truck platooning demonstrations, drivers are still required to attentively monitor the driving environment and be prepared to take over instant manual control upon the detection of operational limits of the system (i.e., Level 2 automation as defined by SAE International, 2016). Studies with passenger cars have shown that this level of automation may not require less attentional demand compared to manual driving (Stapel, Mullakkal-Babu, & Happee, 2019; Stapel, Mullakkal-Babu, & Happee, 2017). Unscheduled and critical transitions of control can be particularly dangerous for drivers in the platoon, because the available time budget is usually very small due to the short following distance used for platooning. In the foreseeable future, the working range and the reliability of these systems will continue to develop towards a situation in which the driver is allowed to be temporarily out of the loop (Merat et al., 2018) and relax (SAE level 3 or level 4 automation). In certain situations, such as approaching the exit of a motorway and thus dissolving the platoon, transitions of control to human drivers may still occur. These transitions will have to be sophisticatedly designed to ensure that drivers will be ready to take over control safely and comfortably. This can be challenging, because increasing levels of automation may have negative impact on drivers' ability to take over control due to factors such as increasing engagements in non-driving tasks (Jamson, Merat, Carsten, & Lai, 2013), loss of situation awareness (Endsley, 1995), boredom, and fatigue (Körber, Schneider, & Zimmermann, 2015). Also, the variability in driver activities and mental states during automated driving can subsequently cause large differences in driver response towards similar stimuli. A good understanding of driver takeover behaviour and performance under various task conditions becomes essential for the development of future intelligent vehicles.

Typically, taking over control includes several mental and physical processing procedures: perception of the take-over stimulus (normally a take-over request (TOR) issued by the system), cognitive processing of the traffic situation, response selection, resuming motor readiness (by repositioning the hands on the steering wheel and feet on the pedals), and implementation of an intervention (Gold & Bengler, 2014; Petermeijer, De Winter, & Bengler, 2016; Zeeb, Buchner, & Schrauf, 2015, 2016). Multiple factors have been suggested to influence the response time (RT) to complete the take-over process (i.e., take-over time) and other take-over performance parameters, such as human-machine interface (HMI) design (Carsten & Martens, 2019; Melcher, Rauh, Diederichs, Widlroither, & Bauer, 2015; Van den Beukel, Van der Voort, & Eger, 2016; Willemsen, Stuiver, Hogema, Kroon, & Sukumar, 2014), driver inattention and engagement in non-driving tasks (Louw, Kountouriotis, Carsten, & Merat, 2015; Petermann-Stock, Hackenberg, Muhr, & Mergl, 2013; Radlmayr, Gold, Lorenz, Farid, & Bengler, 2014; Wandtner, Schömig, & Schmidt, 2018; Zeeb, Härtel, Buchner, & Schrauf, 2017), urgency of the take-over situation (Gold, Damböck, Lorenz, & Bengler, 2013; Ito, Takata, & Oosawa, 2016; Roche & Brandenburg, 2018), and traffic complexity (Gold, Körber, Lechner, & Bengler, 2016; Jamson et al., 2013; Radlmayr et al., 2014). Zhang, De Winter, Varotto, Happee, and Martens (2019) conducted a meta-analysis on take-over times from 129 studies with SAE level 2 automation or higher, showing that the mean values ranged largely between 0.7 s and 20 s across various experimental studies and conditions. The reviewed studies were predominantly car automation studies. Eriksson and Stanton (2017) pointed out that most of the previous efforts focused on driver take-over time in critical situations, given a predetermined take-over lead time (i.e., the maximum time allowed for a driver to response to a critical event). Greater perceived urgency can shorten take-over times, but on the other hand can also force the driver to take over before being well prepared (Gold et al., 2013; Lu, Happee, Cabrall, Kyriakidis, & De Winter, 2016). Therefore, take-over times measured under critical situations cannot reflect the time drivers ideally need to get prepared, and achieving a short take-over time is not always necessary or beneficial (Zhang et al., 2019). Advanced strategies such as multi-step control transitions and adaptive control transitions (e.g., taking driver states into consideration) may help the driver to gradually get back into the loop, and take over more safely and comfortably. To be able to develop such strategies, and to design scheduled (and therefore non-critical) transitions, it's necessary to understand how long it takes for a driver to conduct the selfregulated control transition process. Eriksson and Stanton (2017) exclusively looked into noncritical take-over scenarios without time restrictions in a passenger car mock-up simulator, and reported largely varied take-over times ranging from 1.9 s to 25.7 s. We have not been able to identify studies that have used professional truck drivers as participants in a truck mock-up driving simulator, with the focus on take-over response times and performance when leaving a platoon in non-critical scenarios.

Due to the fact that driver take-over is a complex procedure, it is of interest to understand how long it takes to complete various information processing stages and come to action, and how these take-over time components are influenced by various factors. In previous studies, researchers often reported RTs measured from the TOR onset until the first contact with the steering wheel/pedal to indicate the motor or physical readiness of the driver, and the start of the intervention (by braking, steering, or button pressing) to indicate the initiation of a conscious response to the take-over scenario (e.g., Zeeb et al., 2015, 2016, 2017). If the driver was visually distracted before the TOR onset, some researchers also measured the RTs when the driver first shifted the gaze away from the visual non-driving task, and when the first fixation on the road occurred (e.g., Gold et al., 2013; Körber, Gold, Lechner, & Bengler, 2016; Feldhütter, Gold, Schneider, & Bengler, 2017). This was done to investigate when the driver shifted the visual attention to the road and started to process the information gathered from the driving

environment. In a few studies, additional actions were registered, such as the first mirror check and the first operation of a turn signal (e.g., Gold et al., 2013; Petermeijer, Doubek, & De Winter, 2017).

In studies on manual driving, the concept of perception-response time is frequently utilized to investigate drivers' brake responses (see the review of Green, 2000), which comprises two components: the perception RT and the movement RT. The perception RT can be defined as the time it takes for the driver to perceive a stimulus, cognitively process the situation, and decide on a response. The movement RT is the time required to perform the actual programmed motoric action. For the analysis of driver take-over times, if only considering the basic response to the TOR, the division between the perception RT and the movement RT would be the start of the hand/foot movement towards the steering wheel/pedal. Moving hands or feet indicates that the driver has perceived and recognized the meaning of the TOR (i.e., they need to take over control), and has decided to regain motor readiness, which marks the end of the perception RT and the start of the movement RT. These two RT components are suggested to have different explanatory variables. For instance, the perception RT can be more related with the characteristics of the take-over stimulus (e.g., modality and intensity of the TOR) and driver monitoring strategies (e.g., how long and how frequently did the driver look onto/off the road, see Zeeb et al., 2015, 2016), while the movement RT can be more related to the complexity and the amplitude of the motoric manoeuvre (cf. Fitts, 1954), such as putting away a hand-held device. Note that the movements of hands and feet do not have to be the result of a decision about the required action after the comprehension of the current situation, especially in critical scenarios where the movements could be an automated response (i.e. as a reflex). To regain situation awareness may require a considerable amount of time (Lu, Coster, & De Winter, 2017), the process of which could overlap with, or even exceed the movements to resume manual control.

In the manual driving context, the movement RT is often neglected since it only refers to the foot traveling time from the acceleration pedal to the brake pedal, which normally takes less than 0.5 s (Green, 2000). When studying driver behaviour in highly automated driving, taking over control is more of a task-switching process (i.e., to perform two tasks in succession, see Monsell, 2003) rather than multi-tasking such as being distracted during manual driving (Louw, Merat, & Jamson, 2015; Young & Stanton, 2007; Zeeb et al., 2017). This change of paradigms leads to an expansion of non-driving task classifications from simply listening to the radio to complex tasks incompatible with driving, such as writing emails, playing computer games, or even resting with eyes closed (see Naujoks, Befelein, Wiedemann, & Neukum, 2017 for a review). When engaged in non-driving tasks, drivers' postures become less predictable and may heavily influence take-over times. Therefore, the distinction between the perception RT and the movement RT has its added value for a deeper insight into factors influencing driver take-over times and the individual differences. However, very few researchers reported the start of the hand movement when studying take-over times (Kerschbaum, Omozik, Wagner, Levin, Hermsdörfer, & Bengler, 2017; Kerschbaum, Lorenz, & Bengler, 2015). The perceptionresponse time during driver take-over process is little addressed up to now.

In summary, a closer examination of drivers' take-over process and the variability in noncritical take-over situations, particularly in case of truck platooning, is needed to design safe and comfortable transitions of control. The current study aims to investigate how various task conditions influence truck drivers' perception-response times when receiving a request to leave the platoon and take back control, and to explore the factors influencing the variability in this process based on video observations. The factors explaining the RT components and their variability can be used towards the design of an adaptive transition approach.

# 3.2 Methods

## 3.2.1 Participants

Twenty-two participants (2 females) took part in the experiment. They all held a truck driver's license for at least eight years (M = 28 years, SD = 12) and drove at least 10.000 km per year in a truck (M = 35219 km/year, SD = 27508). On average, the participants were 47.4 years old (SD = 11.5), ranging from 27 to 64 years old. The research was approved by the Ethical Committee for participant studies of TNO.

## 3.2.2 Apparatus

The experiment was conducted at TNO in a high fidelity moving base driving simulator consisting of a DAF truck mock-up (see Figure 3.1). The road and traffic environment were projected on cylindrical screens around the vehicle with an 180 degree field of view. Two screens placed behind the vehicle showed the scenery for the rear view mirrors. Vehicle related data including the input of the control elements and the status change of the automation system were recorded by the simulator at a sampling rate of 50 Hz. In addition, the participants' head motion and eye movements were recorded by a non-obtrusive remote eye tracker (SmartEye AB, Gothenburg, Sweden). The mock-up was also equipped with three cameras to observe the drivers' full body postures, facial expressions, and feet positions, respectively.



# Figure 3.1: TNO truck driving simulator with moving hexapod and cylindrical projection screen

### 3.2.3 Two-truck platoon system

An automated system was simulated that allows a truck to follow a lead truck at a short following distance on public motorways, initially limited to platoons of two trucks. The first truck is intended to be driven by a human truck driver (but is controlled by the simulator in this experiment). Once engaged, the second truck is controlled by the automation system (Figure 3.2). The participant in the driving simulator was the driver of the second truck in the platoon. For the participant, the center view of his/her own lane was blocked by the lead truck, but the view of the adjacent lanes was unblocked, as shown in Figure 3.3.

The automated system was modelled as a combination of CACC and LKAS. The driver could push a button normally used for cruise control on the right side of the steering wheel to switch the automation system on/off. To be able to switch the system on, the driver had to drive in an activation zone behind the lead truck. After activating the system, the vehicle controlled both longitudinal and lateral control and did not have to be monitored, thus simulating SAE L4 automated driving. At the end of the automated driving period, the system would indicate that the driver had to take over control by displaying a text message on the screen (Figure 3.4, in Dutch: "Neem de controle over", in English: "Take over control"), accompanied by an auditory signal. The driver had to push the button again to resume both longitudinal and lateral manual control and leave the platoon.



Figure 3.2: Two-truck platoon system



Figure 3.3: Front view in the second truck. The center of the ego lane was blocked by the lead truck, while the driver was able to perceive traffic information from the adjacent lanes



#### Figure 3.4: Visual interface of the take-over request (text in Dutch: "Take over control")

#### 3.2.4 Experimental design and test procedures

The participants were instructed to drive on the right-hand lane of a two-lane motorway behind a lead truck at an average speed of 80 km/h. The route included slight curves and moderate surrounding traffic, but no entries or exits. All participants started with a training session to get familiar with the driving simulator and with platooning. They were asked to perform the coupling and decoupling procedures at least three times, until they felt familiar and comfortable with the system.

After the training, each participant performed eight platooning trials in different test conditions, in which four variables were manipulated: non-driving tasks (no task, using a handheld tablet PC, and relaxing with eyes closed), criticality of the control transition (self-regulated noncritical transition vs. critical transition due to unexpected system failure), time gaps between the trucks (0.8 s vs. 0.3 s), and whether a brake event was implemented after the control transition back to the driver. The system failure trial was always performed at the end of the experiment, while the other trials were conducted in a counterbalanced order. This paper only discusses the five non-critical trials in which the time gap was set to 0.8 s during automated driving. This distance is the minimum allowed time gap defined in ISO 15622, and is considered as an accepted distance for testing the first platooning concepts with truck drivers. The other trials will not be further described (please see Wilschut, Willemsen, Hogema, & Martens, 2016 for more details on this project, and Zhang, Wilschut, Willemsen, Alkim, & Martens, 2017 for the critical transition trial).

In three trials, no events were programmed after the control transition back to the driver. They serve as the basis of the study to investigate drivers' performance when leaving the platoon and stabilizing the truck in normal, uneventful transitions. In each trial, participants started with a short period of manual driving, then activated the platooning system to follow the lead truck automatically. During the automated driving segment, each participant was subjected to one of the three task conditions as briefly mentioned above: Driver Monitoring (participants were instructed to monitor the surroundings constantly, so hands-off, feet-off, and eyes on the road), Driver Not-monitoring (participants were provided with a tablet PC and were asked to use this, so hands-off, feet-off, and eyes off the road, but they were allowed to scan the outside world if they wanted to) and Eyes-closed (participants were not allowed to open their eyes, so hands-off, feet-off, and eyes off the road). These conditions manipulated participants' attentiveness and represented three activities likely to be performed by the driver in future automated vehicles. In particular, the Eyes-closed condition emulated the situation that drivers take a rest

with their eyes closed and cannot perceive any visual information, which is very likely to occur at higher levels of automation. After four minutes in the Driver Monitoring condition, or eight minutes in the Driver Not-monitoring and Eyes-closed conditions, the TOR was issued and participants were asked to take over control by pressing the button whenever they felt ready to do so without time restrictions. The automation duration was shorter for the Monitoring condition to increase the possibility that drivers were still paying attention and were alert, and longer for the other two conditions increase the chance of the driver being out-of-the loop. After taking over control, participants continued with a short manual driving phase until the end of the scenario.

To explore whether participants possessed sufficient readiness and situation awareness to cope with more complex situations after the self-regulated transition of control, two additional trials (in Driver Monitoring and Driving Not-monitoring conditions, respectively) with a brake event were implemented. Three seconds after participants took over control, the lead truck started to brake at a deceleration rate of 5 m/s2 for two seconds. No forward collision warnings or brake assist systems were available, and participants had to brake to avoid accidents. The experimental scenario before the brake event was identical to the uneventful trials described above.

#### 3.2.5 Dependent variables

For the basic trials without brake events, three take-over RT components were measured and evaluated, namely the total RT (TRT), the perception RT (PRT), and the movement RT (MRT). The TRT was defined as the time interval from the TOR onset until the moment when participants pressed the button to reclaim control. This value could be directly obtained from the driving simulator data. The PRT was measured from the onset of the TOR until the start of participants' hand movement reaching for the steering wheel, indicating the time elapsed for the participant to perceive and understand the message that they need to take over control, and to decide to resume motor readiness. Subtracting the PRT from the TRT, the MRT was retrieved, showing how long it took the participant to get physically ready. Because this study was about non-critical conditions without time restraints, we assumed that the information perception and processing cycle (in terms of perceiving and understanding the TOR) was finished before participants started to move their hands. The timestamps for the start of the hand movement were manually annotated from the video recordings by two independent observers. A high degree of reliability between the observers was found (the average intraclass correlation coefficient (ICC) was 0.912 with a 95% confidence interval from 0.880 to 0.936). Participants' behaviours during the take-over process were also recorded by the observers to explore the individual differences.

To assess take-over performance (i.e., quality) in stabilizing the truck in its lane, standard deviation of lateral position (SDLP), the number of steering wheel reversals (with a gap of 3 degrees, according to the definition described in SAE International (2015)), the longitudinal speed as well as its standard deviation were analysed. Smaller deviations in lateral position and longitudinal speed, as well as fewer strenuous steering wheel reversals indicate more consistent and seamless control. Previous research showed that to stabilize the vehicle after taking over manual control required around 35–40 s (Merat, Jamson, Lai, Daly, & Carsten, 2014). In the current study, we observed the performance metrics for the first 40 s after the transition. The observation was divided into four timeslots of 10 s except for the steering wheel reversals, which did not occur frequently within such short timeslots. This measure was reported for the entire observation period of 40 s. Additionally, we measured the time gap after taking over

control, aiming to understand how drivers regulated the following distance when leaving the platoon.

For the additional trials with brake events, participants' responses to the braking lead truck were evaluated using the brake RT, the minimum time-to-collision (TTC) with the lead truck, and the minimum time headway (THW) to the lead truck. The brake RT was defined as the time interval between the onset of the braking light of the lead truck and the moment when the participant started pressing the brake pedal. The minimum TTC and the minimum THW were calculated for the 10 s after the onset of the braking light of the lead truck. A shorter brake RT, a longer minimum TTC, and a higher minimum THW suggest a safer collision avoidance behaviour and a better performance.

#### 3.3 Results

Due to technical problems, the simulator data in 10 trials (for driving performance assessment), and the video recordings in 13 trials (for perception-response time assessment) were not available for analysis. An overview of data availability per condition is presented in Table 3.1. To make use of all available data on each subject instead of dropping the entire case due to incomplete data, before performing the statistical tests, the missing data were imputed using the expectation maximization (EM) method (Dempster, Laird, & Rubin, 1977) using the 'Missing Data Analysis' command in SPSS 24. The EM method is an iterative procedure that estimates missing data based on available observations using expectation and maximization algorithms, which is suggested to produce more reliable and less biased estimates as compared to traditional missing data handling methods (e.g., mean imputation and regression imputation, see Musil, Warner, Yobas, & Jones, 2002). Results from a preliminary Little's Missing Completely at Random (MCAR) test suggested that the data were missing at random. That is, there appears no systematic or non-random pattern of data omission, and the EM method is considered appropriate. Note that the imputed data were only used for the purpose of statistical tests. All descriptive results were reported based on the raw data without imputation. All statistical analyses were conducted using SPSS 24. The significance level was set to 0.05.

Condition	Total trials	Available datasets	Available datasets	
	performed by	for perception-	for driving	
	the participants	response time	performance	
Driver Monitoring (w/o BE)	22	18	19	
Driver Not-monitoring (w/o BE)	22	17	19	
Eyes-closed (w/o BE)	22	18	20	
Driver Monitoring (with BE)	22	-	21	
Driver Not-monitoring (with BE)	22	-	21	

#### Table 3.1: Overview of data availability

*Note.* BE = Brake event. The perception-response time was only reported for trials without brake events.

#### 3.3.1 Take-over response times

Averaged over all conditions, the participants took 4.45 s (SD = 2.21) to take over control. The TRT varied from 1.75 s to 11.31 s, and in 95% of the trials participants took over within 8.5 s. The stacked bar plot (Figure 3.5) and the descriptive data table below show an overview of PRTs and MRTs per condition. These two components add up to the TRTs.



# Figure 3.5: RT components compared across the three conditions, all measured in second. The bars show the means of MRTs and PRTs per condition; they add up to the TRTs. IQR = Interquartile range.

Because the TRT is not independent from the PRT and the MRT, two separate repeatedmeasures Analyses of Variance (ANOVA) were performed to examine the effects of task conditions on the total take-over time and its sub-components. A one-way repeated-measures ANOVA was first conducted to analyse TRT. Due to the violation of the assumption of sphericity ( $\chi^2(2) = 11.81$ , p = .003), degrees of freedom were corrected using Greenhouse-Geisser estimates of sphericity ( $\varepsilon = 0.69$ ). There was a significant main effect of task conditions on TRTs (F(1.38, 29.04) = 43.13; p < .001,  $\eta^2 = 0.67$ ). The post hoc tests showed a significantly shorter TRT in Driver Monitoring condition, in which participants deactivated the automation system on average 3 s earlier than in the other two conditions. No significant differences in TRTs were found between Driver Not-monitoring and Eyes-closed conditions.

Regarding the sub-components of TRTs, a 3 (task conditions)  $\times$  2 (RT components) repeatedmeasures ANOVA was conducted. Mauchly's test indicated that the assumption of sphericity was violated for task conditions ( $\chi^2(2) = 11.81$ , p = .003), and the interaction between task conditions and RT components ( $\chi^2(2) = 11.04$ , p = .004). Degrees of freedom were corrected using Greenhouse-Geisser estimates of sphericity ( $\varepsilon = 0.69$  and 0.72). Results showed significant main effects for both task conditions (F(1.38, 29.04) = 43.13; p < .001,  $\eta^2 = 0.67$ ) and RT components (F(1, 21) = 85.34; p < .001,  $\eta^2 = 0.80$ ). The interaction between task conditions and RT components was also statistically significant (F(1.40, 29.49) = 17.96; p < .001,  $\eta^2 = 0.46$ ).

Post hoc pairwise comparisons showed the longest PRT in Eyes-closed condition (M = 1.89 s, SD = 1.66), while no significant differences were found between Driver Monitoring and Driver Not-monitoring conditions. Significant differences in MRTs were found between all condition pairs. The longest MRT was found in the Not-monitoring condition in which the drivers were holding the tablet before the transition (M = 4.61 s, SD = 1.40), followed by the Eyes-closed condition (M = 3.53 s, SD = 1.90). The participants in the Driver Monitoring condition showed the shortest MRT (M = 1.67 s, SD = 0.49). The results further showed that MRTs were significantly higher than PRTs in both Driver Monitoring and Driver Not-monitoring conditions. Although a similar trend was found for the Eyes-closed condition, the difference between the PRT and the MRT didn't reach statistical significance.

#### 3.3.2 Take-over performance in uneventful trials

With respect to lateral control of manual driving (SDLP) (Figure 3.6), a 3 (Task condition) × 4 (Timeslots: 0–10 s, 10–20 s, 20–30 s, 30–40 s) repeated-measures ANOVA showed significant effects of task conditions (F(2, 42) = 4.01; p = .026,  $\eta^2 = 0.16$ ) and time (F(3, 63) = 11.29; p < .001,  $\eta^2 = 0.35$ ). Pairwise comparisons showed a significantly smaller SDLP in the Driver Monitoring condition, but only for the first 20 s. Concerning the function of time, a significantly larger SDLP was found for the first 10 s in all conditions. SDLP became stable and didn't show significant changes in the following timeslots. Another repeated-measures ANOVA was performed on the number of steering wheel reversals for the first 40 s after taking over control (Figure 3.7). No significant differences were found between task conditions.



Figure 3.6: Average standard deviation of lateral position (SDLP) for the first 40 s after control was transferred back to the driver. The error bars represent the standard deviations.



# Figure 3.7: Number of steering wheel reversals (>3 deg) for the first 40 s after control was transferred back to the driver. The error bars represent the standard deviations.

Figure 3.8 and Figure 3.9 show longitudinal driving performance with respect to the longitudinal speed and its standard deviation. A main effect of *time* was found for speed (F(3, 63) = 29.58; p < .001,  $\eta^2 = 0.59$ ), but there was no main effect of task conditions (F(2, 42) = 2.89; p = .66,  $\eta^2 = 0.12$ ). The speed dropped significantly between 10 s and 20 s after the transition of control (from 21.71 m/s to 21.24 m/s averaged over all conditions), and increased again afterwards. Pairwise comparisons revealed that participants drove at the lowest speed in the Driver Not-monitoring condition, but only for the first 20 s. No differences were found between the other two task conditions.

Main effects of both *task conditions* (F(2, 42) = 4.97; p = .01,  $\eta^2 = 0.19$ ) and *time* (F(3, 63) = 50.96; p < .001,  $\eta^2 = 0.71$ ) were found for the standard deviation of the speed. The variability in speed was significantly larger for the first 10 s, after which the drivers in all task conditions were driving at a more consistent speed. The differences between task conditions only lasted for 10 s, with the deviation in speed in Driver Monitoring condition significantly smaller compared to the other task conditions. No significant differences were found between Driver Not-monitoring and Eyes-closed conditions for any of the timeslots.



Figure 3.8: Average longitudinal speed for the first 40 s after control was transferred back to the driver. The error bars represent the standard deviations.



# Figure 3.9: Average standard deviation of longitudinal speed for the first 40 s after control was transferred back to the driver. The error bars represent the standard deviations.

Figure 3.10 shows how participants regulated the following distances for the first 40 s after the transition of control. A repeated-measures ANOVA revealed main effects of both *task conditions* (F(2, 42) = 4.38; p = .019,  $\eta^2 = 0.17$ ) and *time* (F(3, 63) = 131.74; p < .001,  $\eta^2 = 0.86$ ). Post hoc tests showed that the following distance continued to increase as time passed by, and that participants in the Driver Monitoring condition drove with significantly smaller following distance compared to the other two conditions except for the first timeslot. In order to know when the driver stopped increasing the distance to the lead truck, we conducted a further observation until 120 s after leaving the platoon. No significant differences between timeslots were found from 90 to 100 s, and the time headway came to a stable status at around 2.7 s.



Figure 3.10: Average time headway for the first 40 s after control was transferred back to the driver. The error bars represent the standard deviations.

#### 3.3.3 Driver response in the brake event

In the two trials with brake events, all participants braked in response to the decelerating lead truck with an average brake RT of 1.07 s (SD = 0.36 s). No collisions occurred. Paired samples t-tests were conducted to examine whether significant differences existed when comparing the three performance measures between the two task conditions (Table 3.2). Despite of the slight tendency of a better performance in the Driver Monitoring condition, the differences didn't reach the significance level.

conditions, as well as results of the paired t-tests between conditions.							
	Driver Monitoring		Driver Not-monitoring		Paired t-test		
	М	SD	Μ	SD	t	df	р
Brake RT (s)	1.02	0.22	1.16	0.45	-1.14	21	0.267
Minimum TTC (s)	2.41	0.83	1.99	0.79	2.01	21	0.057
Minimum THW (s)	0.60	0.13	0.57	0.21	1.19	21	0.248

Table 3.2: Means and standard deviations of the performance measures in the two tas
conditions, as well as results of the paired t-tests between conditions.

#### 3.4 Discussion

This study mainly investigated driver perception-response times to take over control from highly automated truck platooning systems in normal, non-critical operation. We studied how this take-over process was influenced by three task conditions, namely when the driver was monitoring the road, when working on a hand-held tablet, and when resting with the eyes closed. In general, results show that most participants (95% as mentioned in Section 3.3.1) could indicate their readiness to take over manual control within 8.5 s regardless of the task conditions. When comparing with the non-critical take-over times of passenger car drivers measured in Eriksson and Stanton (2017), the median take-over times in the current study with truck drivers were 1.7 s and 0.7 s shorter in monitoring and visually distracted conditions, respectively. Several factors may contribute to the differences. First of all, the small time gap to the lead truck may lead to shorter take-over times. Because the front view of the ego lane was largely blocked by the lead truck, the driver in the truck platoon could not perceive as much traffic environment as in a passenger car that is not in a platoon (Zhang et al., 2017), which may reduce the information processing time. The small time gap might also lead to a higher risk perception, prompting the driver to respond faster (cf. Summala, 2000). Moreover, the take-over time was measured until the first intervention (steering or braking) of the participants in Eriksson and Stanton (2017), while in the current study the participants pressed a button to resume control. The decision making time to determine a proper intervention may contribute to the longer takeover time in Eriksson and Stanton (2017). In addition, our study utilized within-study design with short automation duration for each trial, whereas in Eriksson and Stanton (2017), betweenstudy design was adopted. Repeated scenario exposure and the expectancy on the upcoming event could have led to shorter RTs (Engström, Aust, & Viström, 2010; Green, 2000; Zhang et al., 2019).

Similar mean take-over times were found in the Driver Not-monitoring condition and the Eyesclosed condition, both significantly longer than the Driver Monitoring condition. On a closer examination, different RT patterns were revealed. In the Driver Not-monitoring condition, the longer total take-over time was mainly due to the increase in the hand movement RT rather than the perception RT. This is in line with previous findings that holding a hand-held device increased the take-over time largely by approximately 1-2 s compared to using a hand-free device in car drivers (Wandtner et al., 2018; Zeeb et al., 2017). With respect to the Eyes-closed condition, the perception RT was the longest among all task conditions, but the hand movement RT was shorter compared to the Driver Not-monitoring condition due to the absence of a handheld tablet. The outliers with extremely large perception RT values also occurred in the Eyesclosed condition, possibly due of the tendency of drowsiness. We observed that one participant seemed to have fallen asleep at the moment of the TOR onset, and seemed confused for the first few seconds after "waking up". It is therefore noteworthy that merely closing the eyes does not emulate sleeping, and caution should be made when interpreting the results obtained from such conditions. Worse situations could occur to fatigued drivers who are deeply asleep during automated driving. Multi-step warning systems can be particularly beneficial for sleeping drivers to get back into cognition loop gradually without causing startling effects. Another possible explanation for the slow perception response is that drivers with eyes closed could only rely on the auditory channel to perceive the warning signal, and thus the effect of the multimodal TOR was reduced (Baldwin & Lewis, 2014; Bazilinskyy & De Winter, 2015). In the other conditions, participants could also see the visual signals through their peripheral vision.

In addition, we found higher movement RTs than perception RTs in all task conditions. This differs from the previous findings in manual driving context, whereby the movement RT is normally much shorter than the perception RT (Green, 2000). When driving in an automated vehicle, it would take on average 1.7 s for truck drivers to replace their hands at the steering wheel at a comfortable pace without non-driving tasks, which could be more than 2.5 times longer when drivers were engaged in non-driving tasks with hand-held devices. The variability in movement RT is also suggested to be the main cause for the individual differences in total take-over times. On a closer examination, we observed the postures of the participants at the moment of TOR onset, as well as their activities during the take-over process. Four typical activities were extracted and further combined with the conditions in which one particular activity occurred. The hand movement RTs in the corresponding trials are shown in Table 3.3. In all Driver Monitoring trials and 12 (out of 18) Eyes-closed trials, participants directly moved their hands to the steering wheel without any intermediate activities. The RT to complete this simple process was the shortest, also with the smallest variability. When participants had to put away the hand-held device in order to take over control, this RT was increased by approximately 2.5 s. In two Driver Not-monitoring trials, the participants also had to put on/off their glasses to switch to driving tasks. This behaviour caused approximately 1 s extra movement time in addition to putting away the tablet only. Moreover, we discovered that in six Eyes-closed trials the participants leaned backwards to rest comfortably during automated driving, and had to adjust the seat position to be able to take over control, resulting in a large mean movement RT of 5.68 s. The variability in physical activities during the take-over process considerably influenced the take-over times.

pre-TO Activities	Description	Occurrence	Condition	MPT (	s)
		number		Mean	SD
Direct hand-on	The participant directly pressed the button without any intermediate activities.	30	All Monitoring	1.99	0.87
			Eyes-closed (12/18)		
Put away tablet	The participant put away the hand-held tablet before pressing the button.	15	Not- monitoring	4.50	1.37
Put away tablet + glasses	The participant put away the hand-held tablet, and put on/off the glasses before pressing the button.	2	Not- monitoring	5.42	1.35
Adjust the seat	The participant adjusted the seat before pressing the button.	6	Eyes-closed	5.68	1.28

#### Table 3.3 Activities performed by the participants during the taking over process

With respect to driving performance after the transition, the differences between the task conditions mainly persisted for the first 10 s, with the performance in the Driver Monitoring condition being slightly better compared to the other task conditions. Such differences between conditions disappeared when more time passed by. Driving performance with respect to SDLP, longitudinal speed and its standard deviation was stabilized within 20 s after the transition. Overall, participants could safely stabilize the truck after leaving the platoon in all task conditions. In the additional trials with brake events, all participants were able to respond quickly to avoid collisions with the decelerating lead truck, with no significant differences in performance between task conditions. Participants' take-over times were suggested to represent the time they need to get ready for normal, uncritical transitions, and be prepared for moderately complex driving situation after the transition.

There are several limitations to be discussed. First, this study focused on non-critical transitions, and participants only needed to stabilize the vehicle or respond to a brake event after leaving the platoon, which may require relatively low cognitive and physical load. The perception RT only concerned the time it takes to sense the TOR and interpret its meaning, which cannot reflect the time to build up adequate situation awareness, as mentioned previously. Therefore, caution should be exercised when generalizing the outcomes to highly complex and unexpected scenarios, which involve high hierarchical level of control tasks (e.g., respond to an unexpected traffic accident, see Michon, 1985). Zhang et al. (2017) focused on the critical transition trial in which participants had to take over control upon an unexpected system failure and avoid collisions with a suddenly exposed stationary truck. All participants could take over successfully without causing an accident, suggesting that they were capable to handle even more critical events. However, no non-driving tasks were permitted in this critical trial and the results may only apply for drivers that are still in the loop. Future studies could implement a larger variety of scenarios and task conditions, and investigate drivers' take-over performance in more complex situations. Second, the duration of non-driving task engagement was short (8 min). The participants were free to choose how to interact with the tablet and no highly cognitively demanding tasks were instructed to be performed. These factors may generate better take-over performance compared to that measured after prolonged automated drives and after being engaged in intensive cognitive tasks. Third, the utilization of a within-subject design may increase familiarization and expectancy, and generate shorter RTs. Moreover, our study was conducted in a driving simulator, which generally raises the issues of ecological validity such as reduced risk perception, simplified driving environment and reduced workload (Carsten & Jamson, 2011; De Winter, Happee, Martens, & Stanton, 2014; De Winter, Van Leeuwen, & Happee, 2012; Eriksson, Banks, & Stanton, 2017; Godley, Triggs, & Fildes, 2002; Green, 2000). However, several studies have found relative validity and sometimes even absolute validity, depending on the type of simulator (Kaptein, Theeuwes, & Van der Horst, 1996; Riener, 2010; Risto & Martens, 2014). The high-fidelity driving simulator used in our study is designed to generate high behavioural validity.

# 3.5 Conclusion

This study mainly aims to investigate how long it takes truck drivers to get ready to leave a truck platoon in normal operations and how the take-over process is influenced by three task conditions: monitoring the driving environment, interacting with a hand-held tablet, and relaxing with the eyes closed. Additionally, we explored if drivers could cope with a brake event immediately after taking over control. The positive results suggested a sufficient readiness level for moderately complex driving situations.

We found substantial differences in take-over times between the task conditions and large variability between the individual drivers. By exploring perception-response times instead of merely total take-over times, we found different RT patterns for different task conditions. After interacting with a hand-held tablet, the increase in the total take-over time was mainly caused by the hand movement RT to put away the tablet, whereas after relaxing with eyes closed, both longer perception- and movement RTs contributed to the longer total take-over time. Furthermore, the hand movement RT contributed to the total take-over time to a larger extent compared to the perception RT, which can be mainly explained by drivers' activities to resume physical readiness. The results suggest the importance of adaptive approach with personal driver readiness predictors as input parameters for a safe and comfortable transition, and the importance to focus on cognitive and motoric preparation phases before resuming manual control.

# Acknowledgement

This study was funded by the TNO Early Research Program (ERP) Human Enhancement: Adaptive Automation. The first author and the last author are involved in the Marie Curie Initial Training Network (ITN) project HFAuto – Human Factors of Automated Driving (PITN-GA-2013-605817). We would like to thank Ingmar Stel and Rob van de Pijpekamp for their technical assistance in this study, and all participants for their valuable contribution.

# Reference

Alam, A., Besselink, B., Turri, V., Martensson, J., & Johansson, K. H. (2015). Heavy-duty vehicle platooning for sustainable freight transportation: A cooperative method to enhance safety and efficiency. *IEEE Control Systems*, *35*(6), 34–56. doi:10.1109/MCS.2015.2471046

Baldwin, C. L., & Lewis, B. A. (2014). Perceived urgency mapping across modalities within a driving context. *Applied Ergonomics*, 45(5), 1270–1277. doi:10.1016/j.apergo.2013.05.002

Bazilinskyy, P., & De Winter, J. C. F. (2015). Auditory interfaces in automated driving: an international survey. *PeerJ Computer Science*, *1*, e13. doi:10.7717/peerj-cs.13

Bergenhem, C., Shladover, S., Coelingh, E., Englund, C., & Tsugawa, S. (2012). *Overview of platooning systems*. Paper presented at the 19th ITS World Congress, Vienna, Austria.

Carsten, O., & Jamson, A. H. (2011). Driving simulators as research tools in traffic psychology. In B. E. Porter (Ed.), *Handbook of traffic psychology* (pp. 87-96). Academic Press. doi: 10.1016/B978-0-12-381984-0.10007-4

Carsten, O., & Martens, M. H. (2019). How can humans understand their automated cars? HMI principles, problems and solutions. *Cognition, technology and work.* DOI: 10.1007/s10111-018-0484-0

De Winter, J. F. C., Van Leeuwen, P. M., & Happee, R. (2012). Advantages and disadvantages of driving simulators: A discussion. In A.J. Spink, F. Grieco, O.E. Krips, L.W.S. Loijens, L.P.J.J. Noldus, & P.H. Zimmerman (Eds.), *Proceedings of Measuring Behavior* (pp. 47-50). Retrieved from:

http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.388.1603&rep=rep1&type=pdf#pa ge=47

De Winter, J. C. F., Happee, R., Martens, M. H., & Stanton, N. A. (2014). Effects of adaptive cruise control and highly automated driving on workload and situation awareness: A review of the empirical evidence. *Transportation Research Part F: Traffic Psychology and Behaviour*, 27, 196–217. doi:10.1016/j.trf.2014.06.016

Dempster, A. P., Laird, N. M., & Rubin, D. B. (1977). Maximum likelihood from incomplete data via the EM algorithm. *Journal of the royal statistical society, Series B (methodological)*, 39 (1), 1-38.

Endsley, M. R. (1995). Toward a theory of situation awareness in dynamic systems. *Human Factors: The Journal of the Human Factors and Ergonomics Society*, 37(1), 32–64. doi:10.1518/001872095779049543

Endsley, M. R. (2017). Autonomous Driving Systems: A Preliminary Naturalistic Study of the Tesla Model S. *Journal of Cognitive Engineering and Decision Making*, *11*(3), 225-238. doi:10.1177/1555343417695197

Engström, J., Aust, M. L., & Viström, M. (2010). Effects of working memory load and repeated scenario exposure on emergency braking performance. *Human factors*, *52*(5), 551-559. doi:10.1177/0018720810381072

Eriksson, A., Banks, V. A., & Stanton, N. A. (2017). Transition to manual: Comparing simulator with on-road control transitions. *Accident analysis & prevention*, *102*, 227–234. doi:10.1016/j.aap.2017.03.011

Eriksson, A., & Stanton, N. A. (2017). Takeover time in highly automated vehicles: noncritical transitions to and from manual control. *Human Factors*, *59*, 689–705. doi:10.1177/0018720816685832

European Truck Platooning Challenge (2016). *European Truck Platooning Challenge*. Retrieved from <u>https://www.eutruckplatooning.com/default.aspx</u>

Feldhütter, A., Gold, C., Schneider, S., & Bengler, K. (2017). How the duration of automated driving influences take-over performance and gaze behavior. In C. M. Schlick, S. Duckwitz, F. Flemisch, M. Frenz, & S. Kuz (Eds.), *Advances in ergonomic design of systems, products and processes* (pp. 309–318). Berlin, Heidelberg: Springer. <u>doi:10.1007/978-3-662-53305-5\_22</u>

Godley, S. T., Triggs, T. J., & Fildes, B. N. (2002). Driving simulator validation for speed research. *Accident analysis & prevention*, 34(5), 589–600. doi:10.1016/S0001-4575(01)00056-2

Gold, C., & Bengler, K. (2014). Taking over control from highly automated vehicles. In N. Stanton, S. Landry, G. Di Bucchianico, & A. Vallicelli (Eds.), *Advances in human aspects of transportation: Part II* (pp. 66–69). AHFE Conference. doi:10.1177/0018720816634226

Gold, C., Damböck, D., Lorenz, L., & Bengler, K. (2013). "Take over!" How long does it take to get the driver back into the loop? *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *57*(1), 1938–1942. <u>doi:10.1177/1541931213571433</u>

Gold, C., Körber, M., Lechner, D., & Bengler, K. (2016). Taking over control from highly automated vehicles in complex traffic situations the role of traffic density. Human Factors, 58, 642–652. doi:10.1177/0018720816634226

Green, M. (2000). "How long does it take to stop?" Methodological analysis of driver perception-brake times. *Transportation Human Factors*, 2, 195–216. doi:10.1207/STHF0203 1

Hjälmdahl, M., Krupenia, S., & Thorslund, B. (2017). Driver behaviour and driver experience of partial and fully automated truck platooning–a simulator study. *European transport research review*, *9*(1), 8. doi:10.1007/s12544-017-0222-3

Ito, T., Takata, A., & Oosawa, K. (2016). *Time required for take-over from automated to manual driving* (No. 2016-01-0158). SAE Technical Paper. <u>doi:10.4271/2016-01-0158</u>

Jamson, A. H., Merat, N., Carsten, O. M. J., & Lai, F. C. H. (2013). Behavioural changes in drivers experiencing highly-automated vehicle control in varying traffic conditions. *Transportation research part C: emerging technologies*, 30, 116–125. doi:10.1016/j.trc.2013.02.008

Kaptein, N., Theeuwes, J., & Van der Horst, R. (1996). Driving simulator validity: Some considerations. *Transportation Research Record: Journal of the Transportation Research Board*, 1550, 30–36. doi:10.3141/1550-05

Kerschbaum, P., Lorenz, L., & Bengler, K. (2015). *A transforming steering wheel for highly automated cars.* Paper presented at the Proceedings of the 2015 IEEE Intelligent Vehicles Symposium (IV), Seoul, South Korea (pp. 1287–1292). doi:10.1109/IVS.2015.7225893

Kerschbaum, P., Omozik, K., Wagner, P., Levin, S., Hermsdörfer, J., & Bengler, K. (2017). How does a symmetrical steering wheel transformation influence the take-over process? In D. de Waard, A. Toffetti, R. Wiczorek, A. Sonderegger, S. Röttger, P. Bouchner, T. Franke, S. Fairclough, M. Noordzij, & K. Brookhuis (Eds.), *Proceedings of the Human Factors and Ergonomics Society Europe Chapter 2016 Annual Conference, Prague, Czech Republic* (pp. 96–106).

Körber, M., Gold, C., Lechner, D., & Bengler, K. (2016). The influence of age on the take-over of vehicle control in highly automated driving. *Transportation Research Part F: Traffic Psychology and Behaviour*, *39*, 19–32. doi:10.1016/j.trf.2016.03.002

Körber, M., Schneider, W., & Zimmermann, M. (2015). *Vigilance, boredom proneness and detection time of a malfunction in partially automated driving*. Paper presented at 2015 International Conference on Collaboration Technologies and Systems (CTS) (pp. 70-76), Atlanta, GA. doi:10.1109/CTS.2015.7210402

Louw, T., Kountouriotis, G., Carsten, O., & Merat, N. (2015). Driver Inattention During Vehicle Automation: How Does Driver Engagement Affect Resumption Of Control? Paper presented at the 4th International Conference on Driver Distraction and Inattention, Sydney, New South Wales, Australia. ARRB Group. Retrieved from: http://eprints.whiterose.ac.uk/91858/8/DDI%202015\_%2815381%29\_Driver%20Inattention% 20during%20vehicle%20automation\_TLouw\_GKoutouriotis\_OCarsten.pdf

Louw, T., Merat, N., & Jamson, H. (2015). Engaging with highly automated driving: to be or not to be in the loop? *Proceedings of the Eighth International Driving Symposium on Human Factors in Driver Assessment, Training and Vehicle Design, Salt Lake City, UT* (pp. 190–196). doi:10.17077/drivingassessment.1570

Lu, Z., Coster, X., & De Winter, J. C. F. (2017). How much time do drivers need to obtain situation awareness? A 684 laboratory-based study of automated driving. *Applied Ergonomics*, *60*, 293–304. doi:10.1016/j.apergo.2016.12.003

Lu, Z., Happee, R., Cabrall, C. D. D., Kyriakidis, M., & De Winter, J. C. F. (2016). Human factors of transitions in automated driving: A general framework and literature survey. *Transportation Research Part F: Traffic Psychology and Behaviour, 43*, 183–198. doi:10.1016/j.trf.2016.10.007

Melcher, V., Rauh, S., Diederichs, F., Widlroither, H., & Bauer, W. (2015). Take-over requests for automated driving. *Procedia Manufacturing*, *3*, 2867–2873. doi:10.1016/j.promfg.2015.07.788

Merat, N., Jamson, A. H., Lai, F. C. H., Daly, M., & Carsten, O. M. J. (2014). Transition to manual: Driver behaviour when resuming control from a highly automated vehicle. *Transportation research part F: traffic psychology and behaviour*, 27, 274–282. doi:10.1016/j.trf.2014.09.005

Merat, N., Seppelt, B., Louw, T., Engström, J., Lee, J. D., Johansson, E., ... & Keinath, A. (2018). The "out-of-the-loop" concept in automated driving: Proposed definition, measures and implications. *Cognition, Technology & Work*, 1-12. doi:10.1007/s10111-018-0525-8

Michon, J. A. (1985). A critical view of driver behavior models: what do we know, what should we do? In L. Evans & R. C. Schwing (Eds.), *Human Behavior and Traffic Safety* (pp. 485–524). Boston, MA: Springer.

Monsell, S. (2003). Task switching. *Trends in cognitive sciences*, 7(3), 134–140. Doi:10.1016/S1364-6613(03)00028-7

Musil, C. M., Warner, C. B., Yobas, P. K., & Jones, S. L. (2002). A Comparison of Imputation Techniques for Handling Missing Data. *Western Journal of Nursing Research*, 24(7), 815–829. doi:10.1177/019394502762477004

Naujoks, F., Befelein, D., Wiedemann, K., & Neukum, A. (2017). A review of non-drivingrelated tasks used in studies on automated driving. In N. A. Stanton (Ed.), *Advances in human aspects of transportation*. Cham: Springer. <u>doi:10.1007/978-3-319-60441-1\_52</u>

Petermann-Stock, I., Hackenberg, L., Muhr, T., & Mergl, C. (2013). Wie lange braucht der Fahrer--eine Analyse zu Übernahmezeiten aus verschiedenen Nebentätigkeiten während einer hochautomatisierten Staufahrt [How long does the driver need - an analysis of take-over times from various non-driving tasks during highly automated traffic jam driving]. *6. Tagung Fahrerassistenzsysteme. Der Weg zum automatischen Fahren.* TÜV SÜD Akademie GmbH.

Petermeijer, S. M., De Winter, J. C. F., & Bengler, K. (2016). Vibrotactile displays: A survey with a view on highly automated driving. *IEEE Transactions on Intelligent Transportation Systems*, 17, 897–907. doi:10.1109/TITS.2015.2494873

Petermeijer, S. M., Doubek, F., & De Winter, J. C. F. (2017). Driver response times to auditory, visual, and tactile take-over requests: A simulator study with 101 participants. *Proceedings of the IEEE International Conference Systems, Man, and Cybernetics, Banff, AB, Canada.* doi:10.1109/SMC.2017.8122827

Ploeg, J., Van de Wouw, N., & Nijmeijer, H. (2014). Lp string stability of cascaded systems: Application to vehicle platooning. *IEEE Transactions on Control Systems Technology*, 22(2), 786–793. doi:10.1109/TCST.2013.2258346

Radlmayr, J., Gold, C., Lorenz, L., Farid, M., & Bengler, K. (2014). How traffic situations and non-driving related tasks affect the take-over quality in highly automated driving. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 58, 2063–2067. doi:10.1177/1541931214581434

Riener, A. (2010). *Simulating on-the-road behavior using a driving simulator*. Paper presented at the 2010 Third International Conference on Advances in Computer-Human Interactions, Saint Maarten, Netherlands Antilles (pp. 25–31). doi:10.1109/ACHI.2010.10

Risto, M., & Martens, M. H. (2014). Driver headway choice: A comparison between driving simulator and real-road driving. *Transportation Research Part F: Traffic Psychology and Behaviour*, 25, 1–9. doi:10.1016/j.trf.2014.05.001

Roche, F., & Brandenburg, S. (2018). *Should the urgency of auditory-tactile takeover requests match the criticality of takeover situations?* Paper presented at the 21st International Conference on Intelligent Transportation Systems (ITSC) (pp. 1035-1040). IEEE. doi:10.1109/ITSC.2018.8569650

SAE International. (2015). Operational Definitions of Driving Performance Measures and Statistics (J2944\_201506). doi:10.4271/J2944\_201506

SAE International. (2016). *Taxonomy and definitions for terms related to driving automation systems for on-road motor vehicle* (J3016) [Standard]. Warrendale, PA: SAE International.

Stapel, J., Mullakkal-Babu, F. A., & Happee, R. (2017). *Driver Behaviour and Workload in an On-road Automated Vehicle*. Paper presented at RSS2017: Road Safety and Simulation International Conference 2017.

Stapel, J., Mullakkal-Babu, F. A., & Happee, R. (2018). Attentive monitoring of automated driving requires more effort than manual driving. Manuscript Submitted for Publication.

Summala, H. (2000). Brake reaction times and driver behavior analysis. *Transportation Human Factors*, 2, 217–226. doi:10.1207/STHF0203\_2

Van den Beukel, A. P., Van der Voort, M. C., & Eger, A. O. (2016). Supporting the changing driver's task: Exploration of interface designs for supervision and intervention in automated driving. *Transportation Research Part F: Traffic Psychology and Behaviour, 43*, 279–301. doi:10.1016/j.trf.2016.09.009

Wandtner, B., Schömig, N., & Schmidt, G. (2018). Effects of non-driving related task modalities on takeover performance in highly automated driving. *Human Factors*, *60*, 870–881. doi:10.1177/0018720818768199

Willemsen, D. M., Stuiver, A., Hogema, J., Kroon, L., & Sukumar, P. (2014). *Towards guidelines for transition of control*. Paper presented at FISITA World Automotive Congress 2014, Maastricht, Netherlands.

Willemsen, D., Stuiver, A., & Hogema, J. (2015). Automated Driving Functions Giving Control Back to the Driver: A Simulator Study on Driver State Dependent Strategies. Paper presented at the 24th International Technical Conference on the Enhanced Safety of Vehicles, Gothenburg, Sweden.

Wilschut, E. S., Willemsen, D., Hogema, J., & Martens, M. (2016). *Adaptive virtual tow bar and transition of control: A truck driving simulator study* (TNO Report 2016 R11589). Soesterberg: TNO.

Young, M. S., & Stanton, N. A. (2007). Back to the future: Brake reaction times for manual and automated vehicles. *Ergonomics*, 50, 46–58. doi:10.1080/00140130600980789

Zeeb, K., Buchner, A., & Schrauf, M. (2015). What determines the take-over time? An integrated model approach of driver take-over after automated driving. *Accident Analysis & Prevention*, 78, 212–221. doi:10.1016/j.aap.2015.02.023

Zeeb, K., Buchner, A., & Schrauf, M. (2016). Is take-over time all that matters? The impact of visual-cognitive load on driver take-over quality after conditionally automated driving. *Accident Analysis & Prevention*, *92*, 230–239. <u>doi:10.1016/j.aap.2016.04.002</u>

Zeeb, K., Härtel, M., Buchner, A., & Schrauf, M. (2017). Why is steering not the same as braking? The impact of non-driving related tasks on lateral and longitudinal driver interventions during conditionally automated driving. *Transportation Research Part F: Traffic Psychology and Behaviour, 50*, 65–79. doi:10.1016/j.trf.2017.07.008

Zhang, B., De Winter, J.C.F., Varotto, S., Happee, R., & Martens, M.H. (2019). Determinants of take-over time from automated driving: a meta-analysis of 129 studies. *Transportation Research Part F: Traffic Psychology and Behaviour, 64, 285–307.* doi:org/10.1016/j.trf.2019.04.020

# 4. Transitions to manual control from highly automated driving in non-critical truck platooning scenarios (with mounted NDT)

Unpublished manuscript.

### 4.1 Introduction

This study can be seen as a replication of the truck platooning study described in Chapter 3, Both studies used the same truck driving simulator platform and experimental design, but differed in the set-up of the non-driving task performed on the tablet PC. Also, another group of professional truck drivers was invited for this study. In the truck driving simulator study described in Chapter 3, the drivers held the tablet in their hands and were free to interact with it (hand-held). In this study, the tablet was mounted on the centre console (hands-free) and the driver had to perform a standardized visual-motoric task.

This study aims 1) to examine if the findings of the first truck platooning study hold for a different group of professional truck drivers to be generalized to a larger population (replication study), and 2) to investigate the effects of hand-held tablet use compared to hands-free tablet use on drivers' take-over time and take-over quality. The meta-analysis in Chapter 2 pointed out that hand-holding a device is a major determinant of mean take-over response times above many other influential factors, causing an increase of approximately 1 - 1.5 s compared to hands-free task conditions in passenger car studies (e.g., Zeeb, Härtel, Buchner, & Schrauf, 2017; Eriksson & Stanton, 2017; Miller, Sun, Johns, Ive, Sirkin, Aich, & Ju, 2015; Befelein, Naujoks, and Neukum, 2016). Results of the first truck platooning study further revealed a large variation in driver response to put away the tablet PC and to reposition their hands on the steering wheel, which caused a large variation in response times. It is expected that using a mounted tablet could largely shorten the hand movement response time and reduce its variation between drivers. Interestingly, Zeeb et al., (2017) found that using a handheld tablet PC significantly worsened drivers' lane-keeping performance after the transition to manual driving, compared to using a mounted tablet during automated driving. The authors suggested that high motoric task load, such as holding a device in the hands, may impair drivers' take-over quality in lateral vehicle control.

The main aim of this study is to investigated if a similar phenomenon is also found in truck platooning take-over scenarios.

#### 4.2 Methods

#### 4.2.1 Participants

Twenty-three participants (of which 2 females) took part in the experiment. They all held a truck driver's license for at least two years (M = 23 years, SD = 8.8) and drove at least 1000 km per year in a truck (M = 85600 km/year, SD = 42225). On average, the participants were 45 years old (SD = 8.1), ranging from 27 to 58 years old. The research was approved by the Ethical Committee for participant studies of TNO.

#### 4.2.2 Apparatus

In this experiment, the configuration of the truck driving simulator, the simulation of the twotruck platooning system, and the human-machine interface were identical to the first truck platooning study as described in Chapter 3, and therefore will not be repeated here. Driving performance data related to vehicle motion control and driver response times were recorded by the driving simulator software at a sampling rate of 50 Hz. Three cameras were installed in the truck cabin to observe the drivers' face expression, hand movement, and feet movement, respectively.

#### 4.2.3 Experimental design and test scenarios

Similar to the first truck study, a within-subject design was used and each participant performed eight test trials in different experimental conditions, including two baseline manual driving trials and six non-critical platooning trials. The order of the trials was counterbalanced. In each test trial, the participants drove in the right-hand lane (the slower lane) of a two-lane motorway behind a lead truck that was driving with a speed of 80 km/h. They were instructed to follow this lead truck and not change lanes. There were no entries or exits on the route. Slight curves and surrounding traffic were present to simulate realistic but low traffic volume conditions. The drives were always on the same road layout, but the surroundings were different to simulate a different stretch of road to the driver. Before the formal test, the participants were introduced to the concept of platooning, and practised with activating and deactivating the platooning systems in a six-minute-long training drive.

In the automated trials, the ego truck of the participant followed the lead truck with a time gap of 0.8 s. Four minutes after the scenario started, a visual-auditory take-over request (TOR) was issued and the participants had to disengage the automation and take over manual control by pressing a button normally used for cruise control on the right side of the steering wheel. No time restraints were applied and the participants were instructed to press the button whenever they felt ready to resume manual control. After the transition, the participants drove manually for another 2.5 minutes until the end of the scenario.

During the automated driving, the participants were instructed to either monitor the driving situation, or perform a visual-motoric task '*Arrows*' (Engström, Johansson, & Östlund, 2005), or relax with their eyes closed. Performing a standardized task instead of freely using the tablet controlled for workload and allowed for measures of task involvement. Each task condition was assigned to two test trials. The *Arrows* task was performed on a tablet PC mounted on the centre console of the cabin. On the touch screen, metrics (5×5) of arrows were presented with or without a target arrow pointing upwards (among arrows pointing left or right, see Figure 4.1 as an example). The participants determined whether the target arrow was present, and gave their answers by pressing "yes" or "no" on the touch screen. After providing the answer, a new metric was presented automatically.

In one of the two trials within the same task condition, a brake event was implemented immediately after the control transition. The lead truck braked at the deceleration rate of  $5 \text{ m/s}^2$  for 2 s immediately after the driver pressed the button to switch off the automation. The participants had to brake in time to avoid a collision. This was identical to the first truck study.

In the baseline (manual driving) trials, the participants followed the lead truck manually without any assistance from the automation system during a six-minute trial. In one of the trials, the

same brake event as described above was implemented five minutes after the scenario started. Table 4.1 shows an overview of all test conditions analysed in this study.



Figure 4.1: Metrix of arrows with a target arrow pointing upwards.

		•	
Condition	Task during AD	Brake event	Duration (min)
1 (MD)	-	no	6
2 (MD)	-	yes	6
3 (AD)	Monitoring	no	4 (AD) + 2.5 (AD)
4 (AD)	Monitoring	yes	4 (AD) + 2.5 (AD)
5 (AD)	Arrow task	no	4 (AD) + 2.5 (AD)
6 (AD)	Arrow task	yes	4 (AD) + 2.5 (AD)
7 (AD)	Eyes closed	no	4 (AD) + 2.5 (AD)
8 (AD)	Eyes closed	yes	4 (AD) + 2.5 (AD)

 Table 4.1:
 Overview of test conditions analysed in this study.

#### 4.2.4 Dependent variables

Driver take-over response times, post-transition manual driving performance, and response times and response quality in the brake event were analysed and compared between task conditions. To analyse drivers' take-over process, the total take-over response time (TRT), the perception response time (PRT), and the hand movement response time (MRT) were measured in the same manner as in the first truck platooning study. TRT was defined as the time interval between the onset of the TOR until the moment when the participant pressed the button to reclaim control. PRT was measured from the onset of the TOR until the start of the participant's hand movement to grasp the steering wheel, indicating the time elapsed to perceive and understand the necessity to take over control. MRT was the remaining time measured until the

driver pressed the button, indicating the time elapsed to regain motoric readiness for manual driving. Participants' body postures at the TOR onset, and activities during the take-over process were logged from the video recordings by two annotators. A high degree of reliability between the observers was found (the average intraclass correlation coefficient (ICC) was 0.955 with a 95% confidence interval from 0.943 to 0.965). All take-over response time measures were analysed for experimental trials without brake events.

For the trials without brake events, drivers' post-transition manual driving performance was assessed in terms of standard deviation of lateral position (SDLP), the mean and standard deviation of longitudinal speed, and the mean time headway (THW). Results of the first truck experiment showed that SDLP and speed related performance were stabilized within 10- 20 s after the transition, while the THW continued to increase until 90 – 100 s after the transition. In this study, we therefore analysed the SDLP, and mean and standard deviation of longitudinal speed in time windows of 10 s for 40 s after the control transition. THW was observed until 120 s after the control transition. SDLP and THW were also compared to the performance measures in the baseline manual driving condition to explore carry-over effects of platooning, as suggested in Skottke, Debus, Wang, and Huestegge (2014). The observation time window in the baseline condition was 4.5 minutes after the scenario start during 40 s and 120 s.

For the trials with brake events, we measured brake response times, minimum time-to-collision, and maximum deceleration rate to evaluate participants' response to the braking lead truck. Brake response time was defined as the time interval between the activation of the brake lights of the lead vehicle and the moment when the driver pressed the brake pedal.

### 4.3 Results

Since the data of the first participant were lost, the data of 22 participants were analysed. Eight trials divided over six participants were excluded from the analysis of take-over response times due to missing video recordings (in five platooning trials) or participants not obeying the instructions (starting manual driving before pressing the button, in three platooning trials). Data from all trials were valid for the analysis of post-transition manual driving performance. All statistical analyses were conducted using SPSS 24. The significance level used was 0.05.

#### 4.3.1 Take-over response times

An overview of drivers' take-over response times measured in three task conditions is shown in the stacked bar plot (Figure 4.2). The descriptive data of the three response time measures are presented in the table below.

The total take-over times ranged from 0.9 s to 13.27 s (on average 2.99 s) combining all task conditions. Results of Friedman tests indicated significant differences in TRT ( $\chi^2(2) = 21.50$ , p < .001), PRT ( $\chi^2(2) = 20.67$ , p < .001), and MRT ( $\chi^2(2) = 8.00$ , p < .02), between the three task conditions. Post-hoc Wilcoxon signed-rank tests using a Bonferroni-Holm correction revealed that differences in TRTs were significant between each of the two task conditions (*Monitoring* vs. *Arrows task*: p = .005; *Monitoring* vs. *Eyes-closed*: p < .001; *Arrows task* vs. *Eyes-closed*:

p = .022), which is also the case for PRTs (*Monitoring* vs. Arrows task: p = .019; Monitoring vs. Eyes-closed: p < .001; Arrows task vs. Eyes-closed: p = .011). Significant differences in MRTs were only found between the Monitoring condition and the Eyes-closed condition (p = .003). Additional pairwise comparisons were conducted to compare PRTs and MRTs in each task condition. Results showed that MRTs were significantly larger than PRTs in the Monitoring condition (p = .002) and the Arrows test condition (p = .005). No significant differences between PRTs and MRTs in the Eyes-closed condition were indicated.



Figure 4.2: Driver take-over response times in seconds across the three conditions. The bars show the means of MRTs and PRTs per condition. They add up to the TRTs. IQR = Interquartile range.

#### 4.3.2 Post-transition manual driving performance

Three  $3 \times 4$  repeated measures ANOVA were performed to examine the effects of *task* conditions and *time* elapsed after the transition on drivers' SDLP, mean longitudinal speed, and standard deviation of longitudinal speed, respectively. With respect to SDLP (Figure 4.3), there

were a significant main effect of *time* (F(3, 63) = 7.64, p < .001,  $\eta^2 = .27$ ), no main effect of *task*, and a significant interaction between *task* and *time* (F(6, 126) = 3.41, p = .004,  $\eta^2 = .14$ ). Post-hoc pairwise comparisons revealed a significantly higher SDLP in the first 10 s in the *Monitoring* condition. No differences between timeslots were found in the other two conditions. When comparing between task conditions, no significant differences were found in any timeslots. Additionally, we compared the post-transition SDLP to that measured in the baseline (manual) driving condition. Post-transition SDLP was significantly higher than baseline SDLP only in the first 10 s in the *Monitoring* condition.



Figure 4.3: Mean standard deviation of lateral position (SDLP) in the first 40 s after the control transition back to the driver measured in three task conditions (purple bars) and the baseline driving condition (the grey bar). The error bars represent the standard deviations.

Figure 4.4 and Figure 4.5 depict drivers' performance with respect to longitudinal speed and its variation. When examining the effects on longitudinal speed, we found significant main effects of both *task* (F(2, 42) = 5.84, p = .006,  $\eta^2 = .22$ ) and *time* (F(1.69, 35.54) = 27.66, p < .001,  $\eta^2 = .57$ , degrees of freedom adjusted using Greenhouse-Geisser estimates of sphericity), but no significant interaction between the main effects. Post-hoc pairwise comparisons showed that the mean speed significantly decreased in the first 20 s after the transition, and gradually increased between 30 - 40 s. Effects of *task* conditions became noticeable from 10 s after the transition: Participants drove with a significantly lower mean speed in the *Arrows task condition* and the *Eyes-closed* condition compared to the *Monitoring* condition.

There was a main effect of *time* (*F* (3, 63) = 59.16, p < .001,  $\eta^2 = .74$ ) on the variation in longitudinal speed. The main effect of *task* and the interaction between the main effects were not significant. Post-hoc pairwise comparisons showed that the variation in speed significantly decreased in the first 20 s and came to a stable status in the following timeslots.


Figure 4.4: Mean longitudinal speed in the first 40 s after the control transition back to the driver. The error bars represent the standard deviations.



Figure 4.5: Mean standard deviation of longitudinal speed in the first 40 s after the control transition back to the driver, measured in three task conditions. The error bars represent the standard deviations.

With respect to the regulation of the following distance (Figure 4.6), a  $3 \times 12$  repeated measures ANOVA showed significant main effects of *task* (*F* (2, 42) = 4.94, *p* =.012,  $\eta^2$  = .19) and *time* (*F* (1.43, 29.96) = 55.64, *p* <.001,  $\eta^2$  = .73). The interaction between the main effects was marginally significant (*F* (3.30, 69.36) = 55.64, *p* = .087,  $\eta^2$  = .10). The degrees of freedom for *time* and the interaction were adjusted using Greenhouse-Geisser correction due to the violation

of sphericity. Post-hoc pairwise comparisons revealed that in the *Monitoring* and *Eyes-closed* conditions, THW continued to increase until 70 - 80 s after the control transition. In the *Arrows* task condition, a significant increase in THW was not observed after 50 - 60 s. Differences between task conditions were observed between 40 s to 100 s after the control transition. THW in the *Monitoring* condition was significantly lower than the other two conditions between 40 - 60 s. From 60 s to 100 s, significant differences only existed between the *Monitoring* and the *Eyes-closed* conditions. THW in the *Arrows task* and the *Eyes-closed* conditions did not differ significantly in any of the timeslots, despite the tendency of a higher THW in the *Eyes-closed* condition. We performed additional pairwise comparisons to examine the carry-over effects of platooning on THW, and found that the post-transition THW in the *Monitoring* and the *Arrows task* conditions were significantly lower than the baseline TWH throughout the observation window. In the *Eyes-closed* condition, significant differences between the post transition THW and the baseline THW disappeared 80 – 90 s after the transition.



Figure 4.6: Mean time headway (THW) in the first 120 s after the control transition back to the driver measured in three task conditions (purple bars) and in the baseline driving condition (the grey bar). The error bars represent the standard deviations.

#### 4.3.3 Response in the post-transition brake event

In response to the brake event immediately after the control transition, one participant collided with the decelerating lead truck in the initial *Eyes-closed* condition due to late response. Minimum TTC and maximum decelerating in this trial were therefore not included in the analysis, although interesting from an outlier and traffic safety perspective. Three one-way repeated measures ANOVA were performed to compare brake response times, minimum TTC, and maximum deceleration between task conditions. The main effect of task conditions was not significant with respect to any of the performance measures. Table below showed the descriptive statistics and the results of the ANOVAs.

	Monitoring	Arrows task	Eyes-closed	Repeated measures ANOV		OVA	
	M (SD)	M (SD)	M (SD)	F	df	р	$\eta^2$
Brake response time (s)	1.09 (0.36)	1.02 (0.34)	1.19 (0.38)	1.22	2, 42	n.s.	.06
Minimum TTC (s)	2.50 (1.05)	2.85 (1.25)	2.55 (1.23)	0.60	2, 40	n.s.	.03
Maximum Deceleration (m/s <sup>2</sup> )	5.66 (0.87)	5.75 (0.82)	5.94 (0.70)	1.79	2, 40	n.s.	.08

Table 4.2: Means and standard deviations of the brake response measures, as well as the results of repeated measure ANOVAs.

# 4.4 Discussion

#### 4.4.1 Take-over response times

Take-over response times showed significant differences between the three conditions. The *Monitoring* condition yielded the shortest mean total response time, followed by the *Arrows task* condition and the *Eyes-closed* condition. In line with the first study, hand movement time was longer than the perception time except for the *Eyes-closed* condition, in which the perception time increased largely and did not show significant difference compared to the hand movement time. In both experiments, extremely long take-over times (outside the 95<sup>th</sup> percentile, i.e., larger than 8.6 s) were all generated in the *Eyes-closed* condition, by participants seemingly fallen asleep when the TOR was issued.

As expected, the effect of hand-held or hands-free tablet use showed a large impact on takeover times when comparing between the two studies. When performing a task on a mounted tablet instead of holding the tablet in the hands, truck drivers took over on average 2.8 s faster with a smaller variation. This was mainly due to the largely reduced hand movement time while using the mounted tablet (3 s faster than the hand-held condition in Chapter 3), which no longer significantly differed from the hand movement time in the *Monitoring* condition. Also worth noting was that drivers' perception time while performing the *Arrows* task was significantly longer than the *Monitoring* condition. In the first study, drivers did not show increased perception time when freely interacting with the tablet compared to monitoring the driving situation. It could be that the standardized task in the second experiment required more engagement and generated a higher level of workload than naturalistic task (Shinar, Tractinsky, & Compton, 2005). Another possible explanation is that participants might have checked the driving situation less frequently when performing the *Arrows* task (the next level automatically started after giving the answer) than using the tablet freely, which negatively influenced takeover response times (Zeeb, Buchner, & Schrauf, 2015).

#### 4.4.2 Post-transition manual driving performance

In line with the first study, drivers in all task conditions could stabilize the lateral and longitudinal vehicle motion within 10 - 20 s after the control transition, seeing that the SDLP and the variation of longitudinal speed no longer significantly varied in the successive timeslots (Figure 4.3, Figure 4.5). In the first study, drivers showed worse lane-keeping performance in

the *Tablet* and the *Eyes-closed* conditions compared to the *Monitoring* condition in the first 10 s after the transition, which was not observed in this study. It could be that using a mounted tablet instead of a handheld tablet reduced motoric task load and led to a better steering performance immediately after the transition (the mean SDLP was 0.12 m smaller), as suggested in Zeeb et al., (2017). Nevertheless, this could not explain the differences between the two studies regarding the *Eyes-closed* condition (i.e., the differences could be merely by chance). More studies are needed to examine the relation between motoric task load and take-over quality in lateral control for valid conclusions.

Truck drivers in both studies showed similar behaviours in regulating the longitudinal speed and the following distance after the control transition. Drivers reduced the speed within the first 20 - 30 s to rapidly increase the THW, then gradually increased the speed in the following timeslots. The mean THW continued to increase until 1 - 1.5 minutes after the transition, then reached a relatively stable status. The carry-over effect of platooning on THW was still significant two minutes after the control transition, particularly in the *Monitoring* and the *Tablet* conditions. It has to be noted that the baseline manual driving THW was generated in lowvolume traffic conditions, which is usually 1-2 s higher than that measured in high-volume traffic conditions (Ayres, Li, Schleuning, & Young, 2001). Within two minutes after the transition, THWs in all conditions surpassed the safe time headway requirement of 2 s, and can be considered acceptable even though they still appeared lower than the baseline THW. Results of both studies also pointed to the tendency of a lower post-transition THW in the *Monitoring* condition. The carry-over effect appeared to be stronger when drivers were monitoring the lead truck during the platooning, and therefore may have visually adapted to the small gap between vehicles.

# 4.4.3 Performance in the brake event

When examining drivers' response in the brake event immediately after the control transition, no significant differences in brake response times, minimum TTC, and maximum deceleration were found between the monitoring and the tablet conditions, in line with the findings of the first study. Also noteworthy was that one crash occurred in the *Eyes-closed* condition, but not in the other task conditions. This finding, together with the high perception take-over response times generated in this condition, suggested that resting with eyes closed impairs drivers' cognitive performance when resuming manual control from automated driving.

# 4.5 Conclusion

In general, the findings of this replication study are in line with the results of the first truck platooning study, except for the differences related to the tablet PC. Using a mounted tablet PC instead of holding the device in the hands largely reduced the hand movement response time and its variation between drivers, and therefore largely reduced the total response time. Drivers should not be allowed to use handheld device if the take-over situation can be urgent. The results of this study also suggested that using a handheld tablet may impair lane-keeping performance after taking over control. Further research is needed to draw generic statements and conclusions for a larger truck driver population. This study also pointed to the potential risk of resuming

manual driving after sleeping or relaxing with eyes closed. Extra efforts should be made to either prevent truck drivers from sleeping, or to allow this only in case that safe platooning over longer periods of time could be guaranteed. However, it should be carefully taken into account that more research would need to be done to assist these drivers with a smooth transition and a rapid resumption of situation awareness and cognitive performance after having rested with their eyes closed.

#### Acknowledgement

This study was funded by the TNO Early Research Program (ERP) Human Enhancement: Adaptive Automation. We would like to thank Ingmar Stel and Rob van de Pijpekamp for their technical assistance in this study.

#### References

Ayres, T. J., Li, L., Schleuning, D., & Young, D. (2001). *Preferred time-headway of highway drivers*. Paper presented at the 4th International IEEE Conference on Intelligent Transportation Systems (pp. 826-829), Oakland, CA. doi:10.1109/ITSC.2001.948767

Befelein, D., Naujoks, F., & Neukum, A. (2016). *How do naturalistic non-driving-related tasks influence take-over time and quality during highly automated driving*. Poster session presented at human factors and ergonomics society Europe chapter 2016 annual conference. Prague, Czech Republic.

Engström, J., Johansson, E., & Östlund, J. (2005). Effects of visual and cognitive load in real and simulated motorway driving. *Transportation Research Part F: Psychology and Behaviour*, 8, 97–120. doi:10.1016/j.trf.2005.04.012

Eriksson, A., & Stanton, N. A. (2017). Takeover time in highly automated vehicles: Noncritical transitions to and from manual control. *Human Factors*, *59*, 689–705. doi:10.1177/0018720816685832

Miller, D., Sun, A., Johns, M., Ive, H., Sirkin, D., Aich, S., & Ju, W. (2015). Distraction becomes engagement in automated driving. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *59*, 1676–1680. doi:10.1177/1541931215591362

Shinar, D., Tractinsky, N., & Compton, R. (2005). Effects of practice, age, and task demands on interference from a phone task while driving. *Accident Analysis & Prevention*, *37*, 315–326. doi:10.1016/j.aap.2004.09.007

Skottke, E. M., Debus, G., Wang, L., & Huestegge, L. (2014). Carryover effects of highly automated convoy driving on subsequent manual driving performance. *Human factors*, *56*, 1272-1283. doi:10.1177/0018720814524594

Zeeb, K., Buchner, A., & Schrauf, M. (2015). What determines the take-over time? An integrated model approach of driver take-over after automated driving. *Accident Analysis & Prevention*, 78, 212–221. doi:10.1016/j.aap.2015.02.023

Zeeb, K., Härtel, M., Buchner, A., & Schrauf, M. (2017). Why is steering not the same as braking? The impact of non-driving related tasks on lateral and longitudinal driver interventions during conditionally automated driving. *Transportation Research Part F: Traffic Psychology and Behaviour*, *50*, 65–79. doi:10.1016/j.trf.2017.07

# 5. Taking back manual control after automated platooning: A comparison between car and truck driver's behaviour

This chapter is based on Zhang, B., Hogema, J. H., Willemsen, D. M. C., Wilschut, E. S., & Martens, M. H. (2020). *Taking back manual control after automated platooning: A comparison between car and truck driver's behaviour*. Under review.

# 5.1 Introduction

Thanks to the rapid advancement of technology, intelligent transport systems and advanced driver assistance systems are undergoing an intense development. Cars with automated functions that predominantly rely on on-board sensory systems are already available on the consumer market. Meanwhile, connected and automated vehicles (CAV) with advanced communication technologies are drawing more and more attention for their potential to improve road safety, traffic flow, and energy efficiency (Coppola & Silvestri, 2019; Rios-Torres & Malikopoulos; 2017, Talebpour & Mahmassani, 2016; Shladover, 2018). Combining Vehicle to Vehicle (V2V) and Vehicle to Infrastructure (V2I) communication with on-board control units, CAVs are able to exchange real-time traffic information and make immediate adjustment to the dynamics of other connected vehicles, and travel smoothly with a much shorter following distance compared to regular manually driven vehicles and stand-alone automated vehicles (Coppola & Silvestri, 2019; Shladover, 2018; Talebpour & Mahmassani, 2016).

In recent decades, technical research and development on CAVs mainly focused on platooning of heavy-duty trucks, largely driven by the increasing demand for fuel efficient freight transportation (Bhoopalam, Agatz, & Zuidwijk, 2018; Janssen, Zwijnenberg, Blankers, & De Kruijff, 2015; Turri, Besselink, & Johansson, 2016; Tsugawa, 2013). Simulations and field experiments have demonstrated that platooning with a gap between 4 - 10 m could save up to 10% - 20% fuel consumption for the following trucks achieved by reduced aerodynamic drag (Browand, McArthur, & Radovich, 2004; Tsugawa, Jeschke, & Shladover, 2016), and bring immediate profit to freight carriers once adopted (Janssen et al., 2015). During the European Truck Platooning Challenge in April 2016, truck platoons with automated longitudinal control were brought onto public roads for the first time while crossing international borders, showing the technical and commercial feasibility of platooning technology. Besides truck platoons, passenger car platoons and heterogeneous platoons consisting of heavy vehicles and passenger cars are also realistic use cases when operating platoons in general traffic (Bergenhem, Huang, Benmimoun, & Robinson, 2010; Bergenhem, Shladover, Coelingh, Englund, & Tsugawa, 2012), which receive less attention due to a slower return on investment (Tsugawa et al., 2016; Janssen et al., 2015).

# 5.1.1 Human driver's role in platooning

Various platooning concepts have been discussed and developed that operate at different levels of driving automation. The role and responsibility of the driver also vary. A widely adopted platooning concept is to manually drive the lead vehicle by a professional truck driver, while other vehicles follow automatically with a professional driver at the driving seat, connected and coordinated through wireless communication and radar technology such as Cooperative Adaptive Cruise Control (CACC, Ploeg, Van de Wouw, & Nijmeijer, 2014). In order to come together and get in the right platooning conditions, all vehicles still need to be manually driven. Thus, platoons can be operated on existing road infrastructures without the need of dedicated lanes.

In the early development phase and in many demonstrations on public roads, platooning systems only enabled longitudinal automation of the following vehicles, while steering was still executed by human drivers (Level 1 driving automation according to SAE taxonomy, 2018). At the current stage, partial automated platoons that execute both lateral and longitudinal vehicle control under constant supervision of the driver (SAE Level 2) have been realized on test tracks, and are expected to be commercialized as a next stage (Janssen et al., 2015). Experts have expressed concerns regarding this level (Kyriakidis et al., 2019; Norman, 2015; Onnasch,

Wickens, Li, & Manzey, 2014) because humans are naturally poor at prolonged monitoring tasks (Davies & Parasuraman, 1982; Mackworth, 1948; Parasuraman 1987; Scerbo, 2001) and it can be very risky for the driver in a platoon to respond to longitudinal critical events due to very short inter-vehicular distances (Nowakowski, Shladover, & Tan, 2015; Willemsen et al., 2018). The long-term goal is to implement highly automated platoons that do not require constant monitoring nor require drivers as emergency backups within the operational design domain (i.e., SAE Level 4). It is generally believed that only at such level could drivers truly benefit from driving automation, because they could make use of the travel time to relax or undertake other non-driving tasks (Kyriakidis et al., 2019; Carsten & Martens, 2018). Transitions of control between the system and the driver would still be expected when forming and dissolving the platoon, and occasionally after the system encounters sensors limitations or an upcoming end of the operational design domain.

#### 5.1.2 Transitions of control and related research

Similar to the introduction of cars with automated functions, the deployment of platooning technology on public roads is likely to go through several intermediate phases. As long as human involvement is still required, the human-system interaction involving various types of platoon drivers would need to be a major research focus. The transition of control from the automation function back to the driver, also known as driver take over, is widely recognized as a critical phase where adverse situations may occur (for an overview, see Lu, Happee, Cabrall, Kyriakidis, and De Winter, 2016). Compared to manual driving, taking back control from automated driving asks for additional information processing tasks, including perception of the stimulus that initiates the transition (usually a take-over request issued by the system, or an event in the environment), comprehending the driving situation and regaining situational awareness, making a decision on what to do, and actually perform the take-over action (braking, accelerating, or steering) after repositioning hands back on the steering wheel and feet back on the pedals (Gold & Bengler, 2014; Zeeb, Buchner, & Schrauf, 2015, 2016). The key to a successful control transition is to complete the take-over process with sufficient quality before the driver loses control of the situation, thereby avoiding collisions or endangerment (Naujoks, Wiedemann, Schömig, Jarosch, & Gold, 2018; Nilsson, Falcone, & Vinter, 2015). To understand how long it takes for a driver to adequately take over control provides important input for the development of safe driving automation systems.

A large number of studies have investigated drivers' take over performance under various conditions, mostly in stand-alone automated car scenarios. To address the essential question of take-over response times, Zhang, De Winter, Varotto, Happee, and Martens (2019) conducted an exhaustive meta-analysis of 129 studies that involved control transition from automated driving to manual (of which three were truck automation studies). Results showed that mean (average over different participants or groups) take-over times reported in the studies vary largely from less than 1 s to above 20 s under the interplay of diverse factors related to the driver, the type of automation, the human machine interface (HMI), and the driving environment. The most significant determinants of the take-over time are the urgency of the take-over situation, whether non-driving tasks are performed on a hand-held device, and familiarization and expectancy with the take-over scenarios. The authors, along with a number of researchers, also cautioned that a short take-over response time alone cannot guarantee safe take-over behaviour. In more urgent situations, drivers tend to respond faster but with lower quality, characterized by a higher accident rate (Mok, Johns, Lee, Miller, Sirkin, Ive, & Ju, 2015), more abrupt manoeuvring of the vehicle (Gold, Damböck, Lorenz, & Bengler, 2013; Ito, Takata, & Oosawa, 2016; Clark & Feng, 2017), and poorer hazard perception (Vlakveld, Van Nes, De Bruin, Vissers, & Van der Kroft, 2018). A widely accepted assumption is that 7 - 10 s is sufficient for distracted drivers to take over in response to a critical event without crashing (e.g., Ito et al., 2016; Melcher, Rauh, Diederichs, Widlroither, & Bauer, 2015; Petermann-Stock, Hackenberg, Muhr, & Mergl, 2013; Walch, Mühl, Kraus, Stoll, Baumann & Weber, 2017). However, Gold et al. (2013) found that given a 7 s time budget, drivers still exhibited more risky evasive manoeuvres with fewer mirror checks compared to the baseline manual driving condition. Eriksson and Stanton (2017) assert that take-over times measured in non-critical scenarios could better reflect the time the driver actually needs to resume control, and applied no time restrictions in their driving simulator study with passenger car drivers. Participants displayed a large variation in take-over times even without performing non-driving tasks, ranging from 1.9 s to 25.7 s. Lu, Coster, and De Winter (2017) explored time required to regain situation awareness when taking over control, and found that up to 20 s may be needed for the participants to sufficiently comprehend the traffic situation presented in the video clips.

Research on driver take-over performance in platooning scenarios has received less attention. A series of driving simulator studies rooted in the German project KONVOI investigated drivers' manual driving performance immediately after decoupling from a highly automated passenger car platoon (Eick & Debus, 2005; Wille, Röwenstrunk, & Debus, 2008; Brandenburg & Skottke, 2014; Skottke, Debus, Wang, & Huestegge, 2014). The results generally pointed to a decrease in time headway (THW) after leaving the automation mode compared to drivers' preferred THW in normal manual driving, which may have been a result of behavioral adaptation to the short gaps as suggested by Martens and Jenssen (2012). In particular, Skottke et al., (2014) looked into the change of THW as a function of time, and found that such carryover effects lasted for 10 km (approximately 6 minutes). Impaired lane keeping performance (in terms of increased standard deviation of lateral position, SDLP) after leaving the platoon was also observed in these studies. However, Skottke et al., (2014) argued that this might be due to the long time spent in the experiment (time-on-task effect) rather than the effect of automation, since SDLP increased with similar magnitude in the manual driving condition as the same amount of time passed by. In a recent study, Castritius et al. (2020) compared professional truck drivers' pre- and post-platoon manual driving performance under real traffic conditions on public roads (with the THW of 0.6 s or 0.9 s at the speed of 80 km/h). Drivers in the following truck displayed significantly higher SDLP in the post-platooning section, but no significantly differences in THW between the two sections. The authors cautioned that confounding variables induced by the real road setting, such as the behaviour of surrounding road users and traffic volume, might lead to the inconsistency with the previous findings.

Truck platoon drivers' take-over response in critical system failure scenarios was first tackled in the Japanese Energy ITS project (Zheng, Nakano, Yamabe, Aki, Nakahra, & Suda, 2014; Yamabe, Zheng, Nakano, Suda, Takagi, & Kawahara, 2012). In their experiments, participants were instructed to apply emergency braking as soon as the braking light of the preceding truck was activated (with a THW of 0.45 s at a speed of 80 km/h). Results showed an average takeover response time around 0.6 s, and that in this case (the brake response time being larger than the THW during platoon driving) rear-end collisions could only be avoided when the mean maximum deceleration of the following truck was higher than the preceding truck. Response times here were low since drivers needed to stay alert and knew that they would encounter critical events.

In recent years, several studies conducted by TNO systematically researched drivers' take-over performance from highly automated platooning, aiming to provide input for an adaptive control transition approach. Zhang, Wilschut, Willemsen, and Martens (2019) and Wilschut, Willemsen, Hogema, and Martens (2016) reported truck driving simulator findings, in which professional truck drivers performed eight non-critical control transitions to manual driving

without time restrictions under three task conditions. Results showed that drivers used significantly more time to take over control when interacting with a hand-held tablet or relaxing with their eyes closed compared to monitoring the road without non-driving tasks. A large variation in drivers' response to reposition hands on the steering wheel was also observed, which in turn led to a large variation in the total take-over times.

#### 5.1.3 Difference between passenger car drivers and professional truck drivers

As platooning technology can be applied to both trucks and cars, it is of interest to explore whether professional truck drivers and non-professional passenger car drivers behave differently in a take-over situation after a platooning drive. Heavy trucks and passenger cars foremost differ in physical and operational characteristics such as size, weight, turning radii, power-to-mass ratio, and acceleration/deceleration capability (Mehdizadeh, Shariat-Mohaymany, & Nordfjaern, 2018; Peeta, Zhang, & Zhou, 2005; Ramsay, 1998). To compensate for the vehicle's relatively poor manoeuvrability, heavy-truck drivers normally drive at a lower speed, exhibit a smoother car-following behaviour with a larger distance to the preceding vehicle, and execute lane-changing manoeuvres more slowly with a lower acceleration or deceleration compared to passenger car drivers (Aghabayk, Moridpour, Young, Sarvi, & Wang, 2011; Durrani, Lee, & Zhao, 2016; Moridpour, Rose, & Sarvi, 2010). When driving behind a heavy truck, both car drivers and truck drivers increase following distances compared to following a passenger car (Aghabayk, Sarvi, & Young, 2012; McDonald, Brackstone, Sultan, & Roach, 1999).

Differences between professional truck drivers and non-professional passenger car drivers also exist in their demographics, skill base, and attitude towards safe driving. Compared to average car drivers, truck drivers are predominantly male, have a higher mean age and higher mean annual mileage, spend more hours in traffic working purposes, and experience more practice and training (Rosenbloom, 2011; Rosenbloom, Eldrorb, & Shahara, 2009). Previous research generally points to a more cautious, less risky driving behaviour among truck drivers compared to car drivers, indicated from self-reporting and analysis of naturalistic driving data (Mehdizadeh et al., 2018; Rosenbloom et al., 2009; Walton, 1999). Truck drivers are less likely to commit errors and violations, and are generally well-trained to avoid dangerous situations (Rosenbloom, 2011). Despite the fact that heavy trucks are disproportionally involved in fatal accidents, the large majority of these fatal accidents were caused by cars (Blower, 1998; Rosenbloom, 2011; Thiriez, Radja, & Toth, 2002).

In the context of automated driving, Zhang, Wilschut et al., (2019) and Lotz, Russwinkel, and Wohlfarth (2019) investigated professional truck drivers' take-over performance and compared this to take over performance reported in published passenger car take-over studies (Eriksson & Stanton, 2017; Damböck et al., 2012; Zeeb et al., 2015; Radlmayr, Gold, Lorenz, Farid & Bengler, 2014; Gold, Lorenz & Bengler, 2014). Both studies suggested a shorter take-over response time among professional truck drivers, possibly explained by their higher level of expertise. Nevertheless, the validity of these findings needs further investigation due to the heterogeneity between the studies for comparison (i.e., differences in factors such as experimental design, instruction, the type of simulator that is driven). For example, truck drivers' faster response observed in Zhang, Wilschut et al., (2019) could be due to the effects of practice and familiarization induced by within-subject design, as compared to Eriksson and Stanton (2017) in which between-subject design was used. To date, we are not able to identify studies that directly compare truck drivers' and passenger car drivers' take-over performance controlling for other experimental variables.

# 5.1.4 Research objectives

The literature research pointed to the importance of ensuring a safe control transition to manual in the development of driving automation, and the fact that we are still at an early stage of understanding drive behaviour when decoupling from car and truck platoons. To fill in the research gaps and to enrich the literature on human factors research concerning control transitions in platooning scenarios, this study compares professional truck drivers' take-over performance when leaving a highly automated platoon (as reported in Zhang, Wilschut et al., 2019) to that of car drivers measured in the same TNO driving simulator (with a different vehicle mock-up) using the identical experimental design. The main aim of the study was to investigate behavioural similarities and differences between the two driver groups during the decoupling of platoons and in the subsequent manual driving. The results could be a basis for developing efficient manners to support car and truck platoon drivers to resume manual control safely and smoothly. In addition, research shows that changes in vehicle dynamics during control transitions substantially affect traffic flow efficiency (Varotto, Hoogendoorn, Van Arem, & Hoogendoorn, 2015; Varotto, Farah, Bogenberger, Van Arem & Hoogendoorn, 2020). The results of this study could also provide input for modelling the effects of platooning on traffic flow considering the impact of driver behaviours when dissolving the platoon.

# 5.2 Methods

This paper is based on two separate driving simulator studies conducted in the TNO driving simulator, using an identical experimental design but different simulator configurations (truck and car, with different mock-ups and different vehicle models) and participant groups (truck drivers and car drivers). In the car platooning study, a moving-base car driving simulator experiment was conducted with experienced car drivers. The truck platooning study was a repetition of the car platooning study, with a driving simulator in truck configuration and with professional truck drivers as participants to enable comparisons between the two driver groups. In the following sections, the experimental setup for both experiments are introduced in turn, followed by descriptions of experimental design and dependent measures.

# 5.2.1 Participants

Eighteen participants were involved in the car platooning study. All participants held a driver's license for at least 2 years and drove at least 10.000 km per year. Their average age was 39.5 years old (SD = 9.6). The group consisted of 11 male and 7 female drivers. None of them had experience in driving a truck.

Twenty-two professional truck drivers participated in the truck platooning study. They all held a truck driver's license for at least 8 years (on average 29 years) and drove in a truck at least 10.000 km per year (on average 35218.7 km/year). The average age of the participants was 47.4 years old (SD = 11.5). The group consisted of 20 male and 2 female drivers. Both experiments were approved by the Ethical Committee for participant studies of TNO.

# 5.2.2 Apparatus

Both experiments were conducted in the high fidelity moving-base driving simulator with six degrees of freedom, located at TNO, the Netherlands. In the car platooning experiment, a BMW car mock-up (Figure 5.1 Left), and in the truck platooning experiment a DAF truck mock-up (Figure 5.1 Right), was mounted on the moving base. The two experiments used different mathematical vehicle models to represent the dynamics of a passenger car and a truck,

respectively. Eye heights in the visualisations were also modified to incorporate the differences between the vehicle categories in this respect. In both experiments, the road and traffic environment were projected on cylindrical screens around the vehicle. Three projectors provided the front view with a horizontal viewing angle of 180 degrees and a vertical viewing angle of 41 degrees (22 degrees above and 19 degrees below the neutral viewing direction). The rear view was realised by two screens placed behind the vehicle for exterior mirrors, and one 32-inch LCD screen placed in the back of the car for the interior rear-view mirror (the interior rear-view mirror was not available in the truck cabin). In the car platooning experiment, the human-machine interface (HMI) presenting the status of the automated platooning system was integrated in the dashboard below the speedometer. In the truck platooning experiment, this interface was displayed on an additional tablet PC mounted close to the centre console. Vehicle related data in both experiments were recorded by the driving simulator software at a sampling rate of 50 Hz, including the driver input from the steering wheel and the pedals, and the status change of the automation system. Three cameras from different angles were installed in both cabins to record the participants' facial expressions, full body movement, and feet movement, respectively.



Figure 5.1: Configurations of the TNO moving-base driving simulator and cylindrical projection screen in two experiments. Left: The car platooning experiment with a BMW passenger car mock-up; Right: The same set-up with a DAF truck mock-up.

#### 5.2.3 Automated platooning system

In the study, an automated platooning system was simulated that allowed a vehicle to follow its predecessor at a relatively short following distance, controlling both the longitudinal and lateral motion. The system was designed to operate on public motorways (i.e. without using dedicated lanes), initially limited to platoons of two vehicles. The first vehicle was intended to be driven by a human operator (but was controlled by the simulator scripts in our studies), and the second vehicle was controlled by the automated system once engaged. The automated platooning system was modelled as a combination of Cooperative Adaptive Cruise Control (CACC) and a Lane Keeping System (LKS). The driver could push a button (normally used for cruise control) on the left side of the steering wheel to switch the system on and off. To be able to switch the system on, the driver had to drive in an activation zone behind the lead vehicle. Once activated, the driver did not need to monitor or supervise the system, thus simulating SAE L3/4 automated driving.

Upon approaching the end of the automation zone, the system would issue a take-over request (TOR) by displaying a text message (in Dutch: "*Neem de controle over*", in English: "*Take over control*") on an orange background on the visual interface presenting the automation status,

accompanied with an alert sound. The driver then had to push the button again to deactivate the automated system and take over both longitudinal and lateral control.

#### 5.2.4 Experimental design and test procedures

The experimental design and procedures were identical for both experiments. Prior to driving, the participants read an introduction to the experiment and the automated platooning system, and filled in a questionnaire on their demographics and driving experience. Subsequently, a training session was conducted to familiarize the participants with the driving simulator and with the platooning system. The participants performed the coupling and decoupling procedures at least three times, until they felt familiar and comfortable with the system. After the training, each participants performed one manual driving trial and seven platooning trials. In each trial, the participants were instructed to drive on the right-hand lane of a two-lane motorway, behind a lead vehicle that was driving with an average speed of 100 km/h in the car platoon and 80 km/h in the truck platoon. The participants were instructed to follow this lead vehicle and not to change lanes. There were no entries or exits on the route the participants drove. Slight curves and moderate surrounding traffic were simulated throughout the scenario.

The manual driving trial was always performed first and served as the baseline condition, in which the participants followed the lead vehicle without assistance from automation systems. In the platooning trials, the participants were instructed to drive into the activation zone and switch on the platooning system shortly after the scenario began (approximately 10 - 20 s). In each trial a different experimental condition was implemented by manipulating three independent variables, namely non-driving tasks, the time gap to the lead vehicle in the platoon (0.8 s vs. 0.3 s), and whether a braking event occurred after the participant took over control. The order of the conditions was counterbalanced. This paper only discusses the five trials in which the time gap was set to 0.8 s during automated driving (i.e., the minimum allowed time gap defined in ISO 15622, which is considered to be the accepted distance for testing the first truck platooning concepts).

In each platooning trial, one of three non-driving task conditions was implemented during the automated driving phase. In the *Monitoring* condition, the participants were instructed to constantly monitor the surroundings, so hands-off, feet-off, and eyes on the road; in the *Tablet* condition, the participants were provided with a tablet PC and were asked to use this, so hands-off, feet-off, and eyes off the road, but they were allowed to scan the outside world if they wanted; in *Eyes-closed* condition, the participants were not allowed to open their eyes, so hands-off, feet-off, and eyes off the road. These conditions manipulated participants' visual attention and represented three activities likely to be performed by the driver in future automated vehicles. After four minutes in the *Monitoring* condition, or eight minutes in the *Tablet* and *Eyes-closed* conditions, a TOR was issued and the participants were instructed to take over control by pressing the button whenever they felt ready to do so without time restrictions. The automation duration was shorter for the *Monitoring* condition to increase the possibility that drivers were still paying attention and were alert, and longer for the other two conditions increase the chance of the driver being out-of-the loop. After the transition, the participants continued with manual driving for approximately 90 s until the end of the scenario.

In three trials, no critical events were implemented after the transition to investigate the driver's performance to stabilize the vehicle and regulate the following distance in normal, uneventful situations. In the other two trials (in *Monitoring* and *Tablet* conditions, respectively), a lead vehicle brake event was implemented shortly after the control transition to explore whether the participants possessed sufficient readiness to cope with more complex situations. Three seconds

after the participant pressed the button to deactivate the system, the lead vehicle started braking at a deceleration rate of 5  $m/s^2$  for two seconds. No forward collision warnings or brake assist systems were available, and participants had to respond in time to avoid a potential collision. An overview of all experimental conditions is provided in Table 5.1.

Condition	Task during AD	Post-transition brake event	Duration
0 (MD)	-	-	6.5 min MD
1 (AD)	Monitoring	Ν	4 min AD + 1.5 min MD
2 (AD)	Tablet	Ν	8 min AD + 1.5 min MD
3 (AD)	Eyes-closed	Ν	8 min AD + 1.5 min MD
4 (AD)	Monitoring	Y	4 min AD + 1.5 min MD
5 (AD)	Tablet	Y	8 min AD + 1.5 min MD

Table 5.1: Overview of experimental conditions analysed in this study

*Note*. AD = Automated Driving; MD = Manual Driving.

#### 5.2.5 Dependent measures

In this study, measures regarding take-over response times and take-over quality were assessed and compared between the two experiments, which are introduced in details below.

#### 5.2.5.1 Response times to a take-over request

The total take-over response time was measured from the onset of the TOR until the participant pressed the button to deactivate the platooning system. As in Zhang, Wilschut et al., (2019), the total response time was further divided into perception response time and hand movement response time to analyse the take-over process at a fine-grained level. Perception response time was the response time measured from the onset of the TOR until the participant started to move his/her hands towards the steering wheel, which indicated the time elapsed between perceiving the TOR and understanding the necessity to take over control and starting to take action. Movement response time was the remaining response time measured until the participant pressed the button, indicating the time it took to complete the physical response and to establish motor readiness. The start of the participants' hand movement, the participants' body postures at the moment of the TOR onset, and the activities performed during the take-over process were manually annotated from video recordings to explore underlying factors influencing their takeover response times. In addition, we analysed the gas pedal response time measured from the moment when the participant pressed the button to disengage the automation until the moment when the first input on the gas pedal was made to inspect how the driver regulated speed after overruling the system. All response time measures described in this section were only analysed for experimental trials without brake events.

# 5.2.5.2 Take-over quality

#### 5.2.5.2.1 Post-transition manual driving performance

For the experimental trials without brake events, we analysed the standard deviation of lateral position (SDLP), the mean and standard deviation of longitudinal speed, and the mean time

headway (THW) to assess drivers' performance to stabilize the vehicle in its lane and to maintain a safe following distance after taking over manual control. The SDLP was calculated based on the high-pass filtered (at 0.1 Hz) lateral position signal (to remove effects caused by the driver's low-frequent shifting of lane position, in line with Engström and Markkula, 2007). Merat, Jamson, Lai, Daly, and Carsten (2014) suggested that drivers may need 40-50 s to stabilize the vehicle after the control transition. In our study, the performance measures were analysed in time windows of 10 s for 60 s after the control transition. In addition, we explored carry-over effects of platooning by comparing drivers' mean THW and SDLP after the control transition to their performance in the baseline manual driving condition. To enable a fair comparison, we chose a one-minute timeslot in the baseline condition, starting at five minutes after the start of the scenario, divided in six time windows of 10 s.

#### 5.2.5.2.2 Post-transition brake response

For the experimental trials with brake events after the control transition, participants' responses to the braking lead vehicle were analysed. Relevant measures were brake response time, the maximum longitudinal deceleration, and the minimum time-to-collision (TTC) with the lead vehicle. The brake response time was defined as the time interval between the onset of the braking light of the lead vehicle and the moment at which the participant started pressing the brake pedal.

#### 5.2.6 Data analysis

Several experimental trials were not available for analysis due to missing or incomplete video recordings or driving performance data. An overview of data availability is provided in Table 5.2. Mixed factorial analysis of variance (ANOVA) was conducted to examine the effects of *driver group* (between-subject factor) and *task* (within-subject factor) on each dependent variable related to take-over responses and brake responses. To assess post-transition manual driving performance as a function of time, *time* was included as an additional within-subject factor in the mixed factorial ANOVA. If Mauchly's tests indicated that the assumption of sphericity had been violated, degrees of freedom were corrected using Greenhouse-Geisser estimates. Post-hoc pairwise comparisons were performed using the Bonferroni correction.

	Test conditions	Trials with available video recordings	Trials with available driving performance data	Total number of trials
Car	Monitoring (w/o BE)	17	17	18
	Tablet (w/o BE)	16	15	18
	Eyes-closed (w/o BE)	17	15	18
	Monitoring (BE)	-	13	18
	Tablet (BE)	-	13	18
	Manual driving	-	17	18
Truck	Monitoring (w/o BE)	18	19	22
	Tablet (w/o BE)	17	19	22
	Eyes-closed (w/o BE)	18	20	22
	Monitoring (BE)	-	20	22
	Tablet (BE)	-	17	22
	Manual driving	-	22	22

 Table 5.2: Overview of data availability for analysis. BE = Brake Event.

Because subjects with any missing value would be excluded from repeated measures ANOVA, we first imputed missing values using the expectation maximization (EM) method (Dempster, Laird, & Rubin, 1977) to make sure we were able to at least use all available data. All statistical analyses were performed using IBM SPSS 24; with a significance level of 0.05. Descriptive results were reported based on the original datasets without imputation.

# 5.3 Results

#### 5.3.1 Take-over response times

Combining all task conditions, car drivers took over control on average 3.78 s (SD = 1.79 s) after the TOR onset, ranging from 1.6 s to 9.56 s. For truck drivers, the average take-over response time was 4.45 s (SD = 2.18 s), ranging from 1.75 s to 11.77 s. In both groups, 95 % of the drivers could take over within 8.5 s. Descriptive statistics of car and truck drivers' perception-movement times during the take-over process (total response time, perception response time, and movement response time), and the response time to press the acceleration pedal after automation deactivation (acceleration response time) were measured in three task conditions, as listed in Table 5.3, and visualised in Figure 5.2 and Figure 5.3. Results of 2 (*driver group*) ×3 (*task*) two-way mixed factorial ANOVAs are presented in Table 5.4.

There was a significant main effect of *task* on the total response time, while the main effect of *driver group* was not significant. Total response times were significantly shorter in the *Monitoring* condition compared to the *Tablet* condition and the *Eyes-closed* condition (both at p < .001). Interaction between the effects of *task* and *driver group* was only marginally significant, with car drivers having the tendency to respond faster than the truck drivers in the *Tablet* condition.

A significant main effect of *task* was found on the perception response time, while the main effect of *driver group*, and the interaction effect were not significant. Post-hoc tests revealed significantly longer perception response times in the Eyes-closed condition compared to the Monitoring condition (p <.001) and Tablet condition (p = .001). With respect to the movement response time, significant main effects of *task* and *driver group*, and a significant interaction were found. Post-hoc tests showed significantly higher movement response times of truck drivers than car drivers in the *Eyes-closed* condition (p = .008), and a similar trend in the *Tablet* condition (p = .057). When comparing between task conditions, both driver groups showed significantly higher movement response times in the Tablet condition compared to the Monitoring condition (p < .001 for both driver groups) and the Eyes-closed condition (p < .001for car drivers, p = .006 for truck drivers). Higher movement response times in the *Eyes-closed* condition compared to *Monitoring* condition was only significant for truck drivers (p = .018). In addition, we performed pairwise comparisons to compare the perception response time and the movement response time in all conditions, and found that for both driver groups, the movement response time was significantly larger than the perception response time in the Monitoring and Tablet conditions (all at p < .001). The difference between the perception response time and the movement response time was not statistically significant in Eyes-closed condition.

Both *driver group* and *task* showed significant main effects on the gas pedal response time. The interaction between the main effects was not significant. After the control transition, truck drivers pressed the gas pedal significantly later than the car drivers. Pairwise comparisons further revealed a significantly lower gas response time in the *Monitoring* condition than the *Tablet* condition (p = .014). No significant differences in gas response time were found between other task conditions.



Figure 5.2: Take-over response times of car drivers and truck drivers measured in three non-driving task conditions. The bars show the means of perception response times and movement response times; they add up to the total take-over response times.



Figure 5.3: Car drivers' and truck drivers' mean response times to press the acceleration pedal after automation deactivation in three non-driving task conditions. The error bars represent the standard deviations.

Measure	Task		Car		Truck		
		Mean (SD)	Min.	Max.	Mean (SD)	Min.	Max.
Total response time (s)	Monitoring	2.66 (0.96)	1.60	4.55	2.45 (0.53)	1.75	3.80
	Tablet	4.43 (1.49)	2.15	8.32	5.53 (1.39)	3.62	8.54
	Eyes-Closed	4.26 (2.13)	1.96	9.56	5.42 (2.44)	2.19	11.31
Perception response	Monitoring	0.84 (0.35)	0.55	1.96	0.78 (0.14)	0.43	1.10
time (s)	Tablet	0.84 (0.28)	0.43	1.53	0.91 (0.35)	0.17	1.77
	Eyes-Closed	2.14 (2.08)	0.58	8.39	1.89 (1.66)	0.80	7.73
Movement response	Monitoring	1.82 (0.84)	0.82	3.87	1.67 (0.49)	0.95	2.97
time (s)	Tablet	3.60 (1.43)	1.17	7.07	4.61 (1.40)	2.75	7.74
	Eyes-Closed	2.11 (0.81)	1.05	3.99	3.53 (1.90)	1.19	8.31
Gas pedal response	Monitoring	0.99 (0.89)	0.05	3.45	4.80 (3.47)	0.05	11.90
time (s)	Tablet	1.42 (1.91)	0.04	3.89	8.06 (5.05)	0.50	21.70
	Eyes-Closed	1.28 (1.25)	0.05	5.18	7.92 (4.09)	0.05	14.33

Table 5.3: Descriptive statistics of the four take-over response time measures.

Measure	Factor	F	df	р	Partial η <sup>2</sup>
Total response	driver group	2.51	1 38	122	06
time	task	46.28	1, 50	.122	.00
time	driver group * task	3.16	1.65, 62.66	.059	.08
Perception	driver group	.05	1, 38	.822	.00
response time	task	16.32	1.06, 40.25	<.001	.30
-	driver group * task	.12	1.06, 40.25	.744	.00
Movement	driver group	5.60	1, 38	.023	.13
response time	task	51.65	2,76	<.001	.58
	driver group * task	5.05	2, 76	.009	.12
Gas pedal	driver group	46.29	1, 38	<.001	.55
response times	task	3.80	2, 76	.027	.09
	driver group * task	1.61	2,76	.206	.04

Table 5.4: Results of mixed A	ANOVAs for the four take-ove	er response time measures.
-------------------------------	------------------------------	----------------------------

p < .05 is indicated in boldface

#### 5.3.2 Post-transition manual driving performance

Car and truck drivers' post-transition manual driving performance in terms of SDLP, mean and standard deviation of longitudinal speed, and mean THW in three task conditions were depicted in Figure 5.4, Figure 5.5, Figure 5.6, and Figure 5.7. Table 5.5 showed the results of the 2 (*driver group*)  $\times$  3 (*task*)  $\times$  6 (*time*) three-way mixed factorial ANOVAs conducted for each performance measure.

First, we examined the effects of the three factors on drivers' lateral vehicle motion control performance (SDLP). Results indicated a significant main effect of *time*, and significant two-way interactions between each of the two factors. The effect of *time* was only significant for truck drivers in the *Tablet* condition (p < .001), in which truck drivers' SDLP significantly decreased within 10-20 s after the transition, and remained relatively stable in the remaining timeslots. When comparing between task conditions, truck drivers' SDLP in the *Tablet* condition was significantly higher than the *Monitoring* condition, but only for the first 10 s after the transition (p = .010). Significant differences between driver groups (p = .003) were only found in the *Tablet* condition. For truck drivers, post-transition SDLP in the *Tablet* condition was marginally significantly higher than the baseline SDLP, but only for the first 10 s after transition (p = .06). No significant differences were found in the other two conditions. For car drivers, no significant differences were not observed in any task condition, for any timeslot.

Next, we assessed car and truck drivers' performance with respect to longitudinal speed. There were significant main effects of *driver group* and *time* on mean speed, and a significant two-way interaction between these two factors. In addition to car drivers' higher mean speed than truck drivers (which was a direct result of the experimental condition), we found that truck drivers' speed significantly decreased in the first 20 s after the transition, and then gradually increased in the successive timeslots, while car drivers' speed increased in the first 20 s after the transition, then remained relatively stable in the remaining timeslots. When examining the effects of the three factors on variation in longitudinal speed (SD SPD), we found significant main effects of *driver group* and *time*, and a statistically significant three-way interaction between *driver group*, *task*, and *time*. When comparing between driver groups, car drivers showed a significantly larger variation in speed than truck drivers in all task conditions, in all

timeslots (all at p < .001). The effect of *task* was significant for truck drivers only in the first 10 s after the transition, in which the SD SPD was lower in the *Monitoring* condition compared to the other two task conditions (p < .010). Systematic effects of task were not observed for car drivers. The effect of *time* was only significant for truck drivers (p < .001). In each task condition, truck drivers' variation in speed significantly reduced in the first 20 s after the transition, and remained stable in the remaining timeslots.

Another mixed ANOVA was conducted to examine the effects of the three factors on the mean THW after the control transition. There were significant main effects of *driver group* and *time*, and a significant three-way interaction between *driver group*, *task*, and *time*. In all task conditions, car drivers' mean THW significantly increased only in the first 20 s after the transition. Truck drivers' mean THW continuously increased within the observation window in the *Tablet* and *Eyes-closed* conditions. In the *Monitoring* condition, truck drivers' mean THW only increased in the first 50 s after the transition. When comparing between driver groups, truck drivers' mean THW was significantly lower than car drivers in the first 10 s then overpassed that of car drivers from 20-30 s after the transition, in all task conditions. An exception was that 50 - 60 s after the transition in the *Monitoring* condition, the two driver groups did not show significant differences in THW. The effect of *task* was only significantly lower in the *Monitoring* condition, truck drivers' THW was significantly lower in the transition.

In addition, we assessed car and truck drivers' THW in the baseline (manual) driving condition. Truck drivers' post-transition THW was significantly lower compared to their baseline THW in all timeslots, in all conditions. For car drivers, a significantly lower post-transition THW than the baseline THW was only found in the first 30 s after the transition in the *Monitoring* condition. In the other two task conditions, a significantly lower post-transition THW compared to the baseline THW was found in the first 10 s after the transition, and at the end of the observation window (50 - 60 s).



Figure 5.4: Car drivers' and truck drivers' mean standard deviation of lateral position (SDLP) for the first  $6 \times 10 = 60$  s after the control transition measured in three task conditions, and between 5 min and 6 min after the scenario start in the baseline manual driving condition (MD, dashed lines). The error bars represent the standard deviations.



----Car driver ----Truck driver

Figure 5.5: Car drivers' and truck drivers' mean longitudinal speed for the first  $6 \times 10 = 60$  s after the control transition measured in three task conditions. The error bars represent the standard deviations.



Figure 5.6: Car drivers' and truck drivers' mean standard deviation of longitudinal speed for the first  $6 \times 10 = 60$  s after the control transition measured in three task conditions. The error bars represent the standard deviations.



Figure 5.7: Car drivers' and truck drivers' mean time headway (THW) for the first  $6 \times 10$  = 60 s after the control transition measured in three task conditions (solid lines), and between 5 min and 6 min after the scenario start in the baseline manual driving condition (MD, dashed lines). The error bars represent the standard deviations.

Measure	Factor	F	df	р	Partial $\eta^2$
SDLP	driver group	3.82	1, 38	.058	.09
	task	2.86	2,76	.063	.07
	time	4.62	2.99, 113.53	.001	.11
	driver group * task	4.85	2,76	.010	.11
	driver group * time	3.63	2.99, 113.53	.015	.09
	task * time	3.18	6.43, 244.48	.004	.08
	driver group * task * time	1.03	6.43, 244.48	.409	.03
Speed	driver group	1000.31	1, 38	<.001	.96
	task	.04	1.37, 51.85	.912	.00
	time	11.08	3.41, 129.52	<.001	.23
	driver group * task	.30	1.37, 51.85	.654	.01
	driver group * time	10.20	3.41, 129.52	<.001	.21
	task * time	1.94	5.81, 220.62	.077	.05
	driver group * task * time	1.06	5.81, 220.62	.387	.03
SD SPD	driver group	327.90	1, 38	<.001	.90
	task	1.85	2,76	.165	.05
	time	4.16	3.47, 131.87	.005	.10
	driver group * task	3.47	2,76	.036	.08
	driver group * time	1.21	3.47, 131.87	.309	.03
	task * time	4.38	6.59, 250.27	<.001	.10
	driver group * task * time	3.24	6.59, 250.27	.003	.08
THW	driver group	13.41	1, 38	.001	.26
	task	1.39	2,76	.254	.04
	time	91.35	1.42, 54.29	<.001	.71
	driver group * task	2.40	2,76	.098	.60
	driver group * time	20.94	1.42, 54.29	<.001	.36
	task * time	1.48	3.89, 147.84	.214	.04
	driver group * task * time	5.32	3.89, 147.84	.001	.12

Table 5.5: Results of mixed ANOVAs for the post-transition manual driving performance measures.

p < .05 is indicated in boldface.

#### **5.3.3** Response in post-transition brake events

Car drivers' and truck drivers' brake response times, maximum deceleration rate, and minimum TTC in the post-transition brake event were depicted in Figure 5.8. The descriptive statistics are presented in Table 5.6. In both experiments, all participants successfully avoided a collision with the decelerating front vehicle by braking. Three  $2 \times 2$  mixed ANOVAs were performed to test the effects of *driver group* and *task (Monitoring* vs. *Tablet)* on each performance measure. The results are presented in Table 5.7.

There were significant main effects of both *driver group* and *task* on brake response times (BRT), but no significant interaction between the main effects. In both task conditions, truck drivers responded significantly faster than car drivers (both at p < .001). A significantly shorter BRT in the Monitoring condition compared to the *Tablet* condition was only found for car drivers (p = .023). When examining maximum deceleration during the brake response, we only found a significant main effect of *driver group*. Truck drivers braked less aggressively compared to car drivers in both *Monitoring* (p < .001) and *Tablet* conditions (p = .003). In terms of minimum TTC, both driver groups displayed similar performance in the two task conditions.

The results indicated no significant main effects of *driver group* or *task*, nor significant interaction between the two factors.



Figure 5.8: Brake response measures compared between truck drives and car drivers in three task conditions. Left: means brake response times (BRT); middle: mean maximum deceleration during the brake response; right: mean minimum time to collision (TTC) during the brake response. The error bars represent the standard deviations.

Measure	Task	Car			Truck		
		Mean (SD)	Min.	Max.	Mean (SD)	Min.	Max.
BRT	Monitoring	1.15 (0.36)	0.75	2.40	0.95 (0.20)	0.60	4.15
	Tablet	1.86 (0.44)	1.04	2.41	1.03 (0.38)	0.05	1.60
max. Dec.	Monitoring	7.09 (0.63)	6.33	8.25	6.10 (0.35)	4.72	6.22
	Tablet	6.84 (0.83)	5.42	8.28	6.13 (0.20)	5.42	6.22
min. TTC	Monitoring	2.52 (0.79)	1.26	3.99	2.56 (0.81)	1.53	4.15
	Tablet	2.55 (0.83)	1.42	3.89	2.07 (0.78)	1.13	3.80

Table 5.6: Means, standard deviations and confidence intervals of the brake response times, maximum deceleration rate, and minimum TTC in the post-transition brake event.

 Table 5.7: Results of mixed ANOVAs for the post-transition brake response performance measures.

p < .05 is indicated in boldface.

### 5.4 Discussion

In this study, we took the initiative to compare professional truck drivers' and passenger car drivers' take-over performance when leaving a highly automated platoon measured in two comparable driving simulator experiments. The results outlined similarities and differences between two driver groups regarding take-over response times, post-transition manual driving performance, and post-transition brake. Note that all results should be seen in the light of the fact that truck drivers drove at a lower speed (80 km/h) than car drivers (100 km/h). The gap distance in the truck platoon was also 4.5 m shorter than in the truck platoon. These may cause differences in driver's risk perception, and additionally contribute to the behavioural differences between driver groups. Also to be noted is that the findings were based on comparisons of two independent experiments, which may be another source of differences between driver groups. The main findings are discussed in turn below.

#### 5.4.1 Take-over response times

First of all, task conditions showed similar effects for professional truck drivers' and passenger car drivers' on take-over response times. For both truck and passenger car drivers, monitoring the road lead to lower take-over time compared to using a handheld tablet or relaxing with eyes closed. For both driver groups, the movement response time was the dominant response time component in the *Monitoring* and the *Tablet* conditions, while in the *Eyes-closed* condition, the perception time increased, leading to similar perception times and movement times.

When comparing between the driver groups, professional truck drivers and passenger car drivers did not depict distinct differences in total take-over response times, except for the Tablet condition in which car drivers took over control significantly faster than truck drivers. The analysis of perception-movement response times revealed the main cause to be truck drivers' slower hand movement response. When examining the video recordings for underlying reasons, we discovered that several truck drivers, but none of the car drivers, had to put on or off their glasses in order to change focus from the tablet PC back to the road, and took longer time than others to pause the undergoing task and shut down the tablet screen before putting it aside. These findings may be the result of truck drivers' higher mean age and age-related eye conditions such as presbyopia (for relation between ageing and presbyopia, see Heys, Cram, & Truscott, 2004), and potentially less experience with mobile technological devices (Vaportzis, Giatsi Clausen, & Gow 2017). In our study we did not check for experience with tablet use. Another possible explanation for truck drivers' longer movement time could be a longer spatial distance to put away the tablet PC associated with heavy trucks' larger cabin size. The difference in the cabin layout may also account for truck drivers' higher movement response in the Eyes-closed condition. Video recordings revealed that a number of truck drivers deliberately adjusted the seat while taking over control from the resting posture, which was not observed among car drivers (only one slightly adjusted the seat). Notably, our findings are not consistent with the speculation by Lotz et al., (2019) that professional truck drivers take over faster than passenger car drivers. In their study, that also involved a tablet task, the tablet PC was mounted on the central console rather than holding it in the hands, which did not allow to capture the large effect of movement response on truck drivers (having to put away the tablet). In addition, compared to critical transitions in Lotz et al., (2019), take-over times measured in self-regulated transitions reflect drivers' motivation more than their capability or biological limitations (Zhang, De Winter et al., 2019). Truck drivers in the current study could behave differently in emergency situations and take over before being physically comfortable.

In addition to the regular take-over response times, we examined the response times to press the gas pedal after the control transition, and found quite some differences between driver groups. While car drivers pressed the gas pedal on average 1 - 1.5 s after disengaging the system, professional truck drivers pressed the pedal approximately 4-7 s later. We additionally examined the corresponding THWs at the moment of the acceleration initiation, but did not found significant differences (on average 0.91 s for car drivers and 1.04 s for truck drivers). It might be that both driver groups share a similar THW threshold to initiate car-following manoeuvres after platooning. Truck drivers appeared to be aware of the lower coasting deceleration rate of their vehicles and deliberately delayed the initial input on the gas pedal to increase the THW to this threshold. Another finding worth noticing is that drivers of both groups pressed the gas pedal significantly later in the *Tablet* condition compared to the *Monitoring* condition. This could be that drivers' sensory adaptation to the small gap during platooning was stronger while monitoring the driving environment due to sustained visual stimulation (Skottke et al., 2014; Wark, Lundstrom, & Fairhall, 2007). That is, after keeping the eye on the preceding vehicle for a prolonged time during platooning, the driver might not feel the gap as small as just switching the gaze back to the road, and felt it less risky to start manual car-following at a smaller distance.

#### 5.4.2 Post-transition manual driving performance

Differences between car and truck drivers' post-transition manual driving performance are mainly exhibited in longitudinal vehicle control. Within our observation, truck drivers drove at a more steady speed and in general with a larger THW compared to car drivers. This is in line with previous research (e.g., Aghabayk et al., 2011; Aghabayk et al., 2012; Rosenbloom, 2011) that reported a significant larger THW in the truck-following-truck case than the car-followingcar cases, and truck drivers' smoother and more cautious car-following behavior. When examining the change of speed as a function of time, both driver groups showed the pattern of first decreasing and then increasing the mean speed after the transition in order to increase THW. A similar phenomenon was found in a recently study by Varotto et al., (2020) that investigated passenger car drivers' behavioral adaptation after using a full-range ACC. In this study, car drivers' mean speed reduced very shortly after the transition (and therefore was not depicted in Figure 5.5 due to the large observation window), then rapidly increased to approximately 27 m/s in the following 10-20 s. Truck drivers' mean speed reduced by approximately 1m/s between the first two timeslots, then gradually increased to the previous speed level 40 - 60 s after the transition. This difference in speed regulation corresponds to our finding that truck drivers started pressing the gas pedal at a much later stage than car drivers, and explains the difference in the THW regulation. Car drivers' THW only increased within the first 20 - 30 s after the transition, while truck drivers' THW continuously increased within the observation window and overpassed that of car drivers from 20 s after the transition.

When examining the carry-over effects of platooning in longitudinal control, we found that truck drivers' post-transition THW continued to approach the baseline manual driving THW, but still didn't reach a comparable level at the end of the observation. Truck drivers' THW in the *Monitoring* condition increased more slowly after take-over, and was in general shorter compared to the other two conditions. This supports our speculation that monitoring the preceding vehicle during platooning may increase the carry-over effects, particularly for truck drivers. For car drivers, significant differences between the post-transition. However, we speculate that the carry-over effects may last longer for car drivers, seen from the smaller mean THW compared to the baseline and several drivers' very small THW (< 0.8 s) in the following timeslots. Empirical evidence by Skottke et al., (2014) suggested that carry-over effects on

THW could last for 6 min after the control transition. Unfortunately, the short observation window in our study did not allow to investigate the development of the carry-over effect in further manual driving periods. It also has to be noted that we compared the THW after a take-over with the preferred headway from the manual drive that participants made. However, these mean headways (3.18 s for truck drivers and 1.86 s for car drivers) were the headways chosen in low-volume traffic conditions, which can be considerably larger than that observed under high-volume traffic conditions (typically 1 - 2 s, Ayres, Li, Schleuning, & Young, 2001). Cautions should be exercised when generalizing the outcomes to more complex driving environments.

With respect to lateral control performance, only truck drivers depicted increased SDLP immediately (the first 10 s) after the transition in the *Tablet* condition. This finding suggested a larger impact of platooning on truck drivers' lateral control performance compared to that or car drivers, especially after being engaged in visually and motorically demanded non-driving tasks.

In addition, we suggest that compared to car drivers, truck drivers may require more time to stabilize their control of the vehicle after the transition to manual, as the effect of *time* was significant for truck drivers in all performance measures. In particular, the variations in truck drivers' lateral and longitudinal vehicle control systematically reduced and came to a stable status approximately 30 - 50 s after the transition. For car drivers such patterns were not clearly observed. This could be a result of the difference in simulated vehicle dynamics.

#### 5.4.3 Post-transition brake response

In both experiments, all drivers braked in time in response to the decelerating preceding vehicle immediately after the transition, and no collision occurred. This indicated that both car drivers and truck drivers were indeed ready for moderately critical driving situation when taking over control at a comfortable pace. Significant differences between driver groups were found regarding brake response time and maximum deceleration. Car and truck drivers' minimum TTC did not differ, possibly due to the interplay of truck drivers' faster brake response but less aggressive deceleration compared to car drivers. Differences among task conditions were only found for car drivers, who braked more slowly after using the tablet PC compared to monitoring the driving situation. Truck drivers' more robust and significantly better braking response corresponds to their higher level of training and expertise than average car drivers. In addition, given the similar THW when the brake event occurred (mean THW was 0.90 s for car drivers and 0.86 s for truck drivers), truck drivers may perceive the situation to be more risky than car drivers, knowing heavy vehicle's lower braking capability.

# 5.5 Conclusion

This paper investigated differences in take-over performance between professional truck drivers and car drivers when leaving an automated platoon. During the take-over process, differences between two driver groups mainly lay in car drivers' shorter movement time to resume motor readiness for manual driving, possibly associated with their lower mean age and a smaller cabin size. After disengaging the system, truck drivers showed a smoother and more cautious driving style, suggested by their later input on the gas pedal (to reduce speed and increase THW), slower and more steadily increase in speed, larger THW, and faster but less aggressive response in the brake event. Furthermore, carry-over effects of platooning were suggested for both driver groups in car-following performance, which appeared more pronounced for truck drivers after monitoring the preceding vehicle due to sensory adaptation to the small gaps. Truck drivers also showed impaired lane keeping performance after using a handheld tablet PC, which was recovered rapidly after the control transition. A transition zone between a dedicated automation zone and public roads, or an HMI that reminds drivers of their following distances after platoon driving would be potential measures to ensure safe manual driving immediately after the transition.

In the end, we provide implications for future research. As mentioned above, the driving environment in the current study was relatively simple with low traffic volumes, and drivers may behave differently under heavy-volume traffic conditions. Varotto et al., (2020) reported a larger acceleration and a larger increase in speed when the driver overruled the ACC system at higher traffic densities, which could also be the case after taking over control from CACC systems for platooning. In future research, a more complex driving situation could be simulated to investigate the influence of traffic density and the behaviours of non-platoon road users on platoon drivers' take-over performance. In addition, the platooning durations of our experiment trials were less than 10 min, and only three task conditions were implemented. Future studies could incorporate longer platooning drives without critical events, and a large variation of task conditions to emulate natural platooning environment.

# Acknowledgement

This study was funded by the TNO Early Research Programme (ERP) Human Enhancement: Adaptive Automation. We would like to thank Ingmar Stel and Rob van de Pijpekamp for their technical assistance in this study, and all participants for their valuable contribution.

# References

Aghabayk, K., Moridpour, S., Young, W., Sarvi, M., & Wang, Y. B. (2011). Comparing heavy vehicle and passenger car lane-changing maneuvers on arterial roads and freeways. *Transportation research record*, *2260*, 94-101. doi:10.3141/2260-11

Aghabayk, K., Sarvi, M., & Young, W. (2012). Understanding the dynamics of heavy vehicle interactions in car-following. *Journal of Transportation Engineering*, *138*, 1468-1475. doi:10.1061/(ASCE)TE.1943-5436.0000463

Ayres, T. J., Li, L., Schleuning, D., & Young, D. (2001). *Preferred time-headway of highway drivers*. Paper presented at the 4th International IEEE Conference on Intelligent Transportation Systems (pp. 826-829), Oakland, CA. doi:10.1109/ITSC.2001.948767

Bergenhem, C., Huang, Q., Benmimoun, A., & Robinson, T. (2010). *Challenges of platooning* on public motorways. Paper presented at the 17th world congress on intelligent transport systems (pp. 1-12). Retrieved from: https://pdfs.semanticscholar.org/c5af/e2bfbf86062e00149504fd185c94b8a6d74a.pdf

Bergenhem, C., Shladover, S., Coelingh, E., Englund, C., & Tsugawa, S. (2012). Overview of platooning systems. Paper presented at the 19th ITS World Congress, Vienna, Austria. Retrieved from: <u>http://publications.lib.chalmers.se/records/fulltext/174621/local\_174621.pdf</u>

Bhoopalam, A. K., Agatz, N., & Zuidwijk, R. (2018). Planning of truck platoons: A literature review and directions for future research. *Transportation Research Part B: Methodological*, *107*, 212-228.

Blower, D., (1998). The relative contribution of truck drivers and passenger vehicle drivers to truck-passenger vehicle traffic crashes (Report No. UMTRI-98-25). University of Michigan Transportation Research Institute, Ann Arbor, MI. Retrieved from: https://deepblue.lib.umich.edu/bitstream/handle/2027.42/1246/91183.0001.001.pdf?sequence  $\equiv 2$ 

Brandenburg, S., & Skottke, E. M. (2014). *Switching from manual to automated driving and reverse: Are drivers behaving more risky after highly automated driving?*. Paper presented at the 17th international IEEE conference on intelligent transportation systems (ITSC) (pp. 2978-2983). Qingdao, China. doi:10.1109/ITSC.2014.6958168

Browand, F., McArthur, J., & Radovich, C. (2004). Fuel saving achieved in the field test of two tandem trucks. *UC Berkeley: California Partners for Advanced Transportation Technology*. Retrieved from <u>https://escholarship.org/uc/item/29v570mm</u>

Carsten, O., & Martens, M. H. (2019). How can humans understand their automated cars? HMI principles, problems and solutions. *Cognition, Technology & Work, 21, 3-20.* doi:10.1007/s10111-018-0484-0

Castritius, S. M., Dietz, C. J., Schubert, P., Moeller, J., Morvilius, S., Hammer, S., Tran, C. A., & Haas, C. T. (2020). Truck Platooning Under Real Traffic Conditions: First Insights on Behavioral Adaptations and Gap Preference of Professional Drivers. *Human Factors*. doi:10.1177/0018720820908481

Clark, H., & Feng, J. (2017). Age differences in the takeover of vehicle control and engagement in non-driving-related activities in simulated driving with conditional automation. *Accident Analysis & Prevention*, *106*, 468–479. doi:10.1016/j.aap.2016.08.027.

Coppola, P., & Silvestri, F. (2019). Autonomous vehicles and future mobility solutions. In *Autonomous Vehicles and Future Mobility* (pp. 1-15).

Davies, D. R., & Parasuraman, R. (1982). *The psychology of vigilance*. London: Academic Press.

Dempster, A. P., Laird, N. M., & Rubin, D. B. (1977). Maximum likelihood from incomplete data via the EM algorithm. *Journal of the royal statistical society, Series B (methodological)*, *39*, 1-38.

Durrani, U., Lee, C., & Zhao, P. (2016). Understanding the Differences in Lane Change Maneuvers of Cars and Heavy Vehicles on Freeways. Paper presented at Canadian Transportation Research Forum 51st Annual Conference-North American Transport Challenges in an Era of Change, Toronto, Ontario. Retrieved from: <u>https://ctrf.ca/wp-content/uploads/2016/05/CTRF2016DurraniLeeZhaoTraffic.pdf</u>

Eick, E. M., & Debus, G. (2005). Adaptation effects in an automated car following scenario. In G. Underwood (Ed), *Traffic and transport psychology: Theory and application* (pp. 243-255). Elsevier.

Eriksson, A., & Stanton, N. A. (2017). Takeover time in highly automated vehicles: noncritical transitions to and from manual control. *Human factors*, *59*, 689-705. doi:10.1177/0018720816685832

European Truck Platooning Challenge (2016). *European Truck Platooning Challenge*. Retrieved from <u>https://www.eutruckplatooning.com/default.aspx</u>

Gold, C., & Bengler, K. (2014). *Taking over control from highly automated vehicles*. *Paper* presented at the 5th international conference on applied human factors and ergonomics (pp. 3662–3667).

Gold, C., Damböck, D., Lorenz, L., & Bengler, K. (2013). "Take over!" How long does it take to get the driver back into the loop? *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *57*, 1938–1942. doi:10.1177/1541931213571433.

Gold, C., Lorenz, L., & Bengler, K. (2014). *Influence of automated brake application on takeover situations in highly automated driving scenarios.* Paper presented at the FISITA 2014 world automotive congress, Maastricht, The Netherlands.

Heys, K. R., Cram, S. L., & Truscott, R. J. (2004). Massive increase in the stiffness of the human lens nucleus with age: the basis for presbyopia?. *Molecular Vision*, *10*, 956-963. Retrieved from: <u>https://ro.uow.edu.au/cgi/viewcontent.cgi?article=3676&context=eispapers</u>

Ito, T., Takata, A., & Oosawa, K. (2016). *Time required for take-over from automated to manual driving* (No. 2016-01-0158). SAE Technical Paper. doi.org/10.4271/2016-01-0158.

Janssen, R., Zwijnenberg, H., Blankers, I., & De Kruijff, J. (2015). *Truck platooning: Driving the future of transportation*. Technical report, TNO. Retrieved from: https://www.logistiekprofs.nl/uploads/content/file/janssen-2015-truck.pdf

Kyriakidis, M., De Winter, J. C., Stanton, N., Bellet, T., Van Arem, B., Brookhuis, K., ... & Reed, N. (2019). A human factors perspective on automated driving. *Theoretical Issues in Ergonomics Science*, *20*, 223-249. doi:10.1080/1463922X.2017.1293187

Lotz, A., Russwinkel, N., & Wohlfarth, E. (2019). Response times and gaze behavior of truck drivers in time critical conditional automated driving take-overs. *Transportation research part F: traffic psychology and behaviour*, *64*, 532-551. doi:10.1016/j.trf.2019.06.008

Lu, Z., Coster, X., & De Winter, J. C. F. (2017). How much time do drivers need to obtain situation awareness? A laboratory-based study of automated driving. *Applied Ergonomics*, *60*, 293–304. doi:10.1016/j.apergo.2016.12.003

Lu, Z., Happee, R., Cabrall, C. D. D., Kyriakidis, M., & De Winter, J. C. F. (2016). Human factors of transitions in automated driving: A general framework and literature survey. *Transportation Research Part F: Traffic Psychology and Behaviour*, 43, 183–198. doi:10.1016/j.trf.2016.10.007.

Mackworth, N. H. (1948). The breakdown of vigilance during prolonged visual search. *Quarterly Journal of Experimental Psychology*, *1*, 6-21. doi:10.1080/17470214808416738

Martens, M. H., & Jenssen, G. D. (2012). Behavioural adaptation and acceptance. In A. Eskandarian (Ed.), *Handbook Intelligent Vehicles* (pp. 117-138). Springer. doi:10.1007/978-0-85729-085-4\_6

Mehdizadeh, M., Shariat-Mohaymany, A., & Nordfjaern, T. (2019). Driver behaviour and crash involvement among professional taxi and truck drivers: Light passenger cars versus heavy goods vehicles. *Transportation research part F: traffic psychology and behaviour*, *62*, 86-98. doi:10.1016/j.trf.2018.12.010

Melcher, V., Rauh, S., Diederichs, F., Widlroither, H., & Bauer, W. (2015). Take-over requests for automated driving. *Procedia Manufacturing*, *3*, 2867–2873. doi:10.1016/j.promfg.2015.07.788.

Merat, N., Jamson, A. H., Lai, F. C., Daly, M., & Carsten, O. M. (2014). Transition to manual: Driver behaviour when resuming control from a highly automated vehicle. *Transportation research part F: traffic psychology and behaviour*, 27, 274-282. doi:10.1016/j.trf.2014.09.005

McDonald, M., Brackstone, M. A., Sultan, B., & Roach, C. (1999). Close following on the motorway: initial findings of an instrumented vehicle study. *Vision in vehicles*, 7, 381-389.

Moridpour, S., Rose, G., & Sarvi, M. (2010). The effect of surrounding traffic characteristics on lane changing behaviour. *Journal of Transportation Engineering*, *136*, 1-12. doi:10.1061/(ASCE)TE.1943.-5436.0000165

Naujoks, F., Wiedemann, K., Schömig, N., Jarosch, O., & Gold, C. (2018). Expert-based controllability assessment of control transitions from automated to manual driving. *MethodsX*, *5*, 579-592. doi:10.1016/j.mex.2018.05.007

Nilsson, J., Falcone, P., & Vinter, J. (2014). Safe transitions from automated to manual driving using driver controllability estimation. *IEEE Transactions on Intelligent Transportation Systems*, *16*, 1806-1816. doi:10.1109/TITS.2014.2376877

Norman, D. A. (2015). The human side of automation. In: G. Meyer, S. Beiker (Eds.), *Road Vehicle Automation 2. Lecture Notes in Mobility* (pp. 73-79). Springer, Cham. doi:10.1007/978-3-319-19078-5\_7

Nowakowski, C., Shladover, S. E., & Tan, H. S. (2015). Heavy vehicle automation: Human factors lessons learned. *Procedia Manufacturing*, *3*, 2945-2952. doi: 10.1016/j.promfg.2015.07.824

Onnasch, L., Wickens, C. D., Li, H., & Manzey, D. (2014). Human performance consequences of stages and levels of automation: An integrated meta-analysis. *Human factors*, *56*, 476-488.

Parasuraman, R. (1987). Human computer monitoring. *Human factors*, 29, 695–706. doi:10.1177/001872088702900609

Peeta, S., Zhang, P., & Zhou, W. (2005). Behavior-based analysis of freeway car-truck interactions and related mitigation strategies. *Transportation Research Part B: Methodological*, *39*, 417-451. doi:10.1016/j.trb.2004.06.002

Petermann-Stock, I., Hackenberg, L., Muhr, T., & Mergl, C. (2013). Wie lange braucht der Fahrer–eine Analyse zu Übernahmezeiten aus verschiedenen Nebentätigkeiten während einer hochautomatisierten Staufahrt [How long does the driver need - an analysis of take-over times from various non-driving tasks during highly automated traffic jam driving]. 6. Tagung Fahrerassistenzsysteme. Der Weg zum automatischen Fahren. TÜV SÜD Akademie GmbH

Ploeg, J., Van de Wouw, N., & Nijmeijer, H. (2014). Lp string stability of cascaded systems: Application to vehicle platooning. *IEEE Transactions on Control Systems Technology*, 22, 786–793. doi:10.1109/TCST.2013.2258346

Radlmayr, J., Gold, C., Lorenz, L., Farid, M., & Bengler, K. (2014). How traffic situations and non-driving related tasks affect the take-over quality in highly automated driving. *Proceedings* 

of the Human Factors and Ergonomics Society Annual Meeting, 58, 2063–2067. doi:10.1177/1541931214581434.

Ramsay, E. D. (1998). *Interaction of multi-combination vehicles with the urban traffic environment*. Paper presented at the 5th International Symposium on Heavy Vehicle Weights and Dimensions, Maroochydore, Australia (pp. 151-167).

Rios-Torres, J., & Malikopoulos, A. A. (2017). *Impact of connected and automated vehicles on traffic flow*. Paper presented at 2017 IEEE 20th International Conference on Intelligent Transportation Systems (ITSC) (pp. 1-6), Yokohama, Japan. doi:10.1109/ITSC.2017.8317654

Rosenbloom, T. (2011). Professional drivers. In B. E. Porter (Ed.), *Handbook of traffic psychology* (pp. 389-399). Academic Press. doi:10.1016/B978-0-12-381984-0.10028-1.

Rosenbloom, T., Eldror, E., & Shahar, A. (2009). Approaches of truck drivers and non-truck drivers toward reckless on-road behavior. *Accident Analysis & Prevention*, *41*, 723-728. doi:10.1016/j.aap.2009.03.011

SAE International. (2018). *Taxonomy and definitions for terms related to driving automation systems for on-road motor vehicle* (J3016). Warrendale, PA: SAE International.

Scerbo, M. W. (2001). Stress, workload, and boredom in vigilance: A problem and an answer. In P. A. Hancock & P. A. Desmond (Eds.), *Stress, workload, and fatigue* (pp. 267–278). Mahwah, NJ: Erlbaum.

Shladover, S. E. (2018). Connected and automated vehicle systems: Introduction and overview. *Journal of Intelligent Transportation Systems*, 22, 190-200. doi:10.1080/15472450.2017.1336053

Skottke, E. M., Debus, G., Wang, L., & Huestegge, L. (2014). Carryover effects of highly automated convoy driving on subsequent manual driving performance. *Human factors*, *56*, 1272-1283. doi:10.1177/0018720814524594

Talebpour, A., & Mahmassani, H. S. (2016). Influence of connected and autonomous vehicles on traffic flow stability and throughput. *Transportation Research Part C: Emerging Technologies*, 71, 143-163. doi:10.1016/j.trc.2016.007.07

Thiriez, K., Radja, G., & Toth, G. (2002). *Large Truck Crash Causation Study* (NHTSA Interim Technical Report #DOT HS 809 527). National Center for Statistics and Analysis Advanced Research and Analysis, Springfield, VA.

Tsugawa, S. (2013). An overview on an automated truck platoon within the energy ITS project. *IFAC Proceedings Volumes*, *46*, 41-46. doi: 10.3182/20130904-4-JP-2042.00110

Tsugawa, S., Jeschke, S., & Shladover, S. E. (2016). A review of truck platooning projects for energy savings. *IEEE Transactions on Intelligent Vehicles*, *1*, 68-77. doi:10.1109/TIV.2016.2577499

Turri, V., Besselink, B., & Johansson, K. H. (2016). Cooperative look-ahead control for fuelefficient and safe heavy-duty vehicle platooning. *IEEE Transactions on Control Systems Technology*, 25, 12-28. doi:10.1109/TCST.2016.2542044 Vaportzis, E., Giatsi Clausen, M., & Gow, A. J. (2017). Older adults perceptions of technology and barriers to interacting with tablet computers: a focus group study. *Frontiers in psychology*, *8*, 1687. doi:10.3389/fpsyg.2017.01687

Varotto, S. F., Farah, H., Bogenberger, K., Arem, B. V., & Hoogendoorn, S. P. (2020). Adaptations in driver behaviour characteristics during control transitions from full-range Adaptive Cruise Control to manual driving: an on-road study. *Transportmetrica A: Transport Science*, *16*, 776-806, doi:10.1080/23249935.2020.1720856

Varotto, S. F., Hoogendoorn, R. G., van Arem, B., & Hoogendoorn, S. P. (2015). Empirical longitudinal driving behavior in authority transitions between adaptive cruise control and manual driving. *Transportation Research Record*, 2489, 105-114. doi:10.3141/2489-12

Vlakveld, W., Van Nes, N., De Bruin, J., Vissers, L., & Van der Kroft, M. (2018). Situation awareness increases when drivers have more time to take over the wheel in a Level 3 automated car: A simulator study. *Transportation research part F: traffic psychology and behaviour*, 58, 917-929. doi: 10.1016/j.trf.2018.07.025

Walton, D. (1999). Examining the self-enhancement bias: professional truck drivers' perceptions of speed, safety, skill and consideration. *Transportation research part F: traffic psychology and behaviour*, 2, 91-113. doi:10.1016/S1369-8478(99)00010-8

Wark, B., Lundstrom, B. N., & Fairhall, A. (2007). Sensory adaptation. *Current opinion in neurobiology*, 17, 423-429. doi:10.1016/j.conb.2007.07.001

Wille, M., Röwenstrunk, M., & Debus, G. (2008). KONVOI: Electronically coupled truckconvoys. In D. De Waard, F. O. Flemisch, B. Lorenz, H. Oberheid, & K. A. Brookhuis (Eds.), *Human factors for assistance and automation* (pp. 243–256). Maastricht, Netherlands: Shaker.

Willemsen, D. M. C., Schmeitz, A., Fusco, M., Van Ark, E. J., Van Kempen, E., Söderman, M., Atanassow, B., Sjöberg, K., Nordin., H, Dhurjati, P., Schmidt, C., Daems, F., & Allard, J. (2018). *Requirements Review from EU projects. D2.1 of H2020 project ENSEMBLE* (GA No. 769115). www.platooningensemble.eu

Wilschut, E. S., Willemsen, D., Hogema, J., & Martens, M. (2016). *Adaptive virtual tow bar and transition of control: A truck driving simulator study* (TNO Report 2016 R11589). Soesterberg: TNO.

Yamabe, S., Zheng, R., Nakano, K., Suda, Y., Takagi, T., & Kawahara, S. (2012). *Analysis on behaviors of a driver in the system failure in forming automatic platooning of trucks from manual driving*. Paper presented at the 19th ITS World Congress, Vienna, Austria.

Zeeb, K., Buchner, A., & Schrauf, M. (2015). What determines the take-over time? An integrated model approach of driver take-over after automated driving. *Accident Analysis & Prevention*, 78, 212–221. doi:10.1016/j.aap.2015.02.023.

Zeeb, K., Buchner, A., & Schrauf, M. (2016). Is take-over time all that matters? The impact of visual-cognitive load on driver take-over quality after conditionally automated driving. *Accident Analysis & Prevention*, 92, 230–239. doi:10.1016/j.aap.2016.04.002.

Zhang, B., De Winter, J. C. F., Varotto, S., Happee, R., Martens, M. H. (2019). Determinants of take-over time from automated driving: A meta-analysis of 129 studies. *Transportation* 

Research Part F: Traffic Psychology and Behaviour, 64, 285-307. doi:10.1016/j.trf.2019.04.020

Zhang, B., Wilschut, E. S., Willemsen, D. M., Martens, M. H. (2019). Transitions to manual control from highly automated driving in non-critical truck platooning scenarios. *Transportation Research Part F: Traffic Psychology and Behaviour*, 64, 84-97. doi:10.1016/j.trf.2019.04.006

Zheng, R., Nakano, K., Yamabe, S., Aki, M., Nakamura, H., & Suda, Y. (2014). Study on emergency-avoidance braking for the automatic platooning of trucks. *IEEE Transactions on Intelligent Transportation Systems*, *15*, 1748-1757. doi:10.1109/TITS.2014.2307160

# 6. The effect of see-through truck on driver monitoring patterns and responses to critical events in truck platooning

This chapter is based on Zhang, B., Wilschut, E. S., Willemsen, D. M. C., Alkim, T., & Martens, M. H. (2017). The effect of see-through truck on driver monitoring patterns and responses to critical events in truck platooning. In N. A. Stanton (Ed.), *Advances in human aspects of transportation* (pp. 842-852). Cham: Springer.
# 6.1 Introduction

The technological progress in intelligent transportation systems (ITS) and communication technologies enables rapid development of automated vehicles. Some vehicles already have systems such as Traffic Jam Assist that work at low speeds, which can keep a distance to the vehicle in front (e.g., based on Adaptive Cruise Control) and take control of the lateral position of a car within its lane based on road markings (Lane Keeping Assist System). Drivers are thus relieved from dynamic control tasks in specific situations. There are systems available or under development that work on motorways at higher speeds as well. One example is cooperative platooning, which can be defined as a group of automated vehicles travel closely together for the benefits of energy saving, traffic flow efficiency and safety (Bergenhem, Shladover, Coelingh, Englund, & Tsugawa, 2012; Alam, Besselink, Turri, Mårtensson, & Johansson, 2015). In 2016, the Netherlands initiated a European Truck Platooning Challenge where partially automated trucks of various brands travelled in a platoon to the Netherlands from different European cities (e.g., DAF Trucks participated under the name of the 'EcoTwin' project), showing the possibility and readiness of trucking platooning to be implemented on public roads in the near future.

Despite the promising perspective, considerable efforts still have to be made to realize these high levels of automation where human intervention and supervision are rarely needed. The current commercially available systems are not yet able to cooperate with human drivers/supervisors at an optimal level (Martens & Van den Beukel, 2013), and the drivers are still required to be prepared to take over immediate manual control in case of system boundaries and failures. Maintaining sufficient situation awareness about the surrounding driving environment and system status therefore becomes safety critical. However, this can be challenging even for drivers who are constantly monitoring because this monotonous task can easily cause boredom and vigilance decrement, which can cause slow and inadequate responses (Körber, Schneider, Zimmermann, 2015). The situation is especially disadvantageous for the drivers in a truck platoon when the time headways to the lead truck have to be extremely low (between 0.2 and 0.3 s) in order to effectively save up fuel consumption by reducing air resistance (Bergenhem et al., 2012; Willemsen, Stuiver, & Hogema, 2015). On the one hand, the blocked and monotonous front view may increase difficulties for the platoon drivers to stay "in the loop", and anticipate upcoming traffic situations. Consequently, platoon drivers may not be able to anticipate critical situations until being notified by the system. In extreme cases where timely warning cannot be guaranteed, the drivers are forced to respond immediately and an unsafe situation is most likely to occur due to a lack of preparation. Although it is a highly exceptional situation and should in principle not occur, it is interesting to study how critical this will be and if additional measures may be helpful.

One additional measure may be to capture real-time video streams by a camera installed in the front of the vehicle, and broadcast the image to the following vehicle either through an invehicle device (Belyaev, Vinel, Egiazarian, Koucheryavy, 2013; Gomes, Vieira, & Ferreira, 2012; Olaverri-Monreal, Gomes, Fernandes, Vieira, & Ferreira, 2010; Patra, Arnanz, Calafate, Cano, & Manzoni, 2015) or on a large screen attached to the back of the lead vehicle such as implemented in Samsung Safety Truck (Samsung Newsroom, 2015). These see-through systems (STS) provide drivers with very intuitive traffic information and facilitate safety assessment during overtaking manoeuvres.

Seeing the potential benefits of implementing STS in truck platooning systems, we aimed to investigate whether providing platoon drivers with additional visual information of the forward traffic scene can influence their monitoring pattern and increase awareness of the upcoming

situation. A driving simulator experiment was conducted in the present study to analyze driver gaze behaviour and responses to a sudden and time-critical event when following a see-through lead truck (i.e., with a screen at the rear part, inspired by the Samsung Safety Truck). A normal lead truck served as the control condition. Accordingly, the hypotheses were formulated as follows:

1. When driving behind a see-through truck, the forward scene becomes less monotonous and drivers will allocate more time monitoring the road, especially the see-through screen area.

2. More information about the situation by means of a see-through truck helps a driver to respond more quickly and less severely in case of an emergency situation.

#### 6.2 Method

#### 6.2.1 Participants

Twenty-two participants took part in the experiment. They all held a truck driver's license for at least 8 years and drove at least 10,000 km per year. On average participants were 47.4 years old, with a standard deviation of 11.5. Minimum age was 27 and maximum 64 years old. The group consisted of 20 male and 2 female drivers. The experiment was approved by the Ethical Committee for participant studies of TNO.

#### 6.2.2 Apparatus

The experiment was carried out in a high fidelity moving base driving simulator consisting of a DAF truck mock-up mounted on a 6 DOF moving base. The road and traffic environment were projected on cylindrical screens around the vehicle with a horizontal viewing angle of 180° and two external rear view mirrors. Vehicle and situational parameters such as pedal positions, speed, steering wheel angle and button reaction time were recorded at a sampling rate of 50 Hz. Synchronously, driver eye movements were recorded using a SmartEye nonobtrusive remote eye tracker (SmartEye AB, Gothenburg, Sweden) at a sampling rate of 60 Hz.

#### 6.2.3 Two-truck Platooning System

In this experiment, an automated two-truck platooning system was simulated that allows a truck to follow a lead vehicle at a relatively short following distance, controlling both the longitudinal and lateral motion of the vehicle. The first truck is intended to be driven by a human operator (but is controlled by the simulator in the experiment). Once engaged, the second truck is controlled by the automated system. The automated system was modelled as a combination of a Cooperative Adaptive Cruise Controller (CACC) and a Lane Keeping Assist System (LKA). The driver could push a button normally used for cruise control on the right side of the steering wheel to switch the automated system on/off. To be able to switch the system on, the driver had to drive in an 'activation zone' behind the lead truck at a distance close enough to hook on. After activating the system by pushing the 'on/off' button the system would take over both longitudinal and lateral control. To deactivate the automated system, the driver had to push the 'on/off' button again. The automated system would then transfer both longitudinal and lateral control back to the driver. A tablet display located at the right side of the dashboard indicated status change of the automated systems.

#### 6.2.4 Experimental Design and Test Procedure

Participants drove in the right-hand lane (the slower lane) of a two-lane motorway behind a lead truck that was driving with an average speed of 80 km/h. The participants were instructed to follow this lead truck and not to change lanes. There were no entries or exits on the route the participants drove. Slight curves and surrounding traffic were included in the drive. All participants started with a training session to get familiar with the driving simulator and the automated driving system. During training they were asked to perform the coupling and decoupling procedures at least three times, until they felt comfortable with it. After training, all participants started with a baseline run, that is, without the automated system, hence normal manual truck driving (MD). This was a normal drive on the same road as they would drive on in the conditions with the system. After the baseline trial, the experimental session testing automated driving conditions (AD) began. In total, each participant conducted eight AD trials under different conditions and a critical event were included.

In the critical event trials, 10 participants drove behind a normal lead truck and 12 drove behind a lead truck which had a simulated screen attached to the rear of the truck showing the images of the road before the lead truck, thus giving the impression of a see-through truck (see Figure 6.1). Shortly after the beginning of the trial, the participants pressed the button to activate the automation system and follow the lead truck at a short distance of 0.3 s time headway. In order to have the largest chance of a proper driver response, the participants were instructed to be attentive of the traffic during this scenario. After 6500 m (approx. 5 min after the activation of the system) a visual/auditory warning was issued that the control was switched back to manual control. Immediately afterwards, the lead truck made an emergency manoeuvre either to the left lane (50% of the cases) or to the emergency lane (50%). A stationary vehicle was positioned 200 m from the onset of the warning (approx. 6.5 s to collision if no reaction was taken). After pressing the button to reclaim manual control, the drivers could either brake, or steer to the adjacent lane to avoid an accident. An overview of the experimental conditions is presented in Table 6.1.



Figure 6.1: Screenshot of a normal lead truck (left) and a see-through lead truck (right) in the simulation

	Evading to the right	Evading to the left	N (participants)
Normal truck AD	Х		5
Normal truck AD		Х	5
See-through truck AD	Х		6
See-through truck AD		Х	6
Baseline MD			22

Table 6.1: Overview	of the	experimental	conditions
---------------------	--------	--------------	------------

#### 6.2.5 Dependent variables

#### 6.2.5.1 Gaze behaviour

The analysis of visual behaviour was conducted comparing three conditions: following a seethrough truck (AD), following a normal truck (AD) and baseline drive (MD). For both AD conditions, the time interval for analysis was the period when the automation system was activated. An equivalent time interval was chosen for the manual driving condition to allow the comparison with the AD conditions. Four relevant Areas of Interest (AOIs) were defined: windshield, left mirror, dashboard and secondary screen (where system status and warning were displayed). Glances outside the defined AOIs are labelled "Others". Three basic metrics according to ISO 15007 were utilized to gain an overview about drivers' information acquisition: percent time on AOI, mean glance duration and glance rate. These measures facilitate the understanding of driver monitoring pattern and enable quantitative comparison between the conditions. Since the windshield is the largest and most important AOI representing a driver's forward view, and one glance onto this area may contain multiple fixations, a further analysis was conducted for the fixations located within this area for a deeper insight. A threedimensional figure was plotted for each condition combining the positions of the fixations and the corresponding fixation durations, aiming to show an intuitive picture of drivers' gaze patterns when monitoring the traffic environment.

#### 6.2.5.2 Response to Critical Event

Analyses for the takeover performance was only conducted for two AD conditions. Regarding timing aspects, driver response time (RT) was measured, defined as the time interval from the warning onset until the first conscious intervention was taken, either by pressing the brake pedal for more than 10% or changing the steering angle more than 2°, whichever happened first counted. Quality were evaluated by the maximum brake accelerations, maximum steering angle and the minimum occurring time to collision (min TTC) within each condition. It was assumed that less harsh braking and smaller steering angles are equivalent to a less dynamic manoeuvre. Min TTC is an objective measure to assess the criticality to the situations. Larger min TTC indicates longer time left to avoid a collision with the obstacle, and therefore a safer handling. In summary, early intervention, less severe manoeuvres and larger minimum TTC indicate better performance.

#### 6.3 Results and Discussion

#### 6.3.1 Gaze behaviour

Two participants in the see-through truck condition and five participants in the baseline condition were omitted from the analysis due to missing data in eye-tracking recordings. To gain an overview about drivers' attention allocation, the percent time on AOI was first presented in Figure 6.2, based on total glance times for all participants within each condition. Results

showed when driving manually, the drivers devoted the largest percentage of their attention on the road (85% within all conditions). When following the normal lead truck, the participants showed the lowest percentage of time of eyes on the road (62%), and spent around one third of the total time looking at undefined areas. By checking video recordings, we concluded that they were predominantly looking outside the left window, which was out of the recording range of the eye-tracking system. When following the see-through truck, 73% of their time was spent monitoring the road, which is 11.3% more compared to driving behind a normal truck. In both AD conditions, the participants spent slightly more time checking the mirror, but less time monitoring the dashboard.

Figure 6.3 and Figure 6.4 show mean glance durations and glance rates regarding different AOIs. Due to the fact that not all participants looked at the left mirror, dashboard or secondary screen throughout the experiment, statistical tests were only intended for glances onto the windshield for the three conditions. Two one-way ANOVA tests were conducted, showing significant differences between groups in terms of mean glance duration (F(2,33) = 3.36, p = .047), while no significant differences were found. A Bonferroni post hoc test revealed that when driving behind a see-through truck, the mean glance duration onto the windshield was significantly longer than following a normal truck (p = 0.043). Differences between other conditions were not significant.

Figure 6.5 shows how the drivers distributed their attention within the windshield area. The position of each fixation point was illustrated on an x/y plane according to their world coordination in the eye-tracking system. The blue rectangle on this plane represented the configuration of the windshield. The duration of each fixation was also visualized on the z-axis. As can be seen from the 3D figure, when driving manually and following a see-through truck, the participants concentrated more on the road centre area (the left part of the rectangle) with longer fixation durations. Some extreme values were even above 6 s. When following a normal truck, drivers showed the similar tendency to focus more on the road centre, but the fixations were shorter and more dispersed.



Figure 6.2: Percent time on AOIs compared across groups



Figure 6.3: Mean glance duration on AOIs compared across groups



Figure 6.4: Glance rate on AOIs compared across groups



Figure 6.5: Fixations on the windshield. The positions of each fixation point are illustrated on an x/y plane according to their world coordination. The values on the z-axis indicate the corresponding fixation durations.

Combining the results above, the implementation of the see-through truck significantly increased the amount of attention allocated on the road, especially the centre area where the screen was located. These differences are mainly caused by longer single forward glances rather than more frequent fixations of the eyes. Furthermore, it can be inferred that when following a see-through truck, a driver's gaze behaviour is more similar to that during manual driving. Since the drivers were not instructed to do any secondary tasks, the behaviour of shifting attention away from the road implied a vigilance decrement when conducting the monitoring task. The shorter eyes-off-road time of the drivers in the see-through truck group indicated a higher level of attentiveness and more active monitoring. The first hypothesis is therefore supported and the results suggest a positive effect of the see-through truck on driver monitoring behaviour.

#### 6.3.2 Response to the Critical Event

We measured driver performance metrics within 10 s after the warning onset. All participants successfully avoided a collision with the stationary truck. Table 6.2 gives a descriptive overview of driver reaction types in response to the critical event. Seven out of ten drivers stayed in the right lane and braked in order to avoid the stationary truck when following a normal truck (70%). One of the three drivers who did a lane change braked to full stop in the adjacent lane and didn't pass the obstacle. For the see-through truck, four out of 12 drivers changed lanes (33%).

onation										
	Brake	Lane change without	Lane change with							
		deceleration	deceleration							
Normal Truck	7/10	2/10	1/10							
See-through truck	8/12	2/12	2/12							

 Table 6.2: Numbers of participants who conducted three reaction types in each condition

Table 6.3 contains the means and standard deviations for the takeover quality measures, and the results of the independent t-test for the comparison between groups. For the three vehicle related measures, results are reported only for the participants who did not conduct evasion manoeuvres in order to avoid biases, because the operations on the brake pedal and steering wheel during lane changing are not comparable to those during braking in the ego lane. The only exception is to include the participant that conducted a harsh brake after a lane change in the analysis of max. braking acceleration.

Table 6.3: Means and standard deviations for response times, minimum TTC, maximal brake acceleration and maximum steering wheel angle (SWA), and t-test results for the comparison between groups

	Normal Truck	See-through Truck	t-test results
Response time (s)	<b>M</b> = 2.70	M = 2.89	t(20) = -0.728,
	SD = 0.55	SD = 0.64	p = 0.475
min TTC (s)	M = 4.42	M = 4.13	t(13) = 1.059,
	SD = 0.79	SD = 0.66	p = 0.309
max. $a_{break}$ (m/s <sup>2</sup> )	M = 6.14	M = 5.28	t(14) = 2.280,
	SD = 0.27	SD = 0.96	$p = 0.039^*$
max. SWA (deg.)	M=14.65	M = 12.62	t(13) = 2.439,
	SD = 3.58	SD = 8.03	$p = 0.030^*$

Contrary to our expectation, the drivers in the see-through truck condition did not intervene faster in response to the takeover event. Similarly, no significant differences were found regarding min TTC. Significant effects of the see-through truck were found when comparing

maximum brake acceleration and maximum SWA, with smaller values for both measures in the see-through group. Therefore, a mixed answer was obtained when testing our second hypothesis. It seems that the implementation of see-through lead truck could be associated with less dynamic response manoeuvres, but did not evoke earlier intervention responses. Seen from the min TTC and the fact that no collision occurred, the situation never became really critical for the participants so a faster response might not be necessary. Another explanation can be that the participants in the see-through truck group already detected the obstacle before the warning, so they could better prepare themselves for a moderate response manoeuvre. Previous work of Gold, Damböck, Lorenz, & Bengler (2013) suggested a faster reaction within a short time budget could be accompanied with worse quality. The extra time budget gained from monitoring the see-through screen might have encouraged the participants to take slower, but less severe responses. These assumptions need further elaboration to be tested.

#### 6.4 Conclusion and Outlook

In summary, despite the limited number of participants, a positive effect of the see-through truck can be suggested, with the participants allocating more time monitoring the traffic and responding less severely when encountering a critical event within comparable safety boundaries. This study also provided insight in possible safety issues in case of unforeseen scenarios. Even though drivers showed to be capable of responding appropriately when actively monitoring the environment, there are large individual differences in response times and type of response. Further research will need to focus on the individual differences between truck drivers, on other types of critical situations and on responses when a driver has been out of the loop when distracted by secondary tasks which will be a normal situation in future truck platooning concepts. In addition, the see-through truck tested in the current study has its drawbacks such as high cost and vulnerability to damage. It would be of interest to investigate the effects of in-vehicle see-through displays implemented in truck platooning in future studies.

#### Acknowledgement

This study was funded jointly by the TNO Early Research Program (ERP) Human Enhancement: Adaptive Automation, and Rijkswaterstaat Ministry of Infrastructure and the Environment (in the context of the knowledge agenda automated driving, the full report can be found and downloaded at the corresponding website: <u>http://knowledgeagenda</u>.connekt.nl). The first author and the last author are involved in the Marie Curie Initial Training Network (ITN) project HFAuto – Human Factors of Automated Driving (PITN-GA- 2013-605817).

#### References

Alam, A., Besselink, B., Turri, V., Martensson, J., & Johansson, K. H. (2015). Heavy-duty vehicle platooning for sustainable freight transportation: A cooperative method to enhance safety and efficiency. *IEEE Control Systems Magazine*, 35, 34-56. doi:10.1109/MCS.2015.2471046

Belyaev, E., Vinel, A., Egiazarian, K., & Koucheryavy, Y. (2013). Power control in see-through overtaking assistance system. *IEEE communications letters*, *17*, 612-615. doi:10.1109/LCOMM.2013.012213.122362

Bergenhem, C., Shladover, S., Coelingh, E., Englund, C., & Tsugawa, S. (2012). Overview of platooning systems. Paper presented at the 19th ITS World Congress, Vienna, Austria. Retrieved from: <u>http://publications.lib.chalmers.se/records/fulltext/174621/local\_174621.pdf</u>

European Truck Platooning Challenge (2016). *European Truck Platooning Challenge*. Retrieved from <u>https://www.eutruckplatooning.com/default.aspx</u>

Gold, C., Damböck, D., Lorenz, L., & Bengler, K. (2013). "Take over!" How long does it take to get the driver back into the loop? *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, *57*, 1938–1942. doi:10.1177/1541931213571433

Gomes, P., Vieira, F., & Ferreira, M. (2012). *The see-through system: From implementation to test-drive*. Paper presented at the 2012 IEEE vehicular networking conference (VNC) (pp. 40-47), Seoul, South Korea. doi:10.1109/VNC.2012.6407443

International Organization for Standardization. (2014). Road vehicles — Measurement of driver visual behaviour with respect to transport information and control systems — Part 1: Definitions and parameters (ISO Standard No. 15007-1).

Körber, M., Schneider, W., & Zimmermann, M. (2015). *Vigilance, boredom proneness and detection time of a malfunction in partially automated driving*. Paper presented at the 2015 International Conference on Collaboration Technologies and Systems (CTS) (pp. 70-76), Atlanta, GA, USA. doi:10.1109/CTS.2015.7210402

Martens, M. H., & van den Beukel, A. P. (2013). *The road to automated driving: Dual mode and human factors considerations*. Paper presented at the 16th International IEEE Conference on Intelligent Transportation Systems (ITSC 2013) (pp. 2262-2267), The Hague, Netherlands. doi:10.1109/ITSC.2013.6728564

Olaverri-Monreal, C., Gomes, P., Fernandes, R., Vieira, F., & Ferreira, M. (2010). *The See-Through System: A VANET-enabled assistant for overtaking maneuvers*. Paper presented at the 2010 IEEE Intelligent Vehicles Symposium (pp. 123-128), San Diego, CA, USA. doi:10.1109/IVS.2010.5548020

Patra, S., Arnanz, J. H., Calafate, C. T., Cano, J. C., & Manzoni, P. (2015). Eyes: A novel overtaking assistance system for vehicular networks. In: S. Papavassiliou, S. Ruehrup (Eds.), *Ad-hoc, Mobile, and Wireless Networks. ADHOC-NOW 2015. Lecture Notes in Computer Science, vol 9143.* Springer, Cham. doi:10.1007/978-3-319-19662-6\_26

Samsung Newsroom (2015). *The Safety Truck Could Revolutionize Road Safety*. Retrieved from: https://news.samsung.com/global/the-safety-truck-could-revolutionize-road-safety

Willemsen, D., Stuiver, A., & Hogema, J. (2015). Automated driving functions giving control back to the driver: A simulator study on driver state dependent strategies. Paper presented at the 24th International Technical Conference on the Enhanced Safety of Vehicles (ESV), Gothenburg, Sweden. Retrieved from: <u>https://www-esv.nhtsa.dot.gov/proceedings/24/files/24ESV-000109.PDF</u>

# 7. The effects of monitoring requests on driver attention, take-over performance, and acceptance

This chapter is based on the following two papers:

- Lu, Z\*., Zhang, B\*., Feldhütter, A., Happee, R., Martens, H. M., & De Winter, J. C. F. (2019). Beyond mere take-over requests: The effects of monitoring requests on driver attention, takeover performance, and acceptance. *Transportation Research Part F: Traffic Psychology and Behaviour*, 63, 22 - 37. (\*joint first authors).
- 2) Zhang, B., Lu, Z., Happee, R., De Winter, J. F. C., Martens, M. H. (2019). Compliance with Monitoring Requests, Biomechanical Readiness, and Take-Over Performance: Video Analysis from a Simulator Study. Paper presented at the 13th ITS European Congress, Brainport Eindhoven, The Netherlands.

# 7.1 Introduction

# 7.1.1 Level 2 and 3 automated driving

Automated driving is gradually being introduced to the market and may bring benefits to traffic safety, travel comfort, traffic flow, and energy consumption (Fagnant & Kockelman, 2015; Kühn & Hannawald, 2014; Kyriakidis, Happee, & De Winter, 2015; Meyer & Deix, 2014; Watzenig & Horn, 2017). A number of car manufacturers have released partially automated driving technology (Level 2 automation as defined by SAE International, 2016), combining adaptive cruise control with a lane keeping system. Partially automated driving still requires the driver to monitor the road and be able to take immediate control at all times. Manufacturers and scientists are now working towards a higher level of automation (i.e., SAE Level 3 'conditional automation') in which the system is capable of driving in certain conditions and the driver does not have to monitor the road anymore. In case the system reaches its operational limits, the driver has to take control in response to a take-over request (TOR).

#### 7.1.2 The demanding time budgets of take-over requests

When taking over control, drivers need time to acquire situation awareness (Lu, Coster, & De Winter, 2017; Samuel, Borowsky, Zilberstein, & Fisher, 2016) and physically prepare for taking over control (Large, Burnett, Morris, Muthumani, & Matthias, 2017; Zeeb, Härtel, Buchner, & Schrauf, 2017; Zhang, Wilschut, Willemsen, & Martens, 2019). A large body of research has confirmed the importance of the time budget, defined as the available time between the TOR and colliding with an obstacle or crossing a safety boundary (see Eriksson & Stanton, 2017; Zhang, De Winter, Varotto, Happee, & Martens, 2019, for reviews). While time budgets between 5 and 7 s are often used (Zhang, De Winter et al., 2019), how much time drivers need for taking over control may depend on the driving task and context. Mok, Johns, Miller, and Ju (2017) showed that almost all drivers crashed when the time budget was only 2 s, whereas Lu et al. (2017) showed improvements in situation awareness up to 20 s of time budget.

In on-road settings, a TOR with a long time budget cannot always be provided. If the automation relies on radars or cameras to detect a collision with other road users, the achievable time budget of the TOR depends on the predictability of the unfolding situation and the capabilities of the sensors, which implies that the time budget between the TOR and the collision is usually short. In a review about human-machine interfaces in automated driving, Carsten and Martens (2018) explained that it is often unfeasible for the automated driving system to indicate in sufficient time that human intervention will be needed, which "necessitates constant monitoring by the human, so that a system that is supposed to be relaxing may actually be quite demanding".

#### 7.1.3 Monitoring requests and uncertainty presentation

In a review on transitions in automated driving from a human factors perspective (Lu, Happee, Cabrall, Kyriakidis, & De Winter, 2016), transitions in automated driving were classified into two types: control transitions and monitoring transitions. Lu et al. (2016) argued that much of the human factors literature has focused on control transitions (e.g., studies of take-over time), and pointed out that the two transition types can occur independently. For example, the driver may decide to monitor the road and achieve situation awareness, without necessarily taking over control.

Gold, Lorenz, Damböck, and Bengler (2013) previously implemented the concept of monitoring requests (MRs) in a driving simulator with the aim to achieve a monitoring transition that prepares drivers for a possible TOR. In their study, a TOR was provided if an uncertain situation

became critical (i.e., a pedestrian or object entering the lane of the ego vehicle). The participants were instructed to monitor with their eyes only or keep their hands on the steering wheel in addition. Results showed shorter take-over times and fewer cases of no intervention when the participants were monitoring 'hands on' as compared to visual-only monitoring. By comparing to one of their previous studies (Gold, Damböck, Lorenz, & Bengler, 2013), the authors suggested that the MR concept is effective in terms of safety. Louw, Markkula et al. (2017) and Louw, Madigan, Carsten, and Merat (2017) applied a concept in which an uncertainty alert was implemented upon the detection of a lead vehicle. The lead vehicle could decelerate, accelerate, or change lanes, and participants had to decide themselves whether to take over, as no TOR was provided. The two studies by Louw et al. examined relationships between drivers' eye movement patterns and crashes outcomes. However, an evaluation of the uncertainty alarm was not within their research scope. Summarizing, based on the above studies, it seems that the provision of MRs is viable in automated driving. However, the above studies did not directly compare the effects of the MR concept with a system that provides only a TOR. It would be relevant to make such a comparison and examine whether MRs prepare drivers to take over control safely in response to a subsequent TOR.

Herein, we evaluated a concept where, in addition to issuing a TOR, we provided an MR when approaching a critical location. Such an MR concept would rely not on camera/radar/lidar, but on basic localization (e.g., differential GPS, HD maps). That is, the MR could be applied when approaching a segment of the road where TORs are likely to occur (e.g., an intersection, zebra crossing, or construction works). The automation system thus degrades itself from Level 3 to Level 2 by promoting a temporary monitoring transition when it is uncertain of the (upcoming) environment, instead of changing from Level 3 to manual driving directly. The idea of an MR is that a driver is primed to take-over control but does not necessarily have to take over control.

In the literature, several concepts exist that are similar to MRs. Outside of the domain of driving, likelihood alarm systems (LAS) have been devised, which issue different types of notifications depending on the likelihood that a critical event occurs (e.g., Balaud, 2015; Wiczorek, Balaud, & Manzey, 2015). Also in driving research, concepts have been designed that intermittently or continuously inform the driver and accordingly ensure that drivers are prepared to reclaim manual control. For example, in a driving simulator study, Beller, Heesen, and Vollrath (2013) presented an uncertainty symbol in unclear situations (when the front vehicle was driving in the middle of the two lanes). No TOR was available and the participants had to decide themselves whether to intervene or not. Compared to without such an uncertainty symbol, the participants intervened with a longer time to collision (TTC) in case of automation failure. Other examples are a LED bar on the instrument cluster indicating the momentary abilities of the automation (Helldin, Falkman, Riveiro, & Davidsson, 2013; Large et al., 2017), an ambient LED strip changing colour or blinking pattern based on hazard uncertainty information (Dziennus, Kelsch, & Schieben, 2016; Yang et al., 2017), a continuous verbal notification informing the driver about the state of the ego car and the behaviour of other road users (Cohen-Lazry, Borowsky, & Oron-Gilad, 2017), and a lane-line tracking confidence notification (Tijerina, Blommer, Curry, Swaminathan, Kochhar, & Talamonti, 2017). The results of these studies showed that participants who were provided with the uncertainty indication were better prepared in critical situations (Dziennus et al., 2016; Helldin et al., 2013; Yang et al., 2017). However, there are also a number of potential shortcomings of uncertainty presentations. In particular, continuous displays require driver attention and may hinder engagement in non-driving tasks. Conversely, drivers may neglect such displays when they wish to perform a non-driving task (Cohen-Lazry et al., 2017; Yang et al., 2017).

Finally, it is noted that a number of studies have used the concept of "soft-TOR" or "two-step TOR" to acquire the driver's attention before taking over control (Lapoehn et al., 2016; Naujoks, Purucker, Neukum, Wolter, & Steiger, 2015; Van den Beukel, Van der Voort, & Eger, 2016; Willemsen, Stuiver, & Hogema, 2015; and see Brandenburg & Epple, 2018 for a questionnaire study). Two-step TORs differ from MRs because with a two-step TOR, the driver always has to take over after receiving the notification, whereas this is not necessarily the case with the MR concept.

#### 7.1.4 Compliance

For a warning system to be effective, it is essential to "provide the opportunity for protective behaviour to occur before a threat materializes" (Breznitz, 1984). Compliance plays a central role because it reflects the operator's willingness to perform protective/preparatory behaviour in response to a warning signal (Meyer, 2004; Lees & Lee, 2007). When a warning is issued for something that does not actually occur, a decrease in compliance in the following warning phases might take place, a phenomenon also known as the false alarm effect or cry-wolf effect (Breznitz, 1984; Sorkin, 1988; Dixon, Wickens, & McCarley, 2007; Wickens, Dixon, Goh, & Hammer, 2005; Zabyshny & Ragland, 2003). In Breznitz' experiments that extensively explored the cry-wolf effect, an electric shock threat was announced by a three-minute warning, then cancelled out by the experimenter (i.e., the experimenter announced that the shock would not occur). During the warning phase, the participants could reduce the intensity of the impending electric shock by pressing a pedal at some monetary cost, which represented protective behaviour against the pain. The session was repeated three times. Using three indices to quantify protective behaviour, Breznitz found reductions in the probability and amplitude of protective behaviour (i.e., whether the participants pressed the pedal; if so, how many times the pedal was pressed), as well as an increase in latency between the alarm onset and the initiation of the protective action. Besides being one of the first demonstrations that false alarms reduce compliance, Breznitz further pointed to dynamic patterns of the compliance level: compliance may decrease with consecutive false alarms, and increase significantly after a true alarm.

In the automotive domain, studies have mainly investigated drivers' compliance with imperfect collision warning systems (e.g., Bliss & Acton, 2000, 2003; Cotté, Meyer, & Coughlin 2001; Lees & Lee, 2007; Naujoks, Kiesel, & Neukum, 2016). In line with Breznitz (1984), results suggest that false alarms substantially decrease driver compliance (e.g., lower response frequency, slower braking response, and smaller reductions of speed). A few researchers have distinguished between true false alarms and unnecessary alarms based on the context in which the alarm occurs. While true false alarms are caused by detection errors, unnecessary alarms are issued as intended (e.g., associated with the driving context) but the threat resolves before intervention is needed (e.g., a pedestrian stood on the roadside but later decided not to cross the road). According to Breznitz (1984), both alarm types will cause cry-wolf effects because either the warning or the threat loses credibility. The magnitude of the cry-wolf effect, however, may differ between the two alarm types. In a comparison of false alarms and unnecessary alarms, Lees and Lee (2007) reported more frequent brake responses and larger speed reductions with unnecessary collision alarms and concluded that unnecessary alarms do not diminish compliance because they are comprehensible to the driver. Naujoks et al. (2016) documented a similar differential influence of the two alarm types.

In this study, questions related to driver compliance need to be raised, because only a small portion out of all MRs require an actual driver take-over. The MR implemented in this study notified drivers while entering locations where a take-over was likely to be requested; therefore the MRs that are not followed by TORs are likely to be perceived as unnecessary alarms rather

than false alarms. MRs are also different from the aforementioned collision warnings in that MRs request attention and preparation, rather than driver intervention. It needs to be investigated whether drivers exhibit "cry-wolf" effects when using uncertainty notifications, and how drivers' compliance with a MR affects their take-over performance if a critical event indeed occurs.

#### 7.1.5 Reliance effects

In the cry-wolf effect, Type I errors (false alarms) cause a reduction in reliance. The opposite effect is also possible: if warnings unfailingly require a response, the operator may develop (over)reliance on those warnings, which can be manifested by so-called errors of omission (i.e., not responding when there is no warning) or errors of commission (i.e., complacently responding to a warning that is inappropriate in the given context) (Skitka, Mosier, & Burdick, 1999). Accordingly, it can be argued that any study on in-vehicle warnings ought to include an evaluation of drivers' reliance and trust. An on-road study by Victor et al. (2018) suggests that drivers may fail to act despite being alerted and having their eyes on the road. Thus, there is a certain risk that drivers may not act in a critical situation when the system fails to provide a TOR, despite the fact that an MR is presented beforehand. In the present study, we also examined whether drivers over-relied on the TOR, despite the fact that they were being forewarned by means of an MR.

#### 7.1.6 Aim of the study

In summary, the concepts of uncertainty presentation and MRs are promising, as they can increase situation awareness and cognitively and physically prepare drivers to intervene when needed. However, the literature also points to potential risks in terms of distraction. At present, it is unknown whether an MR works as intended by priming drivers to take-over control if needed. A successful MR system should ensure that drivers respond quickly to a subsequent TOR, and ensure that drivers do not take over if no critical event occurs. Furthermore, it is unknown whether drivers would accept a concept that intermittently requests them to monitor the road.

In this study, a system was implemented that intends to direct the driver's attention to the road by means of an MR when the automation enters a location where a take-over is likely to occur (i.e., a zebra crossing, where pedestrians could sometimes cross the road). The driver's monitoring state (i.e., whether the driver responded by attending to the road and touching the steering wheel), driving performance (braking and steering behaviour in response to a TOR presented after the MR), as well as subjective experience (a variety of human constructs such as workload and trust, Parasuraman, Sheridan, & Wickens, 2008) using such an MR + TOR system were compared with a baseline system which presented only a TOR. Accordingly, the aim of this study was to investigate whether drivers are responsive to the MR by looking at the road when requested, whether drivers do not unnecessarily take over control when no action is needed (when no pedestrians cross the road), and whether drivers have a shorter take-over time when being forewarned by the MR as compared to when receiving only a TOR.

Additionally, we examined if the participants' compliance with MRs changed with previously experienced scenarios, and if they exhibited over-reliance on the TORs. To examine drivers' compliance with MRs, we first looked into how drivers prepared themselves for a potential take-over event in response to the MR. Second, we examined whether drivers' preparatory behaviour decreased after unnecessary MRs (i.e., no take-over required) and increased after experiencing a take-over event. Third, we examined whether drivers' preparatory behaviour was associated with take-over performance.

To examine whether drivers' exhibited over-reliance on the TORs, we included a final trial where an MR was presented, but no TOR followed. This scenario is realistic: As explained above, in some cases, the sensors of the automated driving system may not detect the hazard, and no TOR can be provided. Accordingly, we examined whether drivers failed to respond to a hazard (i.e., an error of omission) in an MR-only scenario in comparison to an MR + TOR scenario.

# 7.2 Methods

# 7.2.1 Participants

Forty-one participants (35 males, 6 females) were recruited through Facebook and university whiteboard advertisements of the Technical University of Munich. Their mean age was 29.6 years (SD = 7.0, ranging from 20 to 57 years). All participants had valid driving licenses (which were held for 11.2 years on average, SD = 7.2). Participants were compensated with 10 euros.

Of the 41 participants, 4 participants had experience with driving in a simulator prior to this study. Furthermore, 18, 12, and 6 participants reported prior experience with adaptive cruise control, a lane keeping system, and partially automated driving, respectively. All participants provided written informed consent, and the research was approved by the Human Research Ethics Committee (HREC) of the Delft University of Technology.

# 7.2.2 Apparatus

The study was conducted in a static driving simulator located at the Technical University of Munich, Germany. The simulator consists of a BMW 6-Series vehicle mock-up, and provides an approximately 180 degrees field of view. Three projectors provided views for the rear-view mirrors. The software for simulating the driving scenarios was SILAB from WIVW GmbH, which recorded the vehicle data at a frequency of 120 Hz. The automated driving system controlled longitudinal and lateral motion, and could be activated and deactivated by pressing a button on the steering wheel. The sound effects of the engine, passing vehicles, as well as warnings, were provided via speakers of the vehicle cabin. A dashboard mounted eye tracking system (Smart Eye) was used to record participants' eye movement at a frequency of 60 Hz. The driver's glance locations were classified into the following areas of interest (AOI): windshield (road in front of the driver), central console, left and right exterior mirror, rearmirror, and instrument cluster. A 9.5 by 7.31-in. handheld tablet (iPad 2) was provided to the participants for performing a non-driving task. The vehicle and the cabin are shown in Figure 7.1.



Figure 7.1: The TU Munich Driving Simulator. Left: full-vehicle mock-up; Right: cabin.

#### 7.2.3 Automation system and human-machine interface

In the basis of the experiment, two automation systems were tested: (1) MR + TOR: automation with take-over requests (TOR) being preceded by monitoring requests (MR) and (2) TOR-only: automation with TOR but without MR. The third condition (MR-only) was presented last to investigate whether the participants had developed over-reliance on the TOR signal.

The MR + TOR system consisted of five automation states, with corresponding status icons shown on the dashboard (Figure 7.2 and Figure 7.3). When the automation was unavailable, a white car on a light blue road was shown in the top centre (Figure 7.2a) and the driver needed to drive manually. When the requirements for automated driving were fulfilled, a verbal notification "Automation available" was issued, and a green steering wheel icon was shown (Figure 7.2b). The driver could press a button on the steering wheel to activate the automation (the icon then changed to Figure 7.2c with an acoustic state changing sound, i.e., a gong). When the automation was active, the participant could take the hands off the wheel and feet off the pedals.

When entering an area in which a critical situation might occur, the system issued an MR. The MR consisted of a gong sound followed by a verbal notification "Please monitor", and a yellow eye-shaped icon (Figure 7.2d). The automation remained fully functional after the MR onset. If no critical event occurred, the MR was dismissed after passing the zebra crossing, and the icon changed back to the 'automation activated' state (Figure 7.2c) accompanied by a gong sound.

If the system detected a situation that it could not handle, a TOR was provided, and the automation was deactivated at the same time, leading to a slight deceleration. The acoustic TOR warning was a sharp double beep (75 dB, 2800 Hz) followed by a verbal take-over request "Please take-over". Figure 7.2e. Figure 7.3 (right) show the visual display for the TOR: an orange hands-on-the-wheel icon in the lower center of the dashboard, and the automation state icon back to "automation unavailable" (Figure 7.2a). Upon receiving the TOR, the driver had to take over by braking and/or steering in response to the situation. After taking over control, the driver had to drive manually until the automation became available again; they could then reactivate the automation. The TOR-only system was identical to the MR + TOR system, except that there was no MR. In addition, the participants drove a third condition (MR-only), in which an MR but no TOR was provided before a critical event.



Figure 7.2: Screenshots of the visual interface for the five system states. a) automation unavailable; b) automation available but not yet activated; c) automation activated; d) monitoring request; e) take-over request.



Figure 7.3: Photos of the instrument cluster with automation status. Left: automation available, corresponding to Figure 7.2 b; Right: take-over request, corresponding to Figure 7.2 e.

#### 7.2.4 Experimental design and test scenarios

A within-subject design was used, meaning that each participant completed all three conditions (MR + TOR, TOR-only, and MR-only) in three separate sessions. The order of the MR + TOR and TOR-only conditions was counterbalanced, whereas the MR-only condition was always presented in the last (i.e., third) session.

The simulated experimental track consisted of rural and city road segments with one lane in each direction. There was moderate traffic in the opposite direction and no traffic in the ego lane. The speed limit was 80 km/h on the rural road and 50 km/h in the city, as indicated by speed limit signs along the road. The automation drove at a constant speed of 80 and 50 km/h in the corresponding segments (except for the deceleration and acceleration between the city and rural roads). The critical events that required driver intervention were pedestrians who were crossing at a zebra crossing in the city road segments. Due to the layout and kinematics of the situation, braking was the required and expected action to avoid a collision, although some optional steering could be applied as well. The participants were not informed about the specific situation, and were told to respond by either steering or braking depending on their judgement. In the MR + TOR condition as well as the TOR-only condition, five zebra crossings were included. At two out of five crossings, two pedestrians stood behind an obstacle (either a bus stop or a truck) on the pavement, 1.5 m from being visible to the participant in the walking direction. The first crossing pedestrian started walking at a speed of 1.5 m/s when the participant's car was 83.33 m away from the zebra crossings (TTC = 6 s at 50 km/h). The other pedestrian crossed the road with a speed of 1 m/s, following the first pedestrian (Figure 7.4 Left). It took around 5 s for the first pedestrian and 9 s for the second pedestrian to cross the road. No pedestrians were present at the other three crossings, and the participants were not supposed to take over (Figure 7.4 Right).

The MR-only condition contained three zebra crossings. There were no pedestrians at the first two crossings. At the last crossing, two pedestrians started crossing the road 7 s after the MR was announced, but no TOR was given. This session ended after the critical event. The session of the MR-only condition lasted approximately 10 min. Figure 7.5 provides an illustration of the order of sessions and events for one participant.



Figure 7.4: Left: Zebra crossing with two pedestrians crossing the road (a take-over scenario). Right: Zebra crossing without pedestrians (here, it was not necessary to intervene). Note that these screenshots were taken from an observer's perspective in the simulator software, not from the driver's perspective.



Figure 7.5: Illustration of the order the sessions and events for one participant. The MR+TOR and TOR-only conditions were counterbalanced, and the MR-only condition was always driven after the first two conditions. The sequences of the five scenarios in MR+TOR and TOR-only conditions were randomized for each participant. The sequence of the three scenarios in the MR-only condition was fixed as shown in c).

#### 7.2.5 Non-driving tasks

The participants were instructed to play Angry Birds or Candy Crush (visual-motor tasks without sound) during automated driving on a handheld tablet PC (iPad 2) provided by the instructor. These games are self-paced and interruptible (Naujoks, Befelein, Wiedemann, & Neukum, 2017), meaning that participants could pause the game whenever they felt necessary to look up to the road.

#### 7.2.6 Procedures

Upon arrival at the institute, the participants were welcomed and asked to read a consent form. The first part of the form contained an introduction to the experiment and the two automation systems. The form mentioned that participants would experience two systems: one with and one without the MR in the first two sessions, and that they would again experience the system with the MR in the third session. Moreover, they were informed that, in all three sessions, the TOR would be available if the critical events are detected successfully. The participants were instructed to keep their hands off the steering wheel and feet off the pedals during highly automated driving. Furthermore, they were asked to play the game during the experiment, and stop playing when the automation requests them to take control. They were also informed to stop playing the game and monitor the surroundings whenever they feel insecure, even when the automation provides no request. Participants were not informed about the specific type of event that would occur (pedestrians crossing the road), nor about the fact that the system would fail to provide a TOR.

After signing the consent form, the participants completed a questionnaire regarding their age, gender, and driving experience. Next, a handout with pictures for each of the automation-status icons was provided, and the non-driving tasks were introduced on the tablet. The participants were then led to the driving simulator. The positions of the seat, mirrors, and the steering wheel were adjusted to each participant's preference, and the eye-tracking system was calibrated.

At the beginning of the experiment, each participant drove a training session of approximately 4 minutes, during which they received verbal explanations from the experimenter. The participants started this training on a rural road and drove manually for around 2 minutes. Upon approaching an urban area, the participants received a notification from the system and pressed the button to activate the automation. In the urban area, the participant experienced an MR when approaching a zebra crossing without a critical event. Shortly afterwards, the participants received another MR and subsequently a TOR because of road construction ahead. The participant had to take over control by braking or steering to avoid a collision with the traffic cones in the ego lane. The training session ended after the participant drove past the construction area.

Next, the participants drove the three experimental sessions described in section 2.4. Before the session, they were informed which of the two systems (TOR-only or MR+TOR) they were about to experience. After each session, the participants took a break and completed a questionnaire about their workload (NASA-TLX) when performing the experiment, and rated the automated driving system they just experienced. The entire experiment lasted approximately 90 min per participant.

#### 7.2.7 Dependent variables

The drivers' behaviour during this study was assessed using the data recorded by the eye tracker, simulator software and self-report questionnaires.

#### 7.2.7.1 Eye movements

Two gaze-based measures were used in this study.

• Eyes-on-road response time: defined as the time interval from the MR onset until the first detected glance on the road. In the TOR-only condition, the eyes-on-road response time is the interval from the TOR onset until the first detected glance on the road.

• The percentage time eyes-on-road: the percentage of time that glances were within the area of the windshield when the automation was active (i.e., periods when the vehicle was within 166.67 m before the zebra crossings were excluded). This measure describes whether participants showed different monitoring behaviour (i.e., voluntarily looking at the road) when using the two automation systems.

Glances shorter than 0.125 s were eliminated from the raw tracking data, in approximate agreement with the minimum possible fixation duration (ISO, 2014).

#### 7.2.7.2 Take-over performance measures

The following measures were used to evaluate how quickly the participants responded to the MR and TOR.

- Hands-on-wheel time: the time interval measured from the moment a pedestrian became visible (i.e., the TOR onset if available) until the participant put at least one hand on the steering wheel, as measured with detection sensors in the steering wheel.
- Brake initiation time: the time interval measured from the moment a pedestrian became visible (i.e., the TOR onset if available) until the first detectable braking movement (first non-zero brake signal).
- Steer initiation time: The time interval measured from the moment a pedestrian became visible (i.e., the TOR onset if available) until the first detectable steering movement before the zebra crossing (exceeding 0.02 radians).
- Minimum TTC: The minimum time to collision (TTC) in scenarios where pedestrians were crossing the road. This measure was calculated after the first moment the driver pressed the brake. The minimum TTC was zero if a collision occurred.
- Maximum longitudinal deceleration: The maximum deceleration in scenarios where pedestrians crossed the road. This measure was calculated for moments the driver pressed the brake.

#### 7.2.7.3 Subjective measures

After each session, participants completed questionnaires concerning workload, acceptance, usability, and trust. All the scores were linearly scaled to percentages.

- Mental workload: the workload was measured using the NASA Task Load Index (NASA-TLX; Hart & Staveland, 1988), which consists of six dimensions: mental demand, physical demand, temporal demand, performance, effort, and frustration. Each of the six items had 20 markers, and ranged from "low" to "high". In the analysis, the score for the performance item was reversed from "low" to "high" to "high" to "low".
- Acceptance: the acceptance scale developed by Van der Laan, Heino, and De Waard (1997) consists of nine questions with items scored -2 to +2 on a 5-point semantic differential scale. Scores were calculated for two dimensions: Usefulness (1. useful-useless, 3. bad-good, 5. effective-superfluous, 7.assisting-worthless, and 9. raising alertness-sleep-inducing) and Satisfaction (2. pleasant-unpleasant, 4. nice-annoying, 6. irritating-likeable, 8. undesirable-desirable). In the calculation of the usefulness and satisfaction scores, the scores for items 1, 2, 4, 5, 7 and 9 were reversed.
- Usability: Usability of the human-machine interface was assessed based on Nielsen's Attributes of Usability (Nielsen, 1994). The participants expressed their degree of

agreement with five statements regarding learnability (learning to operate the system was easy for me), efficiency (my interaction with the system was clear and understandable), memorability (it was easy to remember how to use the system), accuracy (it was easy to use the system quickly without making errors) and subjective satisfaction (the system was easy and comfortable to use) on a seven-tick Likert scale from disagree to agree.

• Trust: Trust in automation system was assessed using five items selected from a questionnaire by Jian, Bisantz, & Drury (2000). The participants expressed their degree of agreement on a seven-tick Likert scale regarding mistrust (the system behaves in an underhanded manner), harm (the system's actions will have a harmful or injurious outcome), suspicion (I am suspicious of the system's intent action, or outputs), confidence (I am confident in the system) and security (The system provides security). Differences between the MR+TOR and TOR-only conditions were compared using paired t-tests, with a significance level of 0.05.

#### 7.2.8 Driver compliance with MRs

The evaluation of driver compliance with MRs only concerns the MR + TOR session. To explore variability in driver compliance, the first three of five MR blocks were analysed. The participants were divided into four groups based on four possible combinations in their first two trials:

- a) 1. MR-only, 2. MR-only
- b) 1. MR-only, 2. MR+TOR
- c) 1. MR+TOR, 2. MR-only
- d) 1. MR+TOR, 2. MR+TOR

Accordingly, their prior experience when receiving an MR in the first three trials was as follows:

- a) 1. First MR, 2. After one MR-only, 3. After two MR-only (see Figure 7.6 for illustration)
- b) 1. First MR, 2. After one MR-only, 3. After TO
- c) 1. First MR, 2. After TO, 3. After one-MR-only
- d) 1. First MR, 2. After TO, 3. After TO.



Figure 7.6: Example of the coding of blocks.

Because the MR aims to prepare the driver for a critical event, we measured the compliance with MR based on drivers' preparatory behaviours, which involved eye, hand, and foot movements. Breznitz's three indices of protective behaviour (see Introduction) were adopted to quantify drivers' preparatory behaviour, namely probability, latency, and amplitude. Detailed descriptions are listed in Table 7.1. The data were retrieved from manual video annotations. All observations started from the onset of the MR, with a duration of 7 s. Two independent raters (the first and second authors) rated the level of preparatory behaviour. The raters came to a consensus on all disagreements.

To determine associations between the level of preparatory behaviour and subsequent take-over performance, three performance measures were compared between preparation levels: 1) take-over time, measured from the onset of the TOR until the moment the driver started to press the brake pedal 2) minimum time to collision, calculated after the moment the driver pressed the brake pedal; 3) maximum deceleration calculated during the braking process.

Measures		Description	
	Eyes	Hands	Feet
Probability	The percentage of participants who looked up (any glance counts, regardless of the duration).	The percentage of participants whose hands were in a more convenient position for taking the steering wheel compared to when no preparation was made at all (i.e., Level of preparation > 0, as described below).	The percentage of participants whose feet were in a more convenient position for taking the steering wheel compared to when no preparation was made at all (i.e., Level of preparation > 0, as described below).
Latency	How long after the MR onset the participant started looking up.	How long after the MR onset the participant started moving their hands to be in a more convenient position for taking the steering wheel.	How long after the MR onset the participant started moving their foot to be in a more convenient position for pressing the braking pedal.
Amplitude	-	Level 0: Both hands were holding/interacting with the iPad/phone at the MR onset, with no obvious movement to put the iPad/phone away.	Level 0: The feet were placed far from the pedals at the MR onset, with no obvious indication to move one foot closer to the pedals.
		Level 1: The iPad was already placed on the lap at the MR onset, or the participant lowered the iPad on the lap after receiving the MR. Both hands were still holding/touching the edge of the iPad/phone, with no obvious movement to free the	Level 1: One foot was already placed close to the pedals at the MR onset, or the participant moved one foot closer to the pedals after receiving the MR, but not hovering above the brake pedal.

#### Table 7.1: Measures of preparatory behaviour

hand(s) to be prepared to take the wheel.

Level 2: There was an obvious attempt to free at least one hand to be prepared to grab the steering wheel (e.g., putting the iPad on the passenger seat, or having one hand in the air). Level 3: At least one hand was

touching the steering wheel at the MR onset, or the participant put at least one hand on the steering wheel after receiving the MR. Level 2: The foot was hovering above the brake pedal at the MR onset, or the participant put one foot on the braking pedal after receiving the MR.

#### 7.3 Results

#### 7.3.1 Missing values and excluded data

Of the 41 participants, two participants experienced severe simulator sickness, and one participant had difficulties understanding the operation of the automation system. These three participants were excluded from all analyses. Furthermore, one participant's eye-tracking data was lost due to an experimenter's error, and the gaze calibration for three participants was not performed properly. Their eye tracking data were excluded from the eye-tracking analysis. Summarising, the data analysis is based on the driving performance data and the self-report data from 38 participants, and the eye tracking data from 34 participants.

One event from one participant in the TOR-only condition was excluded from all analyses, because the automation was deactivated before the event. Furthermore, in the TOR-only condition, one collision with a pedestrian occurred. This collision occurred because the driver intentionally did not brake to determine whether the car could brake automatically, as was discovered during the interview after the experiment. Only the eye tracking data from this event were included in the analysis. In addition, the eyes-on-road response time of one event in the MR+TOR condition was excluded due to missing data. Table 7.2 provides an overview of the number of events and responses for the main part of the experiment, that is, the MR+TOR and the TOR-only conditions. It can be seen that the MR system generally worked as intended, as participants had their eyes on the road at the moment of the TOR in 61 out of 68 cases. In the remaining 7 cases, participants monitored the road but had their attention allocated back to the secondary task when the TOR was provided. Furthermore, in situations without pedestrians, braking occurred in only 1 out of 114 trials, and in situations with pedestrians, participants braked in all cases.

For the assessment of driver compliance with MRs, several video recordings were unavailable due to technical problems or poor visibility. The number of participants with complete probability/amplitude data for the first three blocks was 26 for hand preparation and 27 for foot preparation.

Condition	Pedestrian-	Το	otal	Braking action	Full	Crash	Eyes on the road at the	Eyes on the road at the
	scenarios	Driving data included	Eye gaze data included		stop	Crasn	moment of the MR	moment of the TOR
MR+TOR	MR (i.e., no pedestrians)	114	102	1	0	_	14	
	MR+TOR (i.e., with pedestrians)	76	68	76	50	0	9	61
TOR-only	TOR (with pedestrians)	74	67	74	50	1		15

Table 7.2: Number of events and responses in the MR+TOR and TOR-only conditions.

#### 7.3.2 Gaze behaviour

We analysed the allocation of the participants' eyes on the road and instrument cluster while they were approaching the zebra crossings. Response times were calculated starting with the onset of the TOR and MR. The visualizations were performed using the position of the participant's car on the x-axis, since the TOR/MR was triggered based on the position of the car, which is consistent with how sensors work in real systems. Furthermore, by using distance instead of time on the x-axis, spatial relationships can be assessed intuitively; this would be impossible when using time on the x-axis, as different participants take different amounts of time to complete the scenario, depending on how they brake and use the throttle to accelerate again.

Figure 7.6 shows how the participants shifted their attention back to the road after receiving an MR or TOR as a function of travelled distance, for three scenarios: MR without pedestrians crossing the road, MR followed by a TOR (i.e., pedestrians crossing the road), and TOR in TOR-only conditions (i.e., without an MR).



Figure 7.7: Participants' visual attention allocation on the windshield (upper plot) and instrument cluster (lower plot) for the MR+TOR and TOR-only conditions. Three vertical lines from left to right are the locations of the MR (0 m; time to zebra crossing = 12 s), TOR (97.3 m; time to zebra crossing = 5 s), and zebra crossing (166.7 m).

From Figure 7.7, it can be seen that participants, on the aggregate, showed an eye-movement response towards the road and instrument cluster between 20 m to 40 m after the onset of an MR (in the MR+TOR condition) or a TOR (in the TOR-only condition). After passing the zebra crossing, some participants shifted their attention from the road to the instrument cluster. This attention shift to the instrument cluster may be because participants attempted to assess their speed or the automation status when accelerating again, after having braked for the pedestrians (see Figure 7.8 for a figure with the mean speed).

The mean eyes-on-road response time to MRs in the MR+TOR condition was 1.85 s (SD = 0.51 s), whereas the eyes-on-road response time to the TOR in the TOR-only condition was 1.76 s (SD = 0.73 s) (after removing 23 from 170 events in the MR+TOR condition and 15 from 67 events in the TOR-only condition in which participants already had their eyes on road). According to a paired t-test, this difference in eyes-on-road-time was not statistically significant (see Table 7.3 and Figure 7.9). The maximum eyes-on-road time in the MR+TOR condition was 3.84 s, which means that all participants responded to the MR before the TOR, which was presented 7 s after the MR.

Concerning the eye-gaze behaviour during automated driving in between the zebra crossings, the average percentage of time with eyes on road across the participants for the MR+TOR and TOR-only conditions were 17.71% and 16.43% (SD = 13.98%, 14.05%), respectively, a difference that was not statistically significant between the two conditions (see Table 7.3 and Figure 7.10a). This finding indicates that participants were equivalently distracted in both conditions, as could be expected.

#### 7.3.3 Take-over performance

Figure 7.8 shows drivers' braking actions in the situations where pedestrians were crossing the road and TORs were provided. It can be seen that, on average, participants applied slightly earlier braking, and reduced their speed earlier in the MR+TOR condition than in the TOR condition. Table 7.3 shows the corresponding descriptive statistics for the five take-over measures in the MR+TOR and TOR-only conditions, as well as pairwise comparisons between these conditions. The hands-on-wheel was 3.02 s faster and braking was 0.44 s faster in the MR+TOR condition than in the TOR-only condition. Thus, the results in Figure 7.7 and Table 7.2 indicate that the MRs effectively raised drivers' readiness to make the transition back to manual control of their vehicle. In the MR+TOR condition, the participants even put their hands on the steering wheel on average before the onset of the TOR. In Figure 7.9, the sequence of participants' responses is illustrated for eyes-on-road, hands-on-wheel, braking, and steering. The observed minimum TTC in the MR+TOR condition was 0.27 s longer than in TOR-only condition (consistent with the fact that participants braked earlier), indicating a safer response. However, the maximum deceleration was not significantly different between these two conditions (see Table 7.3, Figure 7.10b and Figure 7.10c).



Figure 7.8: Means and standard deviations across events of the brake position and driving speed in the take-over scenarios in the MR+TOR and TOR-only conditions as a function of travelled distance. The vertical lines mark the start of the TOR (0 m) and the position of the zebra crossing (69.4 m). Note that these are averages, which means that these graphs cannot be used to make inferences about the behaviour of individual participants. For example, the minimum averaged speed in this graph is about 5 m/s, while the majority of the participants came to a full stop.

Table 7.3: Means and standard deviations of participants for gaze behaviour and takeover response times measures in the MR+TOR and TOR-only conditions, and pairwise comparisons between the two conditions.

		Eyes-on- road response time (s)	Eyes-on-road percentage (%)	Hands-on- wheel time (s)	Brake initiation time (s)	Steer initiation time (s)	Maximum deceleratio n (m/s <sup>2</sup> )	Minim um TTC (s)
MR+TOR	М	1.85	17.71	-0.38	1.86	7.91	-8.42	2.83
	(SD)	(0.51)	(13.98)	(3.26)	(0.59)	(5.49)	(0.97)	(0.54)
TOR-only	М	1.76	16.43	2.64	2.30	8.72	-8.72	2.56
	(SD)	(0.73)	(14.05)	(1.88)	(0.61)	(4.32)	(1.00)	(0.72)
	t	1.45	0.75	-5.94	-4.53	-0.54	1.46	3.24
Paired	df	28	33	37	37	29	37	37
t-test	р	0.159	0.462	<0.001	<0.001	0.594	0.152	0.003
	r	0.44	0.75	0.35	0.50	0.086	0.16	0.70



Figure 7.9: Box plots at the level of participants for eyes-on-road, hands-on-wheel, braking, and steering. The figure is created so that the temporal sequence of events is illustrated. The TOR is provided at 0 s, while the MR is provided at -7 s. The eyes-on-road time in the MR+TOR condition is the response to the MR; the other measures are all with respect to the TOR. Negative values indicate that the corresponding behaviour occurred before the TOR onset.



Figure 7.10: Boxplots at the level of participants for the a) percentage time eyes-on-road, b) minimum TTC, and c) maximum deceleration. Three participants who crashed (i.e., minimum TTC = 0 s) were not included in this figure.

#### 7.3.4 Subjective evaluation

#### 7.3.4.1 NASA-TLX

The overall workload is the average score of the six questions in NASA-TLX. There was a statistically significant difference in the scores of the MR+TOR (M = 20.6, SD = 13.4) and TOR-only (M = 26.5, SD = 13.0) conditions, t(37)= -3.39, p = 0.002, r = 0.67. The temporal demand, frustration, and effort items yielded significantly lower scores in the MR+TOR as compared to the TOR-only condition (Table 7.4).

		Overall workload (%)	Mental demand (%)	Physical demand (%)	Temporal demand (%)	Performance (%)	Frustration (%)	Effort (%)
MR+TOR	M (SD)	20.6 (13.4)	21.5 (20.5)	15.0 (14.2)	25.3 (22.3)	14.4 (17.7)	13.7 (19.3)	13.6 (13.7)
TOR-only	M (SD)	26.5 (13.0)	26.0 (21.2)	16.9 (16.1)	36.7 (28.0)	17.0 (19.3)	21.6 (25.7)	22.6 (19.6)
Doined	t(37)	-3.39	-1.73	-0.90	-2.82	-0.89	-2.14	-3.16
r arreu	р	0.002	0.092	0.375	0.008	0.378	0.039	0.003
1-1051	r	0.67	0.70	0.62	0.54	0.52	0.52	0.49

Table 7.4: Means and standard deviations of the self-reported workload per condition.

*Note.* The scores on the items are from low (0%) to high (100%), except for the performance item, which is expressed from high (0%) to low (100%).

#### 7.3.4.2 Usefulness and Satisfaction Scales

The mean usefulness score for the MR+TOR condition (M = 85.0, SD = 10.6) was significantly higher than for TOR-only condition (M= 79.1, SD= 11.3), t(37) = 3.02, p = 0.005, r = 0.39. Similarly, participants were more satisfied with the system in the MR+TOR condition (M = 88.5, SD = 12.3) compared to the TOR-only condition (M = 80.6, SD = 17.1), t(37) = 3.42, p = 0.002, r = 0.57.

#### 7.3.4.3 Usability

The usability score (average of the five usability items) was not significantly different between the MR+TOR condition (M = 97.0, SD = 5.4) and the TOR-only condition (M = 96.1, SD = 5.8), t(37) = 1.25, p = 0.220, r = 0.64.

#### 7.3.4.4 Trust

All trust-related scores for the MR+TOR and TOR-only conditions are shown in Table 7.5. All items showed higher trust in the MR+TOR condition, especially for harm, confidence and security. Additionally, when asked about their preference between the two systems, 31 out of 38 participants preferred the MR+TOR to the TOR-only system.

 Table 7.5: Means and standard deviations of participants for the responses to the trust questionnaire, and results of paired t-tests between conditions

		Mistrust	Harm	Suspicion	Confidence	Security
MR+TOR	M (SD)	30.6 (34.6)	18.4 (23.2)	20.2 (27.2)	84.2 (18.2)	84.2 (15.0)
TOR-only	M (SD)	35.5 (34.5)	28.5 (25.7)	25.9 (27.3)	75.0 (23.8)	73.7 (21.4)
	t	-0.82	-3.38	-1.68	3.39	4.26
Paired	df	36	37	37	37	37
t-test	р	0.419	0.002	0.102	0.002	<0.001
	r	0.54	0.72	0.70	0.71	0.71

#### 7.3.5 Monitoring request without take-over request

The third condition 'MR-only', of which the results were not provided above, was included at the end of the experiment. Because this condition had a different design, the results are discussed separately in the present section. The MR-only condition was included to study whether participants relied on the TOR to follow the MR and to see if participants would still respond to a critical situation if no TOR was provided.

From the 38 participants, three crashed into the pedestrians in the last scenario. Participants' eyes were on the road and hands on the wheel during all three crashes, but participants did not intervene (see also Victor et al., 2018). In a post-experiment interview, all three participants reported their expectation and reliance on the TOR. An overview of the eye movement and braking actions in the pedestrians crossing scenarios in MR+TOR and TOR-only conditions is provided in Figure 7.11. It shows that, on average, participants applied later and harder braking in the MR-only condition than in the MR+TOR condition. Moreover, it is clear that people in the MR-only condition focused on the road rather than on the instrument cluster, presumably because no TOR was shown on the instrument cluster.



Figure 7.11: Participants' mean visual attention allocation across events on the windshield (upper plot) and instrument cluster (middle plot) and means and standard deviations across events of the brake position (lower plot) in the pedestrians crossing scenarios in the MR+TOR and MR-only conditions as a function of travelled distance. Three vertical lines from left to right are the locations of the MR (triggered position = 0 m), TOR (triggered position = 97.3 m), and zebra crossing (166.7 m).

We also compared three performance measures (maximum deceleration, brake initiation time, minimum TTC) in the pedestrian crossing scenarios between the MR+TOR and MR-only conditions (Table 7.6). The three collisions were not included in the comparison because the brakes were not applied. We assessed learning effects by comparing the two scenarios with pedestrians within the MR+TOR condition. Next, we tested whether the learning trend was counteracted by the lack of a TOR, by comparing the MR-only event ('no TOR') with the second MR+TOR event.

As shown in Table 7.6 and Table 7.7, participants braked significantly earlier and with less deceleration after the second TOR compared to the first TOR in the MR+TOR condition. However, this learning effect did not continue into the MR-only condition: In the MR-only condition, participants braked significantly later and harder compared to the second TOR of the MR+TOR condition. No statistically significant difference of minimum TTC was observed in the two pedestrian-crossing events of the MR+TOR condition. However, in the MR-only condition, the minimum TTC was significantly shorter compared to the first and second TOR of the MR+TOR condition. Summarizing, participants braked later in the MR-only condition (TOR only) as compared to MR+TOR condition, despite an expected learning effect in the opposite direction.

	Maximum deceleration (m/s <sup>2</sup> )	Brake initiation time (s)	Minimum TTC (s)		
First TOR	8 84 (0.02)	2.06 (0.71)	2 75 (0 66)		
(MR+TOR condition)	-8.84 (0.93)	2.00 (0.71)	2.75 (0.00)		
Second TOR	8 00 (1 45)	1.82 (0.62)	2.01 (0.60)		
(MR+TOR condition)	-8.00 (1.45)	1.82 (0.03)	2.91 (0.00)		
No TOR	0 10 (0 64)	2 37 (0 55)	1.08 (0.82)		
(MR-only condition)	-9.10 (0.04)	2.37 (0.33)	1.98 (0.82)		

 Table 7.6: Means and standard deviations of participants for the braking measures in the MR+TOR and MR-only conditions

Table 7.7: Results of paired t-tests between performance measures regarding the first TOR in the MR+TOR condition, the second TOR in the MR+TOR condition, and no TOR in the MR-only condition.

	Maximum deceleration (m/s <sup>2</sup> )				Brake initiation time (m)				Minimum TTC (s)			
	Second TOR (MR+TOR condition)		No TOR (MR-only condition)		Second TOR (MR+TOR condition)		No TOR (MR-only condition)		Second TOR (MR+TOR condition)		No TOR (MR-only condition)	
	t(37)	р	t(34)	р	t(37)	р	t(34)	р	t(37)	р	t(34)	р
First TOR (MR+TOR condition)	-3.52	0.001	1.33	0.192	2.36	0.023	-2.96	0.006	-1.44	0.159	6.28	<0.001
Second TOR (MR+TOR condition)			4.94	<0.001			-6.91	<0.001			8.33	<0.001

#### 7.3.6 Compliance with MRs

#### 7.3.6.1 Probability of preparatory behaviour

All participants responded to the MR by looking to the road. Hence, the 'eyes' probability was 1 for each MR block. Figure 7.12 shows the percentages of participants per block in which a hand (left) or foot (right) preparatory behaviour occurred. Results indicate an increase in hand and foot preparation in *After TO* blocks as compared to the previous block. If the participant had experienced no TOR at all, the probability of preparation was overall low and/or showed a decreasing trend (red circular markers).

Regarding hand preparations in the third block: From the three participants who had experienced two take-overs before, all three were prepared (black diamond marker), and from the six participants who had experienced no take-overs before, only three were prepared (red circular marker). Similarly, regarding foot preparations in the third block: From the four participants who had experienced two take-overs before, all four were prepared (black diamond marker), and from the four participants who had experienced two take-overs before, all four were prepared (black diamond marker), and from the four participants who had experienced no take-overs before, none were prepared (red circular marker).



Figure 7.12: Probability of hand (left) and foot (right) preparatory behaviour for the first three blocks.

#### 7.3.6.2 Latency of Preparatory Behaviour

Averaged over all five MR blocks, participants started looking to the road on average 0.70 s (SD = 0.38) after the MR. The start of hand and feet movement (if available) occurred on average 2.13 s (SD = 1.09) and 3.17 s (SD = 1.55) after the MR, respectively. These effects are illustrated in Figure 7.13. Note that all response time data in the analysis of driver compliance were obtained from manual video annotations. Therefore, the mean eyes-on-road times reported in this section were different from those reported in Section 7.3.3 based on SmartEye eye-tracking data.



Figure 7.13: Latency of participants' preparations for Eyes (29 participants based on 115 observations), Hand (26 participants based on 87 observations), Foot (19 participants based on 46 observations)

#### 7.3.6.3 Amplitude of Preparatory Behaviour

Figure 7.14 shows the means of hand and foot preparation levels. Results support the above observations that the mean level of preparation showed an increasing trend in the case of After TO blocks.



# Figure 7.14: Mean level of hand (left) and foot (right) preparation for the four MR block categories.

#### 7.3.6.4 Level of Preparation and Subsequent Take-Over Performance

Table 7.8 shows the means and standard deviations of the take-over performance measures per level of hand and foot preparation. A tendency can be observed that take-over performance improved with higher levels of preparation, where better take-over performance is characterized by a shorter take-over time, a longer minimum TTC, and a lower maximum deceleration. Also, learning effects were observed from the first take-over event to the second, in line with Lu et al. (2019). It is worth noting that a larger number of participants showed the highest level of preparation before the second take-over event as compared to the first event. The trends of preparation level and take-over attempt are illustrated in Figure 7.15.

Table 7.8	3: Means	with	standar	d deviation	s in	parentheses	of	take-over	response	time,
minimun	ı TTC, aı	nd ma	ximum d	leceleratior	ı for	each level of	' ha	nd and foo	t prepara	tion.

Hand preparation	Level 0	Level 1	Level 2	Level 3
First take-over	<i>n</i> = 6	<i>n</i> = 7	<i>n</i> = 7	<i>n</i> = 4
Take-over time (s)	2.49 (0.78)	1.94 (0.79)	2.10 (0.64)	1.89 (0.30)
minTTC (s)	2.45 (0.75)	2.53 (0.52)	2.64 (0.60)	2.76 (0.51)
maxDEC (m/s <sup>2</sup> )	9.49 (0.57)	9.36 (0.27)	8.47 (1.05)	8.79 (0.86)
Second take-over	n = 4	<i>n</i> = 7	<i>n</i> = 8	<i>n</i> = 7
Take-over time (s)	1.90 (0.60)	1.97 (0.80)	2.07 (0.68)	1.53 (0.41)



Figure 7.15: Association between hand and foot preparation levels and take-over time. Standard deviations are shown in Table 7.8.

#### 7.4 Discussion

#### 7.4.1 Main findings

The main aim of this study was to investigate 1) whether drivers are responsive to MRs by redirecting their attention to the road, 2) whether drivers unnecessarily take over control when no action is needed, and 3) whether drivers have a shorter takeover time when being forewarned by the MR as compared to when receiving only a TOR. Accordingly, a systematic comparison of participants' behaviours was made between an MR + TOR system and a traditional TOR-only system. The results indicate that participants showed strong compliance with the MRs: Participants were responsive to the MR by looking at the road, and several participants placed their hands on the steering wheel without specifically being asked to do so. These behaviours indicate that drivers were preparing themselves for a possible take-over. With their eyes on the

road and their hands already on the wheel, the drivers responded faster to TORs in the MR + TOR condition in comparison to the TOR-only condition. The longer minimum TTC values measured in the MR + TOR condition as compared to the TOR-only condition indicate that the MRs helped improve safety. Although the observed improvements (e.g., 0.44 s faster brake response time) may seem modest on an absolute scale, we argue that they can translate into large safety benefits. For example, if decelerating with 8 m/s2, 0.44 s longer braking implies an additional speed reduction of 13 km/h. This speed difference can be expected to yield substantial improvements in the probability of surviving a crash (Joksch, 1993).

Additionally, we found only one unnecessary braking action when no pedestrians were crossing the road, which means the MRs hardly caused unnecessary take-overs when no action was needed. We also found that drivers experienced lower subjective workload, higher acceptance (usefulness and satisfaction), and higher trust for the MR + TOR condition as compared to the TOR-only condition, whereas there were no statistically significant differences in experienced usability. In other words, MRs not only yielded positive effects on behaviour but were generally also experienced as positive. Finally, the presentation of MRs did not change drivers' attention allocation during the automated driving periods, indicating that drivers still felt comfortable to perform the non-driving task in between MRs.

Summarising, the MR concept worked as intended: It permitted drivers to be engaged in a nondriving task (as in a highly automated driving system), and still ensured that participants were attentive and prepared for an upcoming event (as in a partially automated driving system). Thus, our findings show that MRs promote a gradual transition between being disengaged from the driving task and actually taking over control. Put differently, the MRs effectively exploit the idea that automated driving can independently involve driver monitoring transitions and control transitions (Lu et al., 2016). Our results align with previous studies (Cohen-Lazry et al., 2017; Dziennus et al., 2016; Gold et al., 2013; Helldin et al., 2013; Yang et al., 2017), which have shown that MRs and other types of uncertainty indicators stimulate driver to allocate attention to the road when encountering an unpredictable driving environment, in turn yielding improved responses in critical situations.

#### 7.4.2 Driver Compliance with MRs

In addition, we investigated how drivers complied with MRs, how the level of compliance changed with experience within one test session, and how the level of compliance associated with subsequent take-over performance. Compliance was measured by the three indices adapted from Breznitz (1984): probability, latency, and amplitude of eye, hand, and foot preparatory behaviour, retrieved from manual video observation.

The results indicated high overall compliance with MR. Participants looked up onto the road within a short time in all cases, and moved their hands to be better prepared for a possible takeover (e.g., putting down the iPad) in the majority of the cases. In several cases, participants also moved their feet closer to the pedal, but the probability of feet movements was smaller, and the associated latency was higher, than hand preparatory behaviour.

We found a higher preparation probability in response to MRs after a take-over event as compared to MRs after two consecutive MRs without take-over event, in line with the statements of Breznitz (1984) that threat cancellation reduces protective behaviour, whereas hit alarms increase compliance. We observed that after experiencing one or two MRs without critical event, a few participants did not look up as soon as possible, but continued with the non-driving task for a short time. Some participants monitored the road only shortly, and continued with the non-driving task even before the MR was dismissed. Such potentially risky behaviour

may increase as more consecutive MRs without event are experienced in a prolonged drive. Also, drivers may develop alarm fatigue (Cvach, 2012) by excessive 'unnecessary' MRs, and turn off the notification system.

Furthermore, our findings suggest a substantial influence of preparation behaviour on take-over times: when monitoring with a hand on the wheel or with a foot hovering above the pedal, drivers responded faster to the TOR as compared to without any hand or foot preparation action. This is possibly due to a reduction of hand/foot travelling distance (Zhang, Wilschut, Willemsen, & Martens, 2019).

#### 7.4.3 Reliance on the TOR

In the final trial of our experiment, we examined whether people over-rely on the TOR, despite the fact that they have received an MR prompting them to monitor the driving environment. When drivers who were previously exposed to perfectly reliable TORs were provided with only an MR, they showed worse takeover performance as compared to the MR + TOR condition. Three out of 38 participants collided with the pedestrians, whereas the other participants showed higher mean response times, more severe braking, and a smaller minimum TTC as compared to the MR + TOR condition. These effects occurred despite the fact that they were looking at the driving environment and were told that the TOR would be available only if a critical event was detected successfully. This over-reliance may have been caused by the fact that participants were conditioned to respond to the TORs, not to the hazards (i.e., pedestrians) themselves. It is also possible that participants had built inappropriately high trust in the TORs because all preceding pedestrian crossing events came with a TOR. Lee and See (2004) argued that human trust needs to be calibrated according to the context and characteristics of automation. Further research could investigate how to prevent over-reliance on TORs. One idea is to examine whether a variable ratio of the number of TORs over the number of MRs could affect driver trust levels and their responses to the MR.

#### 7.4.4 Limitations

This study has several limitations. First, we presented pedestrian crossing scenarios only, which may have contributed to reduced response times due to familiarity. In future research, a larger variety of scenarios could be tested, including time critical situations and voluntary transitions such as merging or exiting the highway. Future research might also use a between-subjects rather than within-subject design to prevent carry-over effects. However, it is cautioned that between-subjects designs require a substantially larger sample size in order to maintain adequate statistical power. Second, this study used fixed time budgets for monitoring (i.e., 12 s before the collision) and taking over (i.e., 5 s before the collision), which may have led to specific expectations about the timing of taking back control. The time budget between an MR and a TOR could be further investigated. If an MR is provided early, drivers may lose attention again, whereas if an MR is provided late, there may be insufficient time to prepare for taking over. Third, drivers' behaviour was only observed within short experimental sessions. For the assessment of driver compliance, the maximum number of successive cancellations was two. It is likely that a more severe reduction in preparatory behaviour would occur if more MR blocks were implemented. Moreover, as pointed out in a review by Green (2000), expectancy is an important determinant of brake response times. In our experiment, there were no true surprises. That is, the participants were probably expecting at least one take-over event in the entire session. If no event has occurred yet, each transition from one MR to another implies that the take-over event is approaching: Successive MRs without TOR implied greater proximity of danger, which may counteract the decrease in compliance (cf. Chapter 5 in Breznitz, 1984).

Finally, simulator fidelity may be an issue. The absence of physical motion cues may have an effect on how drivers brake (Boer, Yamamura, Kuge, & Girshick, 2000; Siegler, Reymond, Kemeny, & Berthoz, 2001) and may have reduced drivers' awareness of the automation mode (Cramer, Siedersberger, & Bengler, 2017). It is also possible that the presentation of virtual hazards, rather than real hazards, has reinforced the "wait and see" behaviour in the MR-only condition. In addition, participants in an experimental setting tend to 'behave well'; preparatory behaviour may be less in real life settings. In future studies, various reliabilities of TOR and MR should be investigated to reach conclusions that are more credible.

# 7.5 Conclusion

The observed effects of MRs are promising. The MRs directed the drivers' attention to the road and improved their response to a subsequent TOR. Furthermore, the MR + TOR was positively evaluated for workload, usefulness, and satisfaction. We argue that automated driving systems that provide only TORs are not exploiting the richness of sensory information, both of the human and the automation sensor suite. The concept of MR makes use of the fact that automated driving systems have variable certainty about the situation. In our case, we demonstrated the MR concept when the car approaches a zebra crossing, a part of the road entailing a high likelihood that the driver has to take over control.

The simulated MR is realistic in terms of automated driving technology. Differential GPS, HD maps, and traffic data could be used as inputs to the automated driving system to provide an MR when approaching a potentially critical road section, unlike camera and lidar, which are constrained by their detection ranges. However, we caution that the provision of MRs does not guarantee safety. We observed a situation-dependent change in the compliance level, which was associated with driving performance during a subsequent take-over event. That is, the cry-wolf effects may be a concern in the use of MR. We also showed that when the automated driving system fails to detect a hazard and accordingly fails to provide a TOR, a proportion of drivers still crashed. Future research should be focused on measures to counteract the cry-wolf effect such as training and education (Breznitz, 1984; Zabyshny & Ragland, 2003), and conducted on the topic of over-reliance on take-over requests and individual differences in the use of automated vehicles.

#### Acknowledgements

The authors are involved in the Marie Curie Initial Training Network (ITN) HFAuto – Human Factors of Automated Driving (PITN-GA-2013-605817), and would like to thank TU Munich for hosting the experiment.

# References

Balaud, M. (2015). *Likelihood alarm systems: the impact of the base rate of critical events, the cost of alarm validity information, and the number of stages on operator's performance* (Doctoral dissertation). Technische Universität Berlin, Germany. Retrieved from: https://depositonce.tu-berlin.de/bitstream/11303/5217/3/balaud\_magali.pdf

Beller, J., Heesen, M., & Vollrath, M. (2013). Improving the driver–automation interaction: An approach using automation uncertainty. *Human Factors*, 55, 1130-1141. doi:10.1177/0018720813482327
Bliss, J. P. (1993). *The cry-wolf phenomenon and its effect on operator responses* (Unpublished doctoral dissertation). University of Central Florida, Orlando. Retrieved from: https://stars.library.ucf.edu/rtd/3614/

Boer, E. R., Yamamura, T., Kuge, N., & Girshick, A. (2000). Experiencing the same road twice: A driver centered comparison between simulation and reality. *Proceedings of Driving Simulation Conference* (pp. 33–55). Paris, France.

Brandenburg, S., & Epple, S. (2018). Designing the transition from highly automated driving to manual driving: a study of drivers' preferences. *Proceedings of the 6th Humanist Conference*. The Hague, Netherlands. Retrieved from: <u>https://www.humanist-vce.eu/fileadmin/contributeurs/humanist/TheHague2018/22-brandenburg.pdf</u>

Breznitz, S. (1983). *Cry wolf: The psychology of false alarms*. Hillsdale, NJ: Lawrence Erlbaum Associates.

Carsten, O., & Martens, M. H. (2018). How can humans understand their automated cars? HMI principles, problems and solutions. *Cognition, Technology & Work*. 1-18. doi:10.1007/s10111-018-0484-0

Cohen-Lazry, G., Borowsky, A., & Oron-Gilad, T. (2017). The effects of continuous drivingrelated feedback on drivers' response to automation failures. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting* (Vol. 61, pp. 1980–1984). doi:10.1177/1541931213601974

Cramer, S., Siedersberger, K. H., & Bengler, K. (2017). Active vehicle pitch motions as feedback-channel for the driver during partially automated driving. *Workshop Fahrerassistenz und Automatisiertes Fahren, Uni-DAS e. V (Ed.)* (pp. 74-83). Walting, Germany. Retrieved from: <u>https://www.uni-das.de/images/pdf/veroeffentlichungen/2017/08.pdf</u>

Cvach, M. (2012). Monitor alarm fatigue: an integrative review. *Biomedical Instrumentation & Technology*, *46*, 268–277. doi:10.2345/0899-8205-46.4.268

Dixon, S. R., Wickens, C. D., & McCarley, J. S. (2007). On the independence of compliance and reliance: Are automation false alarms worse than misses? *Human Factors*, 49, 564–572. doi:10.1518/001872007X215656

Dziennus, M., Kelsch, J., & Schieben, A. (2016). Ambient light based interaction concept for an integrative driver assistance system - a driving simulator study. *Proceedings of the Human Factors and Ergonomics Society Europe Chapter 2015 Annual Conference* (pp. 171–182). Retrieved from: https://www.hfes-europe.org/wp-content/uploads/2015/12/Dziennus2016.pdf

Eriksson, A., & Stanton, N. A. (2017). Takeover time in highly automated vehicles: noncritical transitions to and from manual control. *Human Factors*, *59*, 689–705. doi:10.1177/0018720816685832

Fagnant, D. J., & Kockelman, K. (2015). Preparing a nation for autonomous vehicles: opportunities, barriers and policy recommendations. *Transportation Research Part A: Policy and Practice*, 77, 167–181. doi:10.1016/j.tra.2015.04.003

Gold, C., Damböck, D., Lorenz, L., & Bengler, K. (2013). "Take over!" How long does it take to get the driver back into the loop? *Proceedings of the Human Factors and Ergonomics Society Annual Meeting* (Vol. 57, pp. 1938–1942). doi:10.1177/1541931213571433

Gold, C., Lorenz, L., Damböck, D., & Bengler, K. (2013). Partially automated driving as a fallback level of high automation. *6. Tagung Fahrerassistenzsysteme*. Munich, Germany. Retrieved from: <u>https://mediatum.ub.tum.de/doc/1187198/file.pdf</u>

Hart, S. G., & Staveland, L. E. (1988). Development of NASA-TLX (Task Load Index): results of empirical and theoretical research. *Advances in Psychology*, *52*, 139–183. doi:10.1016/S0166-4115(08)62386-9

Helldin, T., Falkman, G., Riveiro, M., & Davidsson, S. (2013). Presenting system uncertainty in automotive UIs for supporting trust calibration in autonomous driving. *Proceedings of the* 5th International Conference on Proceedings of the 5th International Conference on Automotive User Interfaces and Interactive Vehicular Applications (pp. 210–217). doi:10.1145/2516540.2516554

International Organization for Standardization (2014). Road vehicles – Measurement of drivervisual behaviour with respect to transport information and control systems – Part 1: Definitionsand parameters(ISO15007-1:2014).Retrieved from:https://www.iso.org/standard/56621.html

Jian, J. Y., Bisantz, A. M., & Drury, C. G. (2000). Foundations for an empirically determined scale of trust in automated systems. *International Journal of Cognitive Ergonomics*, *4*, 53–71. doi:10.1207/S15327566IJCE0401\_04

Joksch, H. C. (1993). Velocity change and fatality risk in a crash—A rule of thumb. *Accident Analysis and Prevention*, 25, 103–104. doi:10.1016/0001-4575(93)90102-3

Kühn, M., & Hannawald, L. (2014). Driver assistance and road safety. In H. Winner, S. Hakuli, F. Lotz, & C. Singer (Eds.) *Handbook of Driver Assistance Systems: Basic Information, Components and Systems for Active Safety and Comfort* (pp. 1–17). Springer International Publishing.

Kyriakidis, M., Happee, R., & De Winter, J. C. F. (2015). Public opinion on automated driving: results of an international questionnaire among 5000 respondents. *Transportation Research Part F: Traffic Psychology and Behaviour, 32*, 127–140. doi:10.1016/j.trf.2015.04.014

Lapoehn, S., Dziennus, M., Utesch, F., Kelsch J., Schieben, A., Dotzauer M., Hesse, T., Köster, F. (2016). Interaction design for nomadic devices in highly automated vehicles. *Mensch und Computer 2016–Workshopband*. Aachen, Germany. doi:10.18420/muc2016-ws08-0006

Large, D. R., Burnett, G. E., Morris, A., Muthumani, A., & Matthias, R. (2017). Design implications of drivers' engagement with secondary activities during highly-automated driving - a longitudinal simulator study. *Road Safety and Simulation International Conference (RSS2017)*. The Hague, The Netherlands. Retrieved from: https://www.researchgate.net/publication/317569487\_Design\_Implications\_of\_Drivers'\_Enga gement\_with\_Secondary\_Activities\_During\_Highly-Automated\_Driving\_-\_\_\_\_A\_Longitudinal\_Simulator\_Study

Lee, J. D., & See, K. A. (2004). Trust in automation: designing for appropriate reliance. *Human Factors*, *46*, 50–80. doi:10.1518/hfes.46.1.50\_30392

Louw, T., Markkula, G., Boer, E., Madigan, R., Carsten, O., & Merat, N. (2017a). Coming back into the loop: Drivers' perceptual-motor performance in critical events after automated driving. *Accident Analysis & Prevention*, *108*, 9–18. doi:10.1016/j.aap.2017.08.011

Louw, T., Madigan, R., Carsten, O., & Merat, N. (2017b). Were they in the loop during automated driving? Links between visual attention and crash potential. *Injury Prevention, 23*, 281-286. doi:10.1136/injuryprev-2016-042155

Lu, Z., Coster, X., & De Winter, J. C. F. (2017). How much time do drivers need to obtain situation awareness? A laboratory-based study of automated driving. *Applied Ergonomics*, *60*, 293–304. doi:10.1016/j.apergo.2016.12.003

Lu, Z., Happee, R., Cabrall, C. D. D., Kyriakidis, M., & De Winter, J. C. F. (2016). Human factors of transitions in automated driving: a general framework and literature survey. *Transportation Research Part F:Traffic Psychology and Behaviour, 43,* 183–198. doi:10.1016/j.trf.2016.10.007

Meyer, J. (2004). Conceptual issues in the study of dynamic hazard warnings. *Human Factors*, 46, 196–204. doi:10.1518/hfes.46.2.196.37335

Meyer, G., & Deix, S. (2014). Research and innovation for automated driving in Germany and Europe. In G. Meyer & S. Beiker (Eds.), *Road Vehicle Automation* (pp. 71–81), Springer. doi:10.1007/978-3-319-05990-7\_7

Mok, B., Johns, M., Miller, D., & Ju, W. (2017). Tunneled in: Drivers with active secondary tasks need more time to transition from automation. *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems* (pp. 2840-2844). doi:10.1145/3025453.3025713

Naujoks, F., Befelein, D., Wiedemann, K., & Neukum, A. (2017). A review of non-drivingrelated tasks used in studies on automated driving. In N. A. Stanton (Ed.), *Advances in Intelligent Systems and Computing, Advances in Human Aspects of Transportation.* Springer. doi:10.1007/978-3-319-60441-1\_52

Naujoks, F., Kiesel, A., & Neukum, A. (2016). Cooperative warning systems: The impact of false and unnecessary alarms on drivers' compliance. *Accident Analysis & Prevention*, *97*, 162-175. doi:10.1016/j.aap.2016.09.009

Naujoks, F., Purucker, C., Neukum, A., Wolter, S., & Steiger, R. (2015). Controllability of partially automated driving functions-does it matter whether drivers are allowed to take their hands off the steering wheel? *Transportation Research Part F: Traffic Psychology and Behaviour*, *35*, 185–198. doi:10.1016/j.trf.2015.10.022

Nielsen, J. (1994). Usability Engineering. San Diego, California: Academic Press.

Parasuraman, R., Sheridan, T. B., & Wickens, C. D. (2008). Situation awareness, mental workload, and trust in automation: Viable, empirically supported cognitive engineering constructs. *Journal of Cognitive Engineering and Decision Making*, 2, 140–160. doi:10.1518/155534308X284417

SAE International. (2016). Taxonomy and definitions for terms related to driving automation systems for on-road motor vehicle (J3016\_201609). Retrieved from: https://saemobilus.sae.org/content/J3016\_201806

Samuel, S., Borowsky, A., Zilberstein, S., & Fisher, D. L. (2016). Minimum time to situation awareness in scenarios involving transfer of control from an automated driving suite. *Transportation Research Record: Journal of the Transportation Research Board*, 2602, 115–120. doi:10.3141/2602-14

Siegler, I., Reymond, G., Kemeny, A., & Berthoz, A. (2001). Sensorimotor integration in a driving simulator: Contributions of motion cueing in elementary driving tasks. *Proceedings of the Driving Simulation Conference* (pp. 21–32). Sophia-Antipolis, France.

Skitka, L. J., Mosier, K. L., & Burdick, M. (1999). Does automation bias decision-making? *International Journal of Human-Computer Studies*, *51*, 991–1006. doi:10.1006/ijhc.1999.0252

Sorkin, R. D. (1988). Why are people turning off our alarms?. *The Journal of the Acoustical Society of America*, 84, 1107–1108. doi:10.1121/1.397232

Tijerina, L., Blommer, M., Curry, R., Swaminathan, R., Kochhar, D. S., & Talamonti, W. (2017). An exploratory study of driver response to reduced system confidence notifications in automated driving. *IEEE Transactions on Intelligent Vehicles*, *1*, 325–334. doi:10.1109/TIV.2017.2691158

Van den Beukel, A. P., Van der Voort, M. C., & Eger, A. O. (2016). Supporting the changing driver's task: Exploration of interface designs for supervision and intervention in automated driving. *Transportation Research Part F: Traffic Psychology and Behaviour, 43,* 279–301. doi:10.1016/j.trf.2016.09.009

Van der Laan, J. D., Heino, A., & De Waard, D. (1997). A simple procedure for the assessment of acceptance of advanced transport telematics. *Transportation Research Part C: Emerging Technologies*, *5*, 1–10. doi:10.1016/S0968-090X(96)00025-3

Victor, T. W., Tivesten, E., Gustavsson, P., Johansson, J., Sangberg, F., Ljung Aust, M. (2018). Automation expectation mismatch: Incorrect prediction despite eyes on threat and hands on wheel. *Human Factors*, *60*, 1095–1116. doi:10.1177/0018720818788164

Watzenig, D., & Horn, M. (2017). Introduction to automated driving. In Watzenig D. & Horn M. (Eds.), *Automated Driving* (pp. 3–16), Springer. doi:10.1007/978-3-319-31895-0\_1

Wickens, C., Dixon, S., Goh, J., & Hammer, B. (2005). Pilot dependence on imperfect diagnostic automation in simulated UAV flights: an attentional visual scanning analysis (Tech Rep. AHFD-05 02/MAAD-05-02). University of Illinois, Urbana-Champaign, IL. Retrieved from:

https://www.researchgate.net/publication/235154940\_Pilot\_Dependence\_on\_Imperfect\_Diag nostic Automation in Simulated UAV Flights An Attentional Visual Scanning Analysis

Wiczorek, R., Balaud, M., & Manzey, D. (2015). Investigating benefits of likelihood alarm systems in presence of alarm validity information. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting* (Vol. 59, pp. 850–854). doi:10.1177/1541931215591255

Willemsen, D., Stuiver, A., & Hogema, J. (2015). Automated driving functions giving control back to the driver: a simulator study on driver state dependent strategies. *24th International Technical Conference on Enhanced Safety of Vehicles (ESV)* (No. 15-0109). Retrieved from: https://www-esv.nhtsa.dot.gov/Proceedings/24/files/24ESV-000109.PDF

Yang, Y., Götze, M., Laqua, A., Dominioni, G. C., Kawabe, K., & Bengler, K. (2017). A method to improve driver's situation awareness in automated driving. *Proceedings of the Human Factors and Ergonomics Society Europe Chapter* (pp. 29–47). Retrieved from: https://www.researchgate.net/profile/Yucheng Yang/publication/335992600 A method to i mprove driver's situation awareness in automated driving/links/5d89f160458515cbd1be5b 44/A-method-to-improve-drivers-situation-awareness-in-automated-driving.pdf

Zabyshny, A. A., & Ragland, D. R. (2003). False alarms and human-machine warning systems. *UC Berkeley: Safe Transportation Research & Education Center*. Retrieved from <u>https://safetrec.berkeley.edu/sites/default/files/publications/escholarship\_uc\_item\_88w8b8g3.</u> <u>pdf</u>

Zeeb, K., Härtel, M., Buchner, A., & Schrauf, M. (2017). Why is steering not the same as braking? The impact of non-driving related tasks on lateral and longitudinal driver interventions during conditionally automated driving. *Transportation Research Part F: Traffic Psychology and Behaviour, 50*, 65–79. doi:10.1016/j.trf.2017.07.008

Zhang, B., De Winter, J. C. F., Varotto, S., Happee, R., & Martens, M. H. (2019). Determinants of take-over time from automated driving: a meta-analysis of 129 studies. *Transportation Research Part F: Traffic Psychology and Behaviour*, 64, 285-307. doi:10.1016/j.trf.2019.04.020

Zhang, B., Wilschut, E. S., Willemsen, D. M. C., & Martens, M. H. (2019). Transitions to manual control from highly automated driving in non-critical truck platooning scenarios. *Transportation Research Part F: Traffic Psychology and Behaviour, 64,* 84-97. doi:10.1016/j.trf.2019.04.006

# 8. Conclusions, discussion and implication for practice

The aim of this thesis was to investigate car and truck drivers' behaviour and performance when resuming manual control back from automated driving, in order to provide input for designing safe and smooth take-over control strategies. The literature and empirical studies addressed the research questions defined in this thesis.

# **RQ1:** What factors influence driver response times in taking back control from automated to manual driving

To answer this research question, a comprehensive meta-analysis was conducted to systematically synthesize the findings of 129 studies that measured take-over response times in a wide range of experimental conditions related to the driver, the vehicle, the take-over situation, and the experimental set-up. Results clearly showed some systematic effects on driver take-over time. Most prominent factors were:

- Urgency of the take-over situation: The mean take-over time was substantially lower when the take-over situation was more urgent (with a shorter time budget available).
- 2) Non-driving task performed during driving automation Mean take-over times were substantially higher when the driver was holding an object in the hands or resting with the eyes closed. Performing a visually demanding nondriving task moderately increased the mean take-over response time.
- 3) Modality used for the take-over request: The mean take-over time was substantially lower when an auditory and/or vibrotactile take-over request was provided compared to a visual-only or no take-over request.
- Experience and familiarization: The mean take-over time was substantially lower when the driver had prior experience with the take-over situation.

5) Anticipation:

Being able to anticipate the take-over request from task-related or environmental cues reduced mean take-over response times.

6) Traffic situation: The mean take-over time was higher when other road users needed to be taken into consideration during the take-over process (e.g., participants had to take over control while driving in the middle lane, while the right/left lane contained traffic.)

Other examined factors, including drivers' mean age, whether the non-driving task demand is auditory or cognitive, whether the take-over request was directional or with peripheral visual stimuli, duration of the driving automation task, the complexity of take-over response, and the fidelity of the driving simulator, only showed minor or inconsistent influence on mean takeover times.

Besides findings of the meta-analysis, the empirical studies in this thesis provided additional insight into driver take-over times and its influencing factors. In the platooning studies, the total take-over time was broken down into two stages:

- Perception response time (to perceive and understand the take-over request) and
- Hand movement response time (to replace hands back on the steering wheel).

It was found that in most cases the hand-movement response time was significantly higher than the perception response time, suggesting movement response time as a dominant component of the total take-over time. Therefore, factors influencing driver motoric take-over process, such as non-driving posture and activities performed when resuming an optimal manual driving position, would substantially influence take-over response times. It was also found that car drivers took over significantly faster than professional truck drivers after using a hand-held tablet, which suggested that either driver type or vehicle type potentially is an influencing factor of take-over times.

This thesis also explored effects of two innovative driver assistant systems on take-over response times. Chapter 6 explored the possibility to overcome the restricted visual anticipation of short-distance platooning in trucks by implementing a see-through lead truck. Despite increasing eyes-on-road time during platooning, providing truck drivers with front view projection did not affect their take-over times in a critical situation compared to not using such technology. Chapter 7 investigated the effect of a monitoring request on driver take-over performance. Provided with both monitoring requests and take-over requests, drivers already regained certain level of mental and motor readiness before an actual take-over was required, and took over significantly faster compared to using a system that only provided take-over requests. This study further suggested that reliance with take-over requests and compliance with MRs could influence drive take-over times.

An overview of factors influencing take-over response times is depicted in Figure 8.1. These factors are mainly based on the meta-analysis that drew relatively more credible conclusions based on at least four studies. Other potentially influencing factors that were not included in the meta-analysis (i.e., too few studies investigated these factors by far) were also suggested based on findings of the empirical studies and previous literature.



Figure 8.1: Overview of factors influencing driver take-over response times (TOT) clustered in four categories: driver related, vehicle relate, situation related, and experiment related. Factors in the three inner layers were examined in the meta-analysis based on at least four studies, so with multiple replicated results. Factors in the innermost layer and the second inner layer have strong effects (marked with two signs) and moderate effects (marked with one sign) on take-over times, respectively. The red bar and the plus sign indicate that the presence of the corresponding factor increases take-over times, while the green bar and the minus sign indicate the opposite effect direction. The factors marked with a circle (in the third inner layer) have minor or inconsistent effects on take-over times. Factors in the outermost layer (in grey) were examples of other potentially influencing factor suggested in the empirical studies and previous literature, which were not included in the meta-analysis because there were no 4 studies available.

• RQ2: How do professional truck drivers and car drivers perform when decoupling from highly automated platoons in normal, non-critical situations under the influence of various task conditions?

The meta-review showed that hardly any studies were performed in a platooning context but rather with stand-alone automation. This thesis filled in this gap and specifically studied professional truck drivers' and car drivers' take-over performance when decoupling from an automated platoon, and how this can be influenced by four representative task conditions, namely monitoring (without non-driving tasks), using a handheld tablet, using a mounted tablet, and resting with eyes closed. The performance metrices concerned take-over response times, response to a brake event (immediately after the transition), and manual driving performance subsequent to the transition.

When required to take over control without time restrictions while platooning, drivers depicted large variation in response times ranging from 1.6 s to 13.3 s (combining all trials performed in the three empirical studies). Drivers took substantially more time to take over control after using a hand-held tablet or resting with eyes closed compared to monitoring the road. Performing a standardized visual-motoric task on a mounted tablet while platooning also generated a higher take-over time compared to the monitoring condition, but of smaller magnitude than holding the tablet in the hands. It was also found that car drivers in platoons took over faster than truck drivers in platoons due to a lower hand movement response time. Video recordings revealed that a number of truck drivers encountered additional barriers in the motoric take-over process such as having to switch reading glasses or taking extra time to shut down the tablet screen.

In response to the decelerating lead truck immediately after the control transition, all drivers could brake in time and avoid a collision after monitoring the road or using a tablet, while one crash occurred to a truck driver in the Eyes-closed condition due to delayed response. These findings suggest that in most cases drivers were ready for moderately complex driving situations after resuming control in non-critical scenarios after platooning with relatively low time headways. Impairing effects of using a tablet on post-transition brake response were only found for car drivers, while truck drivers did not depict differences in performance between task conditions. Significant differences between driver groups were suggested, in that professional truck drivers responded faster and braked more smoothly compared to car drivers. Truck drivers showed overall more robust and better brake responses, possibly due to their higher level of training and expertise than average car drivers.

In the trials without brake events, drivers' longitudinal and lateral manual driving performance were assessed after the transition to gain insight into carry-over effects of automated platooning involving small inter-vehicular gaps. Both car and truck drivers reduced the speed immediately after the control transition to increase following distances (time headways), but still drove with reduced time headways one minute after decoupling compared to their baseline time headway in the manual driving conditions. This implied that carry-over effects of platooning may last for several minutes after the control transition. Such effects appeared to be more profound after monitoring the lead vehicle while platooning, especially for truck drivers.

Impact of platooning on drivers' lateral control performance was only marginally present for truck drivers for the first 10 s after the control transition, in specific conditions. The first truck platooning study showed increased mean SDLP after having used a handheld tablet during platooning. In the second truck platooning study such effect was only found in the monitoring condition, but not in the mounted tablet condition or the eyes-closed condition. The inconsistent findings call for more studies for valid conclusions.

#### • RQ3: Could a monitoring request help driver respond more adequately with takeover performance in critical take-over situations?

In Chapter 7, a design solution was proposed and evaluated that aims to bridge the gap between automated with a driver out of the loop and completely manual driving. The proposed 'monitoring request' is designed to stimulate a dynamic allocation of monitoring tasks to the human driver and the automation system according to the uncertainty of the road segment. Drivers could take their eyes off the road and engage in non-driving tasks in relatively predictable and safe driving situations, and would be requested to monitor the road upon approaching an uncertain road segment for which the vehicle could not reliably predict whether critical take-over events are likely to occur. The system also provides a take-over request upon detection of an actual critical event.

Results of the study in general showed positive effects of the monitoring request. After receiving a monitoring request, all drivers paused the undergoing non-driving tasks on a handheld tablet and looked up on the road. Several drivers also resumed motor readiness by replacing hands on the wheel and hovering the feet above the pedal. These preparatory behaviours led to better take-over performance when a critical event actually occurred compared to the condition that only provided take-over requests, in terms of a shorter response time to the take-over request and a higher minimum time to collision. Drivers also expressed more positive subjective ratings regarding workload, trust, and acceptance after experiencing the innovative system compared to the system that only issued take-over requests. Furthermore, the presence of monitoring request did not affect drivers' eye-gaze behaviour during automated driving (i.e., in between the uncertain areas without system issued requests). In both conditions, drivers were equivalently distracted by the non-driving task and spent approximately 17% time looking on the road.

This study also suggested that the system providing both monitoring requests and take-over requests may induce potential risks. After experiencing the perfectly functioning system, drivers showed significantly worse take-over performance when the system only issued a monitoring request but failed to issue a take-over request upon a critical event, compare to the conditions in which they did receive a take-over request. Drivers appeared to over-rely on the TORs and seemed to have developed a "wait and see" attitude. In addition, because only a small portion out of all monitoring requests required an actual driver take-over, drivers tend to reduce motoric preparatory behaviours after experiencing successive monitoring requests that did not require an actual takeover, and took over more slowly if an actual take-over request and the cry-wolf effect could be concerns in the use of such system, and measures should be applied to stimulate proper trust calibration.

# • RQ4: What explains variability in take-over times? Is an adaptive approach (tuned to a specific driver) a feasible solution to increasing safe and smooth transitions to manual driving?

Besides focusing on mean take-over times, this thesis made efforts to understand within-group variability. The meta-analysis showed a strong positive correlation between mean take-over times and their standard deviations. This means that factors generate high mean take-over times would also generate large variability between drivers. The empirical studies especially addressed that the variation in the motoric take-over process was a main source for the variation in total take-over times. Very slow take-over responses were often associated with complex and random motoric activities when resuming motor readiness, such as looking for reading glasses, having difficulties adjusting seat position, double-checking if iPad was shut properly, putting on shoes that were taken off during automated driving, and stretching arms and necks. Large variation in drivers' perception time was also observed in the Eyes-closed condition. While some drivers could immediately respond to the take-over request, a few drivers seemed to have fallen asleep when the TOR was issued and only responded several seconds later. This situation generated the highest take-over response times measured in the empirical studies (>8.6 s, outside the 95th percentile).

The discussions above infer that dynamically changing driver states (e.g., task condition, posture, and mental states) and driving situation predominantly determine the driver's capability to take over control safely at a specific moment, while the driver's inherent characteristics, such as age, gender, and personality, only play a minor role. Therefore, a driving automation system that can adapt to individual drivers' states has potential to increase driving

safety at intermediate development phases, where the driver would be allowed to perform various non-driving tasks, but is still expected to resume control when requested. For example, upon detection of insufficient driver states, the system could provide pre-warnings to advise the driver away from undesired NDTs, take precautions such as increasing headways to the front vehicle, and allow a longer time budget if a take-over is actually required. However, designers should foresee the limit in providing precise estimations of a specific drivers' take-over readiness, and exercise caution when designing and implementing adaptive automation systems.

One of the first issues would be that by far only a limited number of determinants of take-over times have been identified, merely based on empirical studies in strictly controlled environment. More factors are to be studied in naturalistic driving conditions for a holistic picture, which may require enormous efforts. Furthermore, several substantial determinants of take-over times are not observable and extremely difficult to predict (if not impossible), such as familiarization and expectation with a specific take-over situation, and very importantly, what the driver would actually do during the take-over process. One example is that a participant forgot where he had put his spectacles and searched for several seconds before taking over control. It has to be kept in mind that drivers are most vulnerable in these unpredictable outlier cases, which would occur more often when drivers are less restrained during automated driving. In addition, as the driver learns safe usage of the automation system and calibrates trust predominantly from long-term on-road experiences (Lin, Ma, & Zhang, 2017; Walker, Boelhouwer, Alkim, Verwey, & Martens, 2018), adaptive systems may be unpredictable and confusing for the driver if the set of criteria triggering changes is not transparent and comprehensive (e.g., no specific take-over time budgets). This may hinder the process of establishing a correct mental model and induces potential risks (cf. Billings & Woods, 1994).

The discussion above indicated that to precisely tune to each individual driver would be a rather unrealistic goal. To compensate for the potential limit of adaptive driving automation systems, designers could work on accurate estimations of the system's capabilities and the complexity of the driving environment, and adjust the available functionalities and HMI. If effectively informed about the system status, the drivers would regulate their states (e.g., adjusting nondriving task conditions and non-driving postures) before an actual takeover event occurs. This approach is related to the idea illustrated in Chapter 7, in which drivers established higher levels of mental and motor readiness after receiving a monitoring request (MR) that communicated the situation uncertainty, and exhibited good take-over performance when critical events actually occurred. Also, what the driver is allowed to do could be restricted based on the capability of the car to handle the specific circumstances it drives in. This could also be monitored, as is currently also discussed within international committees such as EuroNCAP, UN-ECE and ISO (SC39 WG8). Interesting aspect here is that the entire system is not aimed at automating as much as possible, but at offering automation when it is reliable, and taking precautions of this is not the case, using the human capabilities without providing a task that does not fit (e.g. continuous monitoring). It seems that driver monitoring systems are expected to be more capable of identifying if a driver's state is in line with minimum requirements. This means that even if a driver is estimated to be in a sufficient functional state (e.g., monitoring with hands on the wheel), (s)he might also fail in takeover due to unobservable and unpredictable factors, so a fallback option or emergency procedure should be available regardless of a driver's estimated state.

#### 8.1 Implications for practice

Based on the findings of this thesis, several recommendations can be provided for researchers, industry and policy-makers towards safer transitions of control from automated driving to manual control. They are listed and explained below.

#### Regaining motor readiness plays a substantial role in the take-over process

This thesis pointed to the substantial influence of driver motoric take-over process on total takeover response times. If a fast take-over response may be requested (e.g., with a time budget shorter than 7 s), the driver should not engage in non-driving tasks involving severe biomechanical distractions, such as holding an object in the hand(s) or out of the driving position. Performing tasks on a mounted device close to the steering wheel would be more recommendable. In addition, the driver's non-driving postures and seating positions should be regulated to avoid undesired delay in takeover. The cabin interior elements could also be designed to regulate driver behaviours according to different automation modes and driver's use scenarios. For example, when driving in a complex area that requires driver monitoring, the vehicle does not allow the driver to transform the interior elements into working or relaxation mode (e.g., recline the seat backwards), and thus providing physical restraints to avoid undesired biomechanical distractions. When a take-over is requested, the interior elements could automatically transform to the manual driving mode, for instance through reconfiguration of the seat and the steering wheel to assist the driver with rapidly establishment of motor readiness.

## Sleeping at the wheel may pose a large safety threat as long as drivers ought to take over within certain time budgets

Studies presented in Chapter 3, 4 and 5 belong to the few existing ones that investigated driver take-over performance after having eyes closed for some period of time. The eyes-closed condition generated the few high take-over times among all trials performed within this thesis, by participants appearing to have fallen asleep before the TOR onset, even though closing their eyes was just for 8 minutes. The only post-transition crash also occurred in this condition although the participant claimed his readiness by pressing the button. These results suggest that sleeping at the wheel may have the largest impact on the drivers' take-over performance among all task conditions. Sleeping drivers would completely lose situation awareness, and need considerable amount of time to be awakened and to restore cognitive and sensory-motor functions (Kaduk, Roberts, & Stanton, 2020; Wörle, Metza, Othersenb, Baumann, 2020; see Tassi & Muzet, 2000 for sleep inertia). The biomechanical distractions that may come along, such as sitting with the seatback in the most reclined position, would add to the difficulties to rapidly resume manual control after sleeping. Strict measures must be taken to prevent driver from sleeping during automated driving as long as critical takeovers may occur. For example, upon detection of extreme sleepiness and sleep onset, the system would issue warnings and require the driver to stop the vehicle and take a rest or only allow this in conditions for which the system knows it can drive reliably for a longer period of time.

#### Drivers should be supported for a period of time after the control transition to manual

Findings from this thesis suggested impact of platoon driving on subsequent manual driving performance. After decoupling from a platoon, drivers may drive with reduced TWH for a couple of minutes. Several previous studies also pointed to carryover effects of driving automation on drivers' attention allocation (e.g., Miller & Boyle, 2019; Vogelpohl, Kühn,

Hummel, Gehlert, &Vollrath, 2018), such as reduced focus on road centre, longer eyes-off-road glances, and inadequate mirror checks. These findings suggest that drivers should be supported through this vulnerable period of performance recovery. Possible means can be HMIs that remind the driver to maintain a safe THW, guide the driver to perform safety checks immediately after resuming control, and direct drivers' attention towards critical areas on the road. However, cautions should be exercised that these HMI interfaces may induce additional distractions, increase driver workload and have their own reliability issues. When decoupling from a platoon, the system could also automatically increase the THW to a safe threshold before asking the driver to take over control. If dedicated lanes were to be implemented, a transition area following the end of the automation zone would allow the driver to recover manual driving performance in a safe environment before entering public roads.

#### Multiple approaches should be combined to ensure road safety of automated driving

The discussions above suggest that multiple approaches need to be combined to manage variability within and between drivers and ensure driving safety at different levels of automation. Driver state monitoring systems could assess if discrepancies between the driver's current state and the desired state exist. If so, the system can issue advisory notifications or warnings, or adjust driving modes to dissolve the discrepancies. Upon detection of extremely risky states according to clearly defined criteria or rules, such as sleeping onset or not having hands on the wheel where imminent takeovers could occur, the system should stop drivers from using driving automation functions until they resume minimum required states. This thesis also cautions that driver state monitoring and prediction has its limits, and additional measures should be taken to ensure road safety. For example, the system should actively and effectively communicate with the driver the current status of the vehicle and uncertainties in the driving environment, and thus prompt the driver to adjust their state accordingly. Besides implementation of digital HMI, the layout of cabin elements can be designed to regulate drivers' non-driving posture and engagement in non-driving activities according to the current task demand, and assist drivers with a fast resumption of motoric readiness if needed. In the end, fallback options should be available whenever possible in case the driver cannot respond adequately in takeover.

#### 8.2 Recommendations for future research

In the end of this thesis, some boundaries of the studies performed within the scope of this thesis are discussed, based on which recommendations for future research are proposed.

First, all empirical studies in this thesis, and the large majority of the take-over studies included in the meta-analysis, were conducted in driving simulators. Virtual and simplified driving environments in a simulator generally raise issues of ecological validity, such as reduced risk perception and workload. Also worth mentioning is that participants may tend to respond in favour of the hypothesis of the study and behave well under observation (see Nichols & Maner, 2008 for "the good-subject effect"). To what extend can the findings be generalized to real-life scenarios needs further investigation, even though various studies have showed at least relative validity (Kaptein, Theeuwes, & Van der Horst, 1996; Riener, 2010; Risto & Martens, 2014; Walker, Hauslbauer, Preciado, Martens, & Verwey, 2019).

In addition, only a small variety of experiment conditions could be examined in the empirical studies, and the periods of automated driving were short (4 - 15 min). Limited insight was gained on drivers' behaviour in naturalistic settings, and the effects of prolonged driving automation on driver states and take-over performance. In the platooning studies, only four task

conditions were implemented under well controlled environment, each lasting for less than 10 minutes. The traffic volume was low throughout the platoon driving simulation, while driver's takeover performance, subsequence car following behaviour, and preferred THW could be substantially different in high traffic volume conditions. In the car driving simulator study investigating effects of monitoring requests, a pedestrian crossing the road was the only scenario presented, and fixed time budgets for the monitoring request (12 s) and the take-over request (5 s) were used, which might have led to participants' familiarization and expectation with the critical situations and reduced take-over response times. In each drive, only five MRs and two TORs were issued over the span of 15 min, and it still remains a question how drivers' compliance with MR change in long term use.

Research in this thesis made a unique contribution to the literature by exploring driver behaviour during and right after decoupling from a highly automated platoon and the effects of monitoring requests that adapt to the uncertainty of the driving environment. Future research could expand the analysis to more realistic situations and test a larger variety of scenarios. Particularly, efforts can be made to investigate driver behaviours in prolonged, naturalistic use of automation, to explore impact of sleeping at wheel on driver take-over performance, and to understand how drivers interact and comply with adaptive automation systems. These will provide valuable input for the development of an effective human-in-the-loop adaptive automation system.

#### References

Gerber, M. A., Schroeter, R., Xiaomeng, L., & Elhenawy, M. (2020). Self-Interruptions of Non-Driving Related Tasks in Automated Vehicles: Mobile vs Head-Up Display. *Proceedings of the ACM Conference on Human Factors in Computing Systems*. Association for Computing Machinery (ACM). doi:10.1145/3313831.3376751

Kaduk, S. I., Roberts, A. P. J., & Stanton, N. A., (2020). I just woke up behind the wheel: Sleep Inertia as a new risk in semi-automated cars. In R. Charles and D. Golightly (Eds.), *Contemporary Ergonomics and Human Factors 2020*. Retrieved from: <u>https://publications.ergonomics.org.uk/uploads/I-just-woke-up-behind-the-wheel-Sleep-Inertia-as-a-new-risk-in-semi-automated-cars.pdf</u>

Kaptein, N., Theeuwes, J., & Van der Horst, R. (1996). Driving simulator validity: Some considerations. *Transportation Research Record: Journal of the Transportation Research Board*, *1550*, 30–36. doi:10.3141/1550-05

Miller, E. E., & Boyle, L. N. (2019). Adaptations in attention allocation: implications for takeover in an automated vehicle. *Transportation research part F: traffic psychology and behaviour*, *66*, 101-110. doi:10.1016/j.trf.2019.08.016.

Nichols, A. L., & Maner, J. K. (2008). The good-subject effect: Investigating participant demand characteristics. *The Journal of general psychology*, *135*, 151-166. doi:10.3200/GENP.135.2.151-166

Riener, A. (2010). *Simulating on-the-road behavior using a driving simulator. Paper* presented at the 2010 Third International Conference on Advances in Computer-Human Interactions, Saint Maarten, Netherlands Antilles (pp. 25–31). doi:10.1109/ACHI.2010.10

Risto, M., & Martens, M. H. (2014). Driver headway choice: A comparison between driving simulator and real-road driving. *Transportation Research Part F: Traffic Psychology and Behaviour*, 25, 1–9. doi:10.1016/j.trf.2014.05.001

Tassi, P., & Muzet, A. (2000). Sleep inertia. *Sleep medicine reviews*, *4*, 341-353. doi:10.1053/smrv.2000.0098.

Thomas, I., & Bruck, D. (2010). Awakening of sleeping people: A decade of research. *Fire technology*, 46, 743-761. doi:10.1007/s10694-008-0065-5.

Vogelpohl, T., Kühn, M., Hummel, T., Gehlert, T., & Vollrath, M. (2018). Transitioning to manual driving requires additional time after automation deactivation. *Transportation research part F: traffic psychology and behaviour*, *55*, 464-482. doi:10.1016/j.trf.2018.03.019

Walker, F., Hauslbauer, A. L., Preciado, D., Martens, M. H., & Verwey, W. B. (2019). Enhanced perception of risk in a driving simulator. *International journal of human factors modelling and simulation*, 7, 100-118. doi:10.1504/IJHFMS.2019.105428

Wörle, J., Metz, B., Othersen, I., & Baumann, M. (2020). Sleep in highly automated driving: Takeover performance after waking up. *Accident Analysis & Prevention*, *144*. doi:10.1016/j.aap.2020.105617

## **Summary**

Driving automation holds great promise for safer and more efficient road transportation, and is becoming a reality thanks to the rapid advancement of technology. However, before full driving automation arrives, the driver would have to take over control of the vehicle when the system fails or reaches its operational limits, which poses new road safety risks at different stages of development. When the system is less capable and reliable, the driver has to closely monitor the system and take over imminent control when necessary. This challenges humans' inherent weak point of staying vigilant over a prolonged period of time. When the technology becomes more mature, the driver would be allowed to engage in a wide range of non-driving tasks, but occasional human interventions still cannot be avoided. How to ensure drivers in various mental and physical states to take over control safely become a major challenge at this stage. A large number of studies have tackled human factors issues related to control transitions, and suggested that no single take-over time budget applies to all drivers in all situations. While an adaptive approach is called for to support individual drivers in taking over control, a better understanding of driver take-over process and the variability between and within drivers is still needed to achieve this goal.

This PhD thesis addresses the challenges stated above and contributes to designing safe and comfortable control transitions to manual. A particular focus was on control transitions in truck platooning scenarios, which have received only limited attention despite of the significance of platooning technology. The first objective is to explore factors influencing the take-over response time and its variability, and to gain a deeper insight in the actual take-over process. The second objective is to systematically study professional drivers' take-over performance in truck platooning scenarios, in order to deliver input for the development of platooning systems and the corresponding infrastructures. The third objective is to design and evaluate potential approaches that prime drivers for a safe and smooth take-over, and to discuss the feasibility of an adaptive and personalized control transition approach.

The first part the thesis (Chapter 2) presents an exhaustive meta-review of 129 studies that reported mean take-over response times, aiming to provide the state of the art on driver take-over research, and to explore determinants of take-over times on an aggregated level. Three complementary approaches were employed: (1) a within-study analysis, in which differences in mean take-over time were assessed for pairs of experimental conditions, (2) a between-study analysis, in which correlations between experimental conditions and mean take-over times were assessed, and (3) a linear mixed-effects model combining between study and within-study effects. The three methods showed that a shorter mean take-over time is associated with a higher urgency of the situation, not using a handheld device, not performing a visual non-driving task, having experienced another take-over scenario before in the experiment, and receiving an auditory or vibrotactile take-over request as compared to a visual-only or no take-over request. It is also found the mean and standard deviation of the take-over time were highly correlated, indicating that the mean is predictive of variability. These findings point to directions for new research, in particular concerning the distinction between drivers' ability and motivation to take over, and the roles of urgency, biomechanical distraction, and prior experience.

The meta-review showed that hardly any take-over studies were performed in platooning context. To fill in the research gap, **the second part** of the thesis (**Chapter 3-6**) presents four empirical driving simulator studies investigating driver take-over performance when leaving highly automated truck and car platoons.

- Chapter 3 and Chapter 4 describe two truck platooning studies investigating professional truck drivers' take-over performance in non-critical scenarios. In both studies, three task conditions were implemented during platoon driving, namely monitoring without non-driving task, interacting with a tablet PC, and relaxing with eyes closed. The difference between the two studies lies in whether the tablet PC was holding in the hands or mounted on the centre console. Results showed substantially longer take-over times with high variability when the driver was using a hand-held tablet or relaxing with eyes closed, compared to monitoring the road or using a mounted tablet PC. By measuring perception-response times instead of merely total take-over times, hand movement response time was found to be the dominant component of the total take-over time, being influenced by the motoric manoeuvres to resume physical readiness before taking over control (e.g., putting away the hand-held tablet, putting on/off reading glasses, adjusting seating position). Analyses of post-transition manual driving performance showed that truck drivers could stabilize the truck within 10-20 s after the transition in all task conditions, and suggested potential carry-over effects involving small gaps during platooning. In response to a brake event immediately after the control transition, all drivers successfully avoided colliding with the decelerating front truck, except for one driver in the eyes-closed condition. These findings suggest the importance to focus on both cognitive and motoric preparation phases before resuming manual control, and pointed to the potential risk of resuming manual driving after sleeping or relaxing with eyes closed.
- As platooning technology can be applied to both heavy and light vehicles, **Chapter 5** explores behavioural similarities and differences between car drivers and professional truck drivers. This study compares professional truck drivers' take-over performance when leaving a highly automated platoon (as reported in Chapter 4) to that of car drivers measured in the same driving simulator (with a different vehicle model and mock-up)

using the identical experimental design. Compared to car drivers, truck drivers took over control more slowly when using a hand-held tablet due to a slower hand movement response; drove more steadily with a larger time gap after the transition; and braked more quickly and less aggressively in response to the decelerating lead vehicle. These findings in general suggest a more cautious manual driving style in professional truck drivers after decoupling from the platoon. In addition, both driver groups showed reduced time headways after the transition, suggesting again carry-over effects of platoon driving involving small gaps.

• For drivers in the truck platoon, monitoring surrounding traffic environment and foreseeing upcoming hazardous situations is very difficult due to very short intervehicular distances and consequently a heavily blocked front view. **Chapter 6** explores whether providing drivers in a truck platoon with additional visual information of the traffic environment can influence their monitoring pattern and increase awareness of upcoming critical situations. Twenty-two professional truck drivers were divided into two groups, either following a see-through lead truck (i.e., with projection of forward scene attached to the rear of the lead truck), or a normal lead truck until the automation system failed unexpectedly in a critical situation. Results showed that when provided with front view projection, the drivers spent 10% more time monitoring the road, and responded less severely to a critical situation, suggesting positive effects of the "see-through" technology. Nevertheless, such technology did not affect their take-over times in a critical situation.

In on-road settings, it is not feasible to always allow sufficient time budgets for drivers in various states to take over safely, neither is it realistic to require prolonged effective driver monitoring. In the third part of the thesis (Chapter 7), a design solution was proposed and evaluated that may bridge the gap between automated with a driver out of the loop and completely manual driving. An HMI concept was designed that provides a monitoring request (MR) when approaching a location where driver take-over is likely to be requested. The MR asked the driver to pause the non-driving task, monitor the traffic environment, and be prepared for a potential take-over. If a critical event was detected, the system provided a take-over request (TOR) as well. The aim is to stimulate a dynamic allocation of monitoring tasks to human and automation that adapts to the system capability and the situation complexity, and to better prepare drivers to take over safely and efficiently. In the simulator-based experiment with 41 participants, the effects of the MR+TOR system were assessed by comparing to a conventional system that only issued TORs. Results showed that the MR+TOR system improved participants' take-over performance in terms of shorter take-over response time and longer minimum time collision, and yielded more positive subjective ratings regarding workload, trust, and acceptance. Because only a small portion out of all MRs require an actual driver take-over, an additional analysis was conducted to investigate how drivers' compliance with MRs was associated with previously experienced scenarios. Although drivers showed good overall compliance by looking up to the road in response to MRs in all cases, hand and foot preparatory behaviour appeared to deteriorate after experiencing an MR without a critical event, and increased after a take-over event. These findings suggest that the cry-wolf effects may be a concern in the use of MR, and measures should be applied to stimulate proper trust calibration.

My PhD thesis provided a comprehensive research on factors influencing driver take-over response times, made an initial contribution to the understanding of driver behaviour during and right after decoupling from a highly automated platoon, and proposed an innovative HMI

design to better prepare drivers for potential critical take-overs. The findings suggested that dynamically changing driver states and driving situation predominantly determine the driver's capability to take over control safely at a specific moment. While a driving automation system that can adapt to individual drivers' states has potential to increase driving safety during transitions, one should also foresee the limit in providing precise estimations of a specific drivers' take-over readiness. Multiple approaches need to be combined to manage variability within and between drivers and ensure driving safety at different levels of automation, such as actively communicating with the driver the current status of the vehicle and uncertainties in the driving environment, designing cabin layout to regulate drivers' non-driving posture and engagement in non-driving activities according to the current task demand, and providing fallback options whenever possible in case the driver cannot respond adequately in takeover.

### Samenvatting in het Nederlands

Volledig automatisch rijden kan in potentie voor veiligere en efficiëntere mobiliteit zorgen. Dankzij technologische vooruitgang wordt volledig automatisch rijden realiteit in de nabije toekomst. Echter, voordat de volledige rij-automatisering er aankomt zal er een fase zijn waarbij de bestuurder de controle over het voertuig incidenteel overneemt, bijvoorbeeld wanneer het systeem uitvalt of de limieten van het systeem worden bereikt. Deze tussenfase van de ontwikkeling van automatisering zullen leiden tot nieuwe veiligheidsrisico's. Wanneer het systeem nog niet volledig automatisch en betrouwbaar is, zal de bestuurder het systeem en het verkeer moeten blijven monitoren om direct de controle over te nemen wanneer dat noodzakelijk is. Hiermee wordt een inherent zwak punt van de mens aangesproken namelijk om gedurende een langere periode waakzaam te blijven (vigilantie) en de status van het systeem en het verkeer te volgen. Wanneer de technologie volwassener wordt, zou de bestuurder een breed scala aan niet-rijdende taken oftewel secundaire taken mogen uitvoeren. Echter incidentele menselijke tussenkomst kan dan nog noodzakelijk zijn. Het wordt een uitdaging om ervoor te zorgen dat in deze fase van de technologieontwikkeling bestuurders in verschillende mentale en fysieke toestanden de controle van het voertuig veilig kunnen overnemen. Een aantal studies hebben de factoren die verband houden met de wisseling van de controle tussen mens en voertuigautomatisering onderzocht, en komen daarbij tot de conclusie dat er geen standaard overnametijd kan worden gebruikt voor een veilige overname van de controle van het voortuig door de bestuurder. Hoewel een adaptieve aanpak wordt gesuggereerd om individuele bestuurders te ondersteunen bij het overnemen van de controle, is een beter begrip van het overnameproces van bestuurders en de variabiliteit tussen en binnen bestuurders nodig om dit doel te bereiken.

Dit proefschrift gaat in op de hierboven genoemde uitdagingen en draagt bij aan het ontwerpen van veilige en comfortabele overgang van automatische naar handmatige voertuigbediening. Bijzondere aandacht ging uit naar controle-overgangen in scenario's voor het platoonen van vrachtwagens, die ondanks het onderschreven belang van platooning slechts beperkte aandacht hebben gekregen. Het eerste doel is om te onderzoeken welke factoren de overnametijd en de variabiliteit daarvan beïnvloeden, om inzicht te krijgen in het daadwerkelijke overnameproces. Het tweede doel is om systematisch de overnameprestaties van professionele chauffeurs te onderzoeken in scenario's voor vrachtwagenplatooning, om zo input te leveren voor de ontwikkeling van platooningsystemen en de bijbehorende infrastructuren. De derde doelstelling is het ontwerpen en evalueren van mogelijke benaderingen om de bestuurder te attenderen en voor te bereiden op een veilige en soepele overname. En tenslotte wordt de haalbaarheid van een adaptieve en gepersonaliseerde overgang van de controle behandeld.

**Het eerste deel** van het proefschrift (**Hoofdstuk 2**) presenteert een uitputtende meta-review van 129 onderzoeken die gemiddelde overnameresponstijden rapporteerden, met als doel de stand van de techniek te verschaffen op het gebied van onderzoek naar het overnemen van de controle door bestuurders en om factoren die de overname beïnvloeden te onderzoeken op een geaggregeerd niveau. Er werden drie complementaire benaderingen gebruikt: (1) een analyse, waarbij verschillen in gemiddelde overnametijd werden beoordeeld van experimentele condities binnen dezelfde studie (2) een analyse tussen studies, waarin correlaties tussen experimentele condities en gemiddelde overname tijden werden beoordeeld, en (3) een linear mixed-effects model die deze twee analyses combineert.

De drie analyses toonden aan dat een kortere gemiddelde overnametijd gepaard gaat met: een hogere urgentie van de situatie, dat er geen technologie gebruikt wordt, dat er geen secundaire visuele taak wordt uitgevoerd, dat men eerder in het experiment een ander overnamescenario heeft meegemaakt, en het ontvangen van een multimodaal (auditief en/of tactiel) overnameverzoek, dit in vergelijking met een visueel of geen overnameverzoek. Ook blijkt dat het gemiddelde en de standaarddeviatie van de overnametijd sterk gecorreleerd waren, wat aangeeft dat het gemiddelde voorspellend is voor variabiliteit. Deze bevindingen geven aanwijzingen voor nieuw onderzoek, in het bijzonder met betrekking tot het onderscheid tussen het vermogen en de motivatie van bestuurders om de controle over te nemen, en de rol van urgentie, biomechanische afleiding en eerdere ervaring van de bestuurder.

Uit de meta-review bleek dat er nauwelijks overnamestudies werden uitgevoerd in de context van platooning. Om de lacune in het onderzoek te dichten, bestaat **het tweede deel** van het proefschrift (**Hoofdstuk 3-6**) uit vier empirische rijsimulatorstudies die de overnameprestaties van bestuurders onderzoeken bij het verlaten van geautomatiseerde auto- en vrachtwagenplatoons.

Hoofdstuk 3 en Hoofdstuk 4 beschrijven twee platooning studies die de overnameprestaties van professionele vrachtwagenchauffeurs onderzochten in nietkritieke scenario's. In beide onderzoeken waren er drie condities tijdens het besturen van een platoon, namelijk monitoren zonder extra secundaire taak, interactie met een tablet, en ontspannen met gesloten ogen zitten. Het verschil tussen de twee studies was de tablet conditie. Waarbij in de ene studie de tablet in de handen werd gehouden, werd in de andere studie de tablet op de middensconsole gemonteerd. De resultaten lieten aanzienlijk langere overnametijden zien met grotere variabiliteit wanneer de bestuurder een draagbare tablet gebruikte of ontspande met gesloten ogen, vergeleken met het monitoren van de weg of het gebruik van een tablet in de console. Door perceptiereactietijden te meten in plaats van louter de totale overnametijden, bleek dat bewegingsreactietijd de dominante component was bij de totale overnametijd. Deze bewegingsreactietijd werden beïnvloed door de motorische handelingen die nodige waren ter voorbereiding om de controle van het voertuig over te nemen (bijv. de draagbare tablet opbergen, een leesbril opzetten/ afzetten, de zitpositie aanpassen). Analyses van het rijgedrag na de overgang naar handmatig rijden toonden aan dat vrachtwagenchauffeurs de truck binnen 10-20 seconden na de overgang in alle condities konden stabiliseren, en suggereerden een mogelijk overdrachtseffect van het rijden in een platoon waarbij iets minder ruimte werd gelaten tot de voorganger na de handmatige overname. Als reactie op een plotselinge remactie onmiddellijk na de overgang van de controle, konden alle chauffeurs met succes een botsing met de vertragende voorste truck vermijden, behalve één chauffeur in de conditie waarbij de ogen gesloten waren. Deze bevindingen suggereren het belang om te focussen op zowel cognitieve als motorische voorbereidingsfasen voordat handmatige controle wordt hervat, en wezen op het potentiële risico van het hervatten van handmatig autorijden na het slapen of ontspannen met gesloten ogen.

- Platooning technologie kan zowel bij vrachtwagens als bij auto's worden toegepast, Hoofdstuk 5 behandelt overeenkomsten en verschillen in gedrag van automobilisten en professionele vrachtwagenchauffeurs. Dit onderzoek vergelijkt de overnameprestaties van professionele vrachtwagenchauffeurs bij het verlaten van een sterk geautomatiseerd platoon (zoals gerapporteerd in Hoofdstuk 4) met die van automobilisten gemeten in dezelfde rijsimulator (met een ander voertuigmodel en mock-up) met behulp van hetzelfde experimentele ontwerp. In vergelijking met automobilisten waren vrachtwagenchauffeurs langzamer bij de overname van de controle wanneer zij een tablet vasthielden vanwege een tragere handbeweging; zij reden gelijkmatiger met een grotere afstand tot de voorligger na de overgang; en remde sneller en minder hard in reactie op de afremmende voorligger, het voorste voertuig van de platoon. Deze bevindingen suggereren over het algemeen een behoedzame handmatige rijstijl bij professionele vrachtwagenchauffeurs na ontkoppeling van het platoon in vergelijking tot de automobilisten. Daarnaast vertoonden beide bestuurdersgroepen een kortere volgtijd tot de voorligger na de overgang naar handmatig rijden, wat duidt op wederom overdrachtseffecten van het rijden in een volledig geautomatiseerd platoon met korte volgafstanden.
- Voor vrachtwagenchauffeurs die rijden in een platoon is het waarnemen van het omringend verkeer en het anticiperen van een mogelijke gevaarlijke situaties lastig. Dit vanwege de zeer korte volgafstanden tussen de voertuigen, waardoor het zicht naar voren sterk belemmerd wordt. In **Hoofdstuk 6** wordt het effect van extra visuele informatie vastgesteld: het monitoren van de verkeerssituatie en het anticiperen op mogelijke kritieke verkeerssituaties. Tweeëntwintig vrachtwagenchauffeurs werden verdeeld in twee groepen, de eerste groep reed achter een doorzichtige voorste vrachtwagen, d.w.z. met een projectie van de verkeerssituatie aan de achterkant van de voorste vrachtwagen. De tweede groep reed achter een normale voorste vrachtwagen. groepen reden in een automatisch rijdend Beide platoon, totdat het automatiseringssysteem onverwachts uitviel. De resultaten toonden aan dat wanneer de chauffeurs zicht hadden op de verkeerssituatie door projectie op de achterkant van de vrachtwagen, de chauffeurs 10% meer tijd besteedden aan het monitoren van de weg en minder abrupt reageerden op een kritieke situatie. Dit wijst op positieve effecten van de projectietechnologie, toch had deze projectietechnologie geen invloed op hun overnametijden in de kritieke situatie.

Op de weg is het niet haalbaar om bestuurders in verschillende staat van alertheid altijd voldoende tijd te geven om de controle over het voertuig veilig over te nemen. En het is evenmin realistisch om van de bestuurder te eisen langdurig alert te blijven en het systeem te monitoren. In het derde deel van het proefschrift (Hoofdstuk 7) wordt een ontwerpoplossing voorgesteld en geëvalueerd die de kloof kan overbruggen tussen automatisch rijden en volledig handmatig rijden. Er is een HMI-concept ontworpen dat zorgt voor een monitoringverzoek (Monitoring

Request; MR) bij het naderen van een locatie waar de bestuurder waarschijnlijk wordt gevraagd om de controle over te nemen. De MR vroeg de bestuurder om de secundaire taak te onderbreken, de verkeerssituatie te inspecteren en voorbereid te zijn op een mogelijke overname. Als er een kritieke verkeerssituatie werd gedetecteerd, stelde het systeem ook een overnameverzoek voor (Take-Over Request; TOR). Het doel is om een dynamische toewijzing van controletaken aan mens en automatisering te stimuleren die zich aanpast aan de systeemcapaciteit en de complexiteit van de situatie, en om bestuurders beter voor te bereiden om het rijden veilig en efficiënt handmatig over te nemen. In dit simulator experiment met 41 bestuurders, waarbij de effecten van het MR+TOR-systeem werden beoordeeld door vergelijking met een conventioneel systeem dat alleen TOR gebruikte. De resultaten toonden aan dat het MR + TOR-systeem de overnameprestaties van de bestuurders verbeterde in termen van een kortere reactietijd bij de overname en een langere minimal time-to-collission, en meer positieve subjectieve beoordelingen opleverde met betrekking tot mentale werklast, vertrouwen en acceptatie. Omdat slechts een klein deel van alle MR's een daadwerkelijke overname van de chauffeur vereistte, is een aanvullende analyse uitgevoerd om te onderzoeken hoe de naleving van MR's door bestuurders beinvloed werd door eerder ervaren scenario's. Hoewel bestuurders over het algemeen een goede naleving lieten zien door in alle gevallen naar de weg te kijken als reactie op MR's, leek het voorbereidende gedrag van handen en voeten te verslechteren na het ervaren van een MR zonder een kritieke gebeurtenis, en het voorbereidend gedrag nam toe na een voorgaand secnario met een overname. Deze bevindingen suggereren dat de foutpositieven een punt van zorg kunnen zijn bij het gebruik van MR, en dat er rekening gehouden moet worden met het stimuleren van een adequate mate van vertrouwen in het systeem.

Mijn proefschrift bevat een uitgebreid onderzoek naar de factoren die van invloed zijn op overname reactietijden tijden van bestuurders. Het levert een bijdrage aan het begrip van het rijgedrag tijdens en direct na het ontkoppelen van een automatisch rijdend platoon en stelt een innovatief HMI-ontwerp voor om bestuurders beter voor te bereiden op mogelijke kritieke overnames. De bevindingen suggereerden dat dynamisch veranderende mentale en fysieke toestanden van bestuurders en verkeerssituaties voornamelijk bepalend zijn voor het vermogen van de bestuurder om op een bepaald moment de controle veilig over te nemen. Hoewel een adaptief systeem voor automatisch rijden zich kan aanpassen aan de toestand van individuele bestuurders, en daarmee de rijveiligheid tijdens overgangen kan verhogen, is dit vooralsnog ook gelimiteerd door het vermogen nauwkeurige accurate schattingen te geven van overname responsetijden van specifieke bestuurders. Meerdere benaderingen moeten worden gecombineerd om variantie binnen en tussen bestuurders mee te nemen tijdens verschillende niveaus van rijautomatisering om de verkeersveiligheid te kunnen garanderen. Zoals actief communiceren met de bestuurder over de huidige status van het voertuig en verkeerssituaties, het ontwerpen van de cabine om de niet-rijdende houding van chauffeurs en betrokkenheid bij secundaire activiteiten in overeenstemming te brengen met de huidige taakvereiste en het bieden van back-up mogelijkheden waar mogelijk, voor het geval de chauffeur niet adequaat kan reageren bij de overname van de controle van het voertuig.

## About the author



Bo Zhang was born on August, 28<sup>th</sup> 1988 in Harbin, Heilongjiang, China. Bo graduated from the Technical University of Munich in September 2015 with the M. Sc. degree in Human Factors Engineering. In her master thesis, she investigated commercial airline pilots' manual skill degradation induced by extensive use of Autopilot systems and a lack of on-the-job practice. Before starting her study in Germany, Bo obtained her B. Sc. degree in Industrial Design at Harbin Institute of Technology, China, in 2011. In January 2016, Bo joined the Department Centre for Transport Studies at the University Twente as a Marie Curie Fellow in the Project HFauto – Human Factors of Automated Driving. She investigated car and truck drivers' behaviour and performance when taking over control from automated driving in various conditions, aiming to provide implications for achieving a safe and smooth control transition back to manual.

In her spare time, Bo is an enthusiast and practitioner of contemporary visual art. Since 2019, she has been undertaking part-time Bachelor Programme in Fine Arts at the Gerrit Rietveld Academy in Amsterdam. Fascinated by how art evokes universal emotions in people from all around the world, Bo wonders what makes us human and forms the basis for universal communication across cultures, and hopes to contribute to safe and seamless human-system interactions that are accessible and beneficial to all people regardless of their social, cultural, or personal characteristics.

#### **Journal Papers**

**Zhang, B.**, Hogema, J. H., Willemsen, D. M., Wilschut, E. S., Martens, M. H. (2020). *Taking over* manual control from automated platoon driving: A comparison between passenger car drivers and professional truck drivers. Under review.

**Zhang, B.**, De Winter, J. C. F., Varotto, S., Happee, R., Martens, M. H. (2019). Determinants of take-over time from automated driving: A meta-analysis of 129 studies. *Transportation Research Part F: Traffic Psychology and Behaviour*, *64*, 285-307.

**Zhang, B.**, Wilschut, E. S., Willemsen, D. M., Martens, M. H. (2019). Transitions to manual control from highly automated driving in non-critical truck platooning scenarios. *Transportation Research Part F: Traffic Psychology and Behaviour*, *64*, 84-97.

Lu, Z.\*, **Zhang, B.**\*, Feldhütter, A., Happee, R., Martens, M., & De Winter, J. C. F. (2019). Beyond mere take-over requests: The effects of monitoring requests on driver attention, take-over performance, and acceptance. *Transportation research part F: traffic psychology and behaviour*, *63*, 22-37.

Haslbeck, A.\*, & **Zhang, B.**\* (2017). I spy with my little eye: Analysis of airline pilots' gaze patterns in a manual instrument flight scenario. *Applied ergonomics*, 63, 62-71.

\*Joint first authors

#### Peer-reviewed conference papers

**Zhang, B.**, Lu, Z., Happee, R., De Winter, J. F. C., Martens, M. H. (2019) *Compliance with Monitoring Requests, Biomechanical Readiness, and Take-Over Performance: Video Analysis from a Simulator Study.* The 13th ITS European Congress, Brainport Eindhoven, The Netherlands.

**Zhang, B.**, Wilschut, E., Willemsen, D., Martens, M. (2017). *Driver Response Times when Resuming Manual Control from Highly Automated Driving in Truck Platooning Scenarios*. Road Safety & Simulation International Conference, The Hague, The Netherlands.

Zhang, B., Wilschut, E. S., Willemsen, D. M. C., Alkim, T., Martens, M. H. (2017). The effect of see-through truck on driver monitoring patterns and responses to critical events in truck platooning. In N. A. Stanton (Ed.), *Advances in human aspects of transportation* (pp. 842-852). Cham: Springer.

#### **Presentations and Scientific Meetings**

Zhang, B., Hogema, J. H., Willemsen, D. M., Wilschut, E. S., Martens, M. H., (2019). *Taking Over Manual Control from Automated Vehicle Platoon: Do Passenger Car Drivers and Professional Truck Drivers Behave Differently?* Presentation at the 6th TRAIL PhD Congress, Utrecht, The Netherlands.

Zhang, B., Lu, Z., Happee, R., Martens, M., & De Winter, J. C. F. (2018). From Passive to Active Drivers' Control Intervention: Exploring the Effects of Uncertainty Monitoring Request on Driver Attention, Take-over Performance and Acceptance. Presentation at the 5th TRAIL PhD Congress, Utrecht, The Netherlands.

Zhang, B., De Winter, J. C. F., Varotto, S., Happee, R., Martens, M. H. (2017). *Factors Influencing Driver Take-over Time when Resuming Manual Control from Automated Driving*. Presentation at the 4th TRAIL PhD Congress, Utrecht, The Netherlands.

Zhang, B., Wilschut, E. S., Willemsen, D. M., Martens, M. H. (2016). *Are you ready to take over? An eye movement study in simulated truck platooning scenarios.* Presentation at the 3nd TRAIL PhD Congress, Utrecht, The Netherlands.

HF-Auto Progress Meeting and Workshop, Chalmers University of Technology and Swedish National Road and Transport Research Institute (VTI), Göteborg, Sweden, Sept. 5–9, 2016 (presenter, hosted by M. Dozza and J. Andersson).

HF-Auto Progress Meeting and Workshop, The French Insitute of Science and Technology for Transport, Development and Networks (IFSTTAR), Lyon, France, Apr. 4—8, 2016 (presenter, hosted by T. Bellet and M. Bruyas).

#### **Technical reports**

Varotto, S.F., Solís-Marcos, I., **Zhang, B.**, De Winter, J. C. F., Happee, R., 2017. Predicting real world effects of automation (Report No. HFAuto D4.1).

#### Outreach

Human factors of automated driving research highlights, AutoMotive Week, Helmond, The Netherlands, Mar. 31, 2017 (Organizer and presenter).

#### **TRAIL** Thesis Series

The following list contains the most recent dissertations in the TRAIL Thesis Series. For a complete overview of more than 250 titles see the TRAIL website: www.rsTRAIL.nl.

The TRAIL Thesis Series is a series of the Netherlands TRAIL Research School on transport, infrastructure and logistics.

Zhang, B., *Taking Back the Wheel: Transition of control from automated cars and trucks to manual driving*, T2021/11, February 2021, TRAIL Thesis Series, the Netherlands

Boelhouwer, A., *Exploring, Developing and Evaluating In-Car HMI to Support Appropriate use of Automated Cars*, T2021/10, January 2021, TRAIL Thesis Series, the Netherlands

Li, X., Development of an Integrity Analytical Model to Predict the Wet Collapse Pressure of Flexible Risers, T2021/9, February 2021, TRAIL Thesis Series, the Netherlands

Li, Z., Surface Crack Growth in Metallic Pipes Reinforced with Composite Repair System, T2021/8, January 2021, TRAIL Thesis Series, the Netherlands

Gavriilidou, A., *Cyclists in Motion: From data collection to behavioural models*, T2021/7, February 2021, TRAIL Thesis Series, the Netherlands

Methorst, R., *Exploring the Pedestrians Realm: An overview of insights needed for developing a generative system approach to walkability*, T2021/6, February 2021, TRAIL Thesis Series, the Netherlands

Walker, F., To Trust or Not to Trust? Assessment and calibration of driver trust in automated vehicles, T2021/5, February 2021, TRAIL Thesis Series, the Netherlands

Schneider, F., Spatial Activity-travel Patterns of Cyclists, T2021/4, February 2021, TRAIL Thesis Series, the Netherlands

Madadi, B., Design and Optimization of Road Networks for Automated Vehicles, T2021/3, January 2021, TRAIL Thesis Series, the Netherlands

Krabbenborg, L.D.M., *Tradable Credits for Congestion Management: support/reject?*, T2021/2, January 2021, TRAIL Thesis Series, the Netherlands

Polinder, G.J., *New Models and Applications for Railway Timetabling*, T2020/18, December 2020, TRAIL Thesis Series, the Netherlands

Scharpff, J.C.D., *Collective Decision Making trough Self-regulation*, T2020/17, November 2020, TRAIL Thesis Series, the Netherlands

Guo, W. Optimization of Synchromodal Matching Platforms under Uncertainties, T2020/16, November 2020, TRAIL Thesis Series, the Netherlands

Narayan, J., *Design and Analysis of On-Demand Mobility Systems*, T2020/15, October 2020, TRAIL Thesis Series, the Netherlands

Gong, X., Using Social Media to Characterise Crowds in City Events for Crowd Management, T2020/14, September 2020, TRAIL Thesis Series, the Netherlands

Rijal, A., *Managing External Temporal Constraints in Manual Warehouses*, T2020/13, September 2020, TRAIL Thesis Series, the Netherlands

Alonso González, M.J., Demand for Urban Pooled On-Demand Services: Attitudes, preferences and usage, T2020/12, July 2020, TRAIL Thesis Series, the Netherlands

Alwosheel, A.S.A., *Trustworthy and Explainable Artificial Neural Networks for choice Behaviour Analysis*, T2020/11, July 2020, TRAIL Thesis Series, the Netherlands

Zeng, Q., A New Composite Indicator of Company Performance Measurement from Economic and Environmental Perspectives for Motor Vehicle Manufacturers, T2020/10, May 2020, TRAIL Thesis Series, the Netherlands

Mirzaei, M., Advanced Storage and Retrieval Policies in Automated Warehouses, T2020/9, April 2020, TRAIL Thesis Series, the Netherlands

Nordhoff, S., User Acceptance of Automated Vehicles in Public Transport, T2020/8, April 2020, TRAIL Thesis Series, the Netherlands

Winter, M.K.E., *Providing Public Transport by Self-Driving Vehicles: User preferences, fleet operation, and parking management,* T2020/7, April 2020, TRAIL Thesis Series, the Netherlands

Mullakkal-Babu, F.A., *Modelling Safety Impacts of Automated Driving Systems in Multi-Lane Traffic*, T2020/6, March 2020, TRAIL Thesis Series, the Netherlands

Krishnakumari, P.K., *Multiscale Pattern Recognition of Transport Network Dynamics and its Applications: A bird's eye view on transport*, T2020/5, February 2020, TRAIL Thesis Series, the Netherlands

Wolbertus, Evaluating Electric Vehicle Charging Infrastructure Policies, T2020/4, February 2020, TRAIL Thesis Series, the Netherlands

Yap, M.D., Measuring, Predicting and Controlling Disruption Impacts for Urban Public Transport, T2020/3, February 2020, TRAIL Thesis Series, the Netherlands

Luo, D., Data-driven Analysis and Modeling of Passenger Flows and Service Networks for Public Transport Systems, T2020/2, February 2020, TRAIL Thesis Series, the Netherlands

Erp, P.B.C. van, *Relative Flow Data: New opportunities for traffic state estimation*, T2020/1, February 2020, TRAIL Thesis Series, the Netherlands